# ANNIER BLUE UNE

# VICTO COOBTILLO

Mouton de Gruyter

# Old Church Slavonic Grammar



# Old Church Slavonic Grammar

# SEVENTH REVISED EDITION

*by* Horace G. Lunt

Mouton de Gruyter Berlin · New York 2001 Mouton de Gruyter (formerly Mouton, The Hague) is a Division of Walter de Gruyter GmbH & Co. KG, Berlin

Printed on acid-free paper which falls within the guidelines of the ANSI to ensure permanence and durability.

Library of Congress Cataloging-in-Publication Data

Lunt, Horace Gray, 1918– Old Church Slavonic grammar / by Horace G. Lunt. - 7th rev. ed. p. cm. Includes bibliographical references and index. ISBN 3110162849 (cloth : alk. paper) 1. Church Slavic language-Grammar. I. Title. PG619.L8 2001 491.8'17015-dc21 2001030588

Die Deutsche Bibliothek – CIP-Einheitsaufnahme

Lunt, Horace Gray: Old church Slavonic grammar / by Horace G. Lunt. - 7., rev. ed. - Berlin; New York : Mouton de Gruyter, 2001 ISBN 3-11-016284-9

© Copyright 2001 by Walter de Gruyter GmbH & Co. KG, D-10785 Berlin All rights reserved, including those of translation into foreign languages. No part of this book may be reproduced or transmitted in any form or by any means, electronic or mechanical, including photocopy, recording, or any information storage and retrieval system, without permission in writing from the publisher. Printing: WB-Druck, Rieden/Allgäu. Binding: Lüderitz & Bauer, Berlin. Printed in Germany.

### PREFACE

This description of the structure of Old Church Slavonic is intended to present fully the important data about the language, without citing all the minutiae of attested variant spellings. The facts have been treated from the point of view of structural linguistics, but pedagogical clarity has taken precedence over the conciseness required for elegant formal description.

Old Church Slavonic was used over a period of some two hundred years and in various geographical parts of the Slavic world precisely at the time when the Slavic languages were undergoing rapid, fundamental, divergent changes. Some of these changes are doubtless reflected in the variant spellings in the few texts which have survived from this period, so that while most variations in grammar and vocabulary are the sorts of stylistic and idiosyncratic differences that are found in the standard or literary language of any single epoch, some important variant details result from different regional dialectal history. It has thus been necessary to include occasional references to historical and comparative linguistics in the first half of this book, although in principle these problems do not fall within the scope of a strictly descriptive, synchronic grammar.

It is necessary to normalize forms to present the grammatical structure as a consistent whole, and the normalization inevitably obscures the differences in the language of the various manuscripts. A clear picture of the different combinations of linguistic elements making up each of the texts is not to be achieved by lists of spelling variants or tables of percentages, but it is worth while to point out some of the striking variations. First-hand acquaintance with the texts and constant comparison of variant readings is the only way to arrive at an understanding both of the underlying unity of the texts as a whole and of the major and minor differences between them.

Little mention is made here of another type of comparison—the relationship of the OCS translated texts to the Greek originals. And yet it is in the Greek and in the translation technique that the explanations of hundreds of tiny problems (expecially of syntax) are to be found, and certain major structural problems need to be posed in terms of the influence of Greek on OCS. However, so few students have enough Greek to profit by such comparisons that it did not seem worth the considerable space that

### PREFACE

would be required. Excellent work in this field is available, though some scholars tend to forget that even a poor translator is governed by the structure of the language into which he is translating. The "Notes on Syntax" in Chapter Six are offered on the premise that something is better than nothing. It is particularly in this area that translation techniques need to be analyzed.

After forty years of teaching OCS and related topics in the history and structure of modern Slavic languages, my views on the nature of language and the models for describing language have evolved away from the Bloomfieldian structuralism of my training. The data of OCS have not changed importantly from the material described by scholars a century ago, although some details from imprecise editions have been discarded and a few new details must be accounted for. I continue to believe that every language is a coherent structure, and that each language can be described in terms of static and dynamic elements and learned by novices who do not have the slightest knowledge of its history.

Departures from tradition in classifying the data in no way change the facts themselves. The OCS verb, for example, is complicated, and classification will not make it less so. *Xvaliti, velěti,* and *želěti* do belong to different paradigms, whether one labels them IV A, IV B, and III 2 with Leskien, or IV, III 1 and III 2 with Diels, or II.8k, II.8e,1k, and I.4a,2b with Koch. I believe that it is most efficient simply to encourage students to learn the form from which the rest of the paradigm can be generated according to rules (*xvali-ti, velě-ti,* but *želěj-qt*<sub>0</sub>) and leave them to study the tables on pp. 114-117 and 136-137 for similarities and differences between paradigms. The present form of description is based on my belief that it is the morpheme that is the basic unit of communication.

A comparison of Old Church Slavonic—a language I believe to be a partially standardized written form of Late Common Slavic—with either its hypothetical ancestors or the descendants or collateral descendants of other forms of LCoS—is not the task of the synchronic description that takes up the first five chapters of this book. In the 1974 edition, I presented an epilogue ("Toward a generative phonology of OCS") that was based on a generative theory that proved to be too ambitious. Chapter Six in this book is an entirely new and relatively traditional sketch of the genesis of OCS (as a representative of Late Common Slavic).

This work was influenced by my teachers of long ago and by the students and colleagues I encountered during my years of teaching. I will not attempt to list them here. I can only express general thanks to the students who asked challenging questions and to their fellow-students and PREFACE

the colleagues throughout the scholarly world who helped me (in direct or indirect ways) find some of the answers. Special gratitude is due to Thomas J. Butler for his help in reading proof.

This edition too I dedicate to the memory of Professor S. H. Cross of Harvard, who introduced me to the study of Slavic, and to Professor G. R. Noyes of the University of California, who gave me my first lessons in Old Church Slavonic.

Horace G. Lunt

## TABLE OF CONTENTS

Prefa	ıce	v
Abbr	eviations	xv
INTR	ODUCTION: EXTERNAL HISTORY AND SOURCES	1
0.0	Definition of OCS	1
0.1	Cyril and Methodius	1
0.2	Manuscripts (.21 The Name of OCS)	3
0.3	Sources (.31 linguistic characteristics of OCS, "dialects"; .32 the "canonical texts"; .33 "*OCS" texts; .34 the study	
	of OCS; bibliography	4
CHAI	PTER ONE: THE OLD CHURCH SLAVONIC WRITING SYSTEMS	15
1.0	Glagolitic and Cyrillic (origin and use; .04 Latin letters)	15
1.1	The letters, general (table; problems of transliteration;	
	normalized and reconstructed forms)	18
1.2	The use of individual letters (.21 consonantal; .23 vocalic:	
	blocked vs. unblocked; .235 <i>i</i> -letters; .236 y; .237 ь ъ; .238 ě;	
	.24 /j/	25
1.3	Diacritics and other signs (.31 indications of iotization;	
	.34 abbreviations)	27
1.4	Punctuation	27
1.5	Numerals	28
CHAI	PTER TWO: THE SOUND SYSTEM	29
2.	Phonemics (.0 Phonology and orthography; .11 vowels;	
	.12 consonants; .2 j; 4. phonotactic constraints; .5 syllable;	
	.6 the jers ( $b$ 3); .61 tense jers; .62 the jer-shift; .625 * $jb$ ;	
	.63 neutral jers; .65 spelling; .7 nasal vowels)	29
3.	Morphophonemics (.2 word structure; .3 consonant adjustment	t;
	.4 palatalization, KI, KAI; .5 alternating vowel morphopho-	
	nemes; .6 iotation; .8 zero desinence; .9 apophony (ablaut))	42
	TER THREE: DECLENSION	52
4.0	Fundamental notions (definitions, possible forms, types of	
	declension, stem)	52

4.1	The twofold nominal declension (.11 vocative; .12 foreign nouns; .13 genitive-accusative; .14 dative sing <i>ovi</i> and other relics of the 'u-stems'; .15 instrumental sing. mascneut.;	
	.16 masc. in $-a$ ; .18 feminines in $-i$ ; .19 comparatives and active participles)	54
4.2	Pronominal declension (words included; suffixes; substitutive softening; .21 vbsb, sicb; .22 sb; .23 kbto; .24 čbto; .25 *jb;	54
	iže, tožde, etc.; 26 contractions)	62
4.3	Compound declension (suffixes, assimilation, contraction;	
	.31 comparatives, active participles; .32 mixture of declensions	64
4.4	Simple nominal declension (tense jers; trie; masculines;	
	.11 anomalous type: jers, gen. ~ acc. sing., nom. pl., četyre,	
	masculines, feminines, neuters)	71
4.5	Mixture of nominal declension types (.51 zvěrь, ogъnjь,	
	gospodь; .52 -telj,-arjь; .53 plane; singinъ;	
	.54 <i>zv</i> -stems; .55 <i>s</i> -stems, <i>oko</i> , <i>uxo</i> )	75
4.6	Declension of personal pronouns (.61 mo-/mb-; .62 ny;	
	.63 short datives; .64 genitive-accusative; .65 na, va)	76
EXCU	JRSUS	
4.7	Formation of the Comparative (7.1 - $i$ ; .72 - $i$ )	77
4.8	Formation of Adverbs (.81 pronominal: position, time,	
	manner; .82 adjectival: $-o/-\check{e}$ ; $-bsky$ )	79
CHAF	TER FOUR: CONJUGATION	81
5.0	Fundamental notions (.1 aspect; .2 inventory of forms; .3 components of a form; prefixes; .4 basic stem; .5 types of basic stem and suffix; .6 truncation, desinence classes, under- lying formulas, types of paradigm; .7 aspect morphology;	
	.8 method of description; .9 person-number desinences)	81
6.	The present tense (.1 suffixes; .2 stem; .3 KI-mutation;	~ -
_	.4–.8 irregularities)	95
7.	The imperative (.1 suffixes, alternations; .2 irregularities) .	98
8.	The present participles (.1 suffixes, alternations;	~~
•	.2 irregularities)	99
9.	The imperfect (.1 suffixes, alternations; .2 irregularities;	100
10	· • •	100
10.	Aorists (types; .1 desinences; .2 truncation; .3 substitutive	
	softening; .4 no; .5 irregularities; .6 unproductive types,	
	root-aorist suffixes, s-aorist suffixes, alternations; x-aorist;	

	.7 replacement of one type by another; .8 attested forms of	
	unproductive aorists: root, s-, x-aorists, variants; .9 irregular	
	verbs)	102
11.	Past participles (definitions; .1 past active participle;	
	.2 resultative participle; .3 past passive participle)	108
12.	Verbal substantive	111
13.	Infinitive and Supine (.1 suffixes; .2 stems and alternations;	
	.3 ambiguities; .4 irregularities	112
14.	Compound tenses (.1 perfect; .2 the pluperfects; .3 the con-	
	ditional; .4 the future perfect)	112
15.	The Individual Classes of Verbs	114
15.1	Verbs with basic stems in - <i>i</i> +	114
15.2	Verbs with basic stems in - <i>ě</i> +	116
15.3	Verbs with basic stems in $-a+$ preceded by a soft consonant	
	other than <i>j</i>	118
15.4	Verbs with basic stems in -j-a+	119
15.5	Verbs with basic stems in -ova+ or -eva+	121
15.6	Verbs with basic stems in $-a+$ preceded by a hard consonant	
	other than $v$	123
15.7	Verbs with the classifier -nq+	127
15.8	Verbs with zero classifier, stems ending in a consonant other	
	than <i>j</i>	131
15.9	Verbs with basic stems in $aj+$ , $-\check{e}j+$ , or $-j-\emptyset+$	136
16.	Irregular verbs	138
~~~		1 40
	PTER FIVE: NOTES ON SYNTAX AND VOCABULARY	142
17.	On adjectives (long and short forms; .1 vocatives)	142
18.	On the use of the cases (.1 nominative; .2 accusative;	
	genitive-accusative; .3 genitive; .4 locative; .5 dative;	143
10	.6 instrumental)	143
19.	On the use of the prepositions (.1 prepositions used with	151
20	only one case; .2 with two cases; .3 with three cases)	151
20.	On the syntax of the numerals	153
21.	On the use of the verbal forms (.1 present; future expressions;	
	.2 past tenses, aorist and imperfect; .3 participles; .4 infinitive	
22.	.5 supine; .6 se-verbs	153
	Some other parts of speech (.1 $da; .2 eda; .3 jako; .4 iže, eže)$	103
23.	On negation (.1 <i>ne</i> , <i>ni</i> ; .2 existential, transitive; .3 rhetorical;	163
	.4 lexical; .5 idiomatic)	103

### TABLE OF CONTENTS

24. Vocabulary and the structure of words (.1 nature of attested vocabulary; .2 meaning; .3 regional variants; .4 word-formation; .5 formants, persons; .7 other formants; .8 adjectives)
166

### CHAPTER SIX: A SKETCH HISTORY:

	FIER SIA. A SKETCH HISTORY.	
FROM	M LATE INDO-EUROPEAN TO LATE COMMON SLAVIC	181
25.	Indo-European and Slavic (.0 IE; .1 Slavic; .2 origins;	
	.4 periodization; .6 modern Slavic	181
26.	Methodology (.0 assumptions; .1 terminology; .12 palatali-	
	zation; .13 iotation; .5 pleophony, nasal vowels)	185
27.	Early IE to Pre-Balto-Slavic (.3 ruki-rule; .4 PBS)	190
28.	Vowels (Early Common Slavic to Middle Common Slavic)	192
29.	Changes (.1 satem palatalization; .2 progressive palatalization	n,
	BdC; .4 first regressive palatalization, KI; .5 sj $zj > \check{s} \check{z}$ ;	
	.7 Vowel Raising, Vowel Adjustment; .8 simplification	
	of syllables; .82 Middle Common Slavic vowel system	
	.9 monophthongization)	193
30.	Word-initial constraints (.2 prothetic w; .3 prothetic j)	203
31.	Second regressive palatalization, KAI	205
32.	Examples of derivation	206
33.	Pre-Balto-Slavic compared with earliest Common Slavic	208
34.	Examples	209
35.	Specific problems (.1 continuant obstruents)	214
36.	Initial vowels (.1 $\check{e}$ -; .2 $a$ - ~ $ja$ -; .4 $q$ -, $wq$ -; $o$ -, $wo$ -;	
	.5 je- ~ o-; 6 ju- ~ u-)	217
37.	IE and OCS morphology (.1 form classes; .2 Slavic ~ IE de-	
	clension; .3 consonant-stems; .4 vocalic stems)	221
38.	Difficulties in history (.1 pronominal vs. nominal;	
	.4 nominative, accusative; .5 accusative plural;	
	.6 nominative plural; .7 u-stems; .8 i-stems; .9 vocative)	224
39.	Case-forms	228
40.	Pronominal forms	230
41.	Numerals	233
42.	Conjugation in IE	235
43.	Present system (.1 desinences; .17 irregular stems)	236
44.	Present markers (.1 apophonic roots; .2 terminology; 3. e, je)	239
45.	Nasal suffix	244
46.	Imperative	246
47.	Imperfect tense	246
48.	Aorist	247

xii

### TABLE OF CONTENTS

49.	Infinitive and supine	247
50.	Apophony	247
5152.	On Slavic accent	248
53.–65.	On the Slavic lexicon	250
Verb Ind	lex	259
Subject	Index	262

### xiii

### ABBREVIATIONS

A = accusativea., act. = active AslP = Archiv für slavische Philologie As = AssemanianusBg = BulgarianByzSI = Byzantinoslavica C = any consonantCl, Cloz = ClozianusCz = Czechcomp. = comparativeD, dat. = dative ECoS = Early Common Slavic Eu, Euch = Euchologium Sinaiticum Ev. = Gospel(s) $f_{...}$  fem. = feminine G, gen. = genitive Gk = GreekGmc = Germanic Go = GothicI, impfv. = imperfective I, instr. = instrumental IJSLP = International Journal of Slavic Linguistics and Poetics imv. = imperative impf. = imperfect inf. = infinitiveJ = St. JohnJF = Južnoslovenski Filolog KF = Kiev Folia L = St. LukeL, loc. = locative LCoS = Late Common Slavic m, masc. = masculineMar = Marianus

MCoS = Middle Common Slavic Mk = St. MarkMt = St. Matthew ms = manuscript mss = manuscripts n = noten. neut. = neuter N, nom. = nominative O = OldOCS = Old Church Slavonic P = perfectiveP. Pol = Polish Ps. = Psalterium Sinaiticum part. = participle pass. = passivepl., plur. = pluralRÉSI = Revue des Études slaves Sa = SanscritSav = Savvina kniga SC = Serbo-Croatiansg., sing. = singularSik = SlovakSln = SloveneSPb = Sanktpeterburg Su, Supr = Suprasliensis Vat = Vatican Cyrillic Palimpsest Zo, Zogr = ZographensisZoF = Zograph Folia/ = or $\sim$  = alternates with; is opposed to

Numeration of the paragraphs is decimal; every number to the right of the decimal point is to be read as a separate unit. Thus 15.642 = 15.6.4.2, i.e. the second subdivision of 15.64, which is the fourth subdivision of 15.6. For personal names in references (Diels, Vaillant, etc.) see the bibliography, 0.341 (pp. 12-14). For details about the codices, see 0.321 ff. (pp. 7-10). Citations are made by page and line for Cloz, Euch, Supr, and KF; by chapter and verse for Zo, Mar, As, Sav, and Vat; and by psalm and verse (Eastern numeration, as in the Septuagint) for Ps.

xvi

# EXTERNAL HISTORY AND SOURCES

**0.0** Old Church Slavonic is the name given to the language of the oldest Slavic manuscripts, which date from the tenth or eleventh century. Since it is a literary language, used by the Slavs of many different regions, it represents not one regional dialect, but a generalized form of early Eastern Balkan Slavic (or Bulgaro-Macedonian) which cannot be specifically localized. It is important to cultural historians as the medium of Slavic culture in the Middle Ages and to linguists as the earliest form of Slavic known, a form very close to the language called Proto-Slavic or Common Slavic which was presumably spoken by all Slavs before they became differentiated into separate nations.

**0.1** The Slavs are mentioned by historians with increasing frequence from the fifth century CE, but there is no reason to believe that they wrote their language down before the ninth century. In 862, Prince Rastislav, ruler of Morava (located somewhere in the Danube Basin), appealed to the Byzantine Emperor Michael III for a teacher who would give instruction in Christian law "in our own language." Michael appointed a priest, the experienced diplomat and able scholar Constantine, called the Philosopher, to the difficult and important mission. Constantine was a native of Salonika, and the Emperor pointed out that all the people of Salonika spoke Slavic well (Gonoynthe Bach vncto chostantine Kethodius, a former civil administrator who had become a monk.

The brothers elaborated an alphabet for the Slavic language, translated the most important liturgical books, and started to train Moravans for the clergy. They travelled to Rome to visit the Pope and have some of their pupils ordained into the priesthood. On the way, the "Slavic apostles" stopped at the court of the Slavic prince Kocel (Kouhab) of Pannonia (in what is now western Hungary), where they were welcomed enthusiastically and acquired more pupils.

The Pope received them favorably and approved of their work, condemning the "three-tongue" heresy of those who claimed that only Greek, Latin, and Hebrew had the right to serve as written and liturgical languages. Constantine, however, fell sick in Rome, and on his death-bed he took monastic vows and assumed the name of Cyril (869). Later he was sainted.

Methodius was now appointed Archbishop of Pannonia (including Morava), and he set out for Morava with his newly consecrated pupils. Rastislav had been deposed and blinded in 870, and the new ruler Sventopulk (GEATONEAKE) was surrounded by Frankish priests who bitterly opposed the Slavic liturgy and the eastern, Greek influences it represented. The Franks had Methodius imprisoned in Bavaria, and only after two years did the Pope come to his aid. The Slavic rite was established in Morava, but on Methodius's death in 885 the Frankish clergy did their best to stamp it out. Driven from Sventopulk's realm, some of the Slavic priests apparently found asylum in Bohemia, and for some time they were able to maintain the Slavic liturgy and the writing that went with it. However, in the eleventh century Slavic culture steadily lost ground in the area and in 1097 the last Slavic monastery was abolished and the Slavic liturgy was formally prohibited.

Meanwhile, the Bulgar ruler Boris had been baptized in 864 and established Christianity as the official religion of his extensive realm. The meager historical sources offer no information about the language used in the new churches, but since Methodius apparently visited Constantinople and left two of his disciples and books in care of the Emperor and Patriarch, it is plausible that some knowledge of OCS existed in the eastern Bulgarian lands. In any case, the main body of Methodius's followers found refuge in Bulgarian territory, and OCS was nourished in two cultural centers, one in the east at the court of the Bulgarian Tsar Simeon (893-927) and one in the west, in Macedonia. Political conditions were favorable, and Slavic culture prospered, but not for long. After the destruction of the Bulgarian state in the east by the Byzantine armies at the beginning of the 970s, a state in Macedonia arose and flourished briefly. The might of Byzantium finally, after a dozen years of warring, crushed the last vestiges of independence by annihilating the armies of King Samuil in 1014. Even after this catastrophe, some degree of learning was maintained in the Bulgarian, Macedonian and Serbian monasteries, and in distant Croatia. When Christianity was accepted by the Rus' prince Volodimer in 988, Slavic books may have found a modest place among East Slavs. In the 1030s Prince Jaroslav "the Wise" apparently adopted

the Slavonic rite, and books and perhaps teachers from the Bulgarian lands made it possible for the East Slavs to adapt Old Church Slavonic for their own use. By the 1050s Kiev and Novgorod were creative cultural centers.

**0.2** The few early manuscripts which have come down to us do not go back to the days of Cyril and Methodius, but date at the earliest from the end of the tenth century and more probably from closer to the 1050s. Being thus the products of the period of turmoil attendant upon the destruction of the Macedonian state, they do not represent a thriving, developing culture, but only remnants. The scribes, it seems, were not well trained, and the manuscripts contain blunders which not even the most ingenious theory can bring into accord with a plausible linguistic system.

It is assumed that most of these manuscripts contain translations made by Cyril and Methodius, and the rest are translations made by their disciples, probably during the first decades after the death of the saints. However, since we lack contemporary manuscripts, or even the immediate copies which were doubtless made in the heyday of the states of Simeon and Samuil, we cannot know in precise detail the language actually written by the Slavic Apostles. Their own works, taken to Bohemia, Bulgaria, and Macedonia, were copied and recopied, edited and modified by generations of workers, and we possess only a few random examples of the copies.

The native dialect of Cyril and Methodius, who were born in Salonika, was presumably southeastern Macedonian. Perhaps Methodius adopted some features of the dialect of the Slavic-speaking province (possibly in the mountains northeast of Salonika) where he was an administrator for a time. In Constantinople the brothers may have become acquainted with the speech of Slavs from other areas. It is not impossible that the local dialects of Morava and Pannonia may have influenced the language of the translations. But in any case all evidence indicates that in the ninth century the difference between Slavic dialects from the Baltic to the Adriatic and Aegean Seas, from the Elbe to the steppes of Kievan Rus', were minimal, and it is probable that the dialect of Salonika was readily understandable to the Moravans and Pannonians of the Danube Basin.

**0.21** Whatever the spoken dialects were, the *church language* appears to have been essentially the same in different areas. Because this language was used in the west and south and then served for centuries in Rus', in the Grand Duchy of Lithuania, and in Muscovy as a literary language (which naturally became modified progressively with the course of time), it is

known as **Church Slavonic**. Since the majority of the early manuscripts which have survived were copied in the Bulgaro-Macedonian area and since there are certain specifically eastern Balkan Slavic features, many scholars have preferred to call the language **Old Bulgarian**, although **Old Macedonian** could also be justified. Early nineteenth-century scholars conjectured that this language was based on the dialect of Pannonia, and accordingly called it **Old Slovenian**. In the earliest sources, the language and letters are referred to by the adjectives **cnostanecka**, Greek  $\sigma \kappa \lambda \alpha \beta \iota \kappa \delta \varsigma$ ,  $\sigma \theta \lambda \alpha \beta \iota \kappa \delta \varsigma$ , or  $\sigma \kappa \lambda \alpha \beta \iota \iota \kappa \delta \varsigma$ , Latin sclavinica, sclavinisca or sclavina, all of which mean simply *Slavic* (or *Slavonic*).

**0.3** The tenth and eleventh centuries witnessed far-reaching changes in the several Slavic macrodialects. Reflections of the changes in the spoken languages appear in the spelling and the grammatical forms in the manuscripts and enable us to identify them as Serbian, Macedonian, Bulgarian, or Rusian (early East Slavic). As a convenient (but arbitrary) date, it is generally reckoned that non-East Slavic manuscripts written (or believed to have been written) before 1100 are Old Church Slavonic, as opposed to the Macedonian-Church Slavonic,<sup>1</sup> Bulgarian-Church Slavonic, or Serbian-Church Slavonic written after that time.<sup>2</sup> Most grammars of Old Church Slavonic exclude the considerable body of manuscripts produced in Rus' before 1100, because they have unmistakeably East Slavic traits. In fact, some Rus' manuscripts come about as close to the theoretical ideal described in grammars as the "classical" manuscripts do.

None of the OCS manuscripts is dated. None can be much older than the year 1000, and some may be considerably younger. It is not easy to establish even the relative age of the manuscript, since a text with archaic phonetic features may present younger morphological forms and vice versa, and the chronologies established on grounds of paleography are not reliable for this earliest period.

<sup>&</sup>lt;sup>1</sup> The line between OCS and post-OCS manuscripts is arbitrary and terminology is varied. The common term "Middle Bulgarian" is usually contrasted to "Old Bulgarian" (= OCS), and loosely used for manuscripts whose language demonstrates a broad spectrum of regional and temporal dialect features, often clearly the result of generations of copying by scribes with different habits.

<sup>&</sup>lt;sup>2</sup> These later forms of Church Slavonic are also known as the Serbian, Russian, etc. recensions of Church Slavonic. There is also a Croatian recension, attested in glagolitic mss throughout the Middle Ages and still used in some Croatian parishes. There is evidence (beside KF, cf. §0.311) for a Bohemian or "Moravian" recension, although only isolated fragments from this area have survived.

It must therefore be emphasized that Old Church Slavonic, as we deal with it in describing the grammatical patterns, is a theoretical, reconstructed language. The manuscripts written over a period of many decades, in different parts of the Balkan peninsula, present numerous variations in spelling, grammar, and vocabulary. It is assumed that the variations are later modifications affecting the originally unified type of language used by Cyril and Methodius and their immediate associates. On this assumption, all grammars of OCS have dealt with the "original" language, with some concessions to the usage of the several manuscripts. Indeed, the myriad attested variants constitute long and essentially uninformative lists. Therefore this grammar too describes a norm, a generalized type of dialect which does not correspond exactly to the facts of any one manuscript. Definable classes of variants are mentioned, and important individual deviations in detail are noted.

**0.301** For the purposes of a grammatical description of the oldest attainable stage of OCS, it is imperative to restrict the data to the oldest manuscripts. I prefer a narrower "canon" than many linguists have defined in recent years (see below). The study of broader cultural problems is quite a different matter. There is no doubt that the scribes who produced the surviving OCS manuscripts were familiar with many texts that are available to us only in copies that were written down decades or centuries later. The language of some of these copies no doubt reflects OCS in many details—but precisely what is old and what is new constitutes a long series of controversial questions. In particular, just what words can be called *Old* Church Slavonic? Lexicographers have admitted a series of texts into the domain of OCS; many more could be justified. But for historical study of the language and related dialects, investigators should be alert to the antiquity of the manuscripts from which each item is cited as evidence.

**0.31** There is no clear-cut set of features which differentiate the language of the manuscripts called Old Church Slavonic from the oldest of the texts termed simply Church Slavonic, but the relatively "correct" usage of certain letters and the relatively high occurrence of certain morphological forms which comparative evidence shows to be old give us some criteria. In the spelling, the use of the letters for the nasal vowels ( $\rho$ ,  $\rho$  and perhaps  $j\rho$ ) are of great importance. It is the "misuse" of these letters which is the clearest sign of an East Slavic scribe and the reason why such manuscripts as the *Ostromirovo Evangelie* (dated 1056-57, the oldest *dated* Slavic ms) are excluded from a description of OCS. Further, the writing of the symbols **b** and **b** more or less where we expect them, and the

consistent usage of the letter  $\check{e}$  ( $\mathfrak{t}$ ) help to identify the language. Chief among the morphological characteristics are the use of the root-aorists (an archaism which was irregular from the point of view of the over-all system of OCS) and the uncontracted forms of the long adjectives. All of these features together, in conjunction with paleographical evidence (the details of the shape of the letters and the style of writing them), and the *absence* of specifically dialect features, serve to mark a manuscript as OCS.<sup>3</sup>

**0.311** Three groups of manuscripts can be distinguished on the grounds of variant phonetic and morphological features. The Kiev Folia, with at least one pervasive Czech trait (see below  $\S0.326$ ), are the only representative of a variant of OCS which was presumably used in Bohemia or perhaps Moravia. (The location of the "Morava" where Constantine and Methodius worked in the 860s is unknown.) It is to be regarded as a "literary dialect", following local norms worked out in a specific area and opposed to the other OCS texts. Two of the cyrillic texts (Sav and Supr, see below) show in general some fairly specific eastern Bulgarian features, but the differences are neither great enough nor consistent enough to make it necessary to oppose a Macedonian "dialect" to a "Bulgarian" dialect of OCS.<sup>4</sup>

**0.32** The "classical" or "canonical" texts of OCS include eight fairly extensive manuscripts (one a palimpsest), two sizeable fragments, and a number of single folia and parts of pages. Best represented is the *Gospel* text, with five manuscripts. The *psalms* are nearly complete, and there is

<sup>&</sup>lt;sup>3</sup> Citations from post-OCS mss that are believed to be reasonably faithful copies of originally OCS texts will be marked \*OCS in this book (e.g. \*OCS snaxa 'daugh-ter-in-law'): the label is marked, not the word.

<sup>&</sup>lt;sup>4</sup> Elaborate schemes of "OCS dialects" have been set up by some scholars on the basis of minute orthographic and morphological details, cf., e.g., Kul'bakin, Vieux Slave 354ff., Marguliés, Codex Supr., 227ff. The chief objection to this method is that it regards each scribe as a careful, trained phonetician who was trying to reproduce his own pronunciation. In reality the scribes were chiefly concerned with writing "correctly"—which sometimes meant copying exactly, and sometimes meant applying slightly different orthographical rules to the text being copied. We can determine literary norms (i.e. the spellings and grammatical forms given scribes or groups of scribes thought were proper), but to determine the phonetic details of pronunciation and from them the local origin of a scribe is impossible from the type of manuscript which we have in OCS. Cf. N. Durnovo, 'Slavjanskoe pravopisanie X-XII vv.', *Slavia* 12.45-84.

a prayer-book, a fragment of a missal, parts of a few hymns, and some sermons and saints' lives. To these manuscripts must be added the oldest dated Slavic text, a short gravestone inscription set up by the Macedonian king Samuil in 993.

The amount is actually quite modest: if the entire body of material were set up in the type and format of this book, it would make a volume well under a thousand pages, including perhaps 350 which represent variants and not separate texts. The individual mss would occupy roughly the following number of pages (for abbreviations see below): Supr 300, Mar 175, Zo 150, Sav, Ps and Euch each 75, Vat 60, Cloz 20, and the fragments another 20.

Perhaps the oldest manuscripts are the two full versions of the 0.321 Gospels, the so-called *tetraevangelia*, both written in glagolitic (see the next chapter, particularly §1.01). The Codex Zographensis (Zo) has 271 folia in OCS, plus 17 in an old Macedonian ChSI glagolitic version (Zo<sup>2</sup>), and some later addenda in cyrillic. The OCS text contains Matthew 3:11 through the end of John (but Mt 16:20-24:20 is later, Zo<sup>2</sup>). Phonetically it is nearest to the theoretical norms posited for the language of Cyril and Methodius, but certain morphological forms (especially aorists) and some textual readings seem to be rather younger. The Codex Marianus (Mar) has 174 folia, containing the Gospel text from Mt 5:23 to John 21:7. Certain deviations from the theoretical norms indicate Macedonian influences, others possibly Serbian (if not northern Macedonian). In the nineteenth century both were still on Mt. Athos, Zo in the Zograph Monastery, and Mar in the skete of the Virgin Mary. Zo is now in the Russian State Library in St. Petersburg, Mar in the Russian National Library in Moscow. Zo may be presumed to have been written in the 1020s, Mar in the 1030s; any dating is guesswork.

**0.322** Quite different arrangements of gospel materials are found in the three gospel lectionaries, where the excerpts from the four gospels are presented as lessons to be read on specific days of the year.<sup>5</sup> The Greek term for such a book is  $\varepsilon \iota \alpha \gamma \gamma \epsilon \lambda \iota ov$ , borrowed into OCS as **EBANFEANE**. In the Greek Orthodox tradition the lectionary, as the primary source for the Word of God, is itself a sacred object that requires special care; it is for this reason that some 25% of all surviving Slavic medieval manuscripts are gospel lectionaries. Yet the individual manuscripts ordinarily vary in content, because—unlike the *tetraevangelion*, which contains the full gospels of Matthew, Mark, Luke, John—the lectionary is a general plan

<sup>&</sup>lt;sup>5</sup> In Rus' a lectionary could be called anpakoes—apparently an adaptation of a Greek designation.

that provides well over 350 slots that designate a particular gospel pericope (reading, lection) that may or should be read. The usual selection provides lections (1) for all Saturdays and Sundays, the weekdays for the six weeks of the Great Fast (Lent) and the seven weeks from Easter to Ascension, and (2) selected Feast Days defined by day of the month. The first part is relatively standard, while the second part varies considerably because each manuscript was written with the needs of a particular region or individual church in mind. The glagolitic Codex Assemanianus (As), with 158 folia, has a chaotic innovating orthography, but retains numerous archaisms; it was written after 1038, and perhaps well after 1050, almost certainly in Macedonia. The newly discovered Vatican palimpsest cyrillic lectionary (Vat) is only partially legible, for the OCS text was washed off sometime near 1200 and a Greek lectionary text was written over the cyrillic lines. Although 96 folia had cyrillic writing, only about half of them contain reasonably legible connected text. Vat seems to be generally conservative, but with enough innovations to place it perhaps in the 1040s, possibly in Macedonia. Sava's Book, or Savvina Kniga (Sav) retains only 129 out of the original 200 or so folia. It is written in cyrillic, and while it retains some old textual readings, the language is definitely innovative, and seems to reflect central or eastern Bulgarian dialects. It probably was written in the 1030s. The Assemanianus was found in Jerusalem in the eighteenth century and taken to Rome, where it is kept in the Vatican Library. Vat is housed in the same library. Sav was in Rus' by the fourteenth century, to judge from the fact that lost folia were replaced by pages written in an East Slavic hand of that time. It was found in a Pskov monastery in the nineteenth century and is now in the Rossijskij Gosudarstvennyj Arxiv Drevnix Aktov in Moscow.

**0.323** The Psalter and Prayer-Book are both still in the Monastery of St. Catherine on Mount Sinai, and are named accordingly. The glagolitic **Psalterium Sinaiticum** contains the 151 psalms plus ten canticles and some common prayers. The text is riddled with faults, but preserves archaisms along with innovating spelling reflecting Macedonian phonetics; it was produced by several scribes who worked together, very likely in the 1040s.<sup>6</sup> From the glagolitic **Euchologium Sinaiticum** (Euch), 137 folia have survived of what must have been a much larger book. Euch

<sup>&</sup>lt;sup>6</sup> A second glagolitic psalter, apparently from the same workshop, was found at St. Catherine's in 1975, but it has not yet been adequately described and is known to the scholarly world from a single photograph.

contains also three damaged folia from an eastern missal (or liturgiarium). The language of the two manuscripts is in many respect similar, and both appear to be from Macedonia.

**0.324** The largest Old Church Slavonic manuscript is the **Codex Supra**sliensis (Supr), with 285 folia. It is a menaeum (четья минея in Russian terminology) for the month of March, that is, a collection of saints' lives for daily reading, and contains also a series of sermons for Holy Week and Easter. The writing is cyrillic, and the language is in every particular younger than that of the other texts, excepting Sav. It seems to have been written in central or eastern Bulgaria. Found in 1823 in a monastery in what is now Poland, it was later broken up: part (1-236, numbering each side) is now in the National Library in Ljubljana (Slovenia), while a second part (237-268) somehow (stolen?) found its way to Russia, where it is now in the Russian State Library in St. Petersburg. The largest section (269-570) remained in Warsaw. Removed from the Zamojski Library during World War II, it reappeared in the US in 1968, was acquired by an American and returned to Poland.

**0.325** Another book which must have contained a large number of homilies (some of which are also in Supr) has survived only in fragments, fourteen folia in glagolitic called the **Glagolita Clozianus** (Cloz). One part of this ms has been demonstrated to be a sermon composed by Methodius. Like Mar, the language of Cloz shows both Macedonian and Serbian influences. Formerly the property of Count Cloz, two of the folia are now in the Ferdinandeum in Innsbruck (Austria), the other twelve in the Museo Civico in Trento, Italy.

**0.326** The remnants of a missal (more precisely, sacramentary) of the western rite is possibly the oldest of our texts. The seven glagolitic folia known as the **Kiev Folia** (KF) are generally considered as most archaic from both the paleographic and the linguistic points of view, but at the same time this text replaces the most characteristically Bulgarian phonetic traits of the other mss with unmistakably Czech features.<sup>7</sup> By this simple modification of the most striking foreign features, the literary language was adapted for local use. Unfortunately the small amount of text of the

<sup>&</sup>lt;sup>7</sup> The Bg št and žd which stem historically from \*tj (and \*kt), \*dj are kept, as a rule, in Rusian and Serbian ChSI, although they stand out as specifically foreign elements. But in KF they are regularly replaced by the Czech c/z, e.g. prosece VIb6 'begging,' pomocb 'help,' podazb 'give' for Bg proseste, pomoštb, podaždb.

KF does not permit far-reading conclusions as to the place of the Czech type of language in the development of the early Slavic literary languages. The ms was taken from Jerusalem (perhaps originally from Sinai) to Kiev in 1870, and is now in the Vernadskyi Central Library.

**0.327** A few isolated pages or fragments which are generally considered OCS are of importance chiefly to confirm the linguistic evidence offered by the larger texts and to demonstrate the early date of some of the literature. The **Shuck Psalter** (Sl), now lost, was five leaves of a cyrillic ms, containing part of the 118th psalm. Two leaves from a glagolitic gospel lectionary are known as the **Ochrid Folia** (Ochr).<sup>8</sup> The Church Fathers are represented by the two cyrillic **Hilandar Folia** (Hil) containing a text of Cyril of Jerusalem and the glagolitic **Rila Folia** (Ril, formerly called *Macedonian Frag.*), with parts of sermons by Ephraim the Syrian. The two cyrillic **Zograph Folia** (ZogrF) are from a monastic code of St. Basil.<sup>9</sup> Eight partially legible pages of liturgical hymns survive in the glagolitic **St. Petersburg Octoich**.

0.33 The mass of later manuscripts that have survived, mostly now in libraries, offer irrefutable evidence that the literature of the Slavs before 1100 must have been far more extensive than this small list. Doubtless Cyril and Methodius themselves translated the Acts and Epistles (Apostolo, in Slavic terminology), and Methodius may well have finished translating the Old Testament. A code of church law (nomokanon) and a patericon (didactic tales about famous monks and hermits) are also attributed to the Slavic Apostles. It is probable that a number of liturgical works were translated from Latin as well as from Greek in the earliest period: the Kiev Folia are an example. Some of the prayers in Euch reflect Old High German versions. The hagiographic Lives (Žitija) of Ss. Cyril and Methodius are early; Methodius may well have written about his brother. At least two poems (Proglass and the Alphabet Acrostic) must be attributed to the immediate pupils of the Slavic Apostles if not to Cyril himself, and another poem (Poxvala Simeonu) is from the early tenth century. Some original hymns surely go back to tenth-century Bulgaria. However, all

<sup>&</sup>lt;sup>8</sup> Undol'skij's Fragments, two folia of a cyrillic gospel lectionary usually called OCS, are rather to be classed with the *Enina Apostol* (discovered in 1960) as representing a slightly more recent kind of language.

<sup>&</sup>lt;sup>9</sup> The badly damaged *Cyrillic Macedonian Folium* is usually called OCS, but it has some later features and in any case supplies no crucial data for grammar. The text appears to be from St. Cyril's preface to his translation of the Gospel.

these works have come down to us in a language which has been modified to suit the tastes of later scribes and which we therefore do not consider in the linguistic study of Old Church Slavonic.

**0.34** The study of Church Slavonic, the literary language of all the Orthodox East and South Slavs (and some Catholic Croats), was begun early by native writers, but their grammars were unoriginal adaptations of Greek and Latin works, wholly inadequate to describe a Slavic language. The best and most famous grammar was published by the Rutherian Meletij Smotryc'kyj in 1619. Modern study of ChSl begins with the great Czech scholar Josef Dobrovský's *Institutiones linguae slavicae dialecti veteris*, 1822. The exploration and description of old manuscripts was continued by the Slovenes Jernej Kopitar and Franjo Miklošič (Miklosich) and the Russian Aleksandr Vostokov (among others), but it was the exemplary editions of the codices Zographensis (1879) and Marianus (1883) by the Croat Vatroslav Jagić that finally made it possible to separate Old Slavonic from later accretions.

The classical description was made by the great leader of the "Young Grammarians", August Leskien, in his *Handbuch der altbulgarischen* (altkirchenslavischen) Sprache. This manual appeared first in 1871, was revised four times, translated into Russian (1890), and has never gone out of use as a textbook. Moreover, its principle of including historical and comparative data beside the synchronic description set the style for nearly all later grammars and textbooks. The reference grammar by Václav Vondrák (1912) is an example. Unquestionably the most important book of this type is the encyclopedic Altkirchenslavische Grammatik by Paul Diels (1932), still an indispensable tool for anyone doing detailed work with OCS, although newer editions of some of the manuscripts show that some of his evidence needs to be modified. The Dutch scholar Nikolaas van Wijk lays an even greater emphasis on the historical factors in his Geschichte der altkirchenslavischen Sprache (1931).

The fundamental discussion of Common Slavic (or Proto-Slavic), with reference to its relations with other Indo-European languages and to the modern Slavic languages, is Antoine Meillet's *Le Slave Commun* (2nd ed., with A. Vaillant, 1934), which is of course based largely on the material of OCS.

An excellent non-historical description of OCS is André Vaillant's *Manuel du Vieux Slave*<sup>2</sup> (1964). It is rich in detail and frequently cites data from later texts to clarify some of the obscure points in OCS, but the treatment of sounds is somewhat old-fashioned for the time. Nikolaj

Trubetzkoy's uneven Altkirchenslavische Grammatik (written before 1938, published 1954) offered stimulating new views on the writing system and the organization of morphological description.

Syntax is given some attention by Vondrák and Vaillant, and more problems are discussed in the 1963 volume edited by Kurz. More comprehensive treatment of many questions is available in Večerka.

The lexicon of the short list of canonical texts, along with a broad selection of words from post-OCS manuscripts whose text is believed to go back to the OCS period, is treated in the *Slovník jazyka staros-lověnského*, published by the Czech Academy, 1958–97. A single-volume distillation of SJS is *Словаръ старославянского языка*, 1994, edited by Raisa Cejtlin and others [reprinted in 1999].

**0.341 Bibliography.** The number of books and articles that deal wholly or in part with OCS is enormous. Here I list only the editions of the OCS texts, some analyses, and some of my own works that provide the background for my decisions. Further titles will appear in footnotes.

A. TEXTS. [The editions are listed first (a) and then studies (b).]

### 1. Glagolitic

KF: (a) Jos Schaeken, Die Kiever Blätter. (= Studies in Slavic and General Linguistics, 9) Amsterdam. (b) H. G. Lunt, 'Once Again the Kiev Folia,' SEEJ 32 (1988): 341-83.

Zo: (a) V. Jagić (ed.), Quattuor evangeliorum Codex Glagoliticus olim zographensis. Berlin, 1879. (b) L. Moszyński, 'Ze studiów nad rękopisem kodeksu zografskiego,' Wroc1aw-Warszawa-Kraków; idem, Język Kodeksu Zografskiego, I (1975), II (1990).

Mar: (a) V. Jagić (ed.), Quattuor evangeliorum ... Codex Marianus glagoliticus. St. Petersburg, 1883 [contains study and lexicon].

As: (a) Josef Kurz, Evangeliarium Assemani II Prague, 1955 [Cyrillic transcription]; I. Dujčev (ed.), Acemanueso евангелие. Sofia, 1981 [photo-reproduction of ms, in color]. (b) H. G. Lunt, 'On the Old Church Slavonic codex Assemanianus,' Makedonski jazik 31-32 (1981-82): 405-16. Christoph Koch, Kommentiertes Wort- und Formenverzeichnis des altkirchenslavischen Codex Assemanianus (= Monumenta Linguae Slavicae Dialekti Veteris, XLIII), Freiburg i. Br., 2000.<sup>10</sup>

<sup>&</sup>lt;sup>10</sup> The extensive comments provide meticulous data that elucidate scores of major and minor details (of spelling, morphology, syntax, meaning, translation technique).

Ps: (a) S. Severjanov, Синайскан псалтырь. Petrograd, 1922 [Cyrillic transcription, with lexicon]; Moshé Altbauer, Psalterium Sinaiticum. Skopje, 1971 [photoreproduction]; F. V. Mareš, Psalterii Sinaitici pars nova (monasterii s. Catharinae codex slav. 2/N) (= Österreichische Akademie der Wissenschaften Phil.-hist. Kl., Schriften der Balkan-Kommission, Philol. Abt., 38), Vienna, 1997 [Cyrillic transcription with lexicon].

Euch: (a) Rajko Nahtigal, Euchologium Sinaiticum I [photo-reproduction], II [Cyrillic transcription, copious notes]. (= Dela 1, 2, of Akad. znanosti in umetnosti, filoz.-filol.-histor.razred), Ljubljana, 1941–42; Ján Frček, Euchologium Sinaiticum, Texte slave avec sources greques et traduction française (= Patrologia Orientalis, XXIV, 5, XXV, 3), Paris, 1933, 1939 [important for Greek sources].

*Cloz*: Antonín **Dostál**, *Clozianus*, *codex palaeoslovenicus glagoliticus*. Prague, 1959 [Cyrillic and Roman transcriptions, Greek texts, translations, lexicon].

Ril: Ivan Gošev, Рилски глаголически листове, Sofia, 1956 [Cyrillic transcription, lexicon]. (b) Н. G. Lunt, IJSLP 1/2 (1959): 16-37.

SPbO: fragmentary, unpublished; preliminary report by Lunt, 'On Slavonic Palimpsests,' American Contributions to the Fourth Internat'l Cong. of Slavicists, Moscow, September 1958 (= Slavistic Printings and Reprintings, XXI), The Hague, 1958, pp. 191-200.

## 2. Cyrillic

Sav: (a) Vjač. Ščepkin, Саввина Книга. StPbg, 1903 [with lexicon]; ed. O. A. Knjaževskaja, L. A. Korobenko, E. P. Dogramadžieva, Саввина книга. Moscow, 1999

Vat: (a) T. Krъstanov, A-M Totomanova, I. Dobrev (eds.), Vatikansko Evangelie. Sofia, 1996. (b) H. G. Lunt, 'On Defining OCS; the Case of the Vatican Cyrillic Palimpsest,' *IJSLP* 43 (2001).

Supr: (a) S. Severjanov, Супрасльская рукопись. SPb, 1904; J. Zaimov and Mario Capaldo, Супрасьлски или Ретков сборник. 2 vol. [photoreproduction, Severjanov's text; Greek texts], Sofia, 1982-83. (b) H. G. Lunt, 'On Editing Early Slavic Manuscripts: the Cases of the Codex Suprasliensis ...,' IJSLP 30 (1984): 7-34, 74-6.

Sl: (a) V. Jagić, Specimina linguae palaeoslovenicae (SPb, 1882)

0.341

<sup>&</sup>lt;sup>11</sup> This edition arrived after the present grammar was already complete; all references to Sav are based on Ščepkin's edition.

Hil: (a) Angelina **Minčeva**, Старобългарски кирилски откъслеци (Sofia, 1978) 24-39.

ZogrFol: (a) Minčeva, pp. 39-45.

[Note: Zo, Mar, Ps, Sav, Supr were reprinted under the auspices of the Seminar für slavische Philologie of the University of Graz, Austria.]

### **B. GRAMMARS**

Paul Diels, Altkirchenslavische Grammatik<sup>2</sup>, I-II, Heidelberg, 1963. André Vaillant, Manuel du vieux slave<sup>2</sup> I-II, Paris, 1964. Граматика на старобългарския език (ed. Ivan Duridanov), Sofia, 1991.

### C. DICTIONARIES

Slovník jazyka staroslověnského, Prague, 1958-1997. Старославянский словарь, ed. R. Cejtlin, Moscow, 1994.

### **D. STUDIES**

On aspect: Antonín Dostál, Studie o vidovém systému v staroslověnštině, Prague, 1954.

On verbal forms: Christoph Koch, Das morphologische System des altkirchenslavischen Verbums, Munich, 1990.

On syntax: Исследования по синтаксису старославянского языка, ed. Josef Kurz, Prague. 1963. Rudolf Večerka, Altkirchenslavische (altbulgarische) Syntax, I-III, Freiburg i. Br., 1989, 1993, 1996.

On the prehistory of Slavic: Antoine Meillet (with André Vaillant), Le slave commun<sup>2</sup>, Paris, 1934. H. G. Lunt, The Progressive Palatalization of Common Slavic, Skopje, 1981; 'Common Slavic, Proto-Slavic, Pan-Slavic: What Are We Talking About? I. About Phonology,' IJSLP 41 (1997) 7-67; 'On Common Slavic Phonology: Palatalizations, Diphthongs, and Morphophonemes,' IJSLP 42 (1998) 7-14; 'Thoughts, Suggestions, and Questions about the Earliest Slavic Writing Systems', Wiener slavistisches Jahrbuch 46 (2000); 'Cyril and Methodius with Rastislav Prince of Morava: Where Were They?' Thessaloniki Magna Moravia, Thessaloniki, 1999. pp. 87-112.

14

### CHAPTER ONE

# THE OLD CHURCH SLAVONIC WRITING SYSTEMS

**1.0** Old Church Slavonic manuscripts are written in two alphabets, *glagolitic* and *cyrillic*, which are functionally equivalent but visually quite different.

1.01 The glagolitic (named from glagols 'word') was probably invented by Constantine-Cyril, perhaps with the aid of his brother Methodius, in or about 863 CE. It is a unique and homogeneous graphic system, despite reminiscences from various styles of Greek, Coptic, and other alphabets. Doubtless the "Slavic Apostles" made the letter-shapes different because they wanted to create a unique system for the new language which was to be used for the praise and glory of God. This is in accordance with the Byzantine tradition allowing autonomy and equality for all of the languages of eastern Christianity, such as Georgian, Armenian, Syriac, and Coptic. Since glagolitic is the work of one man, or one man and his immediate associates, it is pointless to try to trace the gradual *development* of various letters from other symbols in other alphabets.

The cyrillic, a less esoteric alphabet that medieval Slavs attributed 1.02 to St. Cyril, consists of the Greek uncial letters supplemented by symbols for typically Slavic sounds. It surely developed in the border zones where Greek teachers were proselytizing their pagan Slavic neighbors, and possibly represents authoritative decisions made by an "inventor" whose work was to adapt, systematize, and popularize suggestions made by a number of missionaries. It is not impossible that Constantine and his helpers worked out the number and value of symbols first on the basis of the Greek graphic model, and then devised a sharply contrasting set of letters that would proclaim the non-Greek individuality of the language. The earliest cyrillic manuscripts look remarkably like ninth or tenth-century Greek manuscripts. Glagolitic pages look nothing like either Greek or Latin. Perhaps the exotic shapes of glagolitic letters were intended to produce a visual image that might help persuade hostile Frankish or Italian missionaries in Morava that Constantine's mission was not Greek. In any case, the Greek-based alphabet is the ancestor of the cyrillic alphabets used today in the Balkans and among the East Slavs.

**1.03** Glagolitic manuscripts were written in Macedonia well into the thirteenth century, and they were read and copied (in transliteration) in Rus' and the Balkans for some centuries, but the glagolitic alphabet was productive only in Croatia. There it was widely used as late as the seventeenth century, and a few priests in northern Dalmatia still use glagolitic missals to this day.

It is normal for scholars now to publish glagolitic texts in cyrillic or roman transliteration.

The two Slavic alphabets are admirably suited to the language for which they were devised. The dominant principle *one letter for one significant sound* is supplemented by arbitrary spelling conventions, some of which rely on the phonotactic structure of the language.

Like Greek and other eastern alphabets, OCS makes its letters serve as numerals as well as phonetic symbols. The numerical value of the glagolitic letters runs according to the alphabetical order, while the cyrillic numerals are patently borrowed from Greek. Letters representing Slavic phonemes that have no Greek analogue generally have no numerical value in cyrillic.<sup>1</sup>

**1.04** Neither Greek nor Latin contained certain consonants and vowels which were common in Slavonic, and the classical alphabets were unsuitable for Slavic without extensive modification. Their inadequacy is demonstrated by the oldest datable Slavic written on parchment, the Freising texts—three short confessional formulae written in Latin letters and included in a manuscript which can be dated between 998 and 1027. For instance the letter z may stand for any of five different phonemes,  $z \ s \ c \ c$  or  $\dot{z}$ , while the phoneme s may be spelled with s,  $\int$ , z, zz, or  $\int z$  (e.g.  $ze \int z \cos z$  stoko, ocima =  $o \dot{c} ima$ , zinzi = sinci or synci, zla = zla; zloueza = slovesa, sodni =- sodni,  $\int unt = solt$ , bozza = bosa, go $\int zpodi = gospodi$ ).

The inconsistencies and obscurities of the spelling of these brief texts, plus certain marked Slovene dialect features, set them apart from the language of the other old texts, and they will not be treated in this work in spite of their age.

Greek has two numerals (i.e. letters with only numerical value): stigma (5) '6', and koppa (9) '90'. See § 1.5.

Glagolitic	numerical value	Cyrillic	numerical value	transcrip- tion	approximate phonetic value: notes
· <b>h</b>	1	۵	1	a	a
Ľ	2	6	_	b	b
જ	3	6	2	v	v
8	4	r	3	g	go as in go
<u>8</u>	5	A	4	d	d
Э	6	e	5	e	e
*	7	ж	_	ž	azure
\$	8	z, (s)	6	3	adze
θσ	9	z	7	Z	Z
<b>ጞ</b> ,ሞ	10	ı (î)	10	i	i
8	20	и	8	i	i
۸ <b>۴</b>	30	(ħ)	-	-ģ	see below, §1.213
>	40	к	20	k	k
<b>ક</b> .,&	50	۸	30	1	1
<b>3</b> 8	60	M	40	m	m
P	70	N	50	n	n
Э	80	0	60	0	0
f	90	л	70	р	р
Б	_	β	100	r	r
8	200	¢	200	s	S
<b>60</b> , <b>**</b>	300	т	300	t	t
<b>3</b>	400	ov	400	u	u
ተ	500	4	500	f	f (p?) cf. §1.216
¢	-	*	9	th	thin? t? cf. §1.2161
b	600	x	600	х	Ger. ach, R хорошо
0	700	w	800	v	0
8	800?	Ψ	_	št	Eng. sht
V	900	ц	900	с	ts
÷	1000	Y	(§1.34)	č	cheese

Glagolitic	numerical value	Cyrillic	numerical value	transcrip- tion	approximate phonetic value: notes
ш	2000?	ш	_	š	sh
<b>r8</b>	-	ъ	_	Ъ	put, cf. §1.237
8 <b>P</b> 8S		٦LI	-	у	Russian ы
-8	-	Ь	-	Ь	pit, cf. §1.237
A	800?	<b>t</b> *	-	ě	pad, cf. §1.238
P	-	10	-	ju	Eng. you
-	_	IA	-	ja	ya
_	_	ĸ	-	je	ye
<b>3€,€</b>	-	* ~ ~ ~	900	ę	nasalized e
<b>3E</b>	-	Ж	-	Q	nasalized o
Э€	-	(M)	-	ję	y + nasalized e
<b>9€</b>	-	坏	-	jǫ	y + nasalized o
-	-	લા	60	ks	ax
-	_	Ψ	700	ps	apse
<b>ğ</b> .	400?	γv	400	ü	you cf. §1.233
_	-	ç	90	-	see §1.5

1.1 The preceding table lists the two sets of symbols found in the oldest texts. It does not attempt to indicate the exact make-up of the original alphabets, since unambiguous evidence is lacking. It is probable that some of the letters listed here are relatively recent.

**1.10** The theoretical ideal that one symbol = one phoneme (with the implication that one significant sound will always be represented by its own symbol) is violated because the number of phonemes is different from the number of symbols. Some sounds are not unambiguously represented. OCS spelling includes *rules of combination*: the precise significance of certain letters depends on the preceding letter. In particular, certain letters symbolize vowels if they immediately follow a consonant-letter but otherwise (that is, if they are word-initial or follow a vowel-letter) they indicate a syllable that consists of the consonant iod (*j*) plus a vowel (e.g.  $e_{i} = /j_{i}e_{i}/j$ ). The position after a consonant-letter may be called *blocked*; otherwise a vowel-letter is *unblocked*. Details are given below.

The actual spellings of the manuscripts vary widely in detail, because they were written at different times in different places, and the traditions and schooling of the scribes doubtless prescribed different norms. Thus quotations from the manuscripts will not always agree with the "same" forms which are cited in the grammatical discussion. The conventions observed in transliterating from glagolitic into cyrillic, and for rendering both in roman letters, must also be kept in mind. As a rule, cyrillic will be used here in citing cyrillic manuscripts and in sample paradigms. Roman will be used for normalized OCS words, for hypothetical reconstructed forms, and for transliteration from glagolitic manuscripts.

**1.101** Normalized forms mean the spellings which are in accord with the theoretical standard we posit for OCS, although sometimes the "normal" form may occur only rarely in the actual texts. Thus the nominative and accusative singular of 'day' are spelled variously:  $d_{\text{bh}\text{b}}$ ,  $d_{\text{eh}\text{b}}$ ,  $d_{\text{eh}\text{b}$ ,  $d_{\text{eh}\text{b}}$ ,  $d_{\text{eh}\text{b}}$ ,  $d_{\text{eh}\text{b}$ ,  $d_{\text{eh}\text{b}}$ ,  $d_{\text{eh}\text{b}$ ,  $d_{\text{eh}\text{b}}$ ,  $d_{\text{eh}\text{b}$ ,  $d_{\text{eh}\text{b}}$ ,  $d_{\text{eh}\text{b}$ ,  $d_$ 

1.102 Reconstructed forms are those which are never found in the manuscripts, but which we have deduced from all available evidence, including that of modern Slavic dialects and other languages, ancient and modern. Thus the reconstructed gen. pl. of 'day' is  $d_{bnbjb}$ , although neither OCS alphabet has any means of spelling the combination  $j_b$ . Reconstructed forms are always marked with the asterisk (\*).

It may be pointed out here that the asterisk ought to be placed before many forms which are usually not so marked. There is no inventory of all attested forms, and often grammatical discussions adduce words that in fact are not to be found in the manuscripts. For instance, one confidently gives the full declension of *kostb* 'bone' in all 16 forms. Yet in all of OCS only two cases in the singular and five in the plural are attested. Inasmuch as the other forms are found in slightly younger manuscripts, and the grammatical endings are easily established by analogy with other words of the same category, it is not deemed necessary to label these forms as hypothetical. Thus in the paradigms in this book only really hypothetical forms in the less regular categories will be marked with the \*, e.g. certain forms of the irregular verbs 'to be' and 'to give'.

**1.20** The chief evidence for the phonology is the writing, but since the writing is a complex system (with glagolitic and cyrillic variants), the individual symbols are most readily defined in terms of the phonological units we deem pertinent. The following discussion takes for granted the

11 vowels and 24 consonants listed in the tables on pp. 30-31 in the next chapter.

1.21 Seventeen consonants are represented by unambiguous letters:

п	6	4	6	M	т	A	¢	z	ц	ζ	۷	ш	ж	к	r	х
р	b	f	v	m	t	d	S	z	с	3	č	š	ž	k	g	х

The dental sonorants /n l r/ are written with "n l r", but the same letters may stand for the palatals /nj lj rj/, see \$1.22 below.

The front glide /j/ is noted or implied in several ways, see §1.24 below.

**1.211** The voiced affricate 3 was symbolized by a special glagolitic letter called  $3 \\mathbb{e}ilo$ . In many dialects the sound had lenited to z or z and scribes used "z" (named *zemlja*). The chief OCS cyrillic mss, Su and Sav, have no /3/, but z occurs in Hil, ZoF, and Vat. In transliterating from glagolitic to cyrillic it is customary to use the symbol s (which in OCS cyrillic mss functions only as a numeral, '6').

**1.212** The letter  $\psi$  ( $\psi$  in glagolitic) alternates with two-letter spellings, e.g.  $c_{B} \pm \omega t_{A} = c_{B} \pm \omega t_{A}$  'candle'. The pronunciation probably varied from region to region, but functionally we may posit  $\delta t$ , two phonetic units that together serve as a special morphophonological item (see §2.121).<sup>2</sup>

**1.213** Glagolitic  $\mathbf{M}$  "g" (called *djerv*) represents a Greek gamma (nearly always before front vowel): *igemons* = '(Roman) governor'. Its pronunciation in Slavic is unknown. It is conventionally transliterated into cyrillic by  $\mathbf{h}$ , borrowed from later Serbian mss, though SJS normalizes  $\mathbf{r}$ , with the diacritic that will be treated below in §1.31.<sup>3</sup>

1.214 The first glagolitic letter of the word XALMA 'hill' has a special "spider shape" four times in Ps Sin and once in As (but the normal "x" is used in Zogr, Mar, Euch and once in Ps Sin). No plausible phonetic or other reason for this exceptional letter has been found.

**1.215** Cyrillic occasionally writes ks and ps (consonant clusters not permitted in Slavic words) with  $\mathbf{\ddot{g}}$  and  $\boldsymbol{\psi}$ , taken from Greek  $\xi$  (ksi) and  $\psi$  (psi).

20

<sup>&</sup>lt;sup>2</sup> Neither the cyrillic nor the glagolitic shape can be plausibly explained as a compound symbol,  $\omega$  over  $\tau$ .

<sup>&</sup>lt;sup>3</sup> Evidence is too sparse and contradictory to prove or refute any hypothesis about "ψ" and "*djerv*", but as an act of faith I believe that Durnovo and Trubetzkoy were essentially correct in speculating that they were originally devised as separate symbols for the regional reflexes of earlier \**tj* and \**dj* (probably palatal stops). See §26.4. Surviving mss reflect many decades of evolution in orthography and different regional and generational attitudes toward the texts and their pronunciation.

**1.216** The letter "f" (cyr.  $\phi$ ) occurs in borrowed stems spelled in Greek with  $\phi$  (*phi*) or in Latin with "f" or "ph". There is no [f] in native Slavic words, but OCS use of the phi-letter is quite consistent in common names (e.g.  $\phi$  many "Phillip", mocn $\phi$  "Joseph"), and we may assume it was pronounced f by many Slavs.<sup>4</sup>

**1.2161** The letter "th" (cyr.  $\diamond$ ) may correspond to Greek spellings with *theta* ( $\theta$ ), but it is usually replaced by "t" in the oldest texts, e.g. Toma for  $\diamond \circ ma$  'Thomas'. The phone [ $\theta$ ] probably was not part of normal OCS pronunciation.<sup>5</sup>

**1.22** The three palatal sonorants /nj lj rj/<sup>6</sup> may be represented (1) by a diacritic on the letters "n l r" and/or (2) by the use of certain vowel-letters, but often they are not marked; readers are expected to choose the appropriate pronunciation from spellings that are ambiguous; see 1.31. The ubiquitous glide /j/ (iod) has no uniform spelling, but is usually indicated by contextual signals; see below.

**1.230** The nine oral and two nasal vowels are written by means of 12 glagolitic graphs and their combinations or 17 cyrillic graphs and their combinations. These graphs also serve to imply the glide /j/.

**1.2301** The structure of Slavic words is (C)VCVCV—a sequence of open syllables.<sup>7</sup> The symbol C in this formula stands for a limited number

<sup>&</sup>lt;sup>4</sup> Latin had a native f but spelled borrowed Greek stems with "ph".

<sup>&</sup>lt;sup>5</sup> Neither Latin nor the Romance dialects that might have been in contact with Constantine and his mission had phonetic θ, but Latin orthography distinguished "th" in borrowings. It is probable that /f/ was permissible in stem-morphemes in many Slavic 9th-c dialects, but that /θ/ was not. The scribes who wrote the OCS mss surely were familiar with several sets of spelling conventions, but we can only guess what they may have been. Facts such as that Gk φοίνιξ 'palm-tree' appears as *pinik*asə in PsSin but *finik*ə in J 12:13 when we might expect ψγμμα(uc)u (or ψγμμαu) merely underline types of variation that may or may not conceal different types of pronunciation. The Gk phrase *is ta paθi* (είς τὰ πάθη) 'for the passions' (= special lections on Good Friday) comes out *na ta fati* (see SJS sub na+μ).

<sup>&</sup>lt;sup>6</sup> Throughout this book these three units will be spelled with digraphs, to avoid the visual disparity between Croatian ň and *l*, and the inappropriate phonetic associations implied by the symbol ř.

<sup>&</sup>lt;sup>7</sup> Ninth-century Greek had six vowels: *i* (spelled H, I, EI), *ü* (sp. Y, OI), *u* (sp. OY), *e* (sp. E, Al), *o* (sp. O, Ω), *a* (sp. A). Sequences of vowels within a word were common. Word-initial vowels were marked with a diacritic (placed on the second letter of a digraph), e.g. ά-, ά-, i-, 1-, αi-, οὐ-, οὑ-. These "breathings" were merely a visual signal of the beginning of a word. Other diacritics could appear on any vowel to mark the stressed vowel. To write "accents" (*circumflex*, *acute*, *grave*),

of possible consonant clusters (cf. §2.43 below), while the V means only a single vowel. The circumstance that a vowel sound may not be immediately followed by another vowel in the same word underlies a fundamental orthographical convention that distinguishes two possible readings of certain vowel-letters. "VV"—a sequence of two vowel-letters—ordinarily means /VjV/; the glide /j/ is indicated. The syllables /ju, je, jq, je, ja/ are distinguished fairly clearly, but the important difference between /ji/ and /jb/ is not written; it must be deduced by readers from the context, see §1.24 below.

**1.2302** A vowel-letter is *blocked* if it follows a consonant-letter, *unblocked* otherwise. Unblocked position thus means syllable-initial, either within a word or at the beginning of a word.

**1.231** The letters "o" (cyrillic "o") and " $\omega$ " (called *omega*, cyr. "w") stand for the vowel *o*. Omega is rare; its principal uses are in the exclamation 'oh', as a decorative capital letter interchangeable with "o", and in the preposition or prefix  $ot(\mathfrak{z})$ —often  $\mathfrak{W}$ . It may be written in names to imitate Greek spelling.

**1.232** The cyrillic " $\gamma$ " or " $\nu$ " (called *ižica*) functions chiefly as the second element of a digraph  $\varphi \gamma$  for the tense high back rounded vowel *u*. A space-saving variant 8, with the second element above the first, is used sparingly.

In glagolitic, the complex shape of  $i \check{z} i c a$  is maintained when it stands alone, but in the digraph "ou" (cyr.  $\mathfrak{sy}$ , phonetic u) it is usually simplified and fused with the first element; the result looks like a single letter.

Ižica appears independently in a few Greek words and names which were spelled with upsilon, representing a high front rounded  $\ddot{u}$ . Whether the Slavic scribes indeed pronounced this foreign sound, they endeavored to keep the traditional spelling. Thus sümeons comeons  $\sigma_{ij}$  outed 'Symeon', sürië copputa outpica' 'Syria'. Substitution of i or u is not uncommon, comeons, copputa.

**1.233** The unit letter "ju" (cyr.  $\omega$ ) denotes the sequence /ju/. See also \$1.24.

**1.234** The glagolitic "e" denotes e if in blocked position, je if unblocked. Cyrillic " $\epsilon$ " functions the same way in Sav, but the shape  $\kappa$  is usual in Su for unblocked position.

scribes had to memorize general rules and long lists of specific words. A dieresis may be used to show that a vowel-letter is not part of a digraph (e.g.  $\alpha \tilde{i} = ai$  [two syllables], not e).

**1.2341** In cyrillic, the letter " $\pi$ " unambiguously means  $\rho$ , while " $t\pi$ " means  $j\rho$ . That is, one unit stands for the back nasal vowel, and another (albeit visually related) unit stands for *iod* plus the vowel. The unit " $\Lambda$ " (or variants  $\Delta$ ,  $\Delta$ ,  $\Lambda$ ) is  $\rho$  in blocked position (i.e. after a consonant-letter) but  $j\rho$  otherwise.<sup>8</sup>

**1.2342** In glagolitic, a letter  $\clubsuit$  (transliterated *N*) generally serves to indicate nasal quality associated with the preceding vowel-letter: "oN" and "eN" are digraphs usually transliterated "Q" and " $\clubsuit$ " and taken to be equivalent of the cyrillic letters " $\pi$ " and " $\bigstar$ ". A third digraph has a letter that occurs only before "N" and may be represented in roman as " $\circlearrowright$ N" or " $\circlearrowright$ ". Its cyrillic equivalent is the unitary " $\sh\pi$ " and it surely stands for the sequence  $j\varrho$ . (See also §1.241.)

The digraph "eN" is universal in unblocked position, but some scribes used the "N" by itself in blocked position for  $\rho$ . Thus "tN" = /t $\rho$ / 'thee [Acc sg]' ~ "toN" = /t $\rho$ / 'that [Acc sg fem]' (but "toeN" = /t $\rho$ / "that [Gen fem sg]"); cyr. TA ~ TK (TOA). Blocked "eN" is consistent in the Kiev Folia and the Sinai Psalter, and usual in the Rila Folia; there are frequent examples in As and occasional instances elsewhere.

**1.2343** Scholarly tradition dictates that "en" should be spelled in when transliterating from glagolitic, and also when writing normalized OCS in cyrillic. The shape is not found in early South Slavic, but is regular in unblocked position in the East Slavic Ostromir Gospel.<sup>9</sup> The widespread use of is in the authoritative editions of OCS glagolitic manuscripts obscures the nature of the orthographic relationships; cyrillic transliterations like time look strange to Slavists, while toim—a type of spelling unknown in canonical cyrillic OCS manuscripts—appears normal. In this book  $\epsilon$  and h will be used in normalized cyrillic.

**1.235** Glagolitic texts make no clear distinction between the three letters for i ( $\mathfrak{F}, \mathfrak{F}, \mathfrak{P}, \mathfrak{P}$ ), and cyrillic texts use their two letters ( $\mathfrak{u}$ ,  $\mathfrak{l}$ ) almost interchangeably. Readers have to take the orthographic and semantic context into account as they decide the value of each individual spelling. In

<sup>&</sup>lt;sup>9</sup> Here are typical usages of spellings for *e* and *je* in OCS manuscripts:

••	Su	Sav	Hil	Ost	KF	Zo, etc.
blocked	€ ▲	€ ▲	€A	€ A	e en	e n
unblocked	КА	€ ▲	КА	К₩	e en	e en

<sup>&</sup>lt;sup>8</sup> In Russian, ж is called юс большой, к is юс малый, while ж and к (§1.2343) are described as йотиро́ванные, *iotized*. Thus *jus* and *jusy* are used as cover terms in discussing these letters and their variants.

blocked position, the value is always *i*. In unblocked position the value is *i* if it stands for the conjunction or emphatic particle 'and; even' and either *ji* or *jb* otherwise (unless the preceding symbol is  $\mathbf{b}$  or  $\mathbf{b}$ , see next paragraph). It is tempting to speculate that the different shapes had specific contrasting values (perhaps *i*, *jb*) in Cyril's language, but attested usage is unsystematic.

In transliterating glagolitic into cyrillic, it is conventional to maintain the threefold usage of the original by means of a third letter, either i, which is not found in any real cyrillic manuscript, or the  $\ddot{i}$  of Supr (u = g, i = g, i or  $\ddot{i} = q p$ ). In normalized texts only cyrillic u and roman *i* are written. (In historical reconstructions and grammatical discussions, the combinations *ji* and *jb* may be used when pertinent, but this has no support in OCS spelling.)

**1.236** The tense high back unrounded vowel y is written with a combination of two letters,  $\mathbf{b}$  + any of the three symbols for *i* in glagolitic, and  $\mathbf{b}$  + either of the *i*-symbols in cyrillic. In cyrillic,  $\mathbf{w}$  is by far the commonest form.<sup>10</sup> A rare shape  $\mathbf{w}$  (with  $\mathbf{b}$  as first element) occurs in Su. Occasionally the two elements are joined by a line,  $\mathbf{w}$  or  $\mathbf{w}$ . In some contexts it is difficult to decide whether the  $\mathbf{b}$  + i sequences represent one syllable or two, y or yi (with an independent unblocked *i*-letter signifying yji or yj<sub>b</sub>; cf. §4.30121). This sort of ambiguity is rarely an impediment to understanding the words.

**1.237** The letters  $\mathbf{b}$  and  $\mathbf{b}$  stood for high lax vowels in most ninth-century dialects, comparable to the vowels in Eng. *put* and *pit*, respectively. These letters are called *jers*:  $\mathbf{b}$  is the back or hard jer (Russian ëp),  $\mathbf{b}$  is the front or soft jer (R ëpb). The two high lax vowels might appropriately be written  $\mathbf{u}$  and  $\mathbf{t}$  in roman. Yet because their complex development within the OCS period was striking and puzzling to early modern scholars, the jers were put in a special classification as "reduced" or "irrational" vowels symbolized conventionally by the cyrillic letters. Instead of sūnū 'sleep' and dĭnĭ 'day', Slavists usually write sānā, dbnb. The two jervowels were ubiquitous in the language of Cyril and Methodius; they occurred in all sorts of morphemes. By the time the OCS manuscripts

<sup>10</sup> The use of the *i*-letters in KF differs from that of most other scribes. The words slyšati, byti, and \*moji/\*mojb, for example, vary as follows:

Supr/Sav	слъщати	БЛІТИ	мон
ZoMarEuch	слъщати	быти	мон
KF	слъншати	въити	MOI

were written, however, these sounds had disappeared in some instances and changed quality in others. We assume that 9th-century OCS spelling used the jer-letters consistently in accordance with pronunciation. The scribes whose work constitutes our evidence attempted to spell words according to the old pronunciation, but they often erred. One of the major tasks facing students of OCS is to learn the hypothetical shape of morphemes and to recognize the orthographic variables in order to identify the jers that are posited for ideal or normalized words. See §2.51–2.53, below.

**1.238** The letter  $\check{e}$  (cyrillic  $\pounds$ , called " $\pi \intercal \flat$ " in Russian) probably represented a tense low front vowel (similar to that in English *pat*) in the dialects of the Bulgarian lands. The  $\check{e}$  of Western South Slavic regions apparently had a higher, more closed pronunciation. In glagolitic texts the letter stands where historically we expect the vowel  $\check{e}$  after a non-palatal consonant, the sequence *ja* otherwise: thus město 'place', běla*ja* 'white [nom. sg. fem.]', *ja*ko 'as' are spelled "město, bělaě, ěko" (м $\pounds \intercal \bullet$ ,  $\pounds \hbar a \pounds$ ,  $\hbar \kappa \bullet$  in cyrillic transliteration). In blocked position, then, the glagolitic letter " $\check{e}$ " represents the front vowel; in unblocked position it unambiguously represents *j* plus the tense low back vowel *a*. Cyrillic uses  $\pounds$  only in blocked position; the symbols  $\mu$  or  $\bigstar$  (surely based on Greek uncial IA) are reserved for unblocked positions.

**1.24** The glagolitic alphabet has no hint of a letter for the front glide j (called *iod*). This gap is filled (1) by two letters that stand for j plus back vowel, and (2) a convention that interprets a front-vowel letter in unblocked position as j plus the appropriate vowel (with a rule that assigns the value ja to unblocked "ě", see §1.238 above).<sup>11</sup>

Cyrillic lacks a specific symbol for *j* but has three letters with an initial element suggesting Gk *iota*:

phonemic	ju	je	jǫ	ję	ja	јі јь	
glagolitic	ü	e	öN	eN	ě	i <sub>1</sub> i <sub>2</sub> i <sub>3</sub>	unblocked
cyrillic	ю	е (к)	ж	***	IA	нії	unoisekeu

The lack of a device to differentiate  $j_b$  from ji is the most serious defect in the writing systems. It was not remedied for centuries.

<sup>&</sup>lt;sup>11</sup> A sequence of vowel letters ordinarily implies an intervening word boundary. However, in foreign stems and sometime within native words, an unblocked a, u, or q may indicate /j/: farisea Aapucea /fariseja/, bělaa вѣлла /bělaja/.

**1.241** The letters  $\ddot{u}$  and  $\ddot{\varrho}$  (cyr.  $\omega$  and m) may be used after "čšžc 3" and the groups "štžd" (cyr.  $\nu u \approx u \equiv \mu m$  [ $\mu$ ]  $m_A$ ) instead of "u" and " $\varrho$ " (cyr.  $\omega \gamma$  and m). Usage is extremely varied:  $u\omega \omega$  and  $\nu\omega$  are particularly common;  $m\omega$ ,  $\mu\omega$  ( $u\pi\omega$ ),  $m_A\omega$  less so, and  $u\omega$  and  $\Xi\omega$  are rare.  $\mu\pi$  ( $u\pi\pi$ ) is favored. These alternants surely result from scribal rules that varied with place and time. It is unlikely that the spellings give any real help in determining the pronunciation or phonetic nuances of either vowel or consonant.<sup>12</sup>

**1.30** The writing in the manuscripts contains various superscript marks. Most of them are over vowel-letters, usually in unblocked position. No spaces are provided between words; unblocked vowels are usually word-initial. A superscript mark thus may be a reader-friendly visual signal of a word beginning with a vowel, but it is linguistically redundant. It may be a reminiscence of the Greek use of "breathings" that were obligatory on word-initial vowel-letters.<sup>13</sup>

**1.31** One diacritic is used in normalized OCS over or next to the three letters "n l r", to stand for the palatal sonorants nj, lj, rj: thus  $\hat{n}$  l' r<sup>o</sup>or  $\hat{n}$   $\hat{\lambda}$   $\hat{p}$ . This usage is borrowed from the cyrillic Suprasliensis and the glagolitic Zographensis, where the presence of these palatals is usually (if not fully consistently) noted. In Supr, for instance, the theoretical forms \**jeleni* '2 deer [nom. dual]' and \**jelenji* 'deer's [poss. adj. loc. sg.]' are spelled EAENN ~ EAENN. The nominative singular (masculine) forms, which do not

<sup>&</sup>lt;sup>12</sup> Meticulous collections were assembled by scholars before the concept of the phoneme was explicit, but they interpreted the letters as scientific phonetic symbols, carefully chosen by trained linguists; the careful statistics concerning *letter* combinations have little to do with *phonemic* distinctions. Even if one grants a phonemic /ü/ and /ö/ opposed to /u/ and /o/, it is clear that the opposition was neutralized after /š ž št žd/, so that the choice of vowel-letter in these cases was immaterial. The variation in choice of *jers* after palatal consonants (e.g. vb vs. vb) is similarly insignificant.

<sup>&</sup>lt;sup>13</sup> There are supralinear markings in the Kiev Folia that suggest a significant system. Yet the marks are sparse and their distribution so inconsistent that scholars have been unable to agree whether the diacritics signify length, pitch, accent, musical notation for chanting, or perhaps more than one of these. (See Schaeken, Lunt).

Early East Slavic hymnological manuscripts with full Byzantine musical notation surely reflect a lost heritage of OCS mss of this type. It is highly probable that the Slavs chanted scriptural lections in accord with Eastern Orthodox custom. Some of the linear punctuation marks in Vat and the Ostromir Gospel hint at the system of *ekphonetic* notation known from Byzantine lectionaries; no indubitable sample of this type for any period has survived (though the Rusian *Novgorod Folia* are a likely candidate).

happen to be attested, would be similarly distinct:  $\dot{e}_{AENB} \sim \dot{e}_{AENB}$  for  $-n_b \sim -n_j b$ . In other mss, the palatal sonorants may be noted by "ju" (16) and "jq" (167): *ljubljq* ANDEARE 'I love'. For the most part, readers must deduce these three sonorants from the context.

1.32 Normalized texts often use both the diacritic and the "iotized" letters в к ы ю ы: such spelling includes combinations that are not found in any OCS manuscript.

1.  $\kappa$  has no counterpart in glagolitic and does not occur in Sav (one exception) and certain fragments. In Supr it is usual in unblocked position, but  $\epsilon$  is not infrequent.

2. Zo is fairly consistent in marking \*lj \*nj with the diacritic. Supr scarcely ever marks l before ju (AND, not  $\lambda$ ND), and normally writes simply "Q" for "jQ" after a marked consonant ( $\lambda \pi$ ,  $\lambda \pi$ , less often AND, NHT, rarely  $\lambda$ HT,  $\lambda$ HT). \*rj is less well attested because the palatal quality was lost early in South Slavic dialects. It is rarely marked in Supr, somewhat more often in Zo. Other manuscripts do not have this softness-diacritic except for a handful of examples in Mar, plus a couple of doubtful instances in Ps Sin.

In Sav,  $\Lambda m$  is regular for lja, but ht for nja (Sav never uses the diacritics). Supr writes lja and nja inconsistently, e.g.  $\hat{\Lambda}m/\hat{\Lambda}t/\Lambda m/\Lambda t$ . m never occurs after a consonant-letter other than  $\Lambda$  or N.

3. Vat does not use the diacritic. The scribes apparently pronounced sequences \*pj \*bj \*vj \*mj that are foreign to the standard language (see §2.420, below); they write these sequences:  $\Pi K$ , B K, M K,  $\Pi M$ , B J K, B K, M K,  $\Pi M$ , B K, M K,  $\Pi M$ , B K, M K,  $\Pi M$ , H K, B K, M K,  $\Pi K$ , M K, M K, M K,  $\Pi K$ , M K, M K,

1.331 The Hilandar Fragment has a special letter  $\pi$  which looks like a combination of l and the diacritic. Its usage even within the short text is not consistent, and it occurs where no historical \*j is posited. East Slavic mss of the 11th-12th centuries have besides  $\pi$  a parallel symbol  $\mathbf{rr}$  ( $\mathbf{n} + \hat{\mathbf{n}}$ ) and in some texts (e.g. the Čudov Psalter) the two letters are used with great accuracy for \*lj and \*nj. The Hil Fr would suggest that the Rus' borrowed this usage from the Balkan Slavs, though it is not impossible that it was a Rusian innovation.

**1.332** Zogr and Supr often write `over k g x followed by a front vowel, a combination of phonemes that is non-Slavic and therefore occurs only in newly borrowed words, mostly names:  $\hat{\kappa}eca\hat{\rho}b$  'caesar',  $\hat{\kappa}hhreshe$  GK  $\kappa\eta\nu\sigma\sigma\sigma$ 'poll-tax',  $\epsilon BAN\hat{r}enhe$  'Gospel'. However, in this feature, as in many others, the spelling of foreign words shows wide variation. **1.34** In all of the texts there are words which are not spelled out in full. There are abbreviations of two types. The first writes the first and last letter of the stem plus the grammatical ending, and a line is placed over the word:  $\bar{b}gb$ ,  $\bar{b}gu = bogb$ , bogu. In Greek this was originally a means of emphasis rather than a space-saving device, and it was restricted to the *nomina sacra*, the names of Divinity, such as 'God', 'Jesus', 'Spirit'. OCS early extended it to certain other words; it is particularly common with the forms of the verb *glagolati* 'speak, say':  $r\bar{\Lambda}erb$ ,  $r\bar{\Lambda}A = glagoljetb$ , glagolję, etc. The second type is more clearly to save space. A letter of those omitted on the line is written above the others, under a little roof:  $r\bar{\Lambda}B$  glava,  $e\bar{v}$  evvangelie,  $B\bar{h}$  bystb,  $\bar{h}n\bar{\delta}v$  męsopustb, etc. Such abbreviations are increasingly frequent in later manuscripts.

Some of the commonest abbreviations: аплъ = апостолъ; аплъ, англъ = анћелъ; бжи = вожии; блна, блгна = благословена; блка = владъка; гъ = господь; гнъ = господьнь; дбдъ = давъдъ; дша = доуша; ибль, йль = издранлъ; имъ, илмъ = иероусалимъ; ибъ, иб = исоусъ; крстъ = кръстъ; ибскъ = невесьскъ; оща = отъца; спси - съпаси; бъ = христъ, христосъ; цръ = цѣсарь; чкъ, члкъ = чловѣкъ.

1.4 The punctuation in OCS mss is primitive. No space is left between words. Large-size letters may occur in headings, but there is no capitalization in the modern sense. A dot on the line (.) or raised ( $\cdot$ ), or two dots (:), may be written to divide the text into phrases, but they are not systematically used, and occasionally they occur within words. Larger divisions are sometimes marked with more complex symbols (such as : or #). No OCS manuscript uses any of these devices consistently.

**1.5** Numerals are indicated by putting a line above the letter, often also by setting off the letter or letters by a dot on either side:  $\cdot \vec{n} \cdot = 30$ ,  $\cdot \vec{n} \cdot = 39$ ,  $\cdot \vec{n} \cdot = 318$ . The unit ordinarily precedes the ten in the teens:  $\cdot \vec{E}_1 \cdot (= \cdot \vec{E} \cdot) = 12$  (corresponding to d a va na desete, see §20).

The symbol s (or z) has the numeral value of 6 (cf. Gk stigma, s), even in manuscripts where it is not used as a letter. The letter z (= 3, dz) is not used as a numeral.

The symbol  $\varsigma$  (cf. Gk koppa,  $\vartheta$ ) means 90. (After about 1300  $\lor$  replaced it.)

The thousands do not happen to be attested in the OCS glagolitic mss. Later evidence hints that separate letters ( $\xi$ , etc.) had these functions. In cyrillic, however, the thousands are denoted by the units with a preceding special symbol:  $r\vec{s} = 2000$ ,  $rsr\ddot{g}r = 6363$ .

## CHAPTER TWO

# THE SOUND SYSTEM PHONEMICS AND MORPHOPHONEMICS

2.0 The two alphabets, interpreted in the light of information from modern Slavic dialects and ancient and modern related languages, represent the phonemes of OCS very well. There are both too many and too few symbols, but in practice the individual words are clearly delineated; real ambiguities are rare, and readers easily recognize the correct word from the context in which it occurs. While the Greek alphabet and orthography—how the letters are combined to represent Greek words—clearly influenced the selection of OCS symbols and their conventional combinations, Cyril and Methodius created a new system admirably adapted to the phonology of OCS.<sup>1</sup>

**2.01** It is probable that in the frontier zones where Slavs and Greeks were in close contact many Greek words had been adopted into Slavic speech. Such words, along with foreign names, inevitably were used in translating the basic Christian texts. The consonant f appears to have been accepted, surely as a markedly foreign item, in enough words and names so that the letters representing Greek  $\varphi$  (*phi*) are generally correctly used by OCS scribes. This sound was not part of the native phonological inven-

<sup>&</sup>lt;sup>1</sup> Nikolaj Trubetzkoy assumed that glagolitic represented an ideal phonemic alphabet: each letter stands for one phoneme, each phoneme has its own letter. Previous editions of this book adhered to this thesis, and apportioned considerable space to deviations from the expected effects. It is more realistic to speculate that Greek missionaries who undertook to write Slavic did so by attempting to adapt their ingrained Greek habits to the new language. It is plausible that an intelligent experimenter endowed with authority to impose new rules on a group indeed created the system actually recorded in the surviving manuscripts. I assume that Constantine was the decisive figure in the process of inventing the OCS alphabets. For most people writing is a complex process governed by rules that have been learned with great labor; to write "correctly" is a matter of remembering what is learned. Devising a new writing system is an extraordinary feat; it is no wonder that it was seen as a miraculous accomplishment granted by God to Constantine.

tory, however, and p may appear as a substitute. Thus the name 'Phillip' usually is written  $Filip = \Phi_{MAMPL}$ , although occasionally  $Pilip = \Pi_{MAMPL}$  is found. The continuant  $\theta$  is often spelled with its own letter, but much more often the stop t is substituted: Toma rather than  $\Theta_{MA}$  for 'Thomas'. The  $\theta$ sound is negligible among borrowed stems. The voiced velar continuant  $\gamma$  may possibly have been used by some Slavic speakers in Greek stems; it is more plausible to assume that the "g" of glagolitic and the "r" of cyrillic were pronounced as a voiced stop, even in names and foreign words. We assume that /f/ was a separate unit for most users of OCS, but it plays no role in the morphology.  $\theta$  was probably unknown to most; it is a problem not of phonology but of orthography.

**2.02** It is plausible that some scribes pronounced [ü] in words with Greek of or υ, e.g. glagolitic "üsopъ" (cyr. γconъ) 'hyssop, ὕσσωπος', сγκамина 'mulberry tree, συκάμινος', cf. §1.232 and fn. 7, p. 21. This too concerns spelling more than pronunciation.

2.03 OCS spelling, as we have seen, is deficient in four particulars:

- (1) the lack of a specific symbol for the glide j (§1.24);
- (2) no unambiguous way to write the vowel /y/ (§1.236);
- (3) no device for distinguishing the sequence /jь/ from /ji/ (§1.235); and
- (4) the lack of a systematic indicator to distinguish dental /n, l, r/ from the corresponding palatals /nj lj rj/. See §1.3-.331.

**2.04** On the whole, however, the OCS alphabets can be regarded as essentially phonemic. Thus *dams* 'we (will) give' differs from *dams* 'I (will) give' in that in one the phoneme /m/ is followed by the vowel /b/ and in the other the same phoneme /m/ is followed by /b/, any phonetic variation in the *m* being non-significant. There is no reason to assume that there were phonemically palatalized consonants in OCS.<sup>2</sup>

**2.11** OCS had nine oral and two nasal vowels, defined by the distinctive features back/front, high/low, tense/lax, and rounded/unrounded:

	i	У	u	ь	ъ	e	0	ě	a	£	Q
back	-	+	+	-	+	-	+	_	+	-	+
high	+	+	+	+	+	-	-	-	_	(-)	(+)
tense	+	+	+	-	-	-	-	+	+	(+)	(+)
rounded	-	-	+	-	(±)	-	+	-	(-)	-	(±)
nasal	-	-	-	-	-		-	-	-	+	+

<sup>&</sup>lt;sup>2</sup> NB: palatalized consonants are those characterized by a double articulation, palatal + something else (labial, dental, etc.). *Palatal* consonants have only a single - palatal - articulation.

Rounding is distinctive only in the high tense vowels; the  $\sigma$  was probably rounded only in central Macedonia and in Rus'. The back nasal  $\rho$  was nondistinctively rounded in central Late Common Slavic (including Rus' and the southwest [pre-Serbo-Croatian and Slovenian dialects]), but unrounded in the northwest (pre-Polish) and southeast (Macedono-Bulgarian) regions. This is implied by spellings in OCS where "o" is written instead of " $\sigma$ ", "u" and "ju" (cyr.  $\sigma$ ,  $\omega$ ) instead of " $\rho$ " and "j $\rho$ " (cyr.  $\pi$ ,  $\mu\pi$ ).

2.12 The basic consonantal inventory (without the borrowed f) is this:

	labial	dental	palatal	velar
obstruents	рb	t d s z c 3	č [ǯ] š ž	k g x
sonorants	m	nlr	nj lj rj	
glides	w		j	

The obstruents may further be subdivided into continuants (s z š ž [sibilants], x) and stops (p b t d k g vs. affricate [or delayed-release] c  $\mathfrak{z}$  č  $\mathfrak{z}$ ).

The labial continuant is classed here as a glide /w/, though we use the traditional letter v. Like other sonorants it may be preceded by either voiced or voiceless obstruents, e.g. tv, dv. It is the only sonorant that may be followed by another sonorant, in vl and vr. The sequence {ov} in some morpheme-defined positions alternates with u, while in others the {v} behaves like {p b m}. Cf. §3.7, §6.21.

**2.121** The two-phoneme sequences  $\delta t$  and  $\delta d$  function morphophonemically as palatal units.

The underlying shapes may be posited as  $\check{s}\check{c}$  and  $\check{z}\check{3}$ ; a late generative rule converts the affricates ( $\check{c}$  and  $\check{3}$ ) to stops (t and d). Cf. §2.413.

**2.122** The dental affricates  $c \ 3$  may be termed "soft" (and written  $c \ 3$ ) because they have different restrictions of combination from either the "hard" dentals t d or the "soft" palatal  $\check{c}$  (§2.51). In many dialects,  $\check{j}$  had become a continuant  $\check{z}$ . The diacritic marks are not needed for  $c \ 3$ , but  $\check{z}$  will be written to distinguish it from the more common hard dental z.

**2.123** An exceptional phoneme  $\dot{s}$  is to be posited for the anomalous pronouns  $\dot{s}$ -b 'this' and  $vb\dot{s}$ -b 'all'; see §4.201.

**2.2** Throughout this book, the letter *j* will be used for initial and intervocalic /j/ in accord with usual manuscript spellings: *jaže*, *juže*, *jąže*, *tvoja*, *tvoju*, *tvojq* ~ *eže*, *ęže*, *tvoe*, *tvoe*—whereby the "front-vowel letters" *e* and *e* in unblocked position are to be understood as representing /je/ and /ję/. Further, the ambiguous manuscript spellings will be followed

in words like *iže* and *tvoi*, to be interpreted according to the context as /jiže/ or /jьže/, /tvoji/ or /tvojь/; see §1.24.

**2.21** The texts show fluctuation between a and ja at the beginning of certain words (*aviti* ~ *javiti* 'show', agnbcb ~ jagnbcb 'lamb'). This surely reflects dialect variations rather than alternate forms in a single dialect.

**2.22** In some dialects, a *j* between two vowels was lost:  $d\check{e}jati$  'do', *raskajati* 'repent' > d $\check{e}ati$ , *raskaati*. The two vowels could contract ( $d\check{e}ti$ , *raskati*).

Other spellings reflect contractions  $\delta j \delta / y j \delta > y$ ,  $\delta j \delta / i j \delta$  and  $\delta j i / i j i > i$ . The consequences are significant for certain present tenses (see §6.5) and the compound declension (see §4.301).

**2.3** It is probable that there were prosodic features of length and stress which gave an even greater diversity to the OCS vowel system (and greater contrasts among dialects), but since the manuscripts give no information about prosody, we cannot reconstruct the particulars.

2.31 There are, however, certain spelling variations that help to identify some auxiliary morphemes as enclitics or proclitics that are prosodically bound to a major word. Prepositions surely formed an accentual unit together with the following noun; they were doubtless proclitic, as they are in nearly all modern Slavic dialects. The two demonstrative pronominal forms ta 'that' and sb 'this' (nom. acc. sg. masc.) and \*jb 'him' (acc. sg. masc.) seem in certain cases to have functioned as enclitics, forming an accentual unit with the noun or verb they followed. Thus va tamě 'in the dark', sa manojq 'with me', raba ta 'that slave', vidita i 'sees him' were accentual units.

2.4 The following general restrictions on the occurrence and combinations of phonemes obtain.

**2.411** The vowels y,  $\sigma$ , b, e and e do not occur in word-initial position. The vowel  $\check{e}$  in the root  $\check{e}d$ - 'eat' may have been allowed word-initially in some dialects; glagolitic spelling with " $\check{e}d$ -" is perhaps ambiguous. Cyrillic manuscripts usually write ma-(but prefixed osta-, ctmta-, cf. §3.3101).

**2.4111** It is possible that some loan-words may have had initial /e/, but adaptation to the native pattern with /je/ is probable for many. The use of cyrillic  $\epsilon$  vs.  $\kappa$  is not systematic (§1.234) and therefore provides no sure evidence.

**2.412** After k g x only  $y u \ge o \rho a$  may stand; that is, a front vowel (*i*  $\bowtie e \rho e$ ) may not follow a velar consonant.

2.4121 This rule is frequently violated by words of demonstrably recent foreign origin, e.g. kits 'κῆτος, whale', kesarjs 'καῖσαρ, Roman emperor', arxierei ἀρχιερεύς 'archpriest',

xeruvima  $\chi \epsilon \rho o u \beta \mu$  'cherub'. In Greek, these consonants were non-distinctively palatal [K  $\chi$ ]; some Slavs possibly imitated this pronunciation. Words with a Greek gamma before front vowel are spelled sometimes with g, sometimes g (cf. §2.4121): angela ~ angela 'angel'. Doubtless the pronunciation of such words varied in different areas and traditions. What is important is that foreign stems were ordinarily provided with derivational suffixes and inflectional desinences that fitted them into the overall-system of OCS.

**2.413** After palatals ( $\check{s} \check{z} \check{c} \check{s} t \check{z} d nj lj rj$  and j) only  $i e \varrho u a \varrho$  may stand (NOT  $o y \flat \check{e}$ ). Note that the groups  $\check{s} t$  and  $\check{z} d$  are treated as units whose behavior differs from that of t and d, §2.121. (And keep in mind that the digraphs nj lj rj represent unit phonemes, §1.22) Note also that the sequence  $/j\flat/$  is written i (so  $tvoj\flat$  is spelled tvoi, cf. §1.24). The consonants listed here are traditionally called "soft".

**2.414** After c and 3, y o and  $\mathfrak{s}$  cannot stand (but i u e e q e a b may).

**2.415** Any vowel may follow the other consonants (labials, p b v m; and dentals, t d s z n l r).

**2.51** These phonotactic restrictions define four groups of consonants: the velars, the "soft consonants", the pair  $c_3$ , and what may be termed "neutral consonants". Here is a summary in tabular form (*plus* [+] means that a vowel may occur in the position indicated, *minus* [-] that it may not):

	ě	у	ъ	Ь	0	e	i	ę	u	a	ð
1. initial	(-)	-	-	-	+	-	+	-	+	+	+
2. after $k g x$	-	+	+	-	+	-	-	-	+	+	+
3. after š ž č št žd nj lj rj j	-	-	-	+	-	+	+	+	+	+	+
4. after c 3	+	-	-	+	-	+	+	+	+	+	+
5. after $p b t d s z v m n l r$	+	+	+	+	+	+	+	+	+	+	+

This table shows that three vowels  $(u \ a \ q)$  have no restrictions. The contrasts in the first four positions show that  $y \ge o$  are in complementary distribution with  $i \ge e$ . The front nasal vowel e is not so directly opposed to the back nasal q, and the low front tense  $\check{e}$  differs even more from the low back tense a.

**2.52** The syllabic structure is simple: there is a single vowel which may be preceded by a maximum of four consonants (CCCC)V. All syllables are thus open, and any succession of vowels is automatically to be interpreted as a succession of syllables. This occurs when a vowel-final prefix joins a stem-initial vowel (e.g. *naučiti*, voorqžiti), and apparently in the imperfect tense suffix  $\check{e}a/aa$  (cf. §9.1). It seems to be a relatively new development in some desinences of the compound adjectives (aa < aje, uu < uje).

**2.521** Often the expected two-unit groups plj, blj, vlj, mlj are spelled without any l-letter, implying a Bulgaro-Macedonian substitution of /j/ in place of the palatal sonorant /lj/, and therefore groups pj, bj, vj, mj. Thus, e.g., kaplja,  $ljublj\rho$ , avljati, zemlja changed to kapja,  $ljubj\rho$ , avjati, zemja. Most usually this is shown by **b** in place of the l-letter, but before front-vowel letters the **b** may be omitted: e.g. zemba, zemba, zemu, avdbaba, npuctabenue. See also §1.32 (3).

Other dialectal developments affecting the validity of the rules stated here are isolated and unimportant.

**2.522** Consonant clusters are limited to sequences that can be described in a general formula: sibilant + (non-continuant or x) + (v) + (sonorant). *Moreover*: No doubled consonants occur.

And: An initial sibilant must be voiced or voiceless according to the voicing of the following obstruent and palatal before a palatal consonant.

		obst	ruent				sor	noran	t	
s	р	t	k	с	x	v	r	l	m	n
Z	b	d	g	3						
š		č	(š	t)			rj	lj		nj
ž		ž	(ž	d)						

Not all possible combinations occur (see §3.311). Many clusters (printed in italics in the following list) are attested only within a word.

sp	st	sk sc	sx	sv	sr	sl	sm	sn	pr	pl	plj	pn	tv	tr	tn		kr	kl	klj kn		cv
zb	zd	zg	Z3	zv	zr	zi	zm	zn	br	bl	blj	bn	dv	dr	dn	dm	gv	gr	gl gn	gnj	3V
				xv	xr	xl		хn	vr	vl											
spi	· spl	splj stv	str	skr	· skv	skl	sklj	skv	r	scv	sxv	čr	čl		št	šv	šlj	šnj	štvlj	štr	
zbi	· zbl	zblj zdu	, zdr	zgr	zgv	zgl						žr	ž1	žlj	žd			žnj			ždrj

**2.5221** The glide /j/ occurs only as the initial consonant in its syllable. In underlying structure, however, j may follow a consonant or cluster; it may also appear in an intermediate stage of generation. See §3.6.

**2.523** Only one exception to the formula exists: adjacent stops appear in the adverbial suffix -gda (e.g. kogda 'when', togda 'then', cf. §4.812).

**2.53** Borrowed stems with deviant consonant clusters probably inserted jers ( $\mathfrak{b}$ ,  $\mathfrak{b}$ ) to break the clusters, e.g. *pssalsms* 'psalm', *Avsgusts* 'Augustus, A<sup> $\circ$ </sup>yo<sup> $\circ$ </sup>στos' (pron. [avy<sup> $\circ$ </sup>ustos]), *Pavbls* 'Paul, Πα<sup> $\circ$ </sup>λοs' (pron. [pávlos]). The spelling is chaotic and many details are difficult to interpret. See also §2.65.

**2.6** The two high lax vowels—the *jers*— $\mathbf{b}$  and  $\mathbf{b}$  (cf.§1.237), are subject to special processes.

**2.61** Before /j/, the tense/lax opposition is neutralized in the high nonrounded vowels: /b/ is in free alternation with /i/, and /b/ with /y/. Thus \*ljudbje and \*ljudije, \*novbjb and \*novyjb are equivalent; scribes employ alternate spellings like  $\wedge \wedge Abe$  and  $\wedge \wedge AHe$ ,  $\wedge \wedge Bbh$  and  $\wedge \wedge Bbh$ . The vowels in this position will be called *tense jers* (although some scholars prefer the term *reduced y/i*).

The b-letter is used frequently but inconsistently, while the b-letter is rare. This is partly due to the fact that tense /b/ occurs in many more words and categories than tense /b/.

This neutralization applies within a phonological word (or accentual unit, cf. §2.31), so that the jer of a preposition is affected by the /j/ of the following word, and a final jer is affected by an enclitic \*jb; e.g. vs istinq - vy istinq 'in truth', osqdets i - osqdety i 'they will condemn him', prědamb i - prědami i 'I will betray him'.

**2.620** The invention of the jer-letters and their use in the oldest manuscripts guarantee that the two high lax vowels were distinct phonemes in the language of the 9th century. They were surely characteristic of all Slavic dialects at the time. Yet they soon began to change; in certain positions (called *weak*) within a phonological word they simply disappeared, while in other positions (called *strong*) they were pronounced with lower articulation, creating new and regionally varied vowel systems. The early stages of this complex process, dubbed the *jer-shift*, followed the same rules in all of Slavdom, but the details and the eventual results and phonemic accommodations differ from dialect to dialect. OCS attests a generalized type of southeastern Late Common Slavic, but the actual use of the jer-letters in the surviving manuscripts shows that the jer-shift was far advanced or complete at the time the scribes wrote those mss.<sup>3</sup>

Phonologically, the high lax vowels were lost; at a more abstract level, however, they survived as vowel/zero morphophonemes: under specific phonotactic conditions they are vowel phonemes, while under other conditions they are not pronounced—they are phonetic nulls.

**2.621** A jer may be either *strong* or *weak*. A jer is weak in a syllable followed directly by a syllable with a non-jer vowel (i.e. at the end of a

<sup>&</sup>lt;sup>3</sup> The change seems to have started in the southwest (perhaps in Slovene, Czech, or Croatian regions) during the tenth century, spread throughout the Slavic world, reaching Kiev Rus' not earlier than the beginning of the twelfth century, and Novgorod and the northeast somewhat later. Many of the 11th and 12th-century mss of Rus' origin use the jer-letters far more "correctly" than do the OCS texts.

word not followed by an enclitic). A jer is strong only in a syllable directly before a syllable with a weak jer. For example, with weak jers in *italics* and strong jers in **bold-face**: dbnb 'day', dbne (gen. sg.), tbmbnb 'dark (nom. sg. masc)', tbmbno (nom. sg. neut.), sbnb 'sleep', sbnbnb 'of sleep (adj. nom. sg. masc.)', sbnbna (nom. sg. fem.), \*mojb (written moi), \*ljudbjb (written ljudbi); vb tbmě 'in the dark', rabb tb 'that slave', sb monojq 'with me' (cf. §2.31) (Detailed examples are given in §2.65, below.)

In the groups spelled *consonant* + l or r + *consonant*, the jer, in the great majority of cases, is neither strong nor weak, but *neutral*. See §2.63.

**2.622** As the jer-shift progressed, the weak jers ceased to be pronounced. A strong jer was replaced by a non-high vowel. In central Macedonia,  $\mathbf{b} > e, \mathbf{b} > o$ : den, (\*tmen), temno, son, (\*snen, \*sonna), moj, ljudej (written ljudei), vo tmě, rabo t, so mnojǫ. In most Bulgarian dialects  $\mathbf{b}$  became an independent vowel /ə/, still written with the jer-letter, while  $\mathbf{b}$  also became /ə/ in roots, but /e/ in suffixes. In some central and eastern Bulgarian dialects  $\mathbf{b} > \mathbf{a}$  in all positions. In all of Serbo-Croatian and Slovene, the two jers fell together in a single vowel  $\mathbf{a}$ .<sup>4</sup>

**2.623** In the manuscripts these changes are not clearly shown, largely because of the force of written tradition, combined with habits of special—and, surely, often artificial—church pronunciation of sacred texts.<sup>5</sup> The influence of different regional and historical dialects on the texts as they were copied time after time introduced all sorts of modifications, and scribes continually made mistakes. Writing weak jers became a matter of arbitrary rule or random choice. Spellings that deviate fairly systematically from etymological expectations seem to hint at local phonetic usage during stages when weak jers could be pronounced or omitted. In Zogr, for example,  $\mathbf{x}$  usually is written before a syllable with a back vowel,  $\mathbf{b}$  before a syllable with a front vowel. Detailed studies of the work of all the major OCS scribes have failed to prove that orthographical usage reflects the pronunciation, but some general principles can be detected. Jers that

<sup>&</sup>lt;sup>4</sup> In Serbian mss it was written ь (changing to *a* in the 14th century).

<sup>&</sup>lt;sup>5</sup> The maintenance of a vowel even in weak position was supported in some communities by the habit of singing or chanting many liturgical texts to old tunes which were composed to match the musical structure to the vowels (including jers) of archaic texts. See for example the 16th-17th century Russian hymns in E. Koschmieder in *Die ältesten Novgoroder Hirmologien Fragmente* (Abhandlungen d. Bayerisch. Akad., Phil. hist. Kl., NF 53), Munich, 1952, where e/o consistently appears for old jers, e.g. Bece MNPO for VBS6 mirð.

were always in weak position, especially in an initial syllable, are almost regularly omitted by some scribes (e.g. čto, kto, mnogo for čsto, ksto, msnogo). Many scribes consistently write  $\mathbf{b}$  instead of  $\mathbf{b}$  after  $\mathbf{s}$  and  $\mathbf{z}$ . Word-final jers were retained for centuries as a visual signal of the end of the word (since space was not ordinarily left between words).

2.624 In spelling strong jers, the manuscripts also differ: KF writes them correctly—viz. where we expect them on the basis of comparative evidence. They are generally correct in Supr and Sav, perhaps because both jer-letters could be pronounced as /a/. The other mss all have some examples of *e* for strong *b*; in As *e* occurs in nearly 85% of all possible cases, while in Euch it is almost without exception. All the mss have a few instances of *o* for strong *b*; Euch has it in about 30% of the possible cases. It is possible that such spellings in some words reflect conventions from an authoritative tradition. Yet "errors" also occur: thus s**b**tbnik**b** 'centurion' appears as s**b**tbnik**b** as well as sotbnik**b**. Sav consistently omits the strong jer in ANB 'day' and **BCB** 'all (NA masc. sg.)'.

All in all, the use of the jer-letters demonstrates only that scribes felt that these symbols were part of correct spelling.

**2.6241** The phonological calculation of strong jers originally started with the end of a word:  $t_{bmbns}$  but  $t_{bmbna}$ . Examples of the type \**tmen temna* demonstrate the expected development of vowel versus zero in different morphological forms of a single stem. By the end of the OCS period, however the calculation started from the beginning of a word: the recursive rule was *a jer is strong if a jer is in the next syllable*. Therefore underlying

{tbmbnb} > tbmbnb > tbmbnb > tbmbnb > /temen/ {tbmbno} > tbmbno > tbmbno > /temno/ Though the principles of such developments are clearly illustrated in OCS mss, the conservative and inconsistent spellings not only of OCS but of most immediately post-OCS texts obscure the details.

**2.625** The sequence \*jb cannot be expressed in either OCS alphabet, but is written with an i-letter (§1.24). Nonetheless, the presence of the strong or weak jer is sometimes apparent. The adjective /dostojbnæ/ 'wor-thy' is spelled dostoints or (after the strong jer has lowered) dostoents (/dostojen/).

The sequences  $*_{bjb}$  and  $*_{bjb}$  contain tense jers (§2.61) that are strong. The gen. pl. \*ljud**b**jb may be spelled ljudbi or ljudii or (with lowered strong jer) ljudei (for /ljudej/). A tense jer that is weak serves to define a jer in the preceding syllable as strong: tr $\mathbf{b}$ st $\mathbf{b}$ j $\mathbf{q}$  (written also tr $\mathbf{b}$ stij $\mathbf{q}$ ) > trestij $\mathbf{q}$ . See §2.65 for some other examples.

2.626 A great deal has been written about these variant spellings and innumerable and ingenious theories built up, but the fact remains that we are dealing with *spellings* and can only guess at the sounds they represented and speculate about the phonological system or systems. It must be emphasized that NOT ONE SINGLE OLD CHURCH SLAVONIC MANUSCRIPT has the jers written in all cases where the grammars (including this one) posit them. Our chief guide for reconstruction is the East Slavic usage of some of the oldest mss (11th and early 12th century), where the jers seem to have been written according to the older Slavonic tradition. It is even probable that many Rusian scribes, guided by their native speech, corrected the "errors" they found in the South Slavic manuscripts they were copying.

**2.63** Spellings in the groups involving r or l + jer between consonants (traditionally expressed by the formulas trut/trut/tlut/tlut, where t represents any consonant) pose certain problems of interpretation. OCS scribes preferred  $\mathfrak{T}$  in these words, but Sav has only  $\mathfrak{b}$ . We follow etymology in normalizing such words: hence, somrubt 'death', krovb 'blood', slbza 'tear', slonbce 'sun'.

**2.631** Spellings of such words vary in ways that, with the help of evidence from later dialects and related languages, suggest two groups: in one the jer represents a vowel that originally followed the liquid (krъvъ, slъza), while in the other the jer-vowel originally preceded the r/l (\*sъmьтъ, \*sъlnъce). In spelling the first type, some mss (esp. Zo) distinguish the two jer-letters well. Moreover, the jers in these words may when weak serve to make a preceding jer strong (vъ > vo in vo krъvi as in vo vьsěxъ), or if themselves strong may be replaced by o/e: krovь, slezъ [gen. pl.]). In short, these are normal /ь ъ/.

Words whose b or  $\overline{b}$  preceded the liquid are conventionally written with  $\overline{b}$  (but always with b in Sav). We may term this kind of written jer *neutral.*<sup>6</sup> There are no cases indicating the development of such a jer into

<sup>&</sup>lt;sup>6</sup> Early Rus' mss regularly write neutral jers before the liquid: съмъртъ, съмърце, but кръвъ, слъда. Modern Russian equivalents normally indicate the order of phonemes for the etymology: R *er/or* indicates \**br/or* and R *re/ro* shows \**rb/lo*; e.g. smert' < *sombrtb*, krest < *krusto*. That is, if the vowel precedes the *r* in Russian, the jer preceded it in (the early Rusian dialect of) Late ComSI. R/U *le/lo* reflect LCoS *lb/lo* (sleza < *sluza*, plot' < *plotb*), but \**bl* became *ol* in early Rus, and both yielded R *ol* (polnyj - OCS *plunb*).

another vowel according to the rules for strong jers, nor does a jer preceding such a syllable act as if in strong position. It is highly possible that the 9th-century dialect of the original translations still distinguished two separate types of syllable, while the OCS evidence reflects a substantially modified later system. South Slavic developed syllabic liquids in both types. Forms like *smrt* and *slnce* may have appeared slightly earlier, while alternations such as *krov* ~ *krvi* (< krъvь krъvi) and *krest* ~ *krsta* (< krъsta 'cross' krъsta) developed and were eliminated in favor of uniform stems with a syllabic sonorant (*krv krvi, krst krsta*). In any case, arbitrary spellings that position jer-letters after the liquid-letters persisted for centuries.

2.64 The loss of the weak jers meant that the scribes pronounced consonant groups in many places where the older mss had a jer-letter. The ideal visual image included many "silent" letters that separated consonants, and scribes occasionally wrote a jer where it did not belong. For example, when *bbrati* 'take' lost its jer, it became identical with *brati* 'fight'; both are subsequently written *brati* or *bbrati* or *bbrati*. A scribe who pronounced *umreši* 'you will die' but knew it should be spelled *umbreši*, could easily add a spurious jer to the infinitive, writing *umbrěti* or *umbrěti* instead of etymological *umrěti*.

In foreign words and names jers are often added between consonantletters; we cannot be sure just when this is meaningful and when not.

**2.65** Here are some examples of typical spellings of words containing jers. The normalized form (whether actually attested or not) is given first, in roman letters, then the attested variants in cyrillic. Keep in mind these basic general rules—For strong position: a jer-letter is expected (though it may be the "wrong" one; or,  $\mathbf{b}$  is replaced by  $\mathbf{e}$  (often), while  $\mathbf{b}$  is replaced by  $\mathbf{e}$  (rarely). For weak position: non-final letters may be omitted, with or without a supralinear mark; the choice between jer-letters is essentially random. In the work of individual scribes spelling rules (or at least tendencies) may be detected (see §2.523 above):

čыо: чъто, чъто, что. dbnb (nom. sg.): дьнь, дьнъ, день, днь. dbnö (gen. pl.): дьнъ, денъ. \*dbnbjb (gen. pl.): дьньи, дьнии, днеи. dbnbe (nom. pl.): дьнье, денье, дение. otbcb: отьць, отьцъ, отець, отецъ. pravbdbn3: правьдьнъ, правьденъ, правьдей (праведенъ §2.5241). pravbdbn3: правьдьна, праведъна, праведъна. tbmbno: тьмьно, тьмно, темьно. podobbstvbju: подобествью. \*jbm3: имъ, емъ. vbzbm3: възьмъ, въземъ, въземъ. \*božbjb: божьи, божии, божеи. cěsarbstvbe: црествье. prišbstvbe: пришествие. črbnbcb: чръньць, чрънець, чрънецъ. skrbžbt3: скръжьтъ, скръжетъ. skrbžbtanbe: скръжътание. skrbžbšet3: скръжьштетъ, скрежьштетъ. ото skvrылыто pomyšljensi: отъ сквръненъ помъшленен. koto: къто, кто, кто. vo njb въ йь, во нь. umbroši: оумеръши. umbrošb: оумьрьшъ, оумерошъ (§2.6241). monožustvo: мъножъство, мьножъство, множство. ljubovo: любовь. ljubovi: любьви. sotuniko: сътьникъ, сотьникъ, сътъникъ. sozudanue: съзъданье, создание. \*svętoju: сватъи, сватъи, сватои. prědamu \*ju 'I will betray him': пръдамъ и, предамеи, пръдами 1.

Certain borrowings were adjusted to the native pattern; thus in Ps Sin the word 'psalm' is usually *psaloms* or *pssaloms* (but loc. pl. *pssalsměxs*), implying older pssalsms. On the whole, the presence or absence of a jerletter within a consonant cluster in foreign words—particularly names is of no linguistic significance.

**2.70** Occasionally the symbols for the nasal vowels are replaced in the mss by some other vowel-letters, or they stand instead of some other letter. The invention of *jusy*—the letters for nasal vowels—and their generally correct use by OCS scribes (despite the variation in graphs) implies that the two nasal vowels were distinct phonemes in the language of the 9th century. The deviant spellings suggest both regional and historical dialect differences during the OCS period.<sup>7</sup>

**2.701** Most modern Slavic dialects lack nasal vowels, except for Polish. In a small northwestern marginal zone of Slovenian (largely in Austria), and the southern periphery of Macedonian and Bulgarian (in Albania and northern Greece) systematic traces of OCS nasality remain. Though much is unclear about the course of divergent developments, it is certain that the old distribution of \*e and \*e was being modified during the 11th and 12th centuries. In Serbo-Croatian well before 1200 \*e was replaced by e, and \*e by u; northern Macedonian dialects shared these shifts.

2.71 Certain morphemes had nasal and non-nasal variants:

a. the roots \*mud/\*mqd (muditi/mqditi 'be late', mudьпъ/mqdьпъ 'tardy'),\*nud/\*nqd (nuditi/nqditi 'to force', nužda/nqžda 'force'), and \*gnus/ \*gnqs (gnusiti/gnqsiti 'be disgusted', gnusьпъ/gnqsьпъ 'disgusting'). b. the stem \*su-mьn/\*sq-mьn (sumьněti/sqmьněti 'to doubt, suspect').

40

c. the verb poměnqti/pomenqti 'remember'.

<sup>&</sup>lt;sup>7</sup> The two glagolitic symbols "eN" and "oN" for e and q may mistakenly be written without the second element "N". It is not easy to distinguish this kind of scribal error from a more purposeful scribal choice based on a copyist's own pronunciation: the historical change \*e > e took place in many South Slavic dialects; oultimately from \*q is found in certain localities in Macedonia and Bulgaria, but it is improbable that it was present as early as the 12th century.

These surely represent individual adjustments in contexts with nasal consonants.

d. Variation in the borrowed stem sobota/sobota 'Sabbath, Saturday' is probably based on different Greek pronunciations. Sobota is usual; sobota is the only form in Mar, while both forms are in PSSin  $(3 \ q, 4 \ o)$  and As  $(88+\% \ q)$ . Vat has one creation to perhaps 10 legible examples of cost. This distribution probably reflects traditional spelling with "q" versus a local authorization of "o" that was attenuated in later copies.

**2.72** Variant spellings that point to individual dialects provide conflicting information. Thus Mar has some confusion between q and u or jq and ju (e.g. Mar ljublju, for ljubljq 'I love', dat. sg. nemq, for nemu 'to him') that imply a scribe from northern Macedonia or Serbia, regions where u < \*q. But it also has instances of  $\check{e}$  (gr $\check{e}$ di, for gr $\check{e}$ di 'come [imv. sg.]!'), suggesting the pronunciation of southern Macedonia. Perhaps the scribe of one of the copies of the model used for Mar introduced these southern deviations, while the scribe of Mar itself had the more northerly type of dialect. In Ps, o and q are confused: potb, for pqtb 'road'; sqbojo, for sobojq 'self (inst. sg.)'. For e there are several cases where e is written (e.g. ezyci, for qzyci 'tongues'); the opposite is extremely rare: imqni. The number of examples is small, however, and "correct" usage of these letters is one of the criteria for the antiquity of a manuscript.

The Ostromir Gospel of 1056-57 is excluded from the canon of OCS (\$0.32) chiefly because the scribes clearly used **m** and **m** as equivalents of  $\bullet_{Y}$  and **w**, **m** = **a** in certain positions, and **m** = **m**, although in fact the number of "errors" is minimal. In phonological terms, \* $\rho$  merged with \*u, while \* $\rho$  remained a separate unit, redefined as a low front vowel / $\ddot{a}$ / that was distinct from both /e/ (< ComSl \*e) and / $\ddot{e}$ / (< ComSl \* $\check{e}$ ).

2.73 In most Macedonian and Bulgarian regions the front/back distinction between /e/ and /o/ was preserved when the vowel followed a labial or dental consonant (the neutral consonants, §2.51), but was blurred or lost after palatals and  $/c_3/$ . In spelling, the letter "jo" (m) may be replaced by "je" ( $a \ge a$ ), while after spelled "č š ž št žd" or the "n 1 r" that represent /lj nj rj/ "o" may be written "e" ( $a \ge a$ ). The glagolitic manuscripts all have examples, but they are notable only in the glagolitic Assemanianus. Thus As *pomažęts* (for  $-q/t_3$ , 'they annoint' Mk 16:2), *lzžęšte* (for  $-q/t_2$  (is potentially serious, since it involves meaning, but perhaps the scribe intended a plural; this sort of minor textual alteration is not uncommon.<sup>8</sup>

<sup>&</sup>lt;sup>8</sup> Mar J 21:6 has acc. sg. mrěžo 'net', while Zo As have pl. mrěže; at the end of the verse, all three have ne možaaxo privlěšti *ee* 'they were not able to draw *it* in' (with gen. sg. of negation, §23.22)—this fits the singular of Mar (and the Greek) and marks Zo and As as "incorrect". More serious is Mt 19:9 in Vat, творитъ А прѣлювъъ творитъ, 'makes *them* commit adultery' instead of 'her' (see table on p. 63). This sort of confusion becomes more and more frequent in post-OCS manuscripts.

THE SOUND SYSTEM

**2.8** The letters for 3 are absent from some manuscripts and written inconsistently in others. It is apparent that very early 3 was replaced by z' in most of Slavdom (although 3 has survived in Macedonian and Bulgarian dialects). The "soft" z' (§1.211) is not distinguished orthographically from a z of other origins.

### **MORPHOPHONEMICS**

**3.0** The smallest meaningful unit is a *morpheme*. Morphemes are thus the semantic building-blocks of words and sentences. A basic morpheme-shape is posited as a linguistic item in the lexicon. It consists of one or more phonemes, including zero (represented by  $\emptyset$ ).

OCS words belong to two types; simple and complex. *Simple* words are at the same time morphemes and are invariable: most prepositions and conjunctions and some adverbs are of this type. *Complex* words have a stem and an inflectional suffix. A stem must contain a *root* morpheme and it may include one or more *affix*-morphemes—prefixes and suffixes.

Thus the word *bezmilostiv*<sup>5</sup> 'merciless (nom. sg. masc.)' consists of a prefix, a root, a noun-formant suffix, an adjective-formant suffix (which together constitute a meaningful lexical stem) and a case-gender-number inflectional suffix: bez-*mil*-ost-iv- $\mathbf{b}$ . In this book curly brackets will be used to signify underlying morphemes or morpheme-shapes: e.g. {bez-mil-ost-iv- $\mathbf{b}$ }. In this case, the theoretical underlying sequence is unchanged in the surface structure, though the syllable division in the pronounced word differs: *be.zmi.lo.sti.vo.*<sup>9</sup> The word *istekqto* '(they) will run out' consists of a prefix, a root, a zero verb-forming suffix, a present tense-marker that is coordinated with a person-marker to indicate plural: {iz-tek- $\emptyset$ +q-t $\mathbf{b}$ }. Here the underlying structure has been modified: the surface phonology has [s] for {z} and of course nothing at all for { $\emptyset$ }; five morphemes combine in four syllables, *i.ste.kq.to*.

**3.1** Many surface morphemes have more than one shape, depending on the phonological environment. The environment differs chiefly because of different derivational or inflectional affixes that are used to create words.

Root-morphemes normally keep the underlying vowel in all inflected forms; different vowels usually indicate different lexical entries. The final

<sup>&</sup>lt;sup>9</sup> Complex stems with two roots are called compounds, e.g. {mъnog-o+mil-ostiv-ъ} 'greatly merciful'.

consonant(s) of a root may change to adapt to various suffixes. The root rek, for example, appears also as  $re\check{c}$ ,  $r\mathit{bc}$ -, and  $r\check{e}$ - in different conjugational forms of the somewhat anomalous verb {rek- $\emptyset$ +} 'to say' (where zero is a verb-forming suffix with null phonological content): rekots 'they will say', reče 'he said',  $r\mathit{bci}$  'say!', režva 'I said' (whereby the last two shapes are somewhat irregular and need to be noted in the lexical entry for this verb). The variants rok, roc-, roč-, ric-, rič- and rěk- belong in derived words, such as prorok 'prophet' pl. proroci, voc. proroče!, prěrěkati 'contradict', naricati 'to name', 3 sg. naričet .

Prefix-morphemes, however, appear in different shapes as a morphemefinal consonant adapts to the root-initial consonant to which it is attached to make a new lexical word. Thus, for example, the prefix iz has alternate forms *izd-*, *is-* and *i-* (*iz*idqtb 'they will go out', *izd*rekqtb 'they will express', *is*pbjqtb 'they will drink', *is*bxnqtb 'they will dry up').

Suffix-morphemes often have alternating shapes used with different stem-final consonants. The alternating vowels are given in the lexicon. Thus, for example, the neuter nominative singular suffix for the twofold nominal declension is  $\{0/e\}$ , that is, -o in město 'place' but -e in lice 'face'.

**3.2** The permitted structure of OCS words in terms of syllables, consonant groups, and consonant+vowel sequences has been described above.

The underlying morphemic structure may violate some of the surface prescriptions; the differences are eliminated by generative rules.

**3.21** Root-morphemes fit a formula ((((C)V)C)V)C. Thus all roots must end in a consonant. Only the pronominal roots *t*- 'that', \**s*- 'this' (see \$4.201), *k*- 'who (interrogative)', and \**j*- (\$4.25) 'who, which (relative)' consist of a single consonant. Monosyllabic roots without initial consonant are few but include pronouns (*ov*- 'that yonder'), nouns (*ux*- 'ear', *ogl*- 'coal'), and verbs (*or*- 'plow'). The great majority of common roots have the shape CVC—where C may be a cluster—and many have two syllables, CVCVC.

**3.22** Borrowed stems admit more varied structures, but behave like roots. If the stem in Greek ended in a vowel, the Slavic stem usually has *j*: marij-a Mapía [maria], \*isajij-a 'loatas [isa'ias], \*ijudej-b or \*ijuděj-b 'lou $\delta a$  [iud'eos]. See §4.12.

**3.23** Except for borrowed stems, the consonant+vowel sequences within roots conform to §2.51.

3.24 The underlying morpheme { bm } 'take' becomes { jbm } unless pre-

ceded by a consonant: vъzьmъ, but \*jьmъ, \*pojьmъ 'having taken (nom. sg. masc.)'. This contrast is obscured by the spelling, *imъ*, *poimъ*.

**3.25** The morpheme  $\{v \ge p\}$  'cry out' lacks the initial  $\{v\}$  when  $\{v \ge z\}$  is prefixed: imperfective  $v \ge p$  is perfective  $v \ge z$ .

**3.31** Changes in consonants take place at morpheme boundaries; they serve to adjust underlying sequences to the cluster-formula in §2.522. As a rule, if two adjacent elements are incompatible, the first adjusts to the second.

Adjustments at the prefix + stem boundary sometimes differ from those at the stem + desinence boundary.

**3.3101** A few verbal roots prefix an n (the "epenthetic n") when combined with the prefixes so and vo: \*bm-/e-/emlj- 'take'; id- 'go'; ed- 'eat'. Thus sonemore, soneti, sonemljots, sonedets, vonsent, voneti, vonemljots, vonidots.

**3.3103** The stem of the third-person pronoun is j-, nom. sg. masc.  $\{j-\mathbf{b}\}$  spelled i ( $\{1.24\}$ ): gen. sg. masc.  $\{jego\}$ , spelled *ego*. In forms governed by a preposition, j is replaced by nj: ote njego 'from him'.

**3.311** The prepositions/prefixes *bez*, *v*sz, and *iz*, and the prefix *raz* adapt to the voicing and palatal specifications of the cluster:

- z is lost before s, z, or š: e.g. raz+slabiti > raslabiti 'weaken'; vъz+zъvati > vъzъvati 'call'; iz+šьdъ > išьdъ 'having gone out'.
- $z + \tilde{z} > \tilde{z}d$ : e.g. iz+ženotъ > iždeno ' I'll drive out'.
- z > s before p t k x: e.g. iziti 'to exit' but ispadati 'to fall out', istočiti 'pour out', iskopati 'dig out', isxoditi 'go out'.
- z before č either is lost, or else zč > št: e.g. {iz-čist-i+} > ičistiti or ištistiti 'purify, cleanse'; (bez-čisl-ьп-ъ} bečislьпъ or beštislьпъ 'innumerable'.
- z before c either is lost, or else zc > sc or st: e.g. {iz-cěl-i+} > icěliti or iscěliti or istěliti 'heal'.<sup>10</sup>
- z + r > zdr: e.g. {raz-rěš-i+} razdrěšiti 'untie, free'.<sup>11</sup>

<sup>&</sup>lt;sup>10</sup> Historically, the clusters  $\delta c$ ,  $\xi \tilde{z}$ , and sc are to be expected. Shift of the affricate to stop would be normal. Loss of the initial sibilant is hard to explain. In stem-final position (where sk is followed by marked i or  $\check{e}$ ) the cluster sc may alternate with st, but the initial sibiliant always remains. All this variation surely reflects, in haphazard fashion, different regional and temporal dialects.

<sup>&</sup>lt;sup>11</sup> The cluster *zr* is permitted in root-initial position, but not, apparently, across the morpheme boundary.

3.3111-3.313

**3.3111** OCS spelling suggests that these phonetic rules applied to prepositions and prefixes alike. For example: bes tebe 'without you', is kraja 'from the end', v b s krai 'on the edge', b e s t e da 'without a child' (čęda), is trěva/i-crěva 'from the bowels', is crb k b v e 'from church', i-crb k b v b 'from churches', bez-d-razuma 'without understanding', iz-d-rěky 'from the river'; be-zb l a 'without evil', be-srama 'without shame', i-syna 'from the tower', i-svoego domu 'from his house'. Modern editors must choose whether to leave the usual space after the preposition or to use hyphens or a combination of devices.

The scribes of the younger OCS manuscripts (chiefly the Suprasliensis) are inclined to preserve the visual aspect of the prefix/preposition morpheme, and frequently the s or z is written, sometimes followed by a non-etymological jer: e.g. bezs tъštety, bezs čisla, izčazati, izščeze, is črěva, izšedъ, isšedъ, isščedъ, izč стъкъve, bezs razloky, izšrasti, razrušenie. Such spellings in part reflect a shift in orthographical rules, but they also imply that the complex changes of some forms had been abolished by reforming the word.

Spelled assimilation of the type iž njego, bež njego, and v $\bar{z}$ ljubiti is extremely rare; the normal spellings are iz njego, bez njego, v $\bar{z}$ ljubiti.<sup>12</sup>

**3.312** The prefix ob + v > ob: {ob-vlač-i+} > oblačiti 'wrap around', {ob-vbj- $\emptyset$ +ti} > obiti 'wrap, wind around'. The cluster bv does not occur.

**3.3121** The verb *ostopiti* means 'retreat, move back' but also 'besiege, surround', implying {ot-st-} vs. {ob-st-} with loss of the prefix-final consonant before obstruent. Further *okryti* 'open' and *oxoditi* 'surround' have doublets, *otskryti* and *otaxoditi*. Apparently the obscurity of such forms led to a reformation of the prefixes to {ob<sub>b</sub>} and {ot<sub>b</sub>}. As a general rule, *o*- before a consonantal root represents *ob*, but there are often doublets with a spelled consonant.

**3.313** A root-final consonant comes to stand before a consonantal-initial suffix in stem-derivation (with the classifier  $\{-nq+\}$ ), and in conjugation (1-participle, infinitive and supine). If the C is a sonorant, it may include the preceding vowel in the alternation: ov > u; bj/ij > i; bj/yj > y; b + nasal > e (see §11.212, §13.2).

<sup>&</sup>lt;sup>12</sup> Spelling which keeps a single visual image of a morpheme and ignores automatic phonemic changes is called morphophonemic (or, somewhat misleadingly, etymological). Russian is of this type, but treats some prefixes less consistently than did the pre-1917 rules, e.g. бесконечный, бессмертный vs. old безк-, безсм-. and the unchanging preposition in без конца, без смерти. Yet there is a limit to the degree of visual "distortion" allowed, e.g. old *безшумный*, new *бесшумный* but never бешшумный to show the real pronunciation. OCS spelling was apparently consistently phonemic at first, then began to make some use of morphophonemic principles.

**3.3131** The obstruents behave variously. S and z remain (whereby z > s before t). Dental t d > s before t, but drop before n or l. Labial p b drop before t, remain before l, and do either before n. Velar k g combine with t in {šč}, remain otherwise.<sup>13</sup> That is:

a. Before t, the labials are deleted, while the dentals become s:

pt bt vt > t; tt dt zt > st. e.g. {tep- $\emptyset$ +ti} > teti 'to beat', {živ- $\emptyset$ +ti} > žiti 'to live'; {pad- $\emptyset$ +ti} > pasti 'to fall', {lěz- $\emptyset$ +ti} > lěsti 'go'.

Velar k g + t combine in  $\delta t$ : {pek- $\emptyset$ +ti} > pešti 'to cook', {mog- $\emptyset$ +ti} > mošti 'to be able'. Note that this surface / $\delta t$ /-for underlying { $\delta c$ }--is the special sequence that serves as a palatal unit.<sup>14</sup>

b. Before the *l* the labials and velars remain: {tep-Ø+l-i} > tepli '(they) beat', {greb-Ø+l-i} > grebli 'buried'.

The dentals are lost:  $tl \ dl > l$ : {plet- $\emptyset$ +l-i} > pleli '(they) braided', {pad- $\emptyset$ +l-i} > pali '(they) fell'. Note that this is a process limited to this particular morphological category; tl and dl occur in root-initial position.<sup>15</sup>

c. Before the *n* of  $\{nq\}$ , the labials either remain or are deleted, see §15.75. The dentals are lost:  $tn \ dn > n$ . The velars  $k \ g \ x$  remain,  $kn \ gn \ xn$ .

**3.32** The possible CV combinations are defined by three groups of consonants (velars; c and z; soft consonants, §2.51) and eight vowels which are limited as to which consonants they can follow. In specific grammatical morphemes these eight vowels are subdivided into *alternating* versus *non-alternating* morphophonemes. (The morphophonemes include units symbolized as  $i^2$ ,  $\check{e}^2$ , and  $y^2$ , which will be defined below.)

**3.4** The major alternation in stem-final consonants is *palatalization* (or substitutive softening). There are two types: (I) a general or default process that occurs in derivation and inflection, and (II) a limited special process that occurs in declension (and rarely in conjugation) before morphophonemes that are specifically marked.

<sup>&</sup>lt;sup>13</sup> The preterite-marker x becomes s before t. See \$10.11.

<sup>&</sup>lt;sup>14</sup> The historical process surely starts with voicing assimilation: \*gt > \*kt. The agreedon formula for Middle Common Slavic is \*tj. For Southeast Late ComSl (or early Bulgaro-Macedonian) it is safe to posit šč and žž as the immediate forerunners of attested OCS št/žd.

<sup>&</sup>lt;sup>15</sup> For example, *dlsgs* 'long', *dlans* 'palm (of hand)'; *tlskqts* 'they drag'.

3.4-3.5

- I. A velar {k g x} or {c 3} that stands before a front vowel {iь e ě ę}, becomes palatal (č ž š) [except that {sk zg} > šč žš]
- II UNLESS the front vowel is the marked  $\{i^2\}$  or  $\{\check{e}^2\}$ , in which case  $\{k \ g \ x\}$ > c 3 s [whereby  $\{sk \ zg\}$  > sc z3 or st zd].<sup>16</sup>

Type I takes place before any front vowel that is not specially marked and it applies to c and z as well as to k g x. Historically, it changed any velar before all front vowels; we may label it KI. At that time velars followed by a diphthong *ai* were not affected. Subsequently, *ai* became a front vowel (*i* or  $\check{e}$ ); the velars then adapted to these new front vowels, but in a different way. This second regressive palatalization (Type II) may be labelled KAI.

3.5 The alternating and non-alternating morphophonemes are these:

a. Three alternating morphophonemes are purely phonotactic:  $\{b/b\}$ ,  $\{0/e\}$ , and  $\{y/i\}$  choose the second or "soft" variant when they follow  $c \ 3$  or a soft consonant.

{otrok+ъ/ь} {otьc+ъ/ь} {ključ+ъ/ь} {otrok+y/i} {otьc+y/i}	> > > >	оtьсь ključь otroky otьci	'fathers [Ip]'	{měst+o/e} {lic+o/e} {morj+o/e} {měst+y/i} {lic+y/i}	> > > >	lice morje městy lici	'place [Nsn]' 'face [Nsn]' 'sea [Nsn]' 'places [Ipn]' 'faces [Ipn]'
{otbc+y/1} {ključ+y/i}				{iic+y/i} {morj+y/i}			'seas [lpn]'

The tense jers or reduced y/i, (neutralization of  $y \sim z$  and  $i \sim z$  in position before j, §2.61), also alternate in this way.

**b.** A fourth morphophoneme also has y as the basic or hard alternant; we will label it  $y^2$ , to show it belongs with the alternant  $e: \{y^2/e\}$ . Thus, e.g.,

 $\{ otrok+y^2/\varrho \} > otroky `boys [Apm]' \{ r \varrho k+y^2/\varrho \} > r \varrho ky `hands [Apf]' \{ otbc+y^2/\varrho \} > otbc \varrho `fathers [Apm]' \{ ovbc+y^2/\varrho \} > ovbc \varrho `sheep [Apf]' \{ ključ+y^2/\varrho \} > ključ \varrho `keys [Apm]' \{ duš+y^2/\varrho \} > duš \varrho `soul [Apf]' These morphophonemes are lexical elements with specific grammatical meanings: for example, { o/e } serves as nominative singular neuter in the major declensional types, and as initial syllable in certain other substantival desinences (e.g. instrumental sing. { o'_emb}, { o'_ejQ } ).$ 

c. The vowel  $\check{e}$  is idiosyncratic. Though a front vowel, it does not occur after the palatal consonants, but only after the neutral labials and

<sup>&</sup>lt;sup>16</sup> The term for 'Easter' is spelled Pasxa, but surely was pronounced with /sk/, as in spoken Greek; the DL is pascë or pastë. The cluster sx occurs at a prefix-root boundary, eg. isxoditi {iz-xod-i+ti} 'go out', rasxytiti {raz-xyt-i+ti} 'steal, carry off'.

THE SOUND SYSTEM

dentals and the special subgroup  $c_3$ . We may regard it as two underlying entities. The first is the basic or ("hard") alternant in { $\check{e}/a$ }, a morphophoneme that serves as a verb-making formant (§3.5c {sed- $\check{e}/a+ti$ } >  $s\check{e}d\check{e}ti$  'sit' vs. {stoj- $\check{e}/a+ti$ } > stojati 'stand') and in some derivational suffixes.

c1. This  $\check{e}^i$  triggers KI (the First Regressive Palatalization) and then must become a: {kě gě xě} > ča ža ša. See §9.112.

c2. Underlying  $\check{e}^2$  is correlated with  $i^2$  in  $\{\check{e}^2/i^2\}$ , serving as the basic (or "hard") alternant. Both the non-alternating morphophoneme  $\{i^2\}$  and  $\{\check{e}^2/i^2\}$  are specially marked to cause KAI palatalization in a preceding velar  $(k g x); \{i^2\}$  is the nominative plural masculine desinence of the twofold nominal declension and the singular imperative-marker in conjugation (§7.101);  $\{\check{e}^2/i^2\}$  serves as the initial or only morphophoneme in several desinences of the HARD twofold nominal declension, and as a dual or plural imperative marker.

{otrok+i <sup>2</sup> }	>	otroci	'boys [Npm]'	{rqk+ě²/i}	> rǫcě	'hand [LDsf]'
{otьc+i <sup>2</sup> }	>	отьсі	'fathers [Npm]'	{оvьс+ě²/i}	> оvьсі	'sheep [LDsf]'
{ključ+i²}	>	ključi	'keys [Npm]'	{duš+ě²/i}	> duši	'soul [LDsf]'
{dъsk+e <sup>2</sup> /i}	>	dъscě	'board [LDsf]'	{dręzg+ě²/i}	> dręzzě	'woods [LDsf]'
or	>	dъstě		or	> dręzdě	

Non-alternating  $\{i^2\}$  is distinct from nonalternating  $\{i^1\}$ , which is associated with KI mutation, the general type of palatalization ( $\{ok-i\} > oči$  Ndu 'eyes').

The surface sequences cb ce ce cq ca cu (and 3b 3e etc.) unambiguously signify stem-final c 3 + vowel(-initial) desinence {b/b, o/e, y/e, q, a, u}. The sequence ce (3e) is unambiguous in showing stem-final velar (k g) followed by the basic form of desinence-initial { $e^{2/i^2}$ }. In contrast, ci (3i) could be either a c/3 stem followed by { $i^2$ } with the meaning Npm (e.g. otbci, kane3i 'princes') or the soft alternant of a desinence with { $e^{2/i^2}$ } (e.g. Lp otbcixa, kane3ixa).<sup>17</sup>

d. These six alternating morphophonemes above evolved from single vowels or diphthongs that developed differently. The masculine vocative form has -e for "hard stems" and  $c_3$ , but -u for palatal stems. The  $\{e/u\}$  grammatical morpheme that belongs to OCS results from two historically diverse vocative desinences in pre-Slavic. For examples, see §4.11.

<sup>&</sup>lt;sup>17</sup> This assumes that forms like *cěna* 'price', *ʒělo* 'very', and *ocыъ* 'vinegar' are based on lexically given stems of the type {cěn-a}, {ʒělo}, {ocы-ъ}. No verbal basic stem ends in *c ʒ*; the imperatives like *moʒi* 'be able' and *rьci*, *rьcěte* illustrate the singular vs. plural imperative marker, {i<sup>2</sup>} vs. (ě<sup>2</sup>/i<sup>2</sup>}.

3.51-3.72

3.51

To summarize, here are the alternating vowel morphophonemes:

basic	0	ъ	y	y <sup>2</sup>	ě²	ě1	e
"soft"	е	ь	i	e	i <sup>2</sup>	a	u

**3.6** A second process of substitutive softening is *iotation*, which involves the action of an underlying *j*. The *j* may be an underlying suffixal element, but in conjugation it is generated from an underlying *i*, *ĕ*, or *a* followed by a vowel. It is described in  $\S6.13-6.23$ . Here is a summary:

pj bj vj mjtj dj cj ʒj sj zj kj gj xj nj lj rj stj zdjskjzgj па ба ва маџ жду ж ш ж у ж ш ѝ а̂ р̂ џ ждџ жд

Notice that velars (k g x) and c z are affected by iotation in precisely the same way as by KI (type I palatalization, §3.4).

**3.61** A unique example shows  $\xi\xi$  for the expected  $\xi d$  (~ d). In Mk 1:6 the instrumental plural of the possessive adjective derived from *velbods* 'camel' is spelled *velbootdi* in Mar (but normal *velbotdi* in Zo As Sav).

**3.62** The clusters trj and drj sometimes resisted iotation. Evidence is sparse, and complicated by the fact that scribes often fail to distinguish rj from r. Su has **chromotrym** for expected showstrjq 'I look' {sh-motr-i+q}; 3p imperfect **chromotryme** for -mostrjaase {-motr-i-ĕaše}. The cluster trj iotates to strlj, though the lj often disappears in spelling (§2.521): e.g. from {u-mrbtv-i-aj+} 'mortify', pres. passive part. Npm umrbstvlčemi (Ps), but pres. act. part. Nsm umrbstvčęi (Euch). Variation doubtless reflects both regional and temporal dialect differences.

3.71 Labials  $(p \ b \ m \ v)$  are not themselves affected by j; it is the iod that changes—by becoming a unit liquid palatal, lj, see §2.521.

The clusters plj blj mlj vlj surely were standard OCS at the beginning, and efforts to spell pj bj mj vj in their place are to be interpreted as a change in scribal attitudes. Serbo-Croatian and Slovenian and East Slavic have this "epenthetic l" (to use the traditional term despite its misleading implication that l is inserted between labial and iod). Recorded West Slavic never had it, and modern Macedonian and Bulgarian lack it. The Vatican Cyrillic Gospel Lectionary generally avoids it. Post-OCS manuscripts from the Macedono-Bulgarian regions vacillate, but on the whole scribes tried to write l in accord with a tenacious tradition. Perhaps there were indeed regional dialects that supported such spellings; we lack unequivocal evidence for the history.

**3.72** Underlying  $\{v\}$  usually behaves like a palatal obstruent, but the sequence  $\{ov\}$  generally becomes *u* before all consonants (cf. §15.841), including the *j* that is generated in the verbal classifier  $\{ova/eva\}$ . In iverbs, *vj* has the same effect as the other labial consonants and the palatal lj results. See §6.21 with footnote.

**3.721** Stem-final xv is unique to vlaxva 'magician'; the noun is subject to both types of palatalization: voc. vlasve, Np vlasvi. Before a suffixal front vowel in derivation, however, the two units become  $\dot{s}$ : vlasba 'sorcery'.

**3.8** The nominative singular desinence of comparatives, of certain participles (cf. §4.19), and of some anomalous substantives is a zero which has the effect of a consonant. Since a word must end in a vowel (§1.2301), the underlying final consonant must be deleted. Thus {nes- $\emptyset$ + $\pm$ S-C} 'carrying' yields *nes* $\pm$ ; {otrok-et-C} 'child [nom.-acc. sg. neut.]' > *otroče*. Sometimes the VC of the stem is modified before the final consonant is lost: {semen-C} 'seed [nom.-acc. sg. neut.]' > *sěmę*. These noun-forms are essentially irregular morphological forms that belong in the lexical definition of individual words. The masculine {kamen-C} 'stone' has Ns kamy. Other types are  $\exists v \sim y$  ({cr $\pm k \rightarrow v + C$ } > 'church' *cr*bky), *er* ~ *i* ({mater-C} 'mother' > *mati*), *es* ~ *o* ({těl-es-C} 'body' > *tělo*. See pp. 73-74.

**3.90** In the morphology of irregular verbs, and in derivation, there are many more alternations.<sup>18</sup> Here we will point out the chief vowel-alternations (and alternations involving both vowel and consonant) found in the various possible forms of a root which may occur in a single "family".<sup>19</sup> The majority of these root-vowel alternations exemplify an old process called vowel-gradation (or *ablaut* or *apophony*).

**3.911** Perhaps the majority of roots have in some form the vowel e, which is taken as basic for the alternations; e may alternate on the one hand with  $\check{e}$ , on the other with b, and in still other forms there may be o. The  $\check{e}$  and the o may in turn alternate with a, and the b may alternate with i. Thus:

 $\check{e} \leftarrow e \rightarrow b \rightarrow i$   $\downarrow \qquad \downarrow$  $a \leftarrow o$ 

3.912 The alternations  $e \sim o$  and  $\check{e} \sim a$  are {front ~ back};  $e \sim \check{e}$ ,  $o \sim a$ , and  $b \sim i$  are {lax ~ tense};  $e \sim b$  is {low ~ high}.<sup>20</sup> This is all complicated

<sup>&</sup>lt;sup>18</sup> Isolated alternations occur in the roots of irregular verbs:  $\mathfrak{d} \sim u$  (§15.643),  $e \sim \mathfrak{b}$  (§15.644),  $o \sim \mathfrak{b}$  (§15.645),  $o \sim \check{e}$  (§§16.53),  $\check{e} \sim \varrho$  (§16.61),  $e \sim \varrho$  (§16.62), and perhaps  $\mathfrak{d} m \sim u$  (§16.92).

<sup>&</sup>lt;sup>19</sup> "Family" is the term for all possible derivatives of a single root. The example of the root *rek*- given in § 3.1 above illustrates varying shapes of the root, but gives only a small percentage of the words in this particular family.

<sup>&</sup>lt;sup>20</sup> The historical terminology is different: e ~ o is the basic qualitative opposition, ĕ ~ a is "lengthened e-grade vs. lengthened o-grade"; e ~ b is "normal vs. reduced grade", while b ~ i is "reduced vs. lengthened reduced".

by the fact that if the theoretical vowel root was followed by a resonant (r, l; m, n, nj), forms with e, o and b vary according to their position before vowel or consonant. Thus the combination er remained before vowel but became  $r\check{e}$  before consonant, i.e.  $er/r\check{e}$ ; em,  $en/\varrho$ ; or/ra, ol/la,  $on/\varrho$ ; and br/rb, bl/lb,  $bm/\varrho$ . Cf. §16.5ff.

**3.913** No root illustrates all the possibilities, and some roots have only one or two surviving forms. Here are some typical examples. Forms marked \* are not attested in OCS but fit the old patterns and occur in slightly later mss.

- \*ved-: vedǫtъ 'they are leading' vodętъ 'they lead' věsъ 'I led' provaždati sę 'be influenced'
- \*ber-: berç 'I gather' brěmę 'burden' bьrati 'to gather' sъbirati 'to gather' – sъborъ 'a gathering'
- \*mer-: umrěti 'to die' umьrǫ 'I shall die' sъmrьtь 'death' umirajǫ 'I'm dying' – umoriti 'kill' (P) – umarjati 'kill' (I)
- \*pen: \*pьnq 'I stretch' propeti 'crucify' (P) propinati 'crucify' (I) opona 'curtain' – pota 'fetter'

\*zven-: zvonъ 'a sounding noise' – \*zvьněti 'to sound' – \*zvokъ 'sound' \*velk-: vlěko 'I drag' – vlькъ 'having dragged' – privlačiti 'attract'

**3.92** It is probable that this system was fairly vital in the language of Cyril and Methodius, although many of the alternations were represented only by a few lexical items. The jer-shift caused major readjustments, however, so that a number of the older relationships became obscured or lost entirely.

**3.93** The alternations  $o \sim a$ ,  $b \sim y$ ,  $b \sim i$ , and  $e \sim \check{e}$  remained marginally productive in OCS in the formation of imperfective verbs. See the subdivisions of §5.7 for more examples:

 $o \sim a$ 

ukoriukarjaj-'reproach' kosnokasai-'touch' izbadaj- 'stab' omočiomakaj-'moisten' izbodsvobodi- svobaždaj- 'free' sъgarjaj-'burn up' sъgorěь~і ѕъ**бъг**аsъbirai-'collect' зъ**гьd**аsъzidai- 'build' počitaj- 'count, read' ротьпё- ротіпај-'remember' роčьt-'take' zaklьnzaklinaj- 'swear' VЪZ**ьm**vъzimaj 3~ y vъzdъxno- vъzdyxaj-'sigh' роѕъІаposylaj-'send'

## CHAPTER THREE

# DECLENSION

**4.00** The great majority of Old Church Slavonic words are inflected: their form changes to express different relationships. All inflected words consist of a *stem* plus an inflectional suffix—a *desinence*. (The stem may itself be complex—it must contain a *root*, and may contain prefixes and derivational suffixes—but this fact is not important for a discussion of inflection.) In the presence of different desinences, the stem may itself be modified; and conversely, certain desinences have different forms to adapt to different types of stems. Uninflected words are classified, on the basis of their syntactical functions, as adverbs, conjunctions, prepositions, particles, and interjections.

**4.01** Inflected OCS words are of two major categories: *verbs* and *nouns*. The suffixes in both categories nearly always indicate **number**: *singular* (one), *dual* (two) or *plural* (three or more).

**4.02** Nouns have different suffixes to express the relationships of the words to one another in the sentence, that is, **case**. There are six cases in OCS: *nominative, accusative, genitive, locative, dative, and instrumental.* Normally these are expressed by different desinences in singular and plural, but in the dual there are only three possible forms: a nominative-accusative, a genitive-locative, and a dative-instrumental. A separate voca-tive form exists for most masculine and feminine substantives in the singular; otherwise the nominative is used for an appeal.

**4.021** Nouns are divided into three groups on the basis of their expression of gender and their declension-types: *substantives, adjectives,* and *pronouns.* Only substantives have an inherent, unchanging **gender** (*masculine, feminine,* or *neuter*). Other nouns have variable (or syntactic) gender, changing to agree with the substantives they modify in a given sentence. There are two types: *pronouns* do NOT follow the nominal declension, but have a special set of forms we call *pronominal* declension. Adjectives *may* follow the nominal declension and/or a compound declension com-

#### DECLENSION

bining elements of nominal and pronominal declension. Finally, there is the anomalous group of personal pronouns, which have no formal means of expressing gender, and which follow completely idiosyncratic declensions.

**4.0211** A handful of words without declension or gender are defined as adjectives because of their syntactical use and their meanings: e.g. *isplanb* 'full', *svobodb* 'free', *različb* 'different'. They all have synonyms which are inflected as adjectives (e.g. *planb*, *svobodbnb*, *različbnb*).

**4.03** The types of substantival declension generally correspond to gender, but there are outstanding exceptions. The dominant inflection-class, called here *twofold nominal declension* includes: (1) a masculine-neuter type and (2) a feminine type [with a few masculine members], both of which (a) may be used for adjectives as well as substantives, (b) have desinence-variants according to the palatal or non-palatal character of the stem-final consonant, and (c) are productive. Another inflection-class, (3) the *simple nominal declension*, is restricted to substantives, most of them feminine. A series of minor types may be subsumed under this simple nominal declension, and called its *anomalous subtype*. Pronominal stems belong to another paradigm, (4) the *pronominal declension*. Determined or compound adjectives have desinences combining those of (1+2 and 4); this paradigm may be called the *compound* or *adjectival declension*.

The gender and declensional type of a substantive can usually be determined from the nom. sg. form. Substantives in -o or -e are neuter, twofold declension (e.g. *město* 'place', *srbbce* 'heart'; exceptions in  $\S4.414$ ); those in -a are feminine [unless they refer to male persons], of the twofold declension (e.g. *žena* 'woman', *zemlja* 'land'; *vojevoda* 'general'). The few ending in *-nji* (usually spelled simply *-ni*) are also feminine, soft twofold, e.g. *rabynji* 'slave woman'. The nom. sg. desinence -5 indicates masculine hard twofold. The front jer -6 implies masculine soft twofold (e.g. *kostb* 'bone') but masculine is possible (see list in \$4.4032)—or else anomalous (listed in \$4.412). The word-final letter *-i* preceded by a vowel-letter, indicates \**j*-b(\$1.24), soft twofold masculine. Substantives in *-e* are neuters of the anomalous type (e.g. *vrěme* 'time'); those in *-y* are anomalous feminines (e.g. *ljuby* 'love').

**4.04** The stem of every declinable word ends in a consonant. The stem is, by definition, the nominative singular of substantives or the nominative singular masculine (short form) of words with variable gender, minus the final vowel ( $vl_{bk-3}$  'wolf', *žen-a* 'woman', *měst-o* 'place', *on-s* 'he,

that one'). If the removal of the final vowel-letter leaves a vowel-letter, then the phonological stem ends in j (krai = kraj-b 'edge'; pitbe = pitbj-e 'drink', struž crpoyta = struj-a 'stream').

**4.05** If the final stem-consonant is  $\check{s}, \check{z}, \check{c}, j, lj, nj, rj, c, + or the groups <math>\check{s}t$  or  $\check{z}d$ , the stem will be called *soft*; all other stems are *hard*. The desinences may vary accordingly. The basic underlying form is the *hard* (or neutral) desinence; the *soft* form is selected for soft stems. Here are the possible morphophonemes that occur in the twofold nominal and pronominal declensions:

invariable	a	u	ð	i <sup>2</sup>	hard	0	ъ	ě²	у	<b>y</b> <sup>2</sup>	hard
					soft	e	ь	i <sup>2</sup>	i	ę	soft

In addition there is the idiosyncratic e/u of the vocative (see §3.5d). When desinences beginning with  $i^2$  or  $\check{e}^2$  are added to stems in k, g, x, c, or 3, the stem-consonant undergoes KAI (substitutive softening of type II); before the desinence -e these stem-consonants undergo KI (substitutive softening of type I), see §4.11, below.

## 4.1 The twofold nominal declension.

In the table the slash (/) separates the suffix for the *hard* declension (given first) from that of the soft, when there is any difference.

		SINGUI	.AR	DUAL			PLURAL			
Ī	masc	neut	fem	masc	neut	fem	masc	neut	fem	1
Nom			-a		-ĕ/-i		-i		-y²/-e	N
Acc	-ъ/-ь	-0/-е	-9	-a	-6/-1		-y²/-ę	-a	-y-/-ę	Α
Gen	-:	a	-y²/-ę				-ъ/-ь			
Loc	-ё/-і -u -оть/-еть		-ĕ/-i		-u		-ěхъ/	-іхъ	-ахъ	L
Dat			•€/•1	0000	/ ama	-ama	-отъ/-етъ		-атъ	D
Instr			-ojǫ/-ejǫ	-oma/-ema		-ama	-y/i		-ami	Ι
Voc	-e/-u =N/A	=N/A	-о/-е		= N/A			= N/A		v
	masc	neut	fem	masc	neut	fem	masc	neut	fem	

**4.101** Note that in spelling, stem-final j + b or i merges in "i" (§2.625): Nsm (and Gp) \*krajb > krai, краи, Np (and L sing. and pl.) \*kraji > krai, краи.

Velar stems undergo KAI-softening in Ls, Ds fem., NAdu. neut./fem., Lp masc./neut (before  $\{\check{e}/i\}$ ), and Np masc. (before  $\{i\}$ ).

Velar and c 3 stems undergo KI-softening in the vocative (sing.), because the basic shape of  $\{e/u\}$  is a front vowel, see §4.11, below.

**4.1011** Historically the soft feminines contained a CV formant with the shape  $*j\bar{a}$ , and the soft masculines and neuters had a similar formant with a variable vowel e or o, \*je/\*jo. In early Slavic the morphemic segmentation changed; the \*j was perceived as part of the stem, and the vowel took on the role of desinence (or first unit of a desinence): e.g. pre-Slavic  $*lug \cdot j\bar{a}$  became  $*l\bar{s}g \cdot j + a > OCS l\bar{s}\bar{z}a$  'lie' (cf. infinitive { $l\bar{s}g \cdot a + ti$ } 'to lie'). In traditional OCS descriptions therefore the hard feminines are a-stems, the soft are ja-stems. Similarly the masculines and neuters are o-stems (hard) and jo-stems or je-stems. Here we will speak of a-stems and o-stems, hard or soft.

**4.102** Note that in stems ending in -ij-, the *i* of the stem represents the tense b (cf. §2.61) and may be so written: pbsanie/pbsanie "writing", gen. sg. masc. zmija/zmbja 'snake', gen. sg. fem. ladie/ladie 'boat'.

**4.1021** Here belong a few adjectives with stems in -bj. The most important are *velbi/velii* 'great' and *božbi/božii* 'God's, divine'. Note that in nom. sg. masc. and gen. pl. the tense strong **b** (\*-**b**jb) may be written *e*: *božei, velei* (cf. §2.51, 2.525).

**4.1022** A number of formally singular substantives in be or bja are collectives, to be translated as plurals: e.g. tronse 'thorns', kamenbe 'stones', korenbe 'roots', roždbe or raždbe 'branches', bratroja or bratoja 'brothers'.

**4.11** The formation of the *vocative*. Masc. and fem. nouns of this declension have special forms for calling or addressing—the vocative. The desinences are -o/-e for the feminine (*ženo* 'woman', *děvice* 'maiden'), and the highly unusual alternants e/u for the masculine.

The basic variant used with non-palatal stems is the front-vowel e, which is accompanied by KI mutation in velars and c and 3: e.g.  $vlbka \sim vlbče$ ,  $boga \sim bože$ ,  $duxa \sim duše$ ; otbcb 'father'  $\sim otbče$ , kanęzb 'prince'  $\sim kanęže$ . Palatal stems regularly take -u (možb 'man'  $\sim možu$ , cěsarjb 'king'  $\sim cěsarju$ , zmii 'snake'  $\sim zmiju$ ).

Masculine adjectives sometimes take the regular e-desinence, but more often the nominative of the compound declension is used in vocative function; cf. §17.1.

Some variation is to be explained by innovating forms. Syna 'son' occasionally has archaic synu beside regularized syne (cf. §4.145). Su 155.18 has regular KNAXEE, and in the same speech KNAZEY (Su 156.8); apparently 3 was in competition with a newer  $\frac{1}{2}$  (§2.8) that calls for the "soft" desinence.

**4.12** Foreign words of masculine gender (especially names) may have stems ending in a vowel. Probably a *j* was added in conformity with the requirement that declensional stems must end in a consonant. The vowels  $\mathfrak{d}$ , *y*, or  $\check{e}$  cannot stand in post-vocalic position, and the soft variant regularly appears: 'Pharisee' (Greek [farise-os]) thus has *farisei* (Ns and Gp \*-*ejb*; Ls, nom. and instr. pl. \*-*eji*). The Ap *farisee* surely—and Gs *farisea*, Ds *fariseu* probably—may represent -*eje* (-*eja*, -*eju*), but the Ds *fariseovi*, instr. sg. *fariseomb* and Dp

fariseoms seem to introduce a sequence of vowels not found in native words. The variegated spellings of such words indicate disagreements that very likely include pronunciation.

**4.13** Masculine substantives indicating male persons use the genitive singular forms in accusative function, as do pronouns and adjectives referring to such substantives: člověk<sup>5</sup> 'man' – gen./acc. člověka. Usage fluctuates with such words as *bog*<sup>5</sup> 'god', *angel*<sup>5</sup> 'angel', *dux*<sup>5</sup> 'spirit', *rab*<sup>5</sup> 'slave', *otrok*<sup>5</sup> 'child, servant', and words referring to animals. See §18.21.<sup>1</sup>

**4.14** Masculine substantives indicating persons may have beside the regular *-u* desinence of dat. sg. an alternative *-ovi/-evi: synu* or *synovi* 'son', *vraču* or *vračevi* 'doctor'. Sometimes this desinence appears with words that do not refer to persons (*mirovi* 'to the world', *adovi* 'to hell'), possibly conveying a sense of personification.

**4.141** Certain masculine nouns with monosyllabic stems may have in the gen. or loc. sg. (or both) the desinence -u beside the normal -a. Attested with -u in these cases are chiefly: syns 'son', doms 'house', vols 'ox', pols 'half'; but also vraxs 'top', glass 'voice', grams 'bush', dars 'gift', dlags 'debt', dobs 'oak', meds 'honey', mirs 'world', rods 'race', reds 'row', sans 'rank', stans 'camp', syns 'tower', čins 'rank', jads 'poison'.

**4.142** Certain monosyllabic masculines occasionally have (beside the normal nom. pl -i and the gen. -3/-b) bisyllabic desinences, nom. -ove/-eve, gen. -3/-b. Examples are attested for: syns 'son', dome 'house', vole 'ox', pole 'half', vrače 'doctor', gade 'vermin', grozde 'grape', grěxe 'sin', zmii 'snake', sade 'planting'. Supr has several more words with such desinences, but it is probable that they belong to a later stage of development.<sup>2</sup>

**4.143** Some of these same stems (\$4.142) occur with -ox<sup>3</sup> in Lp (beside normal -ex<sup>3</sup>), and/or -ami in Ip. (beside normal -y): synox<sub>3</sub>, syn<sub>3</sub>mi.

**4.144** The same stems rarely have alternate dual forms: NA -y (beside -a), GL -ovu (beside -u), DI -oma (beside -oma): syny, synovu, synoma.

**4.145** These optional suffixes (§§4.14-4.144) must be listed in the OCS lexical entries for the individual substantives. They are remnants of what in early Slavic was surely an independent paradigm, traditionally called "*u*-stems". Some of the desinences began to spread to former *o*-stems; comparative evidence from medieval manuscripts and modern dialects does not suffice to identify the early status of many items. Syna 'son' is best attested and at the same time impeccably documented as an ancient *u*-stem (e.g. by Sanskrit sūnuš): NA sg syna, GL synu, D synovi, I sg \*syname; Voc. synu; NA du. syny, GL synovu, DI synama; Np

<sup>&</sup>lt;sup>1</sup> It is probable that the earliest texts used the gen.-acc. only for substantives indicating a healthy, free, male person; the sick, the crippled, the enslaved, and the supernatural did not count. Surviving manuscripts record a continuing expansion to include all animate masculine singulars; see §18.21.

<sup>&</sup>lt;sup>2</sup> Modern Bulgarian, Macedonian, and eastern Serbo-Croatian use the *-ov-/-ev-* suffix in forming the plural of the great majority of monosyllabic masculines.

synove, Ap syny; G synove; L synove (for \*synaxe), I synami. Some of these desinences were later lost, others redistributed differently in various dialects. Sometimes a desinence could acquire a special meaning, e.g. the "personal dative" of OCS (§4.14).<sup>3</sup>

**4.15** Beside the regular instrumental sing. masc. neut. desinences -omb/-emb, there are occurrences of -bmb/-bmb. They are the only forms in the Kiev Folia; in other mss they are rare and may be simply scribal errors. It is likely that such desinences were characteristic of dialects in Morava and fairly certain for Rus' (they are normal in 11th-century Rus' mss). Neuter stems in -ij and adjectives like božbi, and velbi (§4.1021) occasionally have Is -iimb (<\*-bjbmb) or a contracted -imb.

**4.16** A number of substantives that belong formally to the *a*-declension, which is the typically feminine paradigm, are necessarily of masculine gender because they signify grown male persons: *vladyka* 'ruler, lord', *sluga* 'servant', *junoša* 'young man', *prědsteča* 'forerunner', *qžika* 'relative' (also functions as feminine), *ubiica* 'murderer', *vinopiica* 'wine-bibber'. When the stem is in *-ij/-bj-* the Ns is normally *-ii* (*sqdii* 'judge', *balii* 'doctor'), while foreign personal names had *-ija/-bja* (e.g. Josija 'Josiah', cf. also *mesija* 'messiah'). These words follow the *a*-declension, hard or soft (except that names in *-bja/-ija* have Is in *-ems*; e.g. *Isaiems* 'Isaiah'). Vocatives have *-o/-e: vladyko, junoše, sqdije*.

They have the normal accusative in  $-\rho$ , but note that the pronouns and adjectives referring to them take the genitive masculine form (cf. §4.13); e.g. camoro bradilike 'the lord himself' Su 491/5, юношж красьма 'a handsome youth' Su 187.3.

When these a-masculines are in the plural, modifiers tend to be feminine in form, e.g. dat. pl. starěišinams galileiskams 'chiefs of Galilee' (Mk 6:21).

**4.17** Beside the normal instr. sg. fem. desinence  $-oj\varrho/-ej\varrho$ , there are occasional forms in  $-\varrho$  ( $r\varrho koj\varrho$  and  $r\varrho k\varrho$ ), especially in the Suprasliensis.

**4.18** Some feminine substantives have Ns in *-i* (rather than *-a*). Here belong all with the derivative suffix *-ynj-* (e.g. *rabynji* 'female slave', *bogynji* 'goddess', *pustynji* 'desert, wilderness') and most with stems in *-ij/-bj* (e.g. *ladii* 'boat', *krabii* 'box', *mlbnii* 'lightning'), although there are a few nominatives in *-ija/-bja* (*bratrbja* 'brothers, brethren', and names like *Marija*).

The numeral tysęšti '1000' also belongs here.

Mati 'mother' and dašti 'daughter' belong to another declension; see §4.423, below.

<sup>&</sup>lt;sup>3</sup> Vaillant departs from his descriptive framework by treating these words separately, although he makes it clear that there is not a separate paradigm. He provides details of attestation for both hypothetical ancient *u*-stems and substantives of less clear provenience, \$58-59.

For adjectival forms with -i in nom. sg. feminine, see §4.19.

# Twofold declension paradigms

Masculines: hard-stems grads 'city'; člověks 'man'; soft-stems: možb 'man'; otbcb 'father'

Sing.	Ν	градъ	уловѣкъ	мѫжь	отьць
	Α	градъ	уловѣка	мѫжа	отьца
	G	града	чловѣка	мѫжа	отьца
	L	градъ	чловъцъ	мѫжи	отьци
	D	градоу	чловѣкоу -ови	мжжоу -еви	отьцоу -еви
	I	градомь	чловѣкомь	мѫжемь	отьц€мь
	V	граде	лловфле	мѫжѹ	отьч€
Dual	NA	града	<b>члов</b> њка	мѫжа	отьца
	GL	градоу	чловъкоч	мжжоу	отьцоу
	DI	градома	<b>члов</b> екома	мѫжема	отьцема
Plur.	Ν	гради	уловѣци	мѫжи	отьци
	Α	градъі	уловѣкъі	мжжа	отьца
	G	градъ	чловѣкъ	мѫжь	отьць
	L	градѣхъ	чловъцъхъ	мжжихъ	отьцихъ
	D	градѣмъ	уловъцъмъ	мѫжимъ	отьцимъ
	Ι	градъі	уловѣкъі	мѫжи	отьци
Neuter	rs:	hard-stem <i>měst</i> soft-stems: srb	to 'place'; dьce 'heart'; znd	атепье 'sign'	
Neuter Sing.	rs: NA		-	<i>amenьe</i> 'sign' Знаменье	-H€
		soft-stems: srb	dьce 'heart'; znd	Ū.	-ие -иа
	NA	soft-stems: srь мѣсто	dьce 'heart'; znd срьдьце	<b>Знаменье</b>	
	NA G	soft-stems: <i>srь</i> мѣсто мѣста	<i>dьсе</i> 'heart'; <i>zna</i> срьдьце срьдьца	2наменье 2наменыя	-ига
	NA G L	soft-stems: <i>srъ</i> мቴсто мቴста мቴст <b>а</b>	<i>dьсе</i> 'heart'; <i>zna</i> сръдьце сръдьца сръдьци	с Знаменье Знаменью Знаменьи	-ига -ин
	NA G L D I NA	soft-stems: <i>srb</i> мѣсто мѣста мѣст <b>ѣ</b> мѣстф	<i>dьсе</i> 'heart'; <i>zna</i> срьдьце срьдьца срьдьци срьдьцоу	Знаменье Знаменью Знаменьи Знаменью	-ига -ии -ию
Sing.	NA G L D I	soft-stems: <i>srъ</i> мѣсто мѣста мѣстѣ мѣстоу мѣстомь	<i>dьсе</i> 'heart'; <i>zna</i> срьдьце срьдьца срьдьци срьдьцоу срьдьцемь	Знаменье Знаменыа Знаменьи Знаменью Знаменью	-ига -ии -ию -иемь
Sing.	NA G L D I NA	soft-stems: <i>srb</i> мѣсто мѣста мѣстѣ мѣстъ мѣсточ мѣстъ	<i>dьсе</i> 'heart'; <i>zna</i> срьдьце срьдьца срьдьци срьдьцоу срьдьцоу срьдьцемь	Знаменье Знаменыя Знаменыи Знаменью Знаменьемь Знаменьемь	-ига -ин -ию -иемь -ин
Sing.	NA G L D I NA GL DI NA	soft-stems: <i>sr b</i> мѣсто мѣста мѣстѣ мѣстоу мѣстъ мѣстѣ мѣстъ	<i>dьсе</i> 'heart'; <i>zna</i> срьдьце срьдьца срьдьци срьдьцоу срьдьцемь срьдьци срьдьци	Знаменье Знаменью Знаменью Знаменью Знаменьемь Знаменьемь Знаменью Знаменью	-ига -ин -ию -иемь -ии -ии
Sing. Dual	NA G L D I NA GL NA G	soft-stems: <i>srb</i> мѣсто мѣста мѣстѣ мѣстоү мѣстомь мѣстъ мѣстоү мѣста мѣстъ	<i>dьсе</i> 'heart'; <i>zna</i> срьдьце срьдьца срьдьци срьдьцоу срьдьцемь срьдьци срьдьци срьдьцема	Знаменье Знаменыя Знаменьи Знаменью знаменьемь Знаменьи знаменью знаменью знаменьема	-ига -ин -ию -иемь -ин -ию -иема
Sing. Dual	NA G L D I NA GL DI NA	soft-stems: <i>srb</i> мѣсто мѣста мѣстѣ мѣстоу мѣстомь мѣстъ мѣстоу мѣстома мѣста	<i>dьсе</i> 'heart'; <i>zna</i> срьдьце срьдьца срьдьци срьдьцоу срьдьцемь срьдьци срьдьци срьдьца	Знаменье Знаменью Знаменью Знаменью Знаменьемь Знаменью Знаменью Знаменью Знаменью Знаменью Знаменью Знаменьы	-ига -ин -ию -иемь -ин -ин -ию -ию -иема
Sing. Dual	NA G L D I NA GL NA G	soft-stems: <i>srb</i> мѣсто мѣста мѣстѣ мѣстоү мѣстомь мѣстъ мѣстоү мѣста мѣстъ	dbce 'heart'; znd срьдьце срьдьца срьдьци срьдьцоу срьдьцемь срьдьци срьдьцема срьдьцема срьдьца срьдьць	Знаменью	-ига -ин -ию -иемь -ии -ию -ию -иема -ига -ига
Sing. Dual	NA G L D I NA GL DI NA G L	soft-stems: <i>srb</i> мѣсто мѣста мѣстѣ мѣстоу мѣстомь мѣстъ мѣстъ мѣстъ мѣстъ мѣстъ мѣстъ	dbce 'heart'; znd срьдьце срьдьца срьдьци срьдьцоу срьдьцемь срьдьцема срьдьцема срьдьцема срьдьца срьдьць срьдьць	<ul> <li>Знаменью</li> </ul>	-ига -ин -ию -иемь -ин -ию -иема -иема -ига -ин -ии

58

**Feminines:** hard-stems *žena* 'woman, wife', *rǫka* 'hand'; soft-stem *duša* 'soul'; and the masculine *sǫdii* 'judge'. (for fem. and a-masculines with Ns in -*bji/-iji*)

Sing.	Ν	жена	ржка	доуша	сждъи	-нн
	Α	женж	ржкж	доушж	сждыж	-нж
	G	жены	ржкъі	доуша	сждьм	-на
	LD	женф	ржцѣ	доуши	сждьи	-ии
	Ι	женоіж	ржкож	доушеіж	сждьеіж	-иеіж
	v	жено	ржко	доуше	сждье	-не
Dual	NA	женъ	ржцѣ	доуши	сждьи	-ни
	GL	женоу	ржкоу	доушоу	сждью	-ию
	DI	женама	ржкама	доушама	сждыама	-игама
Plur.	NA	женъі	ржкъ	доуша	сждьа	-на
	G	женъ	, ይጽкъ	доушь	сждьи	-ин
	L	женахъ	ржкахъ	доушахъ	сждыахъ	-иахъ
	D	женамъ	ржкамъ	доушамъ	сждыамъ	-намъ
	I	женами	ржками	доушами	сждыами	-игами

Adjectives: hard-stem nova 'new'; soft-stem ništb 'poor'.

		masc.	neut.	fem.	masc.	neut.	fem.	1
	Ν	новъ		нова	ништь		ништа	N
	Α	=N or G	ново	новж	=N or G	ништ <b>е</b>	ништж	Α
Sing.	G	NO	ва	новъі	HHL	ПТА	ништа	G
	L	ноі	5°B	- NOB'B	NHLL	ПТА	ништи	L
	D	NOE	sor	NAP P	ннш	тоу	имшти	D
	Ι	NOB	OWP	новоня	ништемь ништеж		итенж	Ι
	NA	нова	юва но		ништа ни		ити	NA
Dual	GL		новоу		нништоу			GL
	DI	HQE	новома		ииш,	ништема		DI
	Ν	нови	NOBA		ништи	ништа	ништа	Ν
	Α	новъі	нова	новъј	ништа	ништа	ништм	Α
Plur.	G		новъ			ништь		G
	L	новъ)	(Ъ	новахъ	ништи	ништихъ		L
	D	HOBON	N73.	новамъ	NHUT	ништемъ		D
	Ι	новъі		новами	ниш	ги	ништами	1

4.18

**4.19** The comparatives, the present active participles, and the (first) past active participles follow the twofold nominal declension for their non-definite form (see §4.31 for definite forms), but their nominative and accusative forms in singular and plural are special. The feminine nom. sg. has the desinence -i (cf. §4.18) and the full stem; the masc. and neut. have a shortened stem. The following tables summarizes these cases:

			comp	ii uii i v					
	m sg	n sg	f sg	m pl	n pl	fp	1		
Nom	-ы (-ь)	-е	-ьši	-ьše	-ьšа	-ьš	<u>.</u>	Nom	
Acc	-ьšь	-0	-ьšǫ	-ьšę	-584	-Р2	A Star		c
	stem for other	r forms		- <u>H</u>	sš-		ste	m	

Comparative

Past	Active	Participle
------	--------	------------

		m sg	n sg	f sg	m pl	n pl	f pl	]	
Nor	m		-Ъ	-ъši	-ъše	-ъša	-ъšę	N	lom
Acc	2	-ЪŠЬ	-ъše	-ъšq	-ъšę	- D34	- Бэб	A	cc
	stem for other forms			-ъš-		st	em		

Present A	ctive Partic	ciple (q ~	e present)
-----------	--------------	------------	------------

	m sg	n sg	f sg	m pl	n pl	f pl		
Nom		·y	-ǫšti	-ǫšte	-ošta	-ošte	N	lom
Acc	-ǫštь	-ǫšte	-ęštę	-ǫštę	-บุรเล	-પૃડાદ્	A	cc
stem for other forms			- <b>ošt</b> - st			m		

		$\frac{1}{1} \frac{1}{1} \frac{1}$								
	m sg	n sg	f sg	m pl	n pl	f pl				
Nom		-e -eš		ešti -ešte	-ešta	-ešt	N	lom		
Acc	-ęštь	-ęšte	-ešto	-ešte	-ĘSLA	-251	۶ A	.cc		
s	stem for other forms			-ęšt- ste			stem			

Present Active Participle ( $e \sim i$ pres
---------------------------------------------

**4.191** When the comparative stem ends in  $-\check{e}j$ , the nom. sg. masc. is spelled  $-\check{e}i$  (for  $*-\check{e}j$ -b), neut.  $-\check{e}e$ : nov $\check{e}i$  nov $\check{e}e$  'newer', star $\check{e}i$  star $\check{e}e$ 'older'. The rest of the paradigm has the full suffix  $-\check{e}i\check{s}$ - ( $*-\check{e}jb\check{s}$ -) before the desinences (nov $\check{e}i\check{s}$ -a, etc.). Otherwise the nom. sg. masc. is spelled -bi or -ii (for \*-bj-b), the neuter -e, and the rest  $-b\check{s}$ -: e.g. boljbi or boljii, bolje ~ boljbš-a etc. 'bigger'; veštbi or veštii vešte ~ veštbš-a etc. 'greater'. For the formation of the comparative, see §4.7.

4.192 The two sets of formants for the present active participle are correlated to the shape of the present-markers (cf. §6.11). Verbs with -eta

in the 3 pl pres. have -ext- as participial formant; in nom. sing. masc. (with zero desinence) it reduces to -e. Verbs with -qta in 3 pl. pres. have -ext- as participial formant, which is replaced in nom. sg. masc. by the variable  $\{y/e\}$ : -e appears after a soft (truncated) stem, -y otherwise. E.g. prosi-ti prosets 'beg' ~ prose (mn) prosesti (f.); nes-ti nes-qta 'carry' ~ nesy (mn) nesosti (f); plaka-ti placeta 'weep' ~ place (mn) placesti (f).

4.193 In past active participial stems ending in a soft consonant, the -δof the suffix is everywhere replaced by -b-: prošьši, prošьši, prošše, etc.

**4.194** Variants: Occasionally the accusative form of masc. and neut. participles is used as nominative. Conversely, the comparative nom. sing. masc. in -bi/-ii often serves as acc. sg.; in both instances the tendency is to have a single form for nominative/accusative singular masculine, as do other declined words. In comparatives, the -e of nom. pl. is sometimes replaced by the usual nominal desinence -i, e.g. lučbi for lučbi 'better'.

Examples of **comparatives** (*vęštbi vęštbš*- 'bigger', *nověi nověiš*- 'newer'), **present active participles** (*nesy nesošt*- 'carrying', *vęžę vęžošt*- 'tying', *prosę prosęšt*- 'begging') and **past active partiple** (*nest nestš*- 'having carried'). The other case-forms are like *ništb*, see above.

	masculine	neuter	feminine	masculine	neuter	feminine	
N sg	влштьн	ваште	ваштьши	новъи	новће	новжиши	Nsg
A sg	ваштьшь	вжштс	ваштьшя	новжишь	HVB BC	новжишж	Asg
N pl	Вѫштьш€	ваштьша	<b>8</b> АШТЬША	новжише	новѣнша	новтиша	Npl
A pl	ваштьша	DWM I PMA	BWIILIPI	новтиша	нор риша	ило виши	Apl
	stem for other forms		ваштьш-	новжиш-		stem	

	masculine	neuter	feminine	masculine	neuter	feminine	
N sg	несъі	несжште	несяшти	важа	важа	важѫшти	Nsg
A sg	несжшть		несяштя	важжшть	BWWW	важяштя	Asg
N pl	несяште	несжшта	несяшта	ВѦжѫшт€		важашта	Npl
A pl	несяшта	несмшта	нестати	важжшта	вмллшта	מיו שהששמ	Apl
	stem for other forms несжит-			6A#	stem		

	masculine	neuter	feminine	masculine	neuter	feminine	
N sg	просл	Проса	просмшти	несъ	несъше	несъшн	Nsg
A sg	просмшть	просм	прослштж	несъшь	Net PMC	несъшя	Asg
N pl	прослште	Пеосашта	просашта	несъше	несъша	несъша	Npl
A pl	просмштм		просмати	несъша	пссеши	net allim	Apl
	stem for o	ther forms	прослшт-	NECT	ьш-	stem	

			Singu	ılar		Dual			Plura		
		masc.	neut.	fem.	masc.	neut.	fem.	masc.	neut.	fem.	
1	4	-ъ/-ь	-о/-е	-a	-9	-×/	i	-i	-0	-у/-е	Ν
4	4	- D/ -D	-0/-0	-6	-a -ĕ/-i -:		-y/-ę -a		-y/-e	Α	
0	3	-ogo	/-ego	-oję/-eję	-oju/-eju			-ёхъ/-іхъ			G
Ι	٢	-оть	/-еть	-oi/-ei	-0	ju/-ej	u		- A D/ - I A	Ð	L
Ι	)	-omu	/-emu	-00-01	_ăr	ma/-ima		-ĕ	́тъ/-і	mъ	D
Ι		-ěть	/-imь	-ojǫ/-ejǫ	-61	11a/ *111	14	-ĕ	mi/-in	ni	Ι

## 4.2 The pronominal declension.

**4.201** This is the declension of the pronouns to 'this', ono 'that', ovo 'this close by', into 'another', kto-Ztodo 'each', samo 'self', edino 'one, alone', eterto 'a certain', kako 'of what kind?', nikako Ze 'no (kind of)', někako '(of) some (kind)', inako 'of another kind', jako 'of this kind', tako '(of) such (kind)', vosako or vosěko 'every', \*jo 'he' (i-Ze 'he who'), moi 'my', tvoi 'thy', svoi 'one's own', našo 'our', vašo 'your' čii 'whose', and the numerals which have only dual forms: dowa 'two' and oba 'both'.

The stems s- 'this' and ves- 'all' are anomalous and seem to have a "soft" /ś/. See §4.21 for vese (\*ves-) and sice 'of this kind', §4.22 for se (\*se).

**4.2011** *Tuždb* 'foreign, alien' has pronominal and compound forms: see §4.321.

**4.2012** Kotoryi 'which', *někotoryi* 'a certain' and *nikotoryi* 'none' belong to the compound declension.

**4.202** The desinences for nominative and accusative are the same as in the twofold nominal paradigm, but the other desinences are different.

**4.203** Before desinences beginning with  $\check{e}$  or i, stems in k undergo substitutive softening KAI: e.g.  $tak\mathfrak{F} \sim tac\check{e}m\mathfrak{h}$ ,  $tac\check{e}m\mathfrak{h}$ ,  $tac\check{e}x\mathfrak{F}$ ,  $tac\check{e}mi$ , NA dual neuter and feminine  $tac\check{e}$ , Npm taci.

**4.21** V&S& 'all' and sicb 'of this sort, of (such) kind' take the basic desinences that begin with  $\check{e}$ , otherwise the soft variant; i.e.  $\{\check{e}/i\} \sim \{\check{r}/b, o/e, y/\varrho\}$ . Thus soft v&Sego, v&Semu, v&Semb, v&Sei, v&See, v&Sejo, v&Seju; sicego, etc. ~ v&Semb, v&Semb, v&Sema, v&Semi, v&Seis, sicemb, etc. Forms with the desinence -a are written v&Sa or v&Se in glagolitic, but only **BLCA** (never \***BLCA**) in cyrillic. Asf is usually v&So, rarely v&Sjo (never cyrillic \***BLCA**).

62

4.22-4.26

DECLENSION

**4.22** The pronoun sb (or possibly \*s-) has a suppletive stem sij- that apparently was optional in certain nominative and/or accusative forms. In the following table the more common shorter form is given first, followed by slash and the longer form.

	5	Singular			Dual			Plural	
	masc.	neut.	fem.	m n f masc neut				fem	
N	sь/sii	se/sie	si	sija	s	:	sii/si	ai	cio
A	55/511	SC/SIC	siję	Sija	3		się	SÌ	się

The animate masculine accusative singular is sego.

**4.23** The interrogative pronoun  $k \ge to$  'who' has the normal hard pronominal declension except for the extra word-final suffix -to in the nom.; acc.gen. kogo, loc. komb, dat. komu, and (with mutated root-stem) instr.  $c \ge mb$ .

Instr. cěmb early began to give way to kyimb 'which?', cf. §4.323).

Like  $k \ b t o$  are declined  $n \ b k \ b t o$  'somebody' and  $n \ b k \ b t o \ z e$  'nobody'. The particle z e follows the desinence. (Modern editors often write it as a separate word.)

**4.24** The interrogative pronoun  $\check{c}io$  'what' has the following forms: nom. acc.  $\check{c}io$ , gen.  $\check{c}eso$ , loc.  $\check{c}emb$ , dat.  $\check{c}esomu$ , instr.  $\check{c}imb$ . Similarly  $n\check{e}\check{c}io$  'something',  $n\check{c}io\check{c}e$  'nothing' (where the particle follows the desinence).

Alternate forms are found: gen. čuso, česogo, č'sogo, dat. čusomu, čemu, loc. ni [o/pri] česomuže. Ničutože is found, rarely, as gen. Ničuže appears once in Cloz, and four times in Vat.

**4.25** The pronoun  $*j_b$  'he' is not attested in the nominative; its function as third person pronoun is taken by a demonstrative, usually  $t_b$  'that one', less often  $on_b$  'that one yonder'. The other forms of  $*j_b$  all are attested, including acc. sg. i ( $*j_b$ )—but the root-consonant is never explicitly written (\$1.24): \*jego, \*jeje are spelled ego, ee, etc. After prepositions, j is replaced by the palatal nasal nj:  $k_b njemu$ ,  $s_b njimi$ , na  $nj_b$  (acc. sg. masc.), etc.

With the suffixed particle  $-\check{z}e$ , \*jb serves as a relative pronoun 'who, which', and all forms are attested ( $i\check{z}e$ ,  $e\check{z}e$ ,  $ja\check{z}e$ , etc.).

Note that acc. sg. m \* jb is enclitic and may affect a final jer of a verb form, cf. §2.61.

**4.26** In rare instances, forms of the possessives *moi*, *tvoi*, *svoi* are written without the *e* of bisyllabic desinences: gen. sg. fem *moe* for *moee*, inst. *mojo* for *moejo*. They may be simply errors.

4.26-4.3

		masc.	neut.	fem.	masc.	neut.	fem.	
	N	ፐጌ		ТА	нашь	наше	наша	N
	A	= N,G	то	ТЖ	= N, G	паше	нашж	Α
Sing.	G	т٥	ГО	тол	NAU	цего	нашељ	G
	L	TO	мь	тои	нац	нашемь		L
-	D	тол	nov	тои	наш	emoy	. нашен	D
	I	тѣ	мь	тоія	нашимь		нашеня	I
[	NA	ТА	т	ŧ	наша	NAI	ПН	NA
Dual	GL		тою			нашею		GL
	DI		тѣма				DI	
	N	ти			наши			N
	Α	ты	та	тъ	наша	наша	наша	А
Plur.	GL		тѣхъ			нашихъ	L	GL
	D		<b>ተ</b> ቴለጌ			нашимъ		D
	I		тѣми			нашими		

				Si	ngula	r		Dual			Plural		
				m.	n.	f.	m.	n.	f.	m.	n.	f.	
N	къто		N	(и)	(e)	(m)				(и)			N
Α		чьто	A	=N,G	e	坏	IA	•	•	٨	124	<b>^</b>	Α
G	кого	VECO	G	его		EA							G
L	комь	ч€мь	L	EWF	•			ею			ихъ		L
D	комоу	vecomov	D	EWO	емоу						имъ		D
Ι	цѣмь	чиме	I	нмь е		€ŀÆ	има		A	ими		Ι	

# 4.3 The compound or adjectival declension.

This declension has complex desinences that give to adjectival stems an additional meaning of definiteness roughly equivalent to the English definite article: slěpa žena 'a blind woman' versus slěpaja žena 'the blind woman'. The underlying forms combine the twofold desinence with the case- and gender-equivalent form of the pronoun \**j*<sub>b</sub>, e.g. Nsf slěp-a+j-a, As slěp-q+j-q. Adjectives with these compound desinences are commonly

called "long" or "definite" as opposed to the "short" adjectives that have only the twofold nominal desinences. For the use of long forms, see §17.

The ideal normalized OCS forms are obscured first by the inadequacy of both OCS alphabets in representing the glide j (for j plays a crucial role in these desinences) and second by the complex historical evolution that seems to be reflected in the variegated spellings in the oldest surviving manuscripts. As underlying forms for early OCS, the short adjectival forms plus the post-posed pronoun \*jb provide a solid descriptive basis. Thus, for example, the masculine dative singular of *stars* 'old' is posited as {star-u+j-emu}, ideal OCS *starujemu*, attested also in a shape called "assimilated" and one called "contracted", *staruumu* and *starumu*. Similarly *ništb* 'poor' has *ništujemu*, *ništuumu*, *ništumu*.

**4.301** The inflectional suffixes consist, in principle, of (a) the short or nominal desinence, plus (b) the pronominal root \*j-, plus (c) the soft pronominal desinence.

**4.3011** Two systematic processes of simplification apply to produce the ideal OCS forms.<sup>1</sup> **1**. a desinence-initial sequence -oj- or -ej- is deleted. **2**. a nominal desinence with more than one syllable is replaced by the single syllable y or i. This means that the compound desinence consists of VjV(CV)—two or three syllables.

**4.3012** Within the history of OCS, the intervocalic glide *j* tended to disappear, and contraction could take place. Let us posit successive processes of "assimilation" and "contraction". (**3a**) The syllable *je* assimilates to a preceding tense vowel. (**3b**) *j* disappears between like vowels. In effect, *aja* and *aje* > *aa*, *ĕje* > *ĕĕ*, *uju* and *uje* > *uu*, *iji* > *ii*, *qjq* > *qq*, *eje* > *ęę* (while *oje*, *ĕji* and *yję* remain). Finally, (**4**) two like vowels coalesce.

**4.30121** The sequence  $y_{ji}$  presumably lost the j and  $y_i$  contracted to y. Spellings like HOBTHMT, HOBTHYT are not informative (§4.302), but the shorter spellings (HOBTHMT, HOBTHYT and the like) surely imply y.

**4.3013** The role of j (and its absence) in these forms is crucial; the lack of an unambiguous device for representing j in either OCS alphabet obscures the problems. In the following table the theoretical j is written consistently (with no asterisks). The individual gender-case desinences are numbered in order to facilitate discussion; "s" stands for *soft* when needed. The feminine two-syllable nominal desinences *-ama*, *-ax*<sub>5</sub>, *-am*<sub>5</sub>,

<sup>&</sup>lt;sup>1</sup> These are special morphophonemic adjustments that occur only in this specific environment. The English alternants *don't* and *won't* instead of *do not* and *will not* are formed by similar special rules.

		underlying	OCS	assimilated	contracted	processes
1.	NAsm 1s	Ъ/Ь+ <b>ј</b> -Ь	ЪјЬ ЬјЬ		y∣yj ilij	see §4.3021
2.	NAsn 2s	o/e+j-e	oje eje	ee	e	exempt 3b, 4
3.	Nsf	a+j-a	aja	aa	а	3b, 4
4.	Gsmn	a+j-ego	ajego	aago	ago	3ab, 4
5.	Gsf 5s	у/ę <b>+ј</b> -ејę	yję ęję	ęę	ę	1 3b, 4
6.	Lsmn 6s	ĕ/i+ <b>j</b> -еть	ёјеть ijeть	ёёть ііть	ёть іть	3a, 4 3a, 4
7.	Dsmn	u+j-emu	ujemu	uumu	umu	3ab, 4
8.	LDsf 8s	ě/i+ <b>j</b> -eji	ěji iji	ii	i	1 3b, 4
9.	Ismn 9s	оть+ <b>ј</b> -іть еть+ <b>ј</b> -іть	ујіть іјіть	уіть ііть	уть imь	§4.30121 2, 3b, 4
10.	Isf 10s	oj <b>ǫ+j-e</b> jǫ ejǫ <b>+j</b> -ejǫ	QjQ	ŶŶ	ę	1+1, 3b, 4
11.	NAdm	a <b>+j</b> -a	aja	aa	a	3b, 4
12.	NAdnf 12s	ĕ∕i+j-i	ěji iji	ii	i	exempt 3b, 4
13.	GLd	u+j-eju	uju	uu	u	1, 3a, 4
14.	DId 14s	yma+ <b>j</b> -ima ima+ <b>j</b> -ima	yjima ijima	yima iima	yma ima	§4.30121 3b, 4
15.	Npm	i+j-i	iji	ii	i	3b, 4
16.	Apm 16s	у/ę+ <b>ј</b> -ę	yję ęję	ęę	ę	exempt 3b, 4
17.	NApn	a+ <b>j</b> -a	aja	aa	a	3b, 4
18.	NApf 18s	у/ę+ <b>ј</b> -ę	yję ęję	ęę	ę	exempt 3b, 4
19.	Gp 19s	ъ/ь+ <b>ј</b> -іхъ	ъјіхъ ьјіхъ	уіхъ ііхъ	ухъ іхъ	§4.30121 3b, 4
20.	Lp 20s	ёхъ+ <b>ј</b> -іхъ іхъ+ <b>ј</b> -іхъ	ујіхъ іјіхъ	уіхъ ііхъ	ухъ іхъ	§4.30121 3b, 4
21.	Dp 21s	ěmъ+ <b>j</b> -imъ imъ+ <b>j</b> -imъ	ујітъ іјітъ	уітъ іітъ	утъ ітъ	\$4.30121 3b, 4
22.	Ip 22s	ěmi+ <b>j</b> -imi imi+ <b>j</b> -imi	yjimi ijimi	уітъ іітъ	ymi imi	\$4.30121 3b, 4

-ami—which by rule 2 are replaced by y or *i*—have been omitted from the table, where they might fit into ##14, 20–22; note that the compound desinences have no specifically feminine shapes in dual or plural.

**4.302** The attested glagolitic and cyrillic spelling of variants is chaotic. Since a vowel-letter in unblocked position implies syllable-initial *j*, spellings like *novaa* and *novqq* do not necessarily specify that *j* is not present. The sequences sj and yj, bj and *ij*, are neutralized (§2.61), and the spellings of sequences in desinences 1, 19; 6s, 8s, 9s, 14, 15, 19-22 fluctuate considerably. Normal spelling is -yi/-si or -ii/-bi, cyrillic -un/-un or -un/-un or -un/-un or -un/-un or -un/-un sis found (and could be interpreted as two syllables), while -u in soft-stems is rare. Often it is impossible to know whether a graphic s + i (u, un, un) is to be read as two syllables (sji, sjb, yji, yjb) or one (y).

**4.3021** In desinence #1, the first jer is both strong and tense ( $\S2.525$ ), -**b**jb or -**b**jb, and presumably the final jer had disappeared, leaving word-final *j*—which can be represented only by an *i*-letter. Spellings with -*oi*/-*ei* are found, but they are uncommon.

**4.3022** In desinence #6, the spelled sequence -*ěě*- is sometimes replaced by -*ěa*- (chiefly in As, e.g. *na krilě cîkovoněam* 'on the temple parapet' Mt 4:5, vo životě věčoněamo 'in eternal life' J 23:25). Such spellings surely result from arbitrary orthographic rules.

**4.303** OCS and post-OCS mss demonstrate ever stronger tendencies for the pronominal desinences to influence these compound desinences, or replace them. At the same time it appears that a variety of spelling rules tried to maintain a two-letter sequence (particularly aa, uu, yi) that implies disyllabic pronunciation; it is probable that church tradition of recitation required special syllabification.

4.304 Late OCS shows beginnings of replacement of Dsmn -umu by pronominal -omu (twice in Assem), which is found with ever greater frequency in 11th-12th century Rusian mss and "Middle Bulgarian". The forms slěpoumu J 11:37 and provoumu J 19:32 in Mar are very likely illustrations of scribal uncertainty as to the correct spelling of the words. Isolated examples of -eimo (3 each in Zo Mar Sin, Mutrauteïma 'seeking [Dp]' Su) and -eixo (2 in Sin) in participial forms are very likely artificial. The regular forms become staromu like tomu, ništemu like semu, although by tradition the -umu forms occur also.

**4.305** In contrast, the adjectival *-ago* (or *-aago*) is systematically opposed to pronominal *-ogo/-ego* both in South and East Slavic texts until considerably later.

There is a single exception in OCS: єлѣ живого сжща L 10:30 Sav (єлѣ жива As).

**4.31** The compound forms of the **comparative adjectives** and the **active participles** are somewhat irregular in nom. and acc. (cf. §4.191-192). The Ns masc. long form of the comparative is just like the short: *nověi*, *vęštii*. The NAs neuter long form of comparatives is based on the lengthened stem -*ěiš*- (for \*-*ějьš*-) or -*bš*-, *nověišee*, *vęštee*: there are also occasional examples lacking \*-*bš*- (*tačaee* 'better (sort of)' J 2:10 Zo; въшек 'higher' Su 303.20). The active participles use the longer stems: *prosęštee*, *neszšee*, *prošьšee*.

The Ns fem. is made by adding -*ja* to the short form: *nověišija*, *vęštija*; *nesąštija*, *nesąšija*. Ns masc. of the active participles is regularly formed by adding -i (for \*-*j*<sub>b</sub>) to the short form: *nesyi*, *glagoljei*, *prosei*. In the past active participle, the compound desinence -*ai*/-*bi* may appear as -*yi*/-*ii* (*nesyi*, *prošii*), but spellings with -*ei* (*prošei*) are not uncommon (for \*-*bjb*, cf. §§2.51, 2.522).

		Comp	aratives	P	resent Act	ive Partici	ple	Past. Act. Part.	
		nom.	acc.	nom.	acc.	nom.	acc.	nom.	acc.
	М	-i	-ьšьі	-yi/-ęi	-qštii	-ęi	-ęštii	-yi	-ъšіі
Sg.	N	-ьš	ee	-Qšt	ee	-ę	štee	-ъš	e
	F	-ьšija	-ьšqjq	-qštija	-qštqjq	-ęštija	-ęštǫjǫ	-yi -ъšija -ъšei	-ъšqją
	М	-ьšei	-ьšęę	-qštei	-qštęę	-ęštei	-ęštęę	-ъšei	-ъšęę
Pl.	Ν	-ьš	aja	-QŠ	taja	-ęš	itaja	-ъ	šaja
	F	-ьš	ęę	-QŠ	tęę	-ęš	tęę	-ъ	šęę

Here are the NA long forms (compare the short forms in §4.19):

**4.311** Beside the regular nom. sg. masc. pres. act. part. of the *nesyi* type, there are certain cases where y is replaced by  $\rho$ ,  $\rho$ , or a rare glagolitic symbol which we transcribe p: živyi, živ $\rho$ i, živ $\rho$ i, živ $\sigma$ i. This fluctuation is attested only in a few verbs (principally živyi 'living', syi 'being', gredyi 'coming') and is doubtless a reflection of different dialects and morphological innovation. See Koch 553-59.

**4.312** Beside the expected -ei in nom. pl. masc., -ii (or, contracted, -i) is often found. It is to be regarded as a relatively late form.

**4.313** Very rarely, participial forms in -*e* or -*i* stand for any gender-number-case form.

		masculine	neuter	feminine	]
	N	новъи, новъі		NOBAIA, -AA	N
	Α	= Nom or Gen.	Hoboe	новжих	A
Sing.	G	новаего, новаа	лго, новаго	новъіљ	G
	L	новъемь, новъмь, н	овѣѣмь, новѣамь	новћи	L
	D	новочемоч, новоча	очмоч, новочмоч	HOB BH	D
	I	новъінмь, н	овымь	новжіж	I
	NA	Nobnn	ви	NA	
Dual	GL			GL	
	DI	ł	іовънима, новънма		DI
	N	новии	NOBAIA, NOBAA	новъја	N
	A	новъіл	новаја, новаа		A
Plur.	GL	H	ювънхъ, новъхъ		GL
	D	N	овънмъ, новъімъ		D
	Î	N	ювъними, новънми		Ι

# Compound declension paradigms: hard-stem novyi; soft-stem ništii:

		masculine	neuter	feminine			
	N	ништин, ништи	ништее	ништага, -аа	Ν		
	Α	= Nom or Gen.	миштее	ништжіж	Α		
Sing.	G	ништаего, ништ	ааго, ништаго	ништаа	G		
	L	ништиимь, і	иштимь	ништии	L		
	D	ништоуемоу, ништоу	оумоу, ништоумоу		D		
	I	ништиимь,	ништимь	ништжіж	Ι		
	NA	ништага	ии	NA			
Dual	GL	ништоую		GL			
	DI	NI	ІШТНИМА, НИШТИМА		DI		
	N	ништин	НИШТАГА, НИШТАА	ништаа	N		
	A	ништаа			Α		
Plur.	GL	н	штинхъ, ништихъ		GL		
	D	ни	ШТИИМЪ, НИШТИМЪ		D		
	Ι	ништинми, ништими					

**Compound declension of comparatives and participles,** nominative and accusative. Note that the masc. accusative sing. may also have the genitive form, which like all other declensional forms has the variations listed for *ništii*, above.

		·	· · · · · · · · · · · · · · · · · · ·		1
		masculine	neuter	feminine	
Sg.	Ν	влштьи	ваштьшее	ваштьший	Ν
	Α	ваштьшии	umm i binee	ваштьшяж	Α
Pl.	Ν	ваштьшен	влштьшага	ваштьшаа	Ν
	Α	ваштьшаа		В₩ШІРШ₩₩	Α
	N				N
Sg.	N	новън	новжитее	новъншига	IN
	Α	новжишии		новтиштия	A
Pl.	N	новжишеи	новѣншага	новъншал	N
	Α	новъншал			Α
		1		<b>6</b>	1
<u> </u>		masculine	neuter	feminine	
Sg.	Ν	несън	несжштее	• несжштига	Ν
	Α	несжштни	neemastee	несжштжіж	Α
Pl.	N	несжштен	несжштала	неслштаа	N
	Α	неслштаа	несты гала	нестаноми	Α
	NT	· · · · · ·			NT
Sg.	N	просли	Просмштеє	просмштига	N
	A _	прослштии		прослштлія	A
Pl.	N	просмштен	Просмштага	Просмштмм	Ν
	Α	просмштмм	простана	проскатик	A
Sg.	N	несън, несъі		несъщиа	Ν
55.			несъшее		
	<u>A</u>	иесъшии		№€СЪШѪѬ	A
Pl.	Ν	несъшен	несъшага	несъшаа	N
	Α	несъшаа			A

# 4.32 Mixture of the twofold nominal, pronominal, and compound declensions.

**4.321** Some stems occur with desinences that belong to more than one paradigm. Compound-declension forms of such words are not necessarily definite in meaning.

 $Tu\dot{z}db$  (with alternate shapes  $\dot{s}tu\dot{z}db$  and  $stu\dot{z}db$ ) 'alien, foreign' (the antonym of svoi 'own') is chiefly a soft pronominal stem, but it sometimes appears with twofold nominal or compound desinences. *Edins* 'one; only, sole' has predominantly pronominal desinences, but compound desinences are well attested (see SJS, p. 976).

**4.322** The words *manoga* 'many, numerous', *kolika* 'how big?', *tolika* 'so big', *selika* 'as big as this', *elika*, 'so big (relative)', have a mixture of nominal and pronominal forms. The prominal desinences that begin with -*e*- are selected, accompanied by mutation of stem-final velar ( $kg > c_3$ ): instr. sg. masc. neut. *kolicěma*, gen. loc. pl. *tolicěxa*, dat. pl. *tolicěma*, instr. pl. *tolicěmi*. *Manoga* has both the regular nominal forms and these

pronominal ě-forms: mənogomu/mənozěmu, mənogu/mənozěxu, mənogomu/mənozěmu, mənogy/mənozěmi. All of these words have regular definite forms according to the compound declension.

**4.323** The adjectival interrogative kyi, koe, kaja 'which', and also nikyi, nikyiže 'none' and  $n\check{e}kyi$  'a certain' have the stem k- with (1) long desinences (hard) in nominative and accusative forms, and (2) long desinences beginning with y; and the stem koj- with soft pronominal desinences that begin with e.

		Singu	lar		Dual			Plura	1	]							
	masc.	neut.	fem.	masc.	neut.	fem.	masc.	neut.	fem								
N	кујь	koje	kaja	[kaja]			eXii?		eXii:2	aXiii?		aXiii?	ciji		kaja	kvia	N
A	кујб	којс	kojo	[Kaja]	cěji?		kyję	каја	kyję	A							
G	koj	ego	kojeję	() <sub>r</sub>	[kojeju]		l.	kyjixъ									
L	koj	еть	kajaji		ոյշյայ		Бујав			L							
D	koj	kojemu kojeji		kv	jima		k	ујітъ		D							
Ι	kyj	іть	kojeję	ny.	jina		k	yjimi		Ι							

In the table below, j is written but the forms are not asterisked.

Forms with ko- instead of ky- are attested but rare; they are common in post-OCS mss.

# 4.4 The simple nominal declension.

Two groups of stems belong to this declension: one is regular and productive; the other amounts to a list of individual nouns that are anomalous in varying manners, but includes derivational formants that are productive. The regular type is made up predominantly of feminine substantives (counting also the numerals *petb* '5', *šestb* '6', *sedmb* '7', *osmb* '8', *devetb* '9'), but includes some masculines, and one numeral with syntactic gender, *trbe* '3'. They are usually called, on historical grounds, *i*-stems.

The anomalous stems include substantives of all three genders (counting one numeral of feminine gender, desetb '10') and one numeral with syntactic gender, *četyre* '4'.

The normal i-stems have these underlying desinences:

[	Sing	gular	Du	al	Plu	ral	7	
	masc.	fem.	masc.	fem.	masc.	fem.		
N	_		-i	2	-ьје		Ν	
Α	-Р	<b>-</b> Ь		L	-i -i		Α	
G					-Ы	-ьјь		
L	-i		-Ъј	ju	-ьх	•ЬХЪ		
D			-ы	20	-ьn	пъ	D	
Ι	-ьть	-ьть -ьјо		Па	-ы	ni	Ι	
V	•	i						

**4.401** The tense b in Isf, in GL dual, and Gp usually is spelled *i*. The underlying Gp desinence has a strong and a weak jer (-bjb) and therefore can be spelled -bi or -ii in normalized OCS, or -ei and sometimes merely -i in the mss. Thus Gp \*ljudbjb appears as ljudbi, ljudii; ljudei, ljudi. The strong jer of Ism and Dp may also be spelled *e*: ljudbmb or ljudemb. Cf. §§2.61, 2.625.

**4.402** The numeral tree (trie) is inherently plural. The NA form tri is used for both neuter and feminine agreement. The genitive is trei (\*trbjb), spelled also trei, tri; loc. trexs (trexs); dat. trems (trems); instr. tremi.

**4.4031** The majority of substantives of this pattern were feminine, and some foreign names of women were adapted to it (e.g. Agarb, Tamarb). The productivity of this type is assured in part because of the suffix -ostb, used to form abstract nouns from adjectival roots or stems (e.g. bělostb 'whiteness' ~ bělb 'white'; světblostb 'brightness' ~ světblb 'bright').

**4.4032** Masculine substantives are well attested, but the number is restricted and many individual stems tend to take twofold nominal desinences. Most important are: *bolb* 'sick man', *gvozdb* 'nail', *golqbb* 'dove', *gostb* 'guest', *grstanb* 'throat', *zetb* 'bridegroom', *lystb* 'calf (of leg)', *pqtb* 'road', *tatb* 'thief', *tbstb* 'wife's father', *ušidb* 'fugitive', *črbvb* 'worm', *qglb* 'coal', and *ljudbe* (plural only).

Gladb 'hunger, famine' (Ns gladb krěpškě Zo L 15:14; Is glademb Sin [Deuteronomy 32:24]) is in competition with a better-attested twofold nominal gladb.

Zvěrb 'beast', ognjb 'fire', and gospodb 'lord' are treated in §4.51 below.

*Malomoštb* 'cripple; beggar' takes masculine modifiers, but unexpectedly has the Is feminine desinence, *malomoštijq*, Mk 9:43.

Laksts 'elbow; ell', nogsts 'fingernail', paznegsts 'claw' and pečats 'seal' seem to belong to the anomalous type (cf. Vaillant §70). 4.4032-4.410

## Simple nominal declension paradigms:

feminine kostb'bone' and masculine potb 'road'.

Sing.	NA GLD I	кость Кости Костьіж	-ዘເጽ	ПЖТЬ Пжти Пжтьмь	-(емь)	NA GLD I
Dual	NA GL DI	кости Костью Костьма	-ию	ПЖТИ ПЖТЬЮ ПЖТЬМА	-1110	NA GL DI
Plur.	N A G L D I	кости Костьи Костьхъ Костьмъ Костьми	-ии (-еи) (-ехъ) (-емъ)	ПЖТЬЄ ПЖТИ ПЖТЬИ ПЖТЬХЪ ПЖТЬМЪ ПЖТЬМИ	-ие -ии (-еи) (-ехъ) (-емъ)	N G L D I

# 4.410 The anomalous subtypes of simple nominal declension

About fifty substantives (the most clearly attested of which are listed in the next two pages) have, **in addition to the regular forms** given above, other—older—desinences. From the point of view of OCS, the variant desinences are simply irregularities that must be listed in the lexicon, but in terms of linguistic history they are remnants of formerly distinct paradigms, known as *consonant-stems*. These stems tended to replace the old distinctive endings with simple-stem desinences. Masculines and neuters tend to become even more regular by substituting twofold nominal desinences.

The most striking characteristic is special stems in nominative singulars; they may be described in terms of an underlying zero desinence, see §4.415.

	Singular			Dual			Plural			
	masc.	neut.	fem.	masc.	neut.	fem.	masc.	neut.	fem	
Ν	-ь, -Ø	-ø	-Ø	-i	-ĕ? -i?	-i	-е	-0	-i	Ν
Α	•Ь	-0	-ь		-0:-1:	-1	-i	-a	-1	Α
G		•			• • • • • •			-Ъ		
L		-е			-u			-ехъ		L
D		-i					-етъ		•	D
Ι	-ьт	Ь	-ьјо		-ьта		-у? -ь	mi?	-ьті	I

**4.4111** The GL singular has -e, and the nominative plural masculine has -e; these desinences are unique to anomalous stems and they are frequently used. Thus the masculine stem  $d_{bnb}$  'day' has  $d_{bne}$  as Gs and Ls (while  $d_{bni}$  is the only Ds form, but appears also as Gs and Ls). Np is also  $d_{bne}$  (beside  $d_{bnbe}$ ,  $d_{bnie}$ ,  $d_{enbe}$ , §2.65).

**4.4112** The Gp  $-\delta$  and the GL dual -u are shared with the hard twofold paradigm. *Dunb* normally has i-stem Gp *dunui* (*dunii*, *dunei*), but after numerals almost always *dunb* (*denb*).

**4.4113** The disyllabic desinences of masc. sg. instrumental (- $bmb \sim -emb$ ), Lp (- $bxb \sim -exb$ ), and Dp (- $bmb \sim -emb$ ), are somewhat uncertain. The Is might represent i-stem or soft o-stem desinences: there is no way to distinguish whether written -e- represents historical b or not. The attested plural forms are predominantly spelled -emb and -exb; the historically expected desinences have b.

**4.4114** The numeral *četyre* '4' has a single neuter-feminine NA (pl) form, *četyri*. (All anomalous neuter substantival stems have NAp -*a*, like o-stems.)

**4.412** Masculines are few: *dbnb* 'day', *korenb* 'root', *remenb* 'strap', *elenb* 'stag', *stepenb* 'degree'. *Kamenb* 'stone' and *plamenb* 'flame' have alternate NAs forms, *kamy* and *plamy* (only in Su).

**4.413** Most feminines and neuters have a nominative sg. form with a stem that is shorter than the usual stem. Feminines are of two types, traditionally called "long u-stems" and "r-stems".

The former have a full stem ending in  $-\delta v$ -, but Ns in y: e.g. ljuby 'love', Gs ljubave. Here belong: croky or ciraky 'church', neplody 'barren woman', smoky 'fig', svekry 'husband's mother', loky 'pond', brady 'ax', žrony 'millstone', cěly 'cure', xorqgy 'flag' (Gs crokave or cirakave, neplodave, etc.). Expected Ns \*kry 'blood' is by chance not attested: the accusative form kravo functions as nom. The acc. sg. of other stems is also found in nominative function. Bukavi 'writing, letter' is attested only in the plural, with collective meaning.

The two r-stems *mater*- 'mother' and *dvšter*- 'daughter' have *mati* and *dvšti* in Ns (and vocative), As *materb*, *dvšterb*, Gs *matere*, *dvštere*, etc. The genitive form functions as accusative in Sav and Su.

**4.414** Neuters of the anomalous group are "n-stems", "nt-stems", and "s-stems". Stems in *-en* have NAs in *-e*: e.g. *vrěme* 'time' ~ Gs *vrěmene*,

etc. Also brěmę 'burden', čismę 'numeral', imę 'name', pismę 'letter', plemę 'tribe', sěmę 'seed'.

Stems in -et- also have NAs in -e: e.g. otroče 'child' ~ Gs otročete, etc. This formant was surely productive, serving to create words denoting young living creatures; the list of stems that happen to be attested in OCS is short: osble 'young ass', ovbče 'lamb', kozble 'kid', žrěbe 'foal', kljuše 'draft animal'.

Neuters like NAs *slovo* ~ G *slovese* 'word' are always paralleled by forms without *-es-*; they will be discussed below, §4.55.

**4.415** The shorter nominative shapes represent underlying forms with a zero desinence that is in effect a word-boundary. Stem-final C + desinential  $\emptyset$  violates the requirement that every word must end in a vowel. The productive *nt*-type ({otrok- $et+\emptyset$ } > *otroče*) and the closed group of neuter *n*-stems ({sěm-en+ $\emptyset$ } > *sěme*) involve morphophonemic rules that are required elsewhere in the system. The  $\bar{u}$ -stems ({crьk- $bv+\emptyset$ } > *crьky*) are subject to a special alternation that recurs in certain verbal stems ( $\{11.341\}$ ). The two fundamental kinship terms that constitute the class of *r*-stems ({mat-er+ $\emptyset$ } > *mati*, {dbst-er+ $\emptyset$ } 'daughter') are isolated relics. The masculine *n*-stems ({plam-en+ $\emptyset$ } > *plamy*) and the extremely variable *s*-stems ({těl-es+ $\emptyset$ } > *tělo*) are irregularities that must be listed in the lexicon.

# 4.5 Mixture of the two nominal declension types.

A number of substantives are attested with forms from the twofold nominal and the simple declension (including its anomalous subtypes). From the historical point of view, the variation illustrates the general tendency for the simple nominal declension to be restricted to feminines, and for masculines and neuters to follow the twofold paradigm.

**4.51** Masculine i-stems frequently have singular desinences of the soft twofold declension: zverb wild beast' has Ds and Ap zveri, but Gs zverja and Np zverbe. Ognjb 'fire' also uses chiefly i-stem desinences, but also occasionally twofold soft nominal forms. The extremely common gospodb 'lord' shows great variation, in part obscured by the fact that the stem is abbreviated (e.g.  $\vec{r}_b$ ,  $re\vec{a}_{Mb}$ ). It is clear that i-forms and hard twofold forms are used, e.g. Ds  $gospodi \sim gospodu$ , but spellings like  $\vec{r}_{N}$ , implying gospodju, raise problems of phonological innovation that cannot be treated here.

**4.52** Masculine substantives with the suffixes -telj- and -arj- (both indicating agent or actor) have the normal soft twofold desinences in the singular, but in the plural the stem seems to have been hard (-tel-, -ar-), and desinences are usually those of the anomalous type of simple-declension. With rare exceptions, the Np is -e (učitele 'teachers', rybare 'fisher-

men'), but often the spelling indicates the palatal stem of the singular here as well, at least for -telje,  $-\pi\epsilon\lambda\epsilon$ . The Gp  $-tel\delta$ ,  $-ar\delta$  is less common than the softened stem type -teljb, -arjb (although we must be cautious in interpreting the spelling). Ap has only the soft twofold -e. Instr. pl. is found both with -ly, -ry and with -li (-lji). (Note that many scribes do not distinguish li from lji, and very few distinguish r from rj before any vowel, cf. §§1.31–1.331.)

**4.53** A number of plural nouns with stems in -an--en- ordinarily have the anomalous simple-declension desinences: *izmailitene* 'Ishmaelites' ~ Lp *izmailitenexe*; graždane 'townsmen' ~ Lp graždanexe. Twofold desinences are also found, notably the Ap in -y. (Cf. Vaillant, §71.)

These nouns denote national groups and classes. The corresponding singular and dual forms normally have the derivational suffix -in-, followed by the regular hard twofold desinences: \*izmailitěnint, graždanint.

**4.54** The feminines with stems in - $\delta v$ - (Ns -y, §4.413) regularly have the singular and NAp of the anomalous declension, but other plural forms from the hard twofold feminine (a-stem) paradigm: *crbky* 'church', Gs *crbkove* ~ NAp *crbkovi*, Gp *crbkovo*, Lp *crbkovaxo*, Dp *crbkovamo* (Ip \**crbkovami*, attested in post-OCS mss).

**4.55** Neuter "s-stems" exhibit anomalous simple desinences added to the suffix -es- beside twofold endings without this suffix: NAs tělo 'body' ~ Gs těla/tělese, Ls tělě/tělese, Ds tělu/tělesi, etc. Usage varies with different words and mss (cf. Vaillant §73). Here belong: nebo 'sky, heaven', tělo 'body', drěvo 'tree', slovo 'word, speech', čudo 'wonder', divo 'miracle', dělo 'work', kolo 'wheel', ljuto 'misdeed', istesě, istesa (dual and pl. only) 'kidneys, loins', lice 'face', oko 'eye', uxo 'ear'. Before -es- the stems of lice, oko, uxo undergo substitutive softening: ličese, očese, ušese.

**4.551** Oko and uxo have special dual forms: oči uši, GL očbju/očiju, DI očima ušima. They may take feminine modifiers. Both words are rare in the plural.

# 4.6 Declension of the personal pronouns.

The personal pronouns have no formal expression of gender, and their declensions are unique, with little resemblance to the other declension types. For the third person, the demonstrative \*j- is used (cf. §4.25), suppleted by nominative forms of  $t_{\delta}$  (or, less commonly,  $on_{\delta}$ ).

4.61-4.70

	N	A	G	L	D	Ι	
1st pers. sg.	агъ	mę	mene	m	ьně	тъпоји	ʻI'
2nd pers. sg.	ty	tę	tebe	te	bě	toboję	'thou'
reflexive	-	sę	sebe	se	bě	soboję	'self'
lst pers. du.	vě	na	naju 1		nan	na	'we two'
2nd pers. du.	v	a	va	vaju		vama	
lst pers. pl.	my	ny	na	SЪ	патъ	nami	'we'
2nd pers. pl.	v	y y	vasъ		vатъ	vami	'you'
	N	Α	G	L	D	Ι	

**4.61** Beside  $m \sin e$  and  $m \sin oj e$  there are enough cases of  $m \sin e$  and  $m \sin oj e$  (beside many with m'n-/mn-) to make it impossible to decide which form was really normal. Beside Gs *mene*, Euch and Sin have a number of cases of mne/m'ne.

**4.62** In KF, *ny* functions as Np as well as acc. (According to comparative evidence, this seems to be a Bulgarian-Macedonian feature.)

**4.63** The dative forms  $m \omega n \check{e}$ , teb $\check{e}$ , seb $\check{e}$  are paralleled by mi, ti, si. The difference seems to be that the longer forms are independent, while the short forms are enclitic, and stand after the first accented word of a clause. In a few rare instances short dual and plural forms are found, Ddu va, Dp ny, vy.

**4.64** The genitives *mene tebe nass vass* commonly function as accusative, but this usage is apparently an innovation within the OCS period. Cf. Mt 4:6 pusti *sebe* dolu 'throw thyself down' Zo (As vrb3i *se*, cf. §21.61)

**4.65** The NA dual forms are not needed often; acc. *na* appears only five times. The specifically dual NA *va* and *na* are, on the whole, replaced by the plurals *vy* and *ny*.

# Excursus

Formation of the comparative.

4.70 Comparatives of OCS adjectives end either (1) in -ьі (neut. -e, fem. -ьši) or (2) in -ěi (neut. -če, fem. -čiši). They reflect either (1) the unproductive underlying suffix {jьj-ьš} or (2) the productive {čj-ьš-}.

The declensions are treated in §4.19 and §4.31. The NA short neuter form may be used as an adverb.

OCS has no morphological superlative; the sense of English superlatives is achieved by phraseological means, e.g. L 9:48 iže bo estъ mьnjii vьsěxъ vasъ, lit. 'for he who is the *lesser* one *of* you *all*, i.e. the least; L 13:2 grěšьněiše pače vьsěxъ člověkъ 'more sinful than all men'.

**4.71** The suffix {jbj-bš} is found with a limited but important group of stems. Before the suffix, the stem-final consonant undergoes iotation if possible. Nsm may be analyzed as taking a zero desinence; stem-final š is deleted (cf. §4.415). The NA neuter is unique; the soft desinence *-e* is added to a shortened suffix {j}. Thus { $vet-jbj-bš+\emptyset$ } > vestbjb, {vet-j[bj-bš]+e} > veste.

(1) The comparative stem is unlike the corresponding positive stem: boljbi bolje boljbši 'bigger', vęštbi vęšte vęštbši 'greater' (~ velbi or velika 'big, great'); тъпјы тъпје тъпјьši 'smaller, lesser' (~ mala 'small'); lučbi luče lučbši 'better', unjbi, unje, unjbši 'better' (~ dobra 'good'); gorjbi gorje gorjbši 'worse' (~ zəla 'bad').

Some comparatives are attested only as adverbs, e.g.  $ni\bar{z}e$  'lower' (the positive  $niz\bar{z}k\bar{z}$  'low' by chance is not recorded in OCS).

**4.72** The suffix -*ěi* (neut. -*ěe*, fem. -*ěiši*) is used with all other stems. It causes KI mutation, with automatic shift of *ě* to *a* after palatal consonant (§3.5 c1): stars 'old' ~ starěi starěe starěiši; dobljsi 'brave' ~ dobljai dobljae dobljaiši; monogo 'much, many' ~ monožai; gorjoko 'bitter' ~ gorjočai, and the like.

The underlying suffix is  $\{\check{e}_{j-5}\}$ . The masculine zero-desinence applies (as in §4.71 above). The NA neuter singular also is shorter; it omits the two units  $\delta \check{s}$ .

The underlying {ě} sometimes is spelled with "ě" (cyr. t) instead of "a" (e.g. Mt 21:36 множища 'a larger number of [Apm]' Sav; J 4:1 тъпоžaišę ZoMar).

# On the formation of certain adverbs

**4.80** A large percentage of OCS adverbs must be regarded as given lexical items, historically of diverse formation. Here the adverbs based on pronominal stems will be listed; they constitute an important subsystem. The productive types of adverbs formed from adjectives will be mentioned.

**4.81** Adverbs may be formed by adding special suffixes to the pronominal roots k- (interrogative), t- (pointing, general), s- [or  $\dot{s}$ -] (close by), ov- (pointing, distant), \*j- (relative, usually with added  $\ddot{z}e$ , cf.  $i\ddot{z}e$  'he who'), in- 'another', vis- [or vis-] 'all' (general), nik- (negative).

kъde	where	kamo	whither, to what place	kodu, kodě, otъ kodu	whence, from what place
tu	there	tamo	thither, to that place	tǫdu, tǫdě otъ tǫdu	thence, from that place
sьde	here	sěmo	hither, to this place	sodu, sodě, otъ sodu	hence, from this place
оvъde onъde	there, here yonder, over there	ovamo onamo	thither, hither to that place	ovodu onodu	there over there
ide ideže	in the place where	jamo jamože	to the place where	jǫdu, jǫduže	from the place where
inъde	elsewhere, in another plac	inamo :e	to another place	ínǫdu, inǫdě	from another place
vьsьde	everywhere	vьsěmo	to all places	vьsǫdu, vьsǫdě; otъ vьsodu	everywhere; from every- where
nikъde(ž	e) nowhere	nikamo	to no place		

4.811 Expressing positional relations:

The forms with suffix  $-qdu/-qd\check{e}$  are rare without the preceding  $ot_{\delta}$ , and can mean movement about a place, rather than 'from'.

The meaning of motion is often lost in *tamo*, and it is contrasted to *tu* as more general and distant. Some of the *-de* forms have variants in *-žde* (Su); *vbsbžde*, *inžde*, *ižde* or *iždeže*. Note the compounds *doideže*, *donjbdeže* (rarely *donjbžde*, *doižde*) 'to the place where, up to, until; during'.

To these series may be added voně 'outside', vono '(to) outside', vonějędu/vonějędě 'outside'; ętro 'inside', ętro/vonętro '(to) inside', izętro, ętrojędu, ętrojędě 'from within'; oto obojędu 'from both sides' (cf. oba 'both').

**4.812** Expressing time: kogda 'when', togda 'then', ovogda 'at this time, now', egda 'when' (and vonjegda, cf. §3.3103, 'during'), vosegda 'always', nikogdaže 'never', někogda 'at one time, formerly', togdaže 'at the same time', and inogda 'another time; once'. Beside the forms in -ogda are forms in -ogda (kogda, togda, etc.); this is not a phonetic change of o to o, but a replacement of the old rooots ko-, to- by contemporary ko, to (as in koto and kyi).

**4.813** Expressing manner: kako 'how', tako 'thus', jako/jakože 'in the manner that', *inako* 'otherwise', vbsěko/vbsako 'in all manners', sice 'in this way, thus'.

4.814 Three productive types of adverbs are formed from adjectives.

(a) The short neuter singular accusative and (b) the short neuter locative may function as adverbs. Some stems prefer one or the other, and some stems allow both: *dobro* or *dobrě* 'well', *gorjъko* or *gorjъcě* 'bitterly', *različъno* or *različъně* 'differently'. Comparative evidence shows that both types were current in the oldest forms of Czech and of Balkan Slavic, but it is probable that the usage with individual words varied from dialect to dialect.

(c) Adjectives with the derivational suffix - $\omega sk$ - (§28.41) form adverbs of manner in - $\omega sky$ . Relatively few are attested: e.g. vraž $\omega sky$  'inimically', rab $\omega sky$  'slavishly', mąž $\omega sky$  'like a man, courageously', p $\omega \omega sky$  'like a dog', mir $\omega sky$  'in a worldly (secular) manner',  $gr\omega \delta sky$  'in Greek', evrěisky 'in Hebrew', and latin $\omega sky$  'in Latin'.

# CHAPTER FOUR

# CONJUGATION

# 5. Fundamental notions

**5.1** Every OCS verb belongs to one of two **aspects**, *perfective* or *imper-fective*: the former indicates that the action of the verb is limited by an absolute boundary (that is, that the action or process reaches its culmination), while the latter denotes no absolute boundary of the action. Verbs of motion have a further subdivision within the imperfective aspect: *determined* verbs denote a motion being carried out at one time in one direction, while *non-determined* verbs do not have this denotation. Verbal aspect is inherent (like the gender of nouns), and is not necessarily expressed by any formal means (see §5.70).

**5.2** The system of forms includes three **tenses**—a *present* (which may function also as future) and two past tenses, *imperfect* (or coordinate past) and *aorist* (simple past); **imperative**; five **participles** (*present active* and *passive*, *past active*, *resultative*, *past passive*); a verbal **substantive**; and two inflexible forms, **infinitive** and **supine**. In each tense three **persons** in **singular**, **dual** and **plural** may be expressed, although in certain instances a single form may function for two persons. The imperative has no first person singular, and the third person forms are rare. The participles, except the resultative (§11.2), have full adjectival declensions, both according to the twofold nominal and the compound declensions (cf. §§4.19, 4.31), and the verbal substantive has a full declension, although the dual and plural forms are little used.

**5.3** Every verbal form consists of a *stem* plus a *desinence*—an inflectional suffix. The desinence may be simple, a single suffix, or it may be complex, made up of a terminal suffix (which is in final position) and a non-terminal suffix. Definitions of the various suffixes will be given in the following paragraphs. The stem of any given form is obtained by subtracting the suffix (simple or complex).

In some verbs, one invariable stem is found in all forms (e.g. 'carry'; infinitive nes-ti; 1 sg. pres. nes-q, 3 pl. nes-qtb; imperfect nes-ĕaxb; aorist

nes-охъ; past act. participle nes-ъ; l-participle nes-lъ); in others there are two stems (dělaj-qtъ 'they do' ~ děla-ti 'to do'; plač-qtъ 'they weep' ~ plaka-ti 'to weep'); and in still others there are three stems (vidě-ti ~ viždq 'I see' ~ vid-qtъ 'they see'; dvignq-ti 'to move' ~ dvign-etъ 'he moves' ~ dvig-oxъ 'I moved').

Many verbs have prefixes. For the purposes of conjugation, the prefixes may be ignored. The term *verb*, therefore, will be used in the description of conjugation to refer to all lexical items which have the same stem (as defined above), regardless of prefixes. Thus a statement about the treatment of "the verb *nes-qts* 'carry'" applies equally to  $v_{\overline{v}}$ -nesqts 'bring in', *iz*-nesqts 'carry out', and other prefixed stems. Many verbs do not occur without prefixes. This will be indicated by writing a hyphen before the stem, e.g. -*věštaj-qts* stands for *vzz*-věštajqts 'announce', *otz*-věštajqts 'answer', *pro*-věštajqts 'proclaim', and others. Occasionally the hyphen can be used to show that the forms with prefix behave somewhat differently than the unprefixed verb.

**5.31** The prefixes are:  $v_{\overline{o}}$ ,  $v_{\overline{o}z}$ , do, za, iz, mimo, na,  $nad_{\overline{o}}$ ,  $niz_{\overline{o}}$ , o(b),  $ot(_{\overline{o}})$ , po,  $pod_{\overline{o}}$ , pri, pro, pre,  $pred_{\overline{o}}$ , raz,  $s_{\overline{o}}$ , and u. On the variant shapes of the prefixes ending in consonants, see §3.311 ( $v_{\overline{o}z}$ , iz, raz), §3.312 (ob), and §3.3121 (ot). Note that in a very few verbs the particle ne is prefixed to a verb: videti 'see', nenavideti 'hate'; mosti 'be able', nemosti 'be weak, ill'.

PsSin attests ten forms with the prefix vy-, surely an early western dialect morpheme that was replaced in usual OCS by *iz*-. cf. Lunt *IJSLP* 39–40 (1996): 283–84; Koch p 545.

5.4 The whole conjugation can be described in terms of stems, desinences, and rules governing their combination. Regular verbs are those whose forms can all be predicted from a basic stem, with the aid of appropriate rules. The basic stem is to be found either in the infinitive or in the third person plural present. In the great majority of cases it is the longer of the two stems: for example moli-ti 'to beg' ~ mol-etz; sěja-ti 'to sow'~ sěj-ots; but děla-ti 'to do' dělaj-ots; gre-ti 'to bury' ~ greb-ots (the desinences here are -ti for infinitive and -ots/-ets for 3rd person plural present). In some verbs, however, the two stems are the same length; in such cases the basic stem is the one from which we can gain the most information. For example, the two infinitives ves-ti 'lead' and les-ti 'go' both have stems ending in -s; the corresponding 3 pl. forms are ved-oto and lez-ots, respectively. Now, the sequences dt and zt do not occur: st is found instead. If we regard these infinitive forms as ved-ti and lez-ti, we can predict that in the overt or surface form they will have -sti (§3.3131). Since the third plural present is the form that gives the most information,

5.4-5.5

CONJUGATION

it is the basic stem. This goes for the distinctive forms of the 3 pl.  $\mathbf{\check{zeg}}$ - $\varrho t \mathbf{\check{s}}$ 'burn' and **rek**- $\varrho t \mathbf{\check{s}}$  'say': we can predict the infinitives  $\mathbf{\check{zesti}}$  and *rešti*, but not vice versa (§3.3131a).

In citing verbs, only the basic stem (inf. or 3p pres) will be given; for example, moli-ti, moně-ti, kriča-ti, rinq-ti, věrova-ti, kaza-ti, plaka-ti ~ tep-qt5, greb-qt5, plěv-qt5, čuj-qt5, uměj-qt5, kopaj-qt5, žeg-qt5, pon-qt5.

Thus far we have been dealing with the "real words" as we perceive them in written form and interpret them as strings of morphophonemes. Now we will reanalyze them as theoretical underlying strings of morphemes.

**5.41** Every basic stem may be regarded as containing a suffix that specifies that the stem is a verb and at the same time determines the set of forms that can be derived from the stem. These verb-forming suffixes will be called *classifiers*. There are five overt classifiers that end in a vowel, -i+, -e+, -a+, -ova+, and -nq+, and two that end in the consonant /j/, -aj+ and -ej+. Further there is a zero-classifier,  $-\emptyset+$ ; it has no phonetic value. The symbol + is used to mark the end of the basic stem.

The lexical shape of the basic stems written above in terms of real words is then: mol-i+, mon-e+, kric-a+, ri-nq+, ver-ova+, glagol-a+,  $plak-a+ \sim tep-0+$ , greb-0+, plev-0+, cuj-0+, um-ej+, kop-aj+, zeg-0+, pbn-0+.

5.5 Basic stems end either in a consonant or in a vowel and they are accordingly called consonantal or vocalic basic stems. (Stems with the classifiers  $-a_i + , -e_i + ,$  or  $-\emptyset +$  are consonantal.) Similarly, some desinences begin with a consonant (consonantal suffixes) and others with vowels (vocalic suffixes). Since OCS structure does not permit certain successions of phonemes, the addition of suffix to stem may entail a change: most frequently the stem is altered, but some suffixes may adapt to the stem. For instance, the consonantal basic stem ved-O + (vedots 'they lead') appears as ves- before the consonantal infinitive desinence -ti (vesti) and vebefore the consonantal suffix -l-3 of the resultative participle (vel3). The vocalic basic stem kaz-a+ (kaza-ti 'to show') is kaž- before vocalic suffixes of the present tense (e.g. 3 pl. kažota) but it causes the vocalic imperfect suffix -*ĕax*- (compare *vedĕaxo* 'they were leading') to become -ax-(kazaaxo 'they were showing'). The shortening or modification of the stem is called *truncation*.<sup>1</sup>

<sup>&</sup>lt;sup>1</sup> Some rules of truncation are not simply phonological in nature, for they require special morphological marking for certain morphemes, e.g. the imperfect {ĕa}.

**5.6** Truncation adjusts the theoretical underlying form to a shape that conforms to the structural constraints of the (C)VCVCVCV surface structure (see pp. 32-34). If a vocalic suffix is added to a vocalic stem, the  $\{VV\}$  must be modified, usually by change or elimination of the first vowel. If a consonantal suffix is added to a consonantal stem, the  $\{CC\}$  sequence remains if it fits the patterns for permitted clusters (§2.522), but impermissible sequences are modified. The exact processes to be called for depend on the individual morphemes. The desinences specify the following categories:

- (1) the infinitive and supine;
- (2) the aorist and the resultative (or *l*-) participle;
- (3) the past active participle;
- (4) the past passive participle and the verbal substantive;
- (5) the imperfect tense;
- (6) the first person singular present;
- (7) the other forms of the present tense, the imperative, and the present active and passive participles.

The first three call for C-desinences and the last three for V-desinences, but (4) includes both consonantal and vocalic allomorphs. This structural fact results in pervasive surface formal distinctions traditionally defined as the *present* versus the *infinitive/aorist* stems.

**5.601** The underlying formulas may look very like or very unlike the surface forms they generate. For example, here are the infinitive, the 3rd person singular aorist (both formed with C-desinences), and the 1st person singular present (with a V-desinence) of 'to beat' and 'to testify'. The stem in one is the root with a zero suffix. In the other the root with a suffix and prefix make a verbal stem that is converted into an actor/agent by a noun-forming suffix (*telj*); this in turn becomes an abstract noun (*svěděteljustvo* 'testimony') that is made into a verb (by -*ova*+), and that complex stem is perfectivized by another prefix (*za*-).

```
 \{bbj-Ø+ti\} > biti \{(za-(((sb-((včd)-č+))telj-)bstv-)ova)+ti > zasbvěděteljbstvovati 
 \{bbj-Ø+x-Ø\} > bi \{(za-(((sb-((včd)-č+))telj-)bstv-)ova+x-Ø > zasbvěděteljbstvova 
 {bbj-Ø+Ø-Q} > bbjp {(za-(((sb-((včd)-č+))telj-)bstv-)ova)+Ø-Q} > zasbvěděteljbstvujp 
 Keep in mind that in the surface forms the desinences are usually recognizable; by defini-
tion, the stem (whether truncated or full) is what remains when the desinence is removed. 
 Homographs may represent different words with different underlying structure. Thus prosi 
 can be 3rd singular aorist '(he) begged' or 2 sg. imperative 'beg!'; the context will usually 
 indicate which is more plausible. The aorist 3 sg has past-marker {x} and zero desinence; 
 the imperative sg. has non-terminal -i- plus zero for person: therefore
```

 ${pros-i+x-\emptyset} > prosi \sim {pros-i+i-\emptyset} > prosi.$ 

The former contains the full basic stem-the theoretical  $\{x\}$  of the complex desinence is

5.601-5.602

CONJUGATION

deleted because a consonant cannot stand at the end of a word. The latter has a VV sequence; the general rule is that the second vowel prevails—the first is either lost (as here) or converted to j. The segmentation is therefore aorist <u>prosi</u> versus imperative <u>pros</u>-i. Similarly, 1 sg present *rinq* 'I push' is <u>rin</u>- $q < \{ri-nq+\emptyset-q\}$ , while 3 sg aorist is <u>rinq</u> <  $\{ri-nq+x-\emptyset\}$ .

**5.602** The theoretical basic stems with the desinences and generative rules allow prediction of the number and types of truncated stems. There is a primary division between vocalic and consonantal basic stems. Two other factors affect the classification of conjugational types. First, vocalic stems ending in a front vowel (*i* or  $\check{e}$ ) require a non-terminal vocalic present-marking morpheme g/i; all other basic stems take g/e. And second, some basic stems end in *j* and others when truncated end in a palatal consonant: this affects the selection of certain allomorphs in conjugation. The remaining basic stems are those that have the classifier -nq+ or consist of a root ending in a consonant other than *j* plus zero classifier—thus effectively having stem-final C. Types 6–7 below will be referred to as *hard*, types 8–9 as *soft*.

Here are the possible types of regular verbs (the numbers are for reference to the following illustrative synopses of forms):

A. Basic stem ends in a vowel

1i+	i-verbs; e.g. prositi 'beg', močiti 'torment'	€∕i
2ĕ+	ě-verbs; e.g. тытёtі 'think'	€∕i
$2b. \{X-\check{e}+\} = /\check{s}-a/$	ša-verbs; e.g. slyšati {slyx-ě+} 'hear'	ę/i
	: surface ča ža ša šta žda = {kě gě xě skě zgě}	
3. <i>j</i> -a+	ja-verbs; e.g. dějati 'do'	ǫ∕e
[2b] NB: bojati sę	'fear' and stojati 'stand' belong with 2b, ša-verbs	ę/i
4ova+	ova-verbs; e.g. milovati 'have mercy	ǫ∕e
5. C-a+	Ca-verbs; e.g. glagolati 'speak', vęzati 'tie'	ǫ∕e
6 <b>n</b> q+	no-verbs; e.g. rinoti 'push', dvignoti 'move	'ǫ∕e

B. Basic stem ends in a consonant

7. other than $j-\emptyset+$	C-verbs; e.g. nesǫtə 'carry', rekǫtə 'say	ǫ∕e
8a. aj+	aj-verbs; e.g. dělajęts 'do'	ǫ∕e
8b. ěj+	ěj-verbs; e.g. umějąts 'know how to'	ǫ∕e
9. <i>j-</i> Ø+	j-verbs; e.g. bijęts 'beat', kryjęts 'cover'	ǫ∕e

The table gives samples of typical forms of the major types: a = infinitive, b = 1 sg aorist, c = past passive participle (nom. sg. masc.), d = 1 sg. imperfect, e = 1 sg present, f = 3rd plural and 3rd sing. present, g = present active part., nom. sing. masc.-neuter and feminine, h = present passive participle, i = 2nd person singular imperative

	a	b	с	d	e	f	g	h	i
	inf	ls aor		ls imperf		-	Ns mn/f	pres pass	2s imv
			participle		pres	pres		participle	
1	<b>prosi</b> ti	prosixъ	prošenъ	ргоšаахъ	prošę			prosimъ	prosi
						prositъ	-ęšti		
2	mьněti	<b>тьпё</b> хъ	тьпěпъ	<b>тьпё</b> ахъ	тьпјо	•	mьnę	<b>тьп</b> ітъ	тьпі
						тыпіть	-ęšti		
2b	slyšati	slyšaxъ	slyšanъ	slyšaaxъ	slyšę	slyšętъ	slyšę	slyšimъ	<b>slyš</b> i
						<i>slyš</i> itъ	-ęšti		
3	<b>děja</b> ti	dějaxъ	dějanъ	dějaaxъ	dě <b>j</b> q	dějǫtъ	děję	dějemъ	<b>děj</b> i
						<i>děj</i> etъ	-qšti		
4	milovati	milovaxъ	<b>milova</b> nъ	<b>milova</b> axъ	<b>miluj</b> q	milujqtz	miluję	milujemъ	<b>miluj</b> i
						milujetъ	-qšti		
5	vęzati	vęzaxъ	vęzanъ	<b>vęza</b> axъ	vęžę	vęžǫtъ	vęžę	vęžemъ	vęži
						vęžetъ	-qšti		
6	ringti	rinqxъ	rinovenъ	rinĕахъ	rinq	rinqtъ	riny	<b>rin</b> omъ	<b>rin</b> i
						rinetъ	-Qšti		
7	nesti	nesoxъ	nesenъ	nesĕaxъ	nesq	nesqtъ	nesy	<b>nes</b> omъ	nesi
		něsъ				nesetъ	-qšti		
	rešti	rekoxъ	rečenъ	геčаахъ	rekç	rek qt b	reky	rekomъ	гьсі
		rěxъ			-	rečetъ	-qšti		
8a	dělati	dělaxъ	<b>dĕla</b> nъ	dělaaxъ	<b>dělaj</b> ę	dělajotъ	dělaję	dělajemъ	<b>dělaj</b> i
					- •	dělajetъ	-qšti	-	-
8b	uměti	uměxъ	uměnъ	umĕaxъ	umějo	umějqtъ	uměję	umějemъ	uměji
					2.	umějetъ	-qšti	-	•
9	biti	bixъ	выјепъ	вьјаахъ	въјо	bьjotъ	ьје	<b>ььј</b> етъ	bьji
			-	-	••	bыjetъ	-ošti	•	•
						-	-		

Notice that in this table j is written before front vowel: je je ji (instead of the unblocked "e e i" of the mss.; cf. §1.24).

The C-verbs (type 7 here) have many idiosyncratic variations and irregularities, see §15.8.

**5.70** Before proceeding to the description of the morphology of the verbal categories, let us survey what may be called aspect-morphology, an important part of the word-formation of verbal stems.

The aspect of individual verbs cannot always be determined, since there is no formal criterion, and most contexts permit the use of either aspect, with appropriate modification of the meaning. Indeed, variant readings of a single passage may show different aspects—or at least varying spellings that scholars interpret as signifying different aspects. Even in the earliest OCS period there must have been dialect differences both as to the aspect of certain verbs and as to the details of aspect-morphology, and some variation must have been in the original texts. Further variants were doubtless introduced by scribes from different regions and later periods. But the system as a whole is clear, despite variations and doubtful cases.

(Modern scholars, including native speakers of Slavic languages, not infrequently disagree about the aspect-definition of some OCS verbs.)

5.71 Verbs generally appeared in pairs, one perfective (P) and one imperfective (I). Exceptionally a verb could function in both aspects, and surely there were a few verbs which had no exact mates of the opposite aspect. The relationship between the two members of an aspect pair may be that of a prefixed form opposed to a non-prefixed one (tvoriti I ~ stovoriti P 'to do'), difference in verbal classifier (stop-i-ti P ~ stop-aj-ots I 'tread'), difference in classifier plus modification of the root (prost-i-ti P ~ prašt-aj-ots 'forgive'), or, in rare and not altogether certain cases, of suppletion, i.e. completely different stems (glagola-ti I ~ rek-ots 'say', meta-ti I ~ vrbg-ots 'throw').

[In the following paragraphs, verbs will be cited in basic-stem form without the -ti or -qta used elsewhere in the book; e.g. prost- $i \sim prast-aj$  for prostiti  $\sim prastajqta$ . Hyphens set off prefix and the verbal derivational suffix under discussion.]

The most easily definable types are those where the prefixed verbs have different classifiers, often with a variation in the root as well. In the vast majority of pairs the imperfective has the classifier -aj. The most important pattern shows -i in P versus -aj (with possible root modifications) in I; the underlying structure has two classifiers  $\{-i+aj+\}$  plus rules of root-vowel alternation. In other major patterns the imperfectivizing  $\{-aj+\}$  takes the place of the  $\{-\emptyset+\}$ ,  $\{-nq+\}$ ,  $\{-a+\}$ , or  $\{-\check{e}+\}$  of the perfective. And finally there are some minor types and some individual stems that allow competing forms in one or both aspects.

5.711 Classifier -*i*+ followed by imperfectivizing classifier -*aj*+, plus a rule that root-vowel o is to be replaced by a (and e by  $\check{e}$ , though examples are rare). The underlying sequence of vowels (*i*-a) provides the conditions for iotization (§6.13) if mutation is applicable. This is the most wide-spread and most certainly productive group. Over 180 pairs are attested, while many more are clearly implied by the presence of the two aspect-stems compounded with different prefixes (e.g. on the pattern sb-klěst-i ~ sb-klěšt-aj, is-tbšt-aj and many others, the forms po-těšt-i and u-těst-aj imply both \*potěštaj and \*utěšti). It seems safe to surmise that many verbs attested in only one aspect-stem belonged to this type.

	• 1	*	-		
pri-bliž-i	pri-bliž-aj	near	u-tvrъd-i	u-tvrьžd-aj	make firm
raz-ląči	raz-loč-aj	separate	vъ-gnězd-i	vъ-gněžd-aj	make nest
ostruj-i	ostruj-aj	ruin	u-god-i	u-gažd-aj	please
ras-toč-i	rastač-aj	disperse	pri-gvozd-i	pri-gvažd-aj	nail to
na-poj-i	na-paj-aj	give drink to	pri-měs-i	pri-měš-aj	join
u-krěp-i	u-krĕplj-aj	strengthen	vъ-pros-i	vъ-praš-aj	ask
o-krop-i	o-kraplj-aj	sprinkle	pro-obraz-i	pro-obraž-aj	prefigure
pro-slav-i	pro-slavlj-aj	extol	sъ-xran-i	sъxranj-aj	preserve
iz-bav-i	iz-bavlj-aj	save	sъ-blazn-i	sъ-blažnj-aj	offend
u-mrьtv-i	u-mrьštvlj-aj	mortify	po-xval-i	po-xvalj-aj	praise
u-lov-i	u-lavlj-aj	catch	po-mysl-i	po-myšlj-aj	intend
u-strьm-i	u-strьmlj-aj	rush	ra(z)-šir-i	ra(z)-širj-aj	extend
sъ-mqt-i	sъ-mqšt-aj	disquiet	raz-or-i	raz-arj-aj	ravage
o-cěst-i	o-cěšt-aj	purify			

#### 5.7111 Here are typical examples of possible stem-alterations:

5.7112 A few non-prefixed aspect pairs belong to this class:

av-i	avlj-aj	manifest	prost-i	prašt-aj	forgive
var-i	varj-aj	anticipate	pust-i	pušt-aj	let go
vrěd-i	vrěžd-aj	harm	rod-i	ražd-aj	give birth
gonoz-i	gonaž-aj	free	svobod-i	svobažd-aj	liberate
mьst-i	тьšt-ај	avenge			

**5.7113** Several more pairs of verbs fit this pattern in form, but the meaning of the i-verb is sometimes or always imperfective; thus the formal contrast is not a clearcut reflection of an aspectual opposition. Now, aj-verbs as a class are imperfective (with very few exceptions), while unprefixed i-verbs may belong to either aspect—and some to both. Therefore the presence of an unprefixed aj-verb beside an unprefixed imperfective (or ambiguous) i-verb with the same root represents either a lexical doublet (perhaps reflecting different dialects), or possibly a special aj-verb whose lexical meaning includes iteration of the basic action expressed by the root.<sup>2</sup>

Pairs where both verbs are attested only as imperfectives:  $val-i \sim valj-aj$  'roll',  $velič-i \sim velič-aj$  'magnify', glas-i glaš-aj 'call',  $klon-i \sim klanj-aj$  'bow',  $tvor-i \sim tvarj-aj$  'make'.

#### 88

Note that the term "iterative" is often used for the aj-verbs in varying senses: (1) the *forms* made by this derivational suffix (with or without modifications of the root), (2) the imperfective *meaning* conveyed by such derivatives, and (3) a special *aspect*, a subdivision of the imperfective, characterized by the meaning of iteration or frequentativeness. The evidence is insufficient to posit such a sub-aspect, since the iterative meaning is not always present, although it cannot be denied that this meaning is indeed the characteristic of certain aj-verbs: it is simply a part of their lexical meaning.

Pairs where the i-verb usually, but not always, is perfective:  $ziv-i \sim zivlj-aj$  'give life',  $kr\omega t-i \sim kr\omega t-aj$  'baptize', lis-i lis-aj 'deprive',  $plen-i \sim plenj-aj$  'take prisoner', protiv-i  $se \sim protivlj-aj$  se 'resist',  $svet-i \sim svest-aj$  'sanctify',  $stav-i \sim stavlj-aj$  'place',  $trud-i \sim tružd-aj$  'exert'.

**5.7114** In a few verbs the root in the imperfective does not show the expected changes. *O-pravbd-i* ~ *o-pravbd-aj* 'justify' and *stop-i* ~ *stopaj* (and prefixed forms) apparently have  $\{-aj+\}$  instead of, rather than added to,  $\{-i+\}$ . An unchanged root-vowel is found in *prigotov-i* ~ *pri-gotovlj-aj* 'prepare'; a changed vowel but no iotation in *na-lož-i* ~ *na-lag-aj* 'put on', and *o-moč-i* ~ *o-mak-aj* 'wet'.<sup>4</sup> Doublets are attested in *vas-xyt-i* ~ *vas-xyšt-aj / vasxyt-aj* 'steal'; (*sb)-lom-i* ~ (*sb)-lamlj-aj / (prě)-lam-aj* 'break', *sram-i* ~ *sramlj-aj / sram-aj* 'shame';<sup>5</sup> *sb-motr-i* ~ *sb-motrj-aj / sb-matrj-aj*, cf. *ra-sb-maštrj-aj* 'view'; *u-mqdr-i* ~ *u-mqdrj-aj* 'make wise', cf. *prě-moždrj-aj.*<sup>6</sup> Also *vb-sel-i* ~ *vb-sělj-aj / vb-selj-aj* 'settle'. Other attested variants: *vbz-běs-i* ~ *vbz-běš-aj/vbz-běs-ova* 'make frantic', *iz-měn-i* ~ *iz-měnj-a / iz-měn-ova* 'change', *prě-lbst-i / prě-lbšt-aj / prě-lišt-aj* 'deceive'; and *o-svět-i* ~ *o-svěšt-aj / o-svěšt-av-aj* 'illuminate'.

5.712 The imperfective is formed from the perfective stem by *replacing* the verbal classifier with -aj+; vowel alternation (tense  $a \notin i y$  for lax  $o \ e \ b \ b$ ) if possible, and—with velar roots—mutation (c for k, 3 for g).<sup>7</sup>

a. Perfective stem has zero classifier; a number of these perfective stems are somewhat irregular. Over 40 pairs are attested, many of them very common verbs. The following list gives examples of all the roots attested in OCS. For irregular or special stems, paragraph references to the discussion of each stem are provided. Roots ending in j may replace the stem-final glide with v; see section  $\mathbf{e}$ , below.

sz-bljud- $\emptyset$ + ~ sz-bljud-aj+ 'watch', po-ěd- $\emptyset$ + (§16.22) ~ po-ěd-aj+ 'eat', na-klad ~ na-klad-aj 'load', pad ~ pad-aj 'fall', po-tręs ~ po-tręs-aj 'shake', otz-sěk ~ otz-sěk-aj 'cut off', po-črъp (§15.72) ~ po-črъp-aj 'draw

<sup>&</sup>lt;sup>3</sup> The sole example is Mar *živlěats* in J 6:63, where As has *živets* 'lives' and Zo the same, corrected by a later cyrillic hand to *živits* (the expected reading). It is highly probable that the scribe of Mar (or one of his immediate predecessors) found this same erroneous intransitive verb in his model, and invented *\*živlajets* to make sense of the clause. Many other forms listed in these paragraphs are very likely *ad hoc* inventions of translators or copyists.

<sup>&</sup>lt;sup>4</sup> Underlying {na-log-i+} ~ {nalag-aj+}, {na-mok-i+} ~ {na-mak-aj+}.

<sup>&</sup>lt;sup>5</sup> These may rest on the presence or absence of "epenthetic l" (§3.71).

<sup>&</sup>lt;sup>6</sup> The expected palatal clusters may not appear because a scribe lacked /rj/ or because he expected the reader to derive the proper pronunciation from a morphophonemic spelling. The stem *jazv-i* 'wound' implies {*jazv-i+ъš-*} > \**jažvljъš-* and an imperfective \**jažvlj-aj*: the only attested forms (Su 499.28 μzвъшее сΔ, 436.12 μzвѣкмъш) fail to express *j* or mutation of -*zv-*.

<sup>&</sup>lt;sup>7</sup> In effect, this is Type II palatalization (§3.4), but it has become fully morphologized, see §44.361.

(water)', ot-vrbz- (§15.72)<sup>8</sup>~ ot-vrbz-aj 'open', s-ret (§16.74) ~ s-ret aj 'meet', s-s-sed (§16.61) ~ s-s-sed aj 'coagulate'.

iz-bod- ~ iz-bad-aj 'stab', po-mog ~ po-mag-aj 'help'; po-gnet cf. ugnět-aj 'press, oppress'; pro-cvut (§15.871) ~ pro-cvit-aj 'bloom', po-čut (§15.871) ~ po-čit-aj 'read', za-čun ~ za-čin-aj 'begin', pro-klun ~ proklin-aj 'curse', u-mur ~ u-mir-aj 'die', pro-pun ~ pro-pin-aj 'crucify', istur ~ is-tir-aj 'erase', su-žum ~ su-žim-aj 'squeeze'; na-dum sę ~ na-dymaj sę 'puff up'; here also za-kolj (§16.513) ~ za-kal-aj 'slaughter'.

na-lęk ~ na-lęc-aj 'draw (bow)', is-tek ~ is-těk-aj 'run out'.

With variants: po-greb ~ po-greb-aj/po-grib-aj 'bury'; sz-plet ~ szplět-aj/sz-plit-aj 'braid'; po-strig (§15.874) ~ po-strig-a/po-striz-aj 'tonsure'; szžeg (žsg; §15.875) ~ sz-žag-aj (a for ě, §3.5c1)/sz-žiz-aj 'burn up'.

**b.** Perfective has the classifier **-no+**. Only 23 pairs are attested, although there were surely more.

iz-běg-nq ~ iz-běg-aj 'flee', u-vęz-nq ~ u-vęz-aj 'be caught', u-gas-nq ~ u-gas-aj 'be quenched', iz-gyb-nq ~ iz-gyb-aj 'perish', pro-zęb-nq ~ pro-zęb-aj 'sprout', oto-rig-nq ~ oto-rig-aj 'spew forth'.

 $kos-nq \sim kas-aj se$  'touch', u-top- $nq \sim u$ -tap-aj 'sink'; i-čez- $nq \sim i$ -čazaj (a for ě, §3.5c1) 'disappear'; vzz-dzx- $nq \sim vz$ -dyx-aj 'sigh', po-tzk- $nq \sim po$ -tyk-aj 'knock'; na-vyk- $nq \sim na$ -vyc-aj 'study, learn', szteg- $nq \sim sz$ tez-aj 'gain'

c. Perfective has classifier -a+. Seventeen pairs are attested. Most of the perfective stems have an unpredictable vowel alternation in the root, see \$15.643-.65.

po-maz-a ~ po-maz-aj 'anoint', prě-pojas-a ~ prě-pojas-aj 'gird'; oblobz-a- ~ ob-lobyz-aj 'kiss'; iz-b*s*r-a iz-b*i*r-aj 'select', raz-d*s*r-a ~ razdir-aj 'rend', sz-z*b*d-a ~ sz-zid-aj 'build', po-p*s*r-a ~ po-p*i*r-aj 'trample', po-szl-a ~ po-syl-aj 'send', sz-lzg-a cf. ob-lyg-aj 'lie, deceive'; and with variant, po-kaz-a ~ po-kaz-aj/po-kaz-ova 'show'.

**d.** Perfective has classifier  $-\check{e}+$ . A small group:  $s \eth -gor-\check{e} \sim s \eth -gar-aj$ 'burn up',  $po-m \bowtie -\check{e} \sim po-min-aj$  'remember',  $pr\check{e}-p \bowtie -\check{e} \sim pr\check{e}-pir-aj$  'convince',  $pri-z \bowtie -\check{e} \sim pri-zir-aj$  'view'. In other prefixed forms of  $z \flat -\check{e}+ti$  'to see', there is doubt as to whether the  $\check{e}$ -verbs are always perfective. (L 6:7 Mar nazurčanç i konižnici, if perfective imperfect, would mean that the scribes made a continued series of completed observations. To As negring are is simpler: 'they continuelly

a continued series of completed observations; Zo As naziraaxq is simpler: 'they continually observed him' [Jesus].)

<sup>&</sup>lt;sup>8</sup> The spelled  $r_b$  in  $\check{c}r_bp$  and  $vr_bz$  doubtless represents syllabic r (cf. §2.631). It is highly probable that it was short in the perfectives but long in the imperfectives.

e. Perfective stem ends in **j**, which is replaced by v before the -aj+ classifier. In an older formation, the *j* is the root-final consonant, followed by the zero classifier. A younger formation, surely productive in the innovating type of language attested in Supr, adds the classifier -aj+ to a stem that already has -aj+ or  $-\check{e}j+$ ; the first glide is replaced by v, producing -avaj- or  $-\check{e}vaj-$  (with, however, no apparent change of meaning). The -vaj forms are far more common in the mss.

Some older forms survive beside the new:  $u-bbj-\emptyset + \sim u-bij-aj +$  and u-bivaj 'beat, kill';  $po-vbj \sim po-vij-aj$  and po-vivaj 'wrap'; and  $iz-lbj-\emptyset +$  beside iz-lbj-a+ (§15.46) ~ iz-livaj 'pour out'.

sъ-kryj ~ sъ-kryvaj 'hide', o-myj ~ o-myvaj 'wash', u-nyj ~ u-nyvaj 'lose courage'; u-p<sub>bj</sub> (§15.93) ~ u-pivaj sę 'become intoxicated', vъz-ъp<sub>bj</sub>cf. pri-v-ъpivaj (§3.25) 'call out to'; also the irregular vъs-pě (§16.53) ~ vъs-pěvaj 'sing forth', prefixed forms of byti (§16.11) like iz-by ~ iz-byvaj 'be left over'; o-pljъv-a (§15.52) ~ o-pljъvavaj.

o-del-ěj or o-dol-ěj ~ o-delěvaj or o-dol-ěvaj 'win, be victorious'; uspěj ~ u-spěvaj 'be of use', o-cěpěn-ěj ~ o-cěpěněvaj 'become rigid'; poznaj- ~ po-znavaj 'know', o-klevet-aj ~ o-klevetavaj 'slander', sz-konbč-aj ~ sz-konbčavaj 'finish', podz-kopaj- ~ podz-kopavaj 'dig (under)', oblzgzč-aj ~ ob-lzgzčavaj 'ease', o-tęžbč-aj ~ o-tęžbčavaj 'become heavy'.

Some common stems are both P and I, with corresponding -vaj forms that are only I: razum-ěj 'understand' (IP), po-razuměj (P); razum-ěvaj and pro-raz-uměvaj (I); konьč-aj {kon-ьc-ěj+} 'finish' (IP), prefixed sъkonьč-aj is P and sъkonьčavaj is I; otъ-věšt-aj (IP) 'answer' otъ-věšt-avaj (I)—also sъ-věšt-aj 'convince', u-věšt-aj 'counsel', oběšt-aj 'promise' (§3.312), za-věšt-aj 'bequeath'.<sup>9</sup>

## 5.713 Isolated types

**a.** The small group of verbs having determined ~ non-determined forms (within the imperfective aspect, cf. §5.1) are:  $ved \sim vod-i$  'lead', nes ~ nos-i 'carry', per ~ par-i 'fly', gon-a ~ gon-i 'drive, chase',  $vl\check{e}k \sim vla\check{e}-i$ 

<sup>&</sup>lt;sup>9</sup> These well attested verbs seem to fall into patterns that allow us confidently to predict forms, here the type -svęt-i+ ~ -svět-i-aj+ -svěštavaj+ (§5.7114). The root vět 'solemn speech' underlies synchronic {-vět-i+aj+}; yet the mediating formation -vět-i+ is hypothetical. Although nouns like otsvěts and otsvěštanbe are in common usage, no i-verb is found in early medieval Slavic. The isolated infinitive KPB\*MABBATH SU 188.4 'to traffic dishonestly' suggests a noun krsčbma 'tavern' (SC krčma) and a formation {kbrčbm-i-aj}—which may never have existed—while the meaning is derived from the Gk and the context.

'drag', *id* (\$16.3) ~ *xod-i* 'go, walk', and perhaps *plov* ~ *plav-aj* 'go (by boat)'. With prefixes these pairs become normal P ~ I pairs, e.g. vs-ved(P) ~ *vs-vod-i*(I) 'lead in', *iz-id*(P) ~ *is-xod-i*(I) 'go out'. The pair -*lĕz* ~ *-laz-i* 'go, clamber' is by chance not attested without prefix, cf. ss-*lĕz-* ~ *sslaz-i* 'climb down'.

The rare prefixed stems with *-važd-aj*, *-ganj-aj*, and *-xažd-aj* do not seem to be semantically opposed to the normal imperfectives in *-vod-i*, *-gon-i*, and *-xod-i*. Perhaps some instances represent lexical iteratives (cf. §5.7113), they do not show a systematic opposition of aspect.

**b.** Other types of relationship (P is given first):

kup-i ~ kup-ova 'buy', obraz-i ~ obraz-ova 'form', obišt-i ~ obišt-eva 'associate' (but pri-obišt-i ~ pri-obišt-aj/pri-obišt-avaj).

prě-minǫ {prě-min-nǫ+} ~ prě-min-ova 'pass', po-vinǫ {po-vin-nǫ+} ~ po-vin-ova 'be subject to' (and obinǫ {ob-vin-nǫ} ~ obinova 'avoid', §3.312).

oto-rěz-a ~ oto-rěz-ova 'cut off', voz-isk-a (§15.641) ~ voz-isk-ova 'seek out'; perhaps znamen-aj ~ znamen-avaj/znamen-ova 'signify'.

 $skoč-i \sim skak-a$  'jump',  $tlsk-nq \sim tlsk$  (§15.874) 'knock'; kanq {kap-nq} ~ kap-a 'drip';  $pljunq/plinq \sim pljsv-a$  'spit';  $ots-ri-nq \sim ots-rej/otsrivaj$  'push away'; po-manq {po-maj-nq+} ~ po-maj-a/po-mavaj 'beckon';  $pre-sta-n \sim pre-staj-a$  'stop'; dad- (§16.21) ~ daj-a (but with prefixes  $-dad \sim -daj-a$  or -davaj); imqts eti ({sm-Ø+} §15.83) ~ im-a jemlj-qts (§15.643) 'take' (with consonantal prefixes vsz-sm-a or vsz-sm-aj).

5.721 It is more difficult to determine aspect-pairs whose formal relationship is that of non-prefixed imperfective ~ prefixed perfective (e.g. tvoriti ~ so-tvoriti 'do'), for similarity in form must be supported by identity in lexical meaning. Now, the addition of any prefix to any verb (except the non-determined ones and most imperfectives with the classifier  $-a_{j+}$ ) produces a perfective, but the prefix also adds a semantic element. Thus prě-tvoriti and ras-tvoriti are perfective, but they mean 'transform' and 'dissolve', respectively, and hence are opposed to the simplex tvoriti in lexical meaning as well as in form. In so-tvoriti, however, the prefix sohas no force other than to perfectivize the verb; it is an "empty prefix". Nearly all of the prefixes serve with one verb or another in this purely perfectivizing function (e.g. u-slyšati 'hear', po-gasiti 'quench', vosplakati 'weep'), so that only a semantic analysis of each group of formally related verbs can separate out the prefixed perfective which corresponds in meaning to the simplex imperfective. Information of this kind is often too meager in the texts to permit a clearcut decision.

5.722 The formation of new prefixed forms and of mates of the opposite aspect for various newly-created or already extant verbs must have been an active process in OCS as it is in all modern Slavic languages. It is certain that the classifier -aj (with or without modification of the root), and, to a lesser extent -ova+, were productive for making imperfectives to various other classes of verbs, and it is probable that -nq+ was productive for making perfectives. Patterns of like formations could easily be extended. For example, the pairs  $pr\check{e}$ -tvor- $i+ \sim pr\check{e}$ -tvarj-aj+, pri-tvor- $i+ \sim$ pri-tvarj-aj+, and u-tvor- $i+ \sim$  u-tvarj-aj+, originally opposed to sz-tvor-i+ $\sim$  tvor-i+, gave rise both to a new imperfective sz-tvarj-aj+, and to an unprefixed tvarj-aj+. The preference of different dialects (regional and historical) for specific forms in given contexts doubtless accounts for such doublets in our texts.

**5.8** The possible verb-forms are most economically described in terms of separate morphological categories. After preliminary remarks on the personal suffixes in the next section ( $\S5.9$ ), the present tense will be described (\$6), then the imperative (\$7) and the present participles (\$8). The two past tenses, imperfect (\$9) and aorist (\$10), are followed by the three past participles (\$11), the verbal substantive (\$12), the two invariable forms, infinitive and supine (\$13), and finally a note on the formation of compound tenses (\$14).

NB: the sections on the infinitive and present tense are of particular importance to students because traditionally it is the infinitive, with or without the first person singular present and/or the second or third person singular present, that is cited as the "name-form" in dictionaries and grammars. One must therefore know the relationship of these forms to the *basic stems* used in this book.

A survey of the verbs by classes (\$15) will include the enumeration of irregular verbs; minor irregularities are mentioned in \$6-\$13.

An index of irregular verbs is at the end of the book.

**5.9** The *terminal desinences* expressing the three persons of the dual and the 2nd plural are constant throughout all categories that specify person (present, imperative, imperfect, aorist), while the other persons are expressed by two or more desinences which vary according to the category.

Here is a summary table of all person-number desinences. The term *past* includes the *imperfect* (coordinated past) and the *aorist* (simple past).

	Singular		Dual	Plura	]	
	present	past		present	past	
1	- <b>о</b> (-ть, -ě)	•Ъ	-vě	-тъ	-отъ	1
2	<b>-ši</b> (-si)	-Ø (-tъ)	-ta	-te		2
3	-tъ	-Ø (-tъ)	-te	-tъ	-9, - <del>6</del>	3

- 1sg -ть appears in the present of 5 verbs, -ě is an alternative for one: esmь, damь, ěmь, věmь/vědě, imamь (§16.2).
  - -o is otherwise the universal 1 sg present desinence.
  - -ъ is used for *past* (i.e. imperfect and aorist).
- 2sg -si is present, used with 4 verbs: esi, dasi, ěsi, věsi (§16.2).
  - -ši is the terminal present desinence for all other verbs.
  - -tъ is used in presence of *-imperfect* (i.e. the aorist) in a limited group of specially marked verbs (including some where it is optional); it always is homonymous with 3sg (§10.51-2).
  - -Ø is normal in presence of *past* (i.e. imperfect and aorist, §9.1, 10.1).
- 3sg -tъ is universal in the present tense; it is special in the aorist of certain verbs (§10.51-2).
  - -Ø is normal in presence of *past* (and the same as 2sg). See also §6.61
- 3pl -tъ is universal in the present tense.
  - - $\mathbf{q}$  is used in the imperfect (hence -xq).
  - -e is used in the aorist (whereby  $-xe > -\check{s}e$ ).

**5.91** The third person dual desinence **-te** ('they two') is for the most part clearly opposed to the second person dual **-ta** ('you two'). Thus *third* person dual has the same form as *second* person plural—a distinction maintained in Mar, Ps, Euch, and Cloz. In Zo and As, *-ta* occasionally functions as 3rd person dual, while in Sav and Su *-te* in 3 du is rare.

This means that in late OCS -te signifies 2nd person plural, while 2-3 dual is only -ta; the same form is used for second and third persons in all duals as well as in the imperfect and aorist singular.

Therefore: in every tense-paradigm the third dual desinence should be understood to be -te/-ta.

5.911 Supr (twice) and Sav (6x) use the suffix -tě for 3 person dual with non-masculine subjects, e.g. TEROCT<sup>1</sup> (they (the two Marys) ran' (Mt 28:8 Sav), Vaillant 228.

6.0-6.22

# 6.0 The present tense

6.10 The desinences of the present tense are complex; a vocalic presentmarker preceeds the terminal desinences that indicate person and number, except in the first person singular (where the marker is zero).

6.11 The present-marker is either general,  $\rho$  in 3 pl and e elsewhere, or specific,  $\rho$  in 3 pl, and i elsewhere. The specific  $\rho/i$  is used with basic stems that end in a front vowel (therefore -i+, e+, and the *ša*-verbs, where surface *a* represents underlying { $\check{e}$ }, §15.31);  $\rho/e$  is used with all other verbs. E.g.:

```
nosi-ti 'carry' and moně-ti 'think' + e/i \sim nes-qt 'carry' and dělaj-qt 'do' + q/e
3pl {nos-i+e-tb} {mbn-e+e-tb} ~ {nes-Ø+q-tb} {dėl-aj+q-tb}
```

3 sg	{nos-i+i-tъ}	{mъn-ě+i-tъ}	~	{nes-Ø+e-tъ}	{děl-aj+e-tъ}
lsg	{nos-i+Ø-q}	{mьn-ě+Ø-q}	~	{nes-Ø+Ø-q}	{děl-aj+Ø-q}

6.12 Person-number desinences:

	singular	dual	plural
1st person	-6	-vě	-тъ
2nd person	-ši	-ta	-te
3rd person	-tъ	-te	-tъ

6.13 Vocalic basic stems are truncated before the vocalic desinences: the first vowel  $(V_1)$  either disappears or is replaced by *j*.

(A)  $V_1$  is replaced by j

(1) if it is a, and (2) if  $V_1$  is i or  $\check{e}$  and  $V_2$  is *not* i or  $\varrho$ . Otherwise (B)  $V_1$  is deleted.

The j resulting from the action of rule A creates a new Cj cluster that requires further adjustment.

**6.21** The classifier -*ova*/-*eva* and two stems with -*bv*-*a*+ become *ovj*, *evj*, *bvj*, and then *uj*: e.g. věr-*ova*+ti 'to believe' ~ 3 pl {věr-ova+q-tъ} > věrovjqtъ > věr*uj*qtъ; nepьšt-*eva*+ti 'to suppose' ~ nepьšt*uj*qtъ; plj*bv*-*a*+ti 'to spit' ~ plj*uj*qtъ.<sup>10</sup>

**6.22** A-verb presents have iotation:  $vez \cdot a + ti$  'to tie' ~ { $vez \cdot a + q - tb$ } > vez j q ta > vez j q

<sup>&</sup>lt;sup>10</sup> This is a morphological rule that occurs with these specially-defined morphemes. Like many morphological rules, it has specific phonological effects. Elsewhere in OCS it is normal for *ovj* to become *ovlj*: the *j* after labial consonant becomes palatal *lj*, cf. §6.23: *loviti* 1 sg pres *lovljq* like *lomljq* from *lomiti*.

~ 3 pl {plak-a+q-tb} > plakjqtb > plačqtb, 2 sg {plak-a+e-ši} > plakješi > plačeši. (For a-verbs that have an unpredictable alternation of root-vowel in the present stem, see §15.642.)

In  $s\check{e}j-a+ti$  'to sow' { $s\check{e}j-a+q-ta$ } >  $s\check{e}j-j-qta$ ; double consonants reduce to one:  $s\check{e}jqta$ .

**6.23** In i-verbs and ě-verbs j is generated from i or ě before the first person singular desinence, creating Cj, and triggering iotation: {kup-i+ $\emptyset$ -q} > kupjq by rule A, while {kup-i+ $\varphi$ -tb} > kup $\varphi$ tb y rule B.

kupi-ti kup-ętъ	~ ku <i>plj</i> q	'buy'	ljubi-ti lju <i>b-</i> ętъ	~ lju <i>blj</i> q	'love'
lovi-ti lov-ętъ	~ lovljq	'hunt'	lomi-ti lom-ętъ	~ lomlją	'break'
svęti-ti svęt-ętъ	~ svęštą	'sanctify'	vidě-ti vid-ętъ	~ viždq	'see'
nosi-ti nos-ętъ	~ nošę	'carry'	obrazi-ti obraz-ętъ	~ obražo	'form'
тыně-ti ты <i>п-</i> ętъ	~ ть <i>пј</i> о	'think'	mysli-ti my <i>sl-</i> ętъ	~ my <i>šlj</i> q	'think'
kъsně-ti kъ <i>sn</i> -ętъ	~ kъ <i>šnj</i> ǫ	'be late'	blazni-ti blazn-ętъ	~ blaž <i>nj</i> q	'offend'
pusti-ti pust-ętъ	~ puštą	'let go'	xvali-ti xva <i>l-</i> ętъ	~ xva <i>lj</i> q	'praise'
umori-ti umor-ętъ	~ umorjo	'kill'	sъmotri-ti sъmotr-ętъ	~ sъmoštrją	'look'

**6.3** Root-final velars in C-verbs undergo KI-mutation before the presentmarker *e*: this produces an alternation of consonants within the paradigm.

```
{tek-\emptyset+\varphi-tb} > tek\varphita, 1 sg tek\varphi ~ 2 sg {tek-\emptyset+e-si} > teesi etc. 'run'
{mog-\emptyset+\varphi-tb} > mog\varphita, 1 sg mog\varphi ~ 2 sg {mog-\emptyset-e-si} > mozesi 'be able'
```

**6.41** The verb *xotěti* 'want' (and prefixed forms) takes the non-terminal e-marker in all forms but 3 pl and 1 sg: { $xot-ě+e-t_B$ } by rule B > *xoteta*, while {xot-ě+Ø-q} by rule A2 > *xotjq*, whence *xoštq*. The other forms have {xot-ě+e-}, which by A2 produces *xotje-*; therefore *xošteši*, *xoštema*, etc.

In Su, the root vowel s is found exceptionally for o: 1 sg χτωτπ 169.3, 534.11, 3 sg χτωτπ 153.7, 169.21. These are doubtless dialectal forms; note omission of the 3 sg desinence -to (cf. §6.61).

**6.42** The verb dovblěti 'suffice', which has only 3rd person forms, is attested with 3 pl dovblęts and dovblěto, implying competing classifiers,  $-\tilde{e} \sim -\tilde{e}j$ . The 3 sg dovblets implies irregular selection of the e-marker {dovbl-ě+e-tb} > -lje- (cf. Vaillant 263).

**6.431** The verb *iska*-ti 'seek' has the mutated forms 3 pl *ištǫt*<sup>5</sup> and 1 sg *ištǫ* (< iskj- < isk-a+) beside unmutated *iskǫt*<sup>5</sup> and *iskǫ*. This implies an alternate underlying {isk- $\emptyset$ +}; in the rest of the present, *ište*- results from KI-mutation {isk- $\emptyset$ +e-}.

6.432 The root *met* 'throw' also forms alternate basic stems, {met- $\emptyset$ +} ~ {met-a+} and therefore *meto meto metoši* etc., beside *mešto meštoši* etc.

**6.5** Verbs with present stems in -*aj*- or -*ěj*- have spelled -*ae*-/*ěe*- in all forms except 1 sg and 3 pl: děl*ae*ši, děl*ae*tь; d*ae*ši, d*ae*tь; um*ěe*tь, um*ěe*te; s*ěe*tь, etc. In Mar such forms (particularly 3 sg and 2 pl) are often written

with -aa-/- $\check{e}a$ -, and occasionally (for -aa-) simply -a-: dělaat<sub>b</sub>, dělat<sub>b</sub>, uměat<sub>b</sub>, sěat<sub>b</sub>. This kind of change is extremely rare in other mss.

Similarly, stems in -uj- appear in Mar with -uu (rarely, for expected -ue-): trěbuut<sub>b</sub> for trěbuet<sub>b</sub> (from trěbova-ti 'demand') and the like.

**6.61** The 3rd person desinence  $-t_{\delta}$  (sg and pl) is occasionally omitted. For *est*<sub> $\delta$ </sub> and *něst*<sub> $\delta$ </sub> (§16.101), *e* and *ně* occur. These are surely dialect features. Cf. Vaillant 227.

**6.62** The jer in desinences -mz and -tz may be affected by the following enclitic \**jb* 'him' (§4.25), for the z is both tense and strong. It may be written y(-tyi, -myi), e.g. osodetyi 'they will condemn him', ostavimyi 'we will leave him'. Spellings with -oi also occur, but rarely: e.g. *izbavitoi* 'he will save him'. Cf. Vaillant 43–4.

6.7 Five common verbs have quite irregular present forms; they lack a present-marker (except for e in 3 pl), adding desinences directly to the consonantal root. The 1st and 2nd sg desinences are special: -mb and -si.

6.71 Dati 'give' (P),  $\check{e}sti$  'eat', and  $\check{v}\check{e}d\check{e}ti$  'know' and their prefixed forms have roots in d (dad-,  $\check{e}d$ -,  $\check{v}\check{e}d$ -). The d is retained only before the tense-marker e of the 3 pl; it becomes s before desinences beginning with t (§3.3131a) and drops elsewhere (see §16.2 for details):

damь dasi dastъ, \*davě dasta daste, damъ daste dadętъ; ěmь věmь, ěsi věsi, ěstъ věstъ, \*ěvě věvě ... ědętъ vědętъ.

The archaic and morphologically isolated 1 sg vědě 'I know' is a less frequent equivalent of věmb.

6.72 The verb *iměti* 'to have' forms its present (except for 3 pl) on a stem *ima*- to which desinences are added directly, including the special 1 sg mb. The 3 pl is *imqt*<sup>3</sup> (with a preferred alternant untertain Su):

imamь imaši imatъ, imavě imata imate, imamъ imate imqtъ (imějqtъ).

6.73 Byti 'to be' has a present stem es- except for 3 pl s-: esmb esi estb, esvě esta este, esmb este sqtb.

The negated form is special except in 3 pl: něsmь něsi něstъ, něsvě něsta něste, něsmъ něste, but ne sqtъ.

6.74 The present forms of bqd-qta express the future 'will be'. For other future expressions, see \$14.4, and more especially \$21.11.

6.75 Prefixed derivatives of *byti* (§16.11) use *-bqd-* for the present stem; e.g. *zabyti* 'forget' *zabqdqts*, *pribyti* 'join' *pribqdqts*.

**6.8** The completely isolated form *sets* may be a relic present or aorist, 'he says/said', but in OCS it functions as an adverb meaning 'allegedly'. See SJS.

# 7.0 The imperative

7.100 The imperative desinences are complex, consisting of an imperative-marker  $i^2$  or  $e^2$  plus the personal desinences (including zero).

	singular	dual	plural
1st person	-	-vě	-тъ
2nd person	Ø	-ta	-te
3rd person	Ø	-te	Ø

Before the vocalic imperative-markers, vocalic basic stems are truncated by processes described above in §§6.13-23.

7.101 The marker  $\check{e}$  appears in the dual and plural of basic stems in  $-n\varrho + \text{ or } -\mathcal{O} + (\text{unless preceded by } j)$ ; *i* is used elsewhere—that is, in all singular forms and in the dual and plural of all stems ending in a consonant other than *j* or a vowel other than  $\varrho$ .

7.11 Because these vowels are specifically marked ( $\{i^2\}, \{e^2\}$ ), they trigger KAI-mutation; stem-final velars (k, g) are replaced by  $c \not z \ (\$3.4)$ : mog-qtb 'be able' ~ mozi mozěte; vrbg-qtb 'throw' ~ vrbzi vrbzěte; strěgqtb 'protect' ~ strězi strězěte.

7.111 Stems in -ek and -eg change to  $\frac{1}{6}/53$ : rek-qtb 'say' ~ rbci rbcěmb rbcěte.

**7.2** Irregularities. The verb 'to be' forms the imperative on the stem bqd-: bqdi bqděms bqděte. (The 3 sg bqdi 'may it be, may it happen' is the chief representative of third person imperatives.)

The irregular verbs dati 'give', *ěsti* 'eat', *věděti* 'know', and the otherwise regular vidě-ti 'see' have daždb, ezdb, vezdb and vizdb. (This implies underlying {-d-jb}, and iotation.) In the dual and plural the imperative-marking vowel is *i*: thus dadims dadite, *edite*, *vedite*.

7.201 In Euch, these forms have final i: daždi, viždi, pověždi ('tell', from pověděti).

7.202 KF has the presumably Czech forms with  $z < \{dj\}$ , podazb, otsdazb.

7.21 C-a+ verbs and root *j*-verbs have plural imperative forms with the marker  $\check{e}$  beside regular forms with *i*: pokaza-ti 'show'  $poka\check{z}ite \sim poka\check{z}\check{e}te$  **norametre**, **norametre**, poj-Qtb 'drink'  $piite \sim pi\check{e}te$  **norametre**. The  $a/\check{e}$  is exclusive in Sav, frequent in As, Zo and Supr, less so in Mar, infrequent in Ps, and unknown in Euch and Sav. The spellings are not unambiguous, but the letter-sequence  $\check{z}\check{e}$  in glagolitic and met in cyrillic violates a fundamental OCS rule (§2.413), and  $\check{z}a$  (ma) and the like suggest that underlying

98

 $\{\check{e}^2\}$  (which alternates with  $\{i^2\}$ ) is being replaced with  $\{\check{e}^1\}$  (which alternates with  $\{a\}$ ), cf. §3.5c. Compare Vaillant §149.

7.3 The third plural imperative is attested only by  $b\rho d\rho$  'may they be'. It occurs in L 12:35 (in Zo, Mar, As, Sav), three times in psalm 108, and once in KF.

7.4 The form otspaděmb 'may I fall away' (Ps 7:5), supported by scraps of testimony from post-OCS mss, suggests that in early OCS such forms (with -mb) were more freely used. See Vaillant 232.

7.5 Third person imperative (and occasionally also first person) can be expressed by da plus the present tense:  $da \ pridots$  'may they come, let them come'. See also §22.11.

# 8.0 The present participles

**8.10** The declensional stem of the *present active participle* has a derivational suffix **-ext** or **-oxt** that is correlated to the shape of the present-markers (§6.11). Verbs with -*et* in 3 pl. pres. have -*ext* in pres.act. part., but -*e* in masc./neut. nom. sg. Verbs with -*ot* in 3. pl. pres. have -*oxt* in pres. act. part., but a variable  $\{y^2/e\}$  in nom. sg masculine/neuter; -*e* is used if the (truncated) verbal stem ends in a palatal consonant, -*y* otherwise. Cf. §5.602.

The formant of the *present* **passive** *participle* is **-im-**, or **-em-/-om-**, underlying {i-m-} or {e/o-m}.

For the declension of the active participles see §4.19 and §4.31. The passive participles are regular adjectives that belong to the hard twofold nominal and compound declensions.

8.11 Verbs forming their present tense with the present-marker *i* (i.e. iverbs, ě-verbs, and ša-verbs, §6.11) have present participles in -g and -im-ъ: kupi-ti 'buy' kup-i-ši ~ kupę kupimъ; vidě-ti 'see' vid-i-ši ~ vidę vidimъ; slyša-ti 'hear' slyš-i-ši ~ slyšę slyšimъ. (For these verbs the nom. sg fem. of the active participle is in -gšti: kupęšti, vidęšti, slyšęšti.)

8.12 Other soft present-stems (§5.602) have -e and -em-τ: vęz-a-ti vęž-qtъ 'bind' ~ vęžę vęžemъ; milova-ti 'have mercy' miluj-e-ši ~ miluę miluemъ; dělaj-qtъ 'do' ~ dělaę, dělaemъ; sěja-ti 'to sow' ~ sěę, sěemъ. (For these verbs the nom. sg. fem. of the active participle is in -qšti.)

8.13 All other stems (i.e. C-verbs and no-verbs) take the suffixes -y and -om-b: nes-qtb 'carry' ~ nesy, nesoms; dvigno-ti 'move' ~ dvigny, dvignoms. (For these verbs the nom. sg. fem. of the active participle is in -**Qšti**.) **8.131** Exceptionally, the suffix  $-y^2$  (masc.-neut. nom sg., alternating with -qst- of the rest of the paradigm) is written with a special glagolitic letter we transcribe g: e.g. *nesp*. Such spellings are more commonly found for the -yi of the definite participles. The special letter occurs only for this particular morpheme; the phonetic value is a subject for speculation.

**8.2** The verb byti 'to be' forms the participles sy (f. sǫšti) 'being' and also bǫdy (f. bǫdǫšti), which means 'future, that to come'. Věděti 'know' has vědy (f. vědǫšti).

Gorěti 'burn' has beside regular gorę (goręšti) some forms of gorçšt-: cf. Vaillant §180.

#### 9.0 The imperfect.

9.1 The desinences of the imperfect are complex: the imperfect-marker  $\{\check{e}a\}$  + the past-marker  $\{x\}$  + past person-number desinences. The surface forms are:

	singular	dual	plural	
lst person	-ĕахъ	-ĕaxově	-ĕахотъ	
2nd person	X-X-	-ĕašeta	-ĕašete	
3rd person	-ĕaše	-ĕašete	-ĕaxǫ	

**9.111** The sequence of vowels  $(\check{e}^{t}a)$  interacts with the stem-final vowels as follows:

A. The classifier -i+ before {ěa}- becomes *j*; the resulting iod-cluster (Cj-ěa) undergoes iotation.

B. If the { $\check{e}a$ } follows  $\check{e}$ ,  $\check{e}j$ , a, or aj, the sequence  $\check{j}\check{e}$  or ja is deleted: e.g.  $mbn-\check{e}+\check{e}a > mbn-\check{e}a$ ;  $\check{d}\check{e}l-a\check{j}+\check{e}a > d\check{e}l-aa$ ;  $um-\check{e}j+\check{e}a > um-\check{e}a$ .

**9.112** After consonants, the  $\check{e}^{11}$  behaves as usual (§3.5 c1): (1) it effects KI mutation in velars, and (2) itself becomes *a*. Thus  $rek-\emptyset+\check{e}a->re\check{c}-\check{e}a->re\check{c}aa-;$   $mog-\emptyset+\check{e}a>mo\check{z}-\check{e}a->mo\check{z}aa-.$ 

9.12 The desinences are in effect -čax- plus the desinences of the root aorist, see §10.601.

9.121 Beside the -šet- in the dual and pl. desinences, the innovative forms -ča-sta and -ča-ste are used (see also  $\S5.91$ ). The -šet- forms are

<sup>&</sup>lt;sup>11</sup> The two-syllable person-number desinences of dual and plural may be regarded as /ově, omb/ and /eta, ete/ with a variable "thematic vowel": (1) e is added (a) before t (eta, ete), and (b) before the Ø of 2nd-3rd sing.; (2) o is added otherwise. The past-marker x remains before o (-xově -xomb) but by KI-mutation yields š before e (-šeta -šete). A modification of the rule deletes 1a, and xta and xte (by a more general rule of consonant clustering) yield sta and ste. (In later Serbian dialects, the reformulated rule (1) inserts o before any C, while (2) e is added before Ø; this results in -xota and -xote.)

exclusive in Zo, nearly so in Mar, but only a bare majority in As; in Su they are clearly exceptional. Sav has only *-sta* and *-ste*.<sup>1</sup>

**9.211** Contrary to \$9.111 A, the classifier -i+ in a few examples in Su is simply deleted. Thus prixoděax<sup>±</sup> for prixoždaax<sup>±</sup> (prixodi-ti 'come'), raděax<sup>±</sup> for raždaax<sup>±</sup> (radi-ti 'be pleasing'). This type of form increases in post-OCS mss.

9.212 The classifier  $-\dot{e}$  + appears to convert to *j* and trigger iotation in one isolated example: the form spelled kasnease in L 1:21 Mar appears to represent \*kasnjaase (kasne-ti 'delay'). This type of form is better illustrated in post-OCS Serbian mss.

**9.22** Stems in -nq+ truncate before the  $-\epsilon a$ - suffix: podvignq-ti 'move' ~ podvigněax<sub>b</sub>. Only five examples are attested (four in Su, 1 in Cloz), probably because nearly all nq-verbs are perfective, and perfective imperfects are rarely required (cf. §21.2).

**9.3** In all texts there are forms spelled with only the initial  $\check{e}$  or a of the suffix:  $v\check{e}d\check{e}ax\flat \sim v\check{e}d\check{e}x\flat$  'knew',  $bo\check{e}ax\varrho s\varrho \sim bo\check{e}x\varrho s\varrho$  'feared',  $xot\check{e}a\check{s}e \sim xot\check{e}\check{s}e$  'wanted'. The longer forms usually predominate, but in Sav they are rare exceptions. These spellings apparently indicate that contraction of the two vowels into a single syllable began during the OCS period; most scribes considered it proper to write two vowels.

In Su there are spellings with *ěě* and *jaja* (MAYLN'E KXONE 'we began', CTPORRAUE 'set') that probably represent artificial attempts at restoring the older, non-contracted forms.

9.4 Verbs whose forms are not predictable from a basic stem (§16) usually use the present stem for the imperfect: obrěsti obręštǫtə 'find' obręštaaxə; idǫtə iti 'go' ~ iděaxə; dadętə dati 'give' ~ daděaxə; duti dəmǫtə 'blow'' dəměaxə; žrəti žərǫtə 'sacrifice' ~ žərěaxə; gənati ženǫtə 'drive' ~ ženěaxə; pěti pojǫtə 'sing' ~ pojaaxə; mlěti meljǫtə 'grind' ~ meljaaxə; klati koljǫtə 'stab' ~ koljaaxə; brati borjǫtə 'fight' ~ borjaaxə; \*ěxati ědǫtə 'ride' ~ ěděaxə.

9.5 A few verbs have competing forms from the infinitive/aorist and from the present stem: 'spit' *plsvati pljsvaaxq* (Zo Mar) ~ *pljujqt5 pljujaxq* (As; Mk 15:19); *besědovati* 'converse' ~ **bectAoytame** Su 304.18, 569.30; *trěbovati* 'need' ~ **tptsoytame** Su 307.19; *radovati sę* 'rejoice' ~ *paAoytame* cA Su 550.11 [all other imperfects from ova-verbs, even in Su, have -ovaa-]; 'call' zovati zovaaxq etc., Su 3bbame ~ zovqt5 30btaxt Su 322.12; 'receive' *priemljqt5* прикмываше Su 274.24 ~ *priimati priimaše* etc.

**9.6** The imperfect of *byti* is defective, being found only in the third person: *běaše*, *běašete*, *běaxq* (contracted *běše*, *běxq*). There is some confusion of these forms with the imperfective aorist *bě*, *běste*, *běše* (§10.91).

underlying	i > j	iotation	ějěa > ĕa	KI	ě > a	OCS
pros-i+ěa-хъ	pros-j-ĕa-хъ	proš-ĕахъ	>	>	ргоšаахъ	ргоšаахъ
тып-ĕ+ĕа-хъ	>	>	тып-ё+а-хъ	>	>	тыněахъ
děl-aj+ěa-хъ	>	>	děl-a+a-хъ	>	>	dělaaxъ
daj-a+ĕa-хъ	>	>	daj-a+a-хъ	>	>	dajaaxъ
mil-ova+ĕa-хъ	>	>	mil-ova+a-хъ	>	>	milovaaxъ
vęz-a+ěa-хъ	>	>	vęz-а+а-хъ	>	>	vęzaaxъ
rek-Ø+ĕа-хъ	>	>	>	reč-ěa-хъ	геčаахъ	геčаахъ
bьj-Ø+ĕа-хъ	>	>	>	>	ьъјаахъ	ььјаахъ
						<b>bijaaxъ</b>
nes-Ø+ěa-хъ	>	>	>	>	>	nesĕaxъ

Samples of typical imperfect derivation:

#### 10. Aorists

There is one productive type of aorist in OCS that is used in nearly all verbs, plus two archaic formations that are (or may be) used with C-verbs. The older aorists were apparently in use in Croatia and Macedonia well into the fifteenth century, though they seem to have become obsolete in eastern Bulgaria during the OCS period. See §10.7, below.

10.1 The desinences are:

	singular	dual	plural
lst person	-(0)ХЪ	-(o)xově	-(о)хотъ
2nd person )		-(o)sta	-(o)ste
3rd person	-(e)ø	-(o)ste	-(o)šę

**10.11** The vowels given in parentheses appear only when the basic stem ends in an obstruent  $(p \ b \ t \ d \ k \ g \ s \ z \ x)$ : nes-Qtb 'carry' ~ nesoxo nese. (Cf. also §10.4 below).

The past-marker  $x \sim s \sim \check{s}$  is underlying  $\{x\}$ : it becomes s before t but  $\check{s}$  before  $\varrho$ . 2-3 sg. underlying  $\{nes-\check{O}+x\} > nes-e\cdot x > nese$  (because no word-final C is allowed.)

**10.12** Otherwise (i.e., for the vast majority of verbs ), the suffixes (including the zero of 2-3 sing.) are consonantal. They are added directly to vowel-stems: prosi-ti 'beg' ~ prosixo prosi; moně-ti 'think' ~ moněxo moně; vęza-ti 'bind' ~ vęzaxo vęza, daja-ti 'give' dajaxo daja.

10.2 Sonorant-stems are truncated as follows:

**10.21** *j* is deleted: *dělaj*-qtъ 'do' ~ *dělaxъ děla*; *razuměj*-qtъ 'understand' ~ *razuměxъ razumě*; *bъj*-qtъ 'beat' ~ *bixъ bi bišę*; *kryj*-qtъ 'hide' ~ *kryxъ kry*; *po-ču*-qtъ '*feel'* ~ *počuxъ poču*.

102

**10.22** stem-final b + nasal sonorant is replaced by  $e: raspun-qtb \{raz-pbn+\emptyset\}$  'crucify' raspexs raspe; po-zbnj-qtb 'reap' poze; vszbm-qtb { $vbz-bm-\emptyset+$ } take' ~ vszexs vsze vszes.

10.23 stem-final br is replaced by rě: umbr-qtb 'die' ~ umrěxo umrěšę.

10.24 stem-final ov is replaced by u: natrov-qtb 'feed' ~ natruxomb natru.

10.25 The isolated verbs *melj*-qtъ 'grind', *borj*-qtъ 'fight', *kolj*-qtъ 'slaughter' become *mlěxъ*, braxъ, klaxъ (§16.511-3).

**10.3** Before the -e of 2-3 sg., stem-final k g x become  $\check{c}\check{z}\check{s}(\$3.4 \text{ I})$ : rekotb 'say' ~rekoxb re $\check{c}e$ ; mog-qtb 'be able' ~ mogoxb može.

**10.4** Some nq-verbs with consonantal roots drop the classifier before the aorist desinences: pogybnq-ti 'perish' ~ pogyboxt pogybe; dvignq-ti 'move' ~ dvigoxt dvize (cf. §10.3). Some other verbs retain the nq: umltknq-ti 'be silent' ~ umltknq. A number of verbs are attested with both types; the nq is particularly likely to be omitted in 2–3 sg. See §10.812 and §15.72–6.

10.41 Note that there are a few cases where the surface stem in -nq does not contain the root-final C of the underlying form; when nq is dropped, the C appears in the surface form, e.g. {-spp-nq-ti} > u-spq-ti ~ usppe '(he) went to sleep'. See also §15.73, 17.75.

10.51 Three verbs regularly have the anomalous desinence -sts in 2-3 sg.; byti 'be' ~ bysts, dati 'give' ~ dasts, ësti 'eat' ~ ësts. Both dasts and ests are ambiguous forms, since they are identical with 3 sg. present (cf. §66.21-2).

**10.52** A few root-verbs whose stem ends in a sonorant (and aorist-stem ends in a vowel) usually take a terminal desinence -ts in 2–3 sg.:<sup>12</sup> pbj-qtb 'drink' ~ pit, po-vbj-qtb 'wrap' ~ povit, {ob-vbj-Qt+} obbj-qtb 'wind' ~ obits; \*{nbr-} ~ ponrět5 'sank'; u-mbr-qtb 'die' ~ umrět5; pro-stbr-qtb 'spread' ~ pro-strět5; po-zbr-qtb 'consume' ~ pozrět5; vbz-bm-qtb 'take' ~ vzzet5 (and the simplex {bm-} im-qtb e-ti ~ et5); na-cbn-qtb, za-cbn-qtb 'begin' ~ -cet5; klbn-qtb 'swear' ~ klet5; {raz-pbn-} 'crucify' ~ raspet5.

<sup>&</sup>lt;sup>12</sup> Comparative materials from many sources establish that such root-verbs were inherently unaccented (or, in traditional terms, were "circumflex", presumably with a long vowel with falling intonation). The same verbs also had -t- in the past passive participle (cf. §11.321). Accented stems (called "acute"; presumed to have rising intonation) had no -to in 2-3 sg. aor. and -en- in past pass. part. Thus pbj-'drink' had pito, but bbj- 'strike, beat' had only bi and bbjen-/bijen-. These relationships are somewhat obscured in some OCS examples, particularly in Supr.

Here also the irregular *pojots pěti* 'sing' ~ *pěts* (and *vъspěts* 'started to sing'). Živ-qtъ 'live' has 2-3 sg. žive, but also (Su) жи and прижитъ 'bore'.

10.53 The verb *truti turqta* 'rub' has the expected form -trb and also *-ture* (cf. §16.522).

10.60 The unproductive types of aorist

**10.601** The most widespread type of older aorist is the "*root-aorist*", attested by over 650 examples with some 28 verbs (listed below, \$10.81–.84), including the common *idqts* 'go'. They have no aorist-marker, only the following person-number desinences:

	singular	dual	plural
lst person	-Ъ	*-ově	-отъ
2nd person ]		*-eta	-ete
3rd person	-e	-ete	-9

The stem is a root ending in an obstruent, with zero-classifier or a basic stem followed by -nq+ (which is deleted in the aorist): e.g. **pad**-qtb 'fall' ~ padb pade padq; mog-qtb 'be able' ~ mogb može mogq; vbz-**dvig**nq-ti 'lift' ~ vbzdvigb vbzdviže vbzdvigq.

**10.602** The forms of the "*s*-aorist" occur with certain C-stems (in *b*, *t*, *d*, *s*, *z*) and those in b + nasal (see below, §10.82). The desinences are like the normal type except that they have *s* instead of *x* or  $\check{s}$ :

	singular	dual	plural
lst person	-5Ъ	*-sově	-somъ
2nd person )		*-sta	-ste
3rd person	• <b>-(e)</b> s	-ste	-sę

The desinences are consonantal; stem-final consonants are therefore subject to truncation.

10.6021 Nasal stems truncate per §10.22: {ьm} vъz-ьm-qtъ im-qtъ 'take' ~ vъzęsъ vъzę vъzęsę, ęsъ ę ęsę; {raz-pьn-} raspьn-qtъ 'crucify' ~ raspęsъ raspę raspęsę.

**10.6022** Stem-final obstruents are deleted except in 2-3 sg., where the vowel e is inserted: {sbljud-Ø-s-b} 'I observed' > sbljus; {-bljud-Ø+s} > -bljudes > sbljude.

Moreover, the low lax root-vowels e and o are replaced by tense ě and a: ved-qtь 'lead' ~ věsъ, vede, věsomъ, věsę; bod-qtь 'pierce' ~ basъ, bode, basomъ basę. 10.603-10.7

**10.603** The "x-aorist" is attested for stems in k and g (§10.84). The desinences are those of the normal aorist except that they do not admit the initial vowel o. Stem-final obstruents are deleted, and root-vowel e becomes  $\check{e}$  (except in 2-3 sg., where the vowel e is inserted): rek-qtb 'say' ~  $r\check{e}x\mathfrak{z}$ ,  $r\check{e}\check{c}e$ ,  $r\check{e}ste$ ,  $r\check{e}se$ .

**10.604** Note that the 2-3 singular has the same form in all types of aorist.

**10.7** Historically, there were two aorist formations. The "root aorist" added special non-present person-number desinences directly to the verbal root; the OCS examples are archaic relics that have survived in very common verbs. The "sigmatic aorist" had an explicit tense-sign *s* (written with the letter *sigma* in Greek) before the desinences: OCS displays two obsolescent forms of this paradigm, the "*s*-aorist" and the "*x*-aorists", and a productive type, the "*ox*-aorist".<sup>13</sup> During the OCS period some dialects were eliminating most of the unproductive forms. The root-aorists and the *s*- and *x*-aorists of consonantal stems were replaced by the ox-type (*pads, věss, rěxs ~ padoxs, vedoxs, rekoxs*), while s-aorists of nasal stems gave way to the x-type (*ess ~ exs*). For example:

Root > 0x: идъ > ндохъ; 2-3s иде, 3du идете > идосте (-та); идомъ > идохомъ, идете > идосте, идж > идоша. двигъ > двигохъ, 2-3s движе; 3du движете > двигосте (-ста); двигомъ > двигохомъ, двигж > двигоша

S > OX: вѣсъ > ведохъ, 2-3s веде; 3du вѣсте > ведосте (-ста), вѣсомъ > ведохомъ, вѣсте > ведосте, вѣсм > ведошм

S > x: мсъ > мхъ, 2-3s мтъ (м); 3du мсте (мста), мсомъ > мхомъ, мсте, мсм > мшм

S > ox: рѣхъ > рекохъ, 2-3s рече; 3du рѣсте > рекосте (-ста), рѣхомъ > рекохомъ, рѣсте > рекосте, рѣша > рекоша

To illustrate the distribution of these forms in the texts, here are the percentages of the non-productive forms in each ms. Figures in parentheses are based on fewer than five examples.<sup>14</sup>

<sup>&</sup>lt;sup>13</sup> What we call the s-aorist represents the inherited sigmatic form, with the x-aorist as a variant that utilized the Slavic reflexes of old s (s before t, š before front vowel, x otherwise). The ox-aorist is a further development that inserts a vowel between stem-final obstruent and the x/s/s desinences.

<sup>&</sup>lt;sup>14</sup> It is worth noting that the common verb rekots 'say' accounts for all of the old x-forms in Su and more than 95% of them everywhere but Ps, which has 40 x-forms from rek- and 16 from 5 other roots. Of all attested root-aorists, the verb *idots* accounts for a full two-thirds, -rět- 'come upon' (§16.55) another 12.5%, mog-, pad- and vrbg- another 10%, while the last 10% includes 26 verbs. These facts are an excellent illustration of the principle that irregularities persist in words of highest frequency.

	Ps+2N	Mar	Cloz	Euch	As	Zo	Sav	Supr
type <i>rěx</i> ъ	100	100	(100)	100	<del>99</del>	99+	89+	72
type ids	100	100	100	76	85+	70	66	-
type něsz	100	100	100	(50)	96	43	7+	-
type ęsъ	95+	93	(-)	(75)	64+	10	-	-

**10.80** The following lists signal the attested forms of the unproductive aorists. When no specific reference is given after a form, it means that more than five occurrences are attested; further, the forms may have different prefixes. Chapter-verse citation without indication of ms means that the form occurs in more than one of the Gospel texts in the given passage. A citation followed by + means that there are one to three other occurrences. Otherwise the lists are exhaustive.

#### 10.81 Root-aorists

**10.811** pad-qts 'fall': падж. krad-qtь 'steal': оукрадж Mt 28:13 Mar, -*l*ěz-qtь 'go' излѣӡж J 21:9, вълѣӡж J 6:24. tres-qtь 'shake': сътрасъ са Ps 108:23 (see also §10.84). -grez-qtь 'sink' поградж Ps 2N Ex 15:5, 8. mog-qtь 'be able': възмогъ Ps 39:12; 3du изнеможете Ps 17:37, 87:10, 108:24; възмогомъ Mk 9:28+; могж. -ret- 'come upon': обрѣтъ; 3du обрѣтете, сърѣтете; обрѣтомъ J 1:42+; обрѣтж; сърѣтж J 4:51. sesti seqdqts 'sit down': сѣдъ Ps 25:4, сѣдомъ Ps 136:1, сѣдомъ Ps 136:1, сѣдж Ps 118:23+. -*lešti* -*legqt*s'lie down': възлегж Mk 6:40 Mar. vrbgqts vrěšti'throw': -бръгж. idqtsiti 'go': -идъ; 3du изидете; идомъ; идете; идж. edqts\*exati 'ride' прѣѣдж Zo = въѣдж Mar. u-nbz-qtь 'pierce': оуньъж Ps 37:3 = Euch 76a4. \*o-xrsm- 'get lame' (§15.771): охръмж Ps 17:46.

10.812 u-glbb-nq-ti 'get stuck' (оуглебъ Ps 68:3; оугльбж Ps 9:16); \*vъskys-nq-ti 'get sour' (въскъисж L 13:21 As); and \*{svęd-nq+} 'be scorched' (присвадж Mt 13:6) happen to be attested with no alternate forms.

With root and regular forms: -běg-nq-ti 'flee' (прибътъ Euch 85а6, идбътъ Cloz 6b34; отъбътъ Eu 48а6 ~ отъбътоша Su 229.20). na-vyk-nqti 'learn' (навъікня Ps 105:35 ~ навъікоша Su 488.12). vъs-krьs-nq- 'be resurrected' (въскръся Su 471.4 ~ въскръсоша 386.16). -nik-nq-ti (възнікя Ps 91:8 ~ възникоша Su 39.12), {iz-čez-nq+} (§3.311) 'disappear' (1 sg инезъ Is 38:12, 3du инезете Ps 68:4, инезя Ps 36:20, 63:7, Cloz 13а33, ичезя Ps 101:4 ~ ищезоша Euch 62а22).

With root forms, regular forms without nq, and regular forms with nq: dvig-nq-ti 'move' (by a big to be the state of the stat ~ 3 sg оусъхня Su 343.28+). -top-nq-ti 'drown' (истопя Mt 8:32 Sav = оутопя Mar As ~ оутопоша Zo, истопоша Su 401.3 ~ истопияша Su 197.10). po-tok-nq-ti sę 'stumble against' (потъкя са Mt 7:25, 27 Sav; Cloz 12b20, 23 ~ потъкияща са Su 448.16).

10.82 S-aorist. bljud-qtb 'watch': съблюсъ J 15:10, съблюсомъ Ps 2N Deut 3:30, съблюса J 15:20. bod-qtb 'pierce': пробаса J 19:37+. nes-qtb 'carry' възнъсъ Ps 65:17+; 3 du. възнъсте Ps 103:1, L 2:27 As; възнъса L 2:22+; принъс. ved-qtb 'lead': привъсъ Mk 9:17+; 3 du. привъсте Mt 21:7+, въвъсте L 2:27 Zo Mar, Ps 42:3, изъсте Ps 118:136; въвъсомъ Mt 25:38; привъсте L 23:14, J 7:45; въс., привъс. сvbt-qtb cvisti 'bloom': процвиса Cloz 13b4. -vrbz-qtb -vrěsti 'tie' отъбръсъ Ps 38:10, 118:131; 3 du. отвръсте са J 9:10+; отвръс. {raz-sup-} 'scatter': расоуса са Ps 140:7 2N. greb-qtb 'bury': погреса Mt 14:12+. klon-qtb 'curse': класъ са Ps 88:4+. -p6n-qtb 'stretch, crucify': пропаса, распаса. -čbn-qtb 'begin': начасъ Ps 76:11; начаса Ev. {Бт-} im-qtъ ęti 'take': бъсъ Ril VIII<sup>1</sup> 17; побасъ L 14:20 Mar; приясъ J 10:18 Mar As, приясъ Vat, бъсомъ L 5:5 Mar; бъса, въздаса Ev, Ps.

For met-qtb, čbt-qtb, ěd- 'eat', and tres-qtb, see §10.84.

10.83 X-aorists. rek-qtъ 'say': pkxъ; 3du pkcte/pkcta; pkxomъ, pkcte, pkum. tek-qtъ 'run': tkxъ Ps 58:5, 118:32; 3du tkcte Mt 28:8; tkum. vlěk-qtъ 'drag': быбакхъ Ps 118:131, събакшм, избакшм, обакшм. -lękqtъ 'bend': сълахъ Ps 37:7 = Euch 76a5; сълашм Ps 55:7, налашм Ps 10:2, 36:14, 63:4. sěk-qtъ 'cut': раскшм Ps 73:5. vъz-žeg- 'enkindle': въжким Ps 73:7 (for expected vžaše, §3.5c1).

**10.84** A few verbs have more than one attested unproductive aorist.

tres-qtъ 'shake': root-form сътрася са Mt 28:4 As, s-form сътраса са Mar ~ сътрасоща са Zo Sav.

*męt*-qtъ 'stir, disturb': s-forms съмасъ са Ps 76:5+; съмасомъ Ps 89:7; възмаса, съмаса са Ps ~ x-forms възмаша Mk 6:50 Mar, съмаша Mt 14:26+.

čы-qtъ čisti 'read': s-forms нука Ps 21:18, унса J 19:20 Mar As ~ x-form унша Zo (~ уьтоша Sav).<sup>15</sup>

ěd-ętъ 'eat': s-forms ѣсъ Ps 101:10, ѣсмь [for ѣсомъ? L 13:26 Mar], ѣсм Mk 8:8 Mar, поѣсм Ps 77:45+, сънѣсм Ps 105:28+.

<sup>&</sup>lt;sup>15</sup> SJS lists orbeta, orbetaua 'answer' as a defective verb with only 3rd pers. aor. forms. No root-final C is deleted in 2-3 sg aor (except for irregular dad-, ěd- ~ dasto/da, ěsto/ě), and, moreover, no C-verb with the root vět is otherwise attested; it is safe to regard the 4 examples from As and the 2 from Sav as scribal errors— both scribes are prone to omit syllables. All six instances correspond to normal otověšta(šę) forms in the corresponding verses in other mss, and both forms are correctly written scores of times.

**10.90** In irregular verbs, the aorist stem is normally (i.e. except as noted above) the same as the infinitive stem: *b*<sub>b</sub>*r*<sub>a</sub>-ti berqtb 'gather' ~ *b*<sub>b</sub>*r*<sub>a</sub>xs. gbna-*ti ženqtb* 'drive' ~ gbnaxb; *br*<sub>a</sub>-ti borjqtb 'fight' ~*br*<sub>a</sub>xs: *p*ě-ti pojqtb 'sing' ~ *p*ěxtb: {leg-Ø+} lešti lęgqtb 'lie down' ~ *legoxb*; {sèd-Ø+} sěsti sędqtb 'sit down' ~*s*ědoxb; {-rět/ręt} -rěsti -ręštqtb 'come upon' ~*-r*ětoxb. But the present stem serves for at least the regular aorist in: *vrbg*-qtb vrěšti 'throw' ~ *vrbgoxb* (and *vrbgb*), *čb*-qtb čisti 'read' ~ *čbtoxb* (but *čisb*), *cvbt*-qtb cvisti 'bloom' ~ *cvbtošę* (but *cvise*), -*vrbz*-qtb -vrěsti 'tie' ~ *vrbzoxb* (but *-vrěsb*).

10.91 The unprefixed verb byti 'to be' has two sets of a orist forms: from the stem  $b\check{e}$ -  $(b\check{e}x \, b\check{e} \dots b\check{e}\check{s}e)$  and from the stem by-  $(byx \, byst \, byst \, byse)$ . The  $b\check{e}$ -forms were imperfective and the by-forms perfective (see also §16.1, §21.21). 2-3 sg by is occasionally found for byst b.

**10.911** Prefixed forms -byti (e.g. zabyti 'forget', prěbyti 'remain') have only -byxz, -bystz, etc.

# 11. Past participles

OCS has a *past active participle*, a *resultative participle* (often called "second past active participle") and a *past passive participle*.

11.11 The (first) past active particple has the formant - $\mathbf{b}\mathbf{\dot{s}}$ - or - $\mathbf{v}\mathbf{b}\mathbf{\dot{s}}$ plus the soft twofold nominal or compound desinences. The nominative singular masculine-neuter surface forms end in - $\mathbf{a}$ , - $\mathbf{b}$ , or - $\mathbf{v}\mathbf{a}$  (representing underlying - $\mathbf{\ddot{s}}$ - $\mathbf{\emptyset}$ ; word-final consonant is deleted). For the declension see \$4.18-10, 4.31.

**11.12** The vocalic suffix  $-a\delta$ - is used with i-verbs, and with C-verbs (including nq-verbs that lose the nq), excepting stems in j or the group ov. The consonantal suffix  $-va\delta$ - is used with all other verbs (i.e. those in -ov, -j, -nq,  $-\delta$ , -a).

11.13 The classifier -i+ becomes j before the vocalic suffix  $-\delta\delta'$ , triggering iotation in the stem-final consonant(s), and the back  $\delta$  becomes front  $b: \{\text{pros-}i+\delta\delta'i\} > \text{pros}\delta\delta'i$  'having begged (Nsg fem)'.

Before the consonantal suffix, stem-final j is deleted, and -ov > u: {dělaj+vъš-i} > dělavъši 'having done (N sg fem)', {kryj- $\emptyset$ +vъš-i} > kryvъši 'having hidden (Nsg fem)', {ot-plov- $\emptyset$ +vъš-i} > otъpluvъši 'having sailed away (Nsg fem)'.

Note: roots like сы-qtb 'read', -сыл-qtb 'begin', {ьm} im-qtb ęti 'take', u-mы-qtb 'die, have strong jers before the 55 and are often written with e: наченъше, емъ, въземъшн, оумеръшааго.

**11.14** An innovation within the OCS period is that i-verbs could (like other vocalic stems) take the consonantal suffix  $-vz\bar{s}$ -: *pusti-ti* 'abandon' *puštbši* Mk 10:12 Mar ~ *pustivzši* Zo. Such forms are rare except in Supr, where they constitute the norm.

-*iv* $\delta$  ( $\delta$ -) is absent from As, Ps, Cloz; occurs once in Mar (~ 186x - $b[\delta$ -]), 4x in Zo (~ 163), 3x in Sav [noroyenets] Mt 10:39, in 2 different lections, As *iže pogubits*], nontermets [in a reworded phrase] (~ 69x), under 10% in Euch. In Su, however, 598x ~ 117x, whereby the older forms occur only in a few of the 48 component texts. Further, younger forms of masc. neut nom. sg (*stavorivo*) seem to have been favored over the old forms (*stavorjo*).

11.15 Irregularities. Troti tыrqtā 'rub' has tыrā. Vlěkqtā 'drag' has regular -vlěkā beside -vlbkā. Nebrěgqtā 'neglect' has nebrugā (only in Su) or nebrěgā. Pro-stur-qtā 'spread' seems to have \*prostruvā (Su 311.16 простръвъ) beside regular prostrěvā. Byti 'to be' has byvā. Idqtā iti 'go' has šudā (prišudā, etc.). Dadętā 'give' has davā, ědęt 'eat' has ědā; ědqtā 'ride' has both -čvā and -čxavā.

**11.2** The resultative participle, conveniently called the l-participle, is formed by means of the suffix -l- plus the hard desinences of the twofold nominal desinences. It is found only in nominative short forms.

11.211 The suffix is added directly to the basic stem; since it is consonantal, it may cause truncation of consonantal stems. Stem-final obstruents, except the dental stops (t d, §3.3131), remain. E.g. greb-qtb 'bury' greblo; nes-qtb 'carry' neslo; rek-qtb 'say' reklo; mog-qtb 'be able' moglo.

**11.212** Stem-final t d j is dropped; b + nasal consonant > e; ov > u; br > rb:<sup>16</sup> plet-qtb 'braid' plels: ved-qtb 'lead'; klbn-qtb 'curse' klels; vszbm-qtb 'take' vszels; plov-qtb 'sail' pluls; tbr-qtb 'rub' trbls.

**11.213** No-verbs that lose the classifier in other past forms (\$15.76) may lose it here too: v = z d v i g n q-ti 'lift' v = z d v i g n q l = v = z d v i g n q-ti 'learn' obykle.

11.221 Irregular verbs normally use the infinitive stem for the l-participle: bra-ti borjots 'fight' brals: {leg} lešti lęgots 'lie down' legls; {sěd} sěsti sędots 'sit down' sěls; {ěd} ěsti ědęts 'eat' ěls.

However: vrbg-qtb vrěšti 'throw' vrbglo; cbt-qtb čisti 'read' čblo.

<sup>&</sup>lt;sup>16</sup> Note that the 1-participle stem is identical with the aorist stem (aside from the automatic loss of t/d before l) except for the 6 verbs in br which alternates with rě before all other consonantal suffixes (umbroth umrěti; §15.86), the irregular vlěkoto vlbklo, and the anomalous idoto iti.

11.222 The irregular *idqts iti* 'go' uses the suppletive root *šud: šuls.* 

**11.23** The l-participle is used only in compound verb-forms: the perfect (cf. \$14.1), the pluperfects (\$14.2), the conditional (\$14.3), and the future perfect (\$14.4).

The traditional name "second past active participle" is better applied to some usages in post-OCS texts where non-nominative declensional forms are used. The sole example in OCS is acc. sg fem. [\*aroax] warnaaw '[make its fruit] rotten' (cf. \*izgnijoto 'to rot') in Sav, Mt 12:33, where the other mss have [plods] zəlo '[make its fruit] bad'.

# 11.30 The past passive participle

Only transitive verbs can form this participle, and it is rare in imperfective verbs.

11.31 The suffixes are -t-, -n- or -en- plus the hard desinences of the twofold nominal or the compound declensions.

**11.32** The suffix -t- is restricted to certain sonorant-stems, and it effects truncation. It is regular with stems in  $b + nasal: -p_{bn}-qtb$  'crucify' raspets, propets: {bm} impts eti 'take' ets, vbz-bm-qtb vbz-e-ti ~ vbzets.

It is used also with  $-v_{bj}$ -qtb 'wind'  $\sim -vit$ , pro- $l_{bj}$ -qtb 'pour out'  $\sim$  prolits; pěti pojqtb 'sing'  $\sim p$ ěts; požbr-qtb 'swallow'  $\sim p$ ožrbts (§15.86: but požrents 'sacrificed' §16.521); -vrbz-qtb -vrěsti 'tie'  $\sim otvrbsts$  'open'; and uvęsts 'crowned' from uvęzqts.

**11.33** The suffix **-n**- is used with the classifiers  $a, aj, \check{e}$ , and  $\check{e}j$ , whereby the *j* is truncated: *veza*-ti 'tie' ~ *vezane*, *sěja*-ti 'sow' ~ *sějane*; *pomeně*-ti 'remember' ~ *pomeněne*; *dělaj*-qtb 'do' ~ *dělane*: *razuměj*-qtb 'understand' *razuměne*.

11.34 The suffix -en- is used with all other stems.

It effects the generation of j from the -i+ classifier, and therefore iotation: pros-i+ 'beg' ~ prošenz; sz-lom-i+ 'smash' ~ szlomljenz;  $mq\ddot{c}$ -i+ 'torment' ~  $mq\ddot{c}enz$ .

Note that the "epenthetic 1" does not always develop and the spellings may fail to indicate the /j/ we expect, see §2.4521.

**11.341** In a few verbs the vocalic suffix is preceded by a v (which may be interpreted as a variant of stem-final j in some stems):

Nq-verbs that retain the nq in past forms have -nov-en: rinq-ti 'push' ~ rinovenz. Root-verbs ending in back-vowel +j: kryj-qtz/krzj-qtz 'hide' ~ krzvenz, myj-qtz/mzj-qtz 'wash' ~ u-mzvenz, obuj-qtz 'put on shoes' ~

#### 110

obuvenz. The verb šuj-/šij- 'sew' has šuvenz. Zabyti 'forget' (§16.11) has zabzvenz.

11.351 Irregular verbs normally form the past passive participle on the infinitive stem: stbla-ti steljqtь 'spread' ~ postblanz; bbra-ti berqtь 'collect' ~ -bbranz; gzna-ti ženqtь 'drive' ~ -gznanz.

11.352 However, the verbs dom-qtb duti 'blow', vrbg-qtb vrěšti 'throw', and tlok-qtb tlěšti 'knock' use the present stems: nadomeno, -vrbženo, -tločeno. Vlok-qtb vlěšti 'drag' has both -vločeno and -vlěčeno. Kla-ti kolj-qtb 'stab' has both zaklano and zakoleno 'slaughtered'.

#### 12.0 Verbal substantive

The verbal substantive is a neuter; it follows the soft type of the normal twofold nominal declension.

12.1 The formant -**ьj**-/-**ij**- is added to the stem of the past passive participle to make the substantival stem.

raspыn	crucify	raspętъe	crucifixion	děl-aj	do	dělanьe	doing
um-ěj	know	итěпье	ability	glagol-a	speak	glagolanьe	speaking
тып-ё	think	тыпёпые	opinion	rek	say	rečenьe	statement
pros-i	beg	ргоšепье	plea	dvig-nq	move	dviženьe	movement
rod-i	bear	roždenьe	birth	ri-nq	push	rinovenьe	throwing

Keep in mind that the spelling -be means -bje or -ije.

**12.2** The verbal substantive or name of the action may have variants (perhaps with slightly different meaning) involving the distribution of "participial" -t- and -v-. E.g.:

sěj-a	sow	sějanьe sětie	sowing; planting	и-ты	die	umrыie	death
trov	feed	otrovenьe	poisoning	slov	be known	slutie	fame
by	be	bytьe	being	rek	say	rečenьe	statement
za-by	forget	zabytьe zabъvenьe	oblivion, forgetting	pě-ti poj-qtz	sing	рётье рёпье	singing
bыj	strike	ььепье	striking	и-въј	kill	ubitie	murder
bra-ti borj-qtъ	fight	branьe borjenьe	fighting	žыr-qtъ žrь-ti	sacrifice	žrыie	sacrifice
vrbz-	untie	otvr ьstie	opening	i- id-,	go	sъn-itьe	descent;
vrěz		razvrъzenie		šьd-		ŠЬSTЬ€	going
		pouvrъzenie	2			šьstvье	

It is probable that dialects differed in how they created and retained this kind of derivative; in particular Supr has many distinctive formations.

# 13.0 Infinitive and supine

13.1 The infinitive desinence is -ti; the supine desinence is -tb. Since the only consonant that can stand before t is s, these t- suffixes cause truncation in consonantal stems (except s).

13.2 Some stems ending in vowel plus glide (j v) or sonorant replace the VC with a vowel:

**a**. ov > u (cf. §3.72): slov-qtb ~ sluti slutb 'be renowned'; plov-qtb ~ pluti plutb 'sail'; vzzdrov-qtb 'bellow' ~ vzzdruti vzzdrutb (§15.841).

**b**. bj/bj > i/y (§3.313): bbj-qtb or bij-qtb ~ biti bitb 'strike'; mbj-qtb or myj-qtb ~ myti mytb 'wash'.

c. Otherwise, j and v are deleted:  $\check{c}uj$ -qt $\triangleright \sim \check{c}uti \check{c}ut\sigma$  'sense, hear';  $\check{z}iv$ -qt $\triangleright \sim \check{z}iti \check{z}it\sigma$  'live' (§15.842).

d. b + nasal > e (§3.313): klon-qtb ~ klęti klęto 'curse'; žonj-qtb ~ žeti žęto 'reap'; žom-qtb ~ žeti žęto 'squeeze'(§15.83).

e. ы > rь (§16.521): žы-qtъ ~ žrыi žrыъ 'sacrifice'

**13.3** Root-final velar (k g) combines with desinential t in the group *št* that functions as a palatal unit (§3.3131a). The final  $\mathfrak{s}$  of the supine desinence automatically becomes b: rek-qtb ~ rešti reštb; mog-qtb ~ mošti moštb 'be able' (§15.85).

13.31 The stems *tlbk*-qtъ 'knock' and vrьg-qtъ 'throw' have infinitives *tlěšti* and vrěšti. Cf. §15.874.

13.32 The stem strig-qtb 'shear' has postrěšti.

13.4 Root-final labial p and b are deleted: tep- $Qtb \sim teti tetb$  'beat', greb- $Qtb \sim greti gretb$  'dig' (§15.824).

13.5 There are some cases where the same written form serves as infinitive for two basic stems: *Pomazati* 'anoint' may contain the full basic stem of perfective {po-maz-a+}, or represent a truncated shape of imperfective {po-maz-aj+} (derived by \$13.2c above). The corresponding third person plurals are *pomažqts* and *pomazajqts*.

# 14. Compound tenses

14.01 The resultative (l-)participle may be used with the forms of the verb byti 'to be' in constructions which can be considered compound

14.01-14.2

tenses; the perfect, the pluperfects, the conditional, and the rare future perfect.

**14.02** The past passive participle frequently occurs with forms of *byti* in constructions which may translate a Greek passive and be rendered by an English passive, but they cannot be regarded as compound tenses. Similarly the active participles with *byti* are not compounds, but merely copula plus verbal adjective. For example, *bě* že eterь *bolę* Lazarь (J 11:1) "there was a certain ailing man Lazarus" ~ Marija ... eęže bratrъ Lazarь *bolěaše* (J 11:3) "Mary ... whose brother Lazarus ailed."

#### 14.1 The perfect

The forms of the l-participle are frequently used with the present forms of *byti (esmb,* etc., §6.73) to express an action which took place in the past, but whose results are still significant: e.g. ašte ty *esi vozels* i, pověždb mbně kbde i *esi položilo* (J 20:15) "if thou have borne him hence, tell me where thou hast laid him"; otrokovica *něsto umrbla* nb spitb (Mk 5:39) "the damsel is not dead, but sleepeth"; *něste* li čoli jako sbtvorjbi iskoni mozbsku polb i ženbsku sotvorilo ja esto (Mt 19:4) "have ye not read that he which made them at the beginning made them male and female?"

In the third person singular the auxiliary is omitted fairly frequently in certain of the texts within the Suprasliensis, but there are no parallels in the other codices.

The perfect is not common, chiefly because the texts which we have do not need to express this particular relationship very often.<sup>17</sup>

# 14.2 The pluperfects

Two pluperfect tenses are formed by the l-participle in conjunction with the imperfect or imperfective aorist of byti (běaše/běaxq or běxz etc.). E.g. mzno3i že otz ijudei běaxq prišuli kz Martě i Marii da utěšętz i (J 11:19) "and many of the Jews were come to Martha and Mary to comfort them"; ne bě že ne u Isusz prišulz vz vzsz (J 11:30) "now Jesus was not yet come into the town". The use of the imperfect auxiliary shows that the past moment is coordinated with some other moment, mentioned or simply

<sup>&</sup>lt;sup>17</sup> There are about 600 attested examples of the perfect, against some 10,000 aorists and 2300 imperfects. The precise meaning in individual cases is open to a variety of interpretations, depending on assumptions about aspect, tense, and the degree of dependence on (or independence from) the Greek verb-form underlying the OCS translation. For data, literature, and discussion, see Dostál (Studie, pp. 603 ff.); also K. Trost, *Perfekt und Konditional im Altkirchenslavischen* (Wiesbaden, 1972).

implied by the context; the use of the aorist states an independent action, simply a moment in the past. The participle in both cases shows an action which had started even previous to the past moments implied by the auxiliaries, but whose results were still pertinent.

# 14.3 The conditional

The following auxiliary forms (from the verb byti) are used with the *l*-participles to express a conditional mood: 1 sg. bimb, 2-3 sg. bi; 1 pl. bims, 2 pl. biste, 3 pl. bq or bise. The dual forms are not attested. In Su and Sav these forms are rare; the perfective aorist forms of byti (byxs, etc.) are used instead, whereby only by (never bysts) serves for 2-3 sg. The conditional of 'to be' is usually  $bimb \ byls$  (or  $byxs \ byls$ ), but occasionally the participle is omitted. Very rarely a passive participle is used with bimb.

14.31 The forms are used either in conditions or in purpose clauses: ašte bi věděla darь božьi, i kъto estь glagoljęi ti daždь mi piti, ty bi prosila u njego i dalъ ti bi vodo živo (J 4:10) "If thou knewest the gift of God, and who it is that saith to thee, Give me to drink; thou wouldest have asked of him and he would have given thee living water". Ašte otь sego mira bi bylo cěsarjьstvo moe, slugy ubo moę podvizaly sę bišę da ne prědanь bimь ijuděomь (J 18:26) "If my kingdom were of this world, then would my servants fight, that I should not be delivered to the Jews." In Sav, аще ... бы было ... подвиглы са быша ... прѣдань быхь.

In Su there are five instances of  $a \pm i$  and one of  $a \pm i \pm i$ , which apparently are contractions of  $a \pm i$  ('if') with bi and  $bi \pm i$  (Vaillant 256).

# 14.4 The future perfect or "futurum exactum"

In seven instances l-participles are used with forms of the future  $b \rho d \rho t \sigma$  to signal an action which is viewed as completed before some future moment and whose results are important for that moment; e.g. ašte ključitъ sę da  $b \rho det \sigma s \sigma gnilo$  vъpadъšee, dostoitъ proliěti (Euch 20a25) "If it happens that what has fallen in (i.e. into the wine) has become rotten, it (the wine) should be thrown away."

# 15.1 Verbs with basic stems in -i+

Infinitive		просити 'b	eg'	поустити 'let go'			
		Pres	sent	Imperative			
Sing.	1	прошж	поуштя	-	-	1	
	2	просиши	поустиши	проси	поусти	2	
3 проситъ			поуститъ	проси	поусти	3	

15.1–15.12 CONJUGATION						
Dual	1	просивѣ	поустивѣ	просивк	поустивѣ	1
	2	просита	поустита	просита	поустита	2
	3	просите	поустите	_	-	3
Plur.	1	просимъ	поустимъ	просимъ	поустимъ	1
	2	просите	поустите	просите	поустите	2
	3	просатъ	поустатъ	-	-	3
Pres. a	ct. pa	rt. nom. sg.	masc.	neut.	fem.	
			просм		просмшти	
			поуста		поусташти	l
Pres. p	ass. p	oart. nom. sg.	masc.	neut.	fem.	
			просимъ	просимо	просима	
			поустимъ	поустимо	поустима	
		Impe	rfect	Aori	st	
Sing.	1	прошаахъ	поуштаахъ	просихъ	поустихъ	1
	2-3	прошааше	поуштааше	проси	поусти	2-3
Dual	1	прошааховѣ	поуштааховѣ	просиховъ	поустиховѣ	s 1
	2	прошаашета	поуштаашета	л просиста	поустиста	2
	3	прошаашете	поуштаашете	е просисте	поустисте	3
Plur.	1	прошаахомъ	поуштаахомч	ь просихомъ	поустихома	
	2	прошаашете	поуштаашете	е просисте	поустисте	2
	3	прошаахж	поуштаахж	просишљ	поустиша	3
Resulta	ative	part. nom. sg.	masc.	neut.	fem.	
			просилъ	просило	просила	
			поустилъ	поустило	поустила	
Past ac	tive p	oart. nom. sg.	masc.	neut.	fem.	
			•	просивъ)	прошьши	
_			•	(поустивъ)	поуштьши	
Past pa	issive	part. nom. sg	, masc.	neut.	fem.	
			прошенъ	прошено	прошена	
** • •	_		поуштенъ	поуштено	поуштена	
Verbal	subs	tantive	прошенье	Supine	проситъ	
			поуштенье		поуститъ	

**15.11** The stem-final consonant or sequence of consonants undergoes iotation, if possible, in the first person singular present ( $\S$ 6.23), all forms of the imperfect ( $\S$ 9.111), in the past active participle ( $\S$ 11.12), and the past passive participle ( $\S$ 11.34) and the verbal substantive.

**15.12** Beside the older, more widely used past active participle in  $-b(\check{s})$ - (with truncation and iotation), the younger form with  $-v\sigma(\check{s})$ - added to the basic stem is well attested, particularly in the codex Suprasliensis (§11.14).

115

**15.13** This productive class is represented by about 350 verbs (or, counting the derivatives made with different prefixes, over 800 lexical verbs). It includes a large number of verbs which occur only once or twice in the manuscripts, however, and thus will constitute only about 20% of the verbs occurring in a given text.<sup>18</sup>

#### 15.2 Verbs with basic stems in -ě+

Infinitive мынати		менфти "	think'	трыпѣти 'suf		
		Prese	ent	Impe	erative	
Sing.	1	мьйж	трыпліж	_	-	1
	2	мриипи	тръпиши	мени	трыпи	2
	3	мьнитъ	трыпитъ	мени	тръпи	3
Dual	1	<b>мринв</b> ф	трыпивѣ	меннвф	трьпивъ	1
	2	мьнита	тръпита	мьнита	трыпита	2
	3	мьните	трыпите	-	-	3
Plur.	1	менимъ	трьпимъ	мьнимъ	трыпимъ	1
	2	мьните	трыпите	мьните	трыпите	2
	3	мьнатъ	трыпатъ	-	-	3
Pres.	act.	part. nom. sg	. masc.	neut.	fem.	
		менж		мьнашти		
			трыпа		трыплшти	
Pres.	pass	. part. nom. s	g. masc.	neut.	fem.	
			мьнимъ	мьнимо	мьнима	
			трыпимъ	трыпимо	трыпима	
		Impe	erfect	Aoris	t	
Sing.	1	мьнњахъ	трыпѣахъ	мьнѣхъ	трыпѣхъ	1
2	-3	мьнжаше	трыпѣаше	менф	трыпњ	2-3
Dual	1	мьнѣаховѣ	трыпњаховњ	мьнѣховѣ	трыпѣховѣ	1
	2	мьнѣашета	трыпћашета	мьнѣста	трыпъста	2
	3	мьнѣашете	трыпѣашете	мьнѣсте	трыпѣсте	3
Plur.	1	мьнѣахомъ	трыпћахомъ	мьнѣхомъ	трыпѣхомъ	1
	2	мьнвашете	трыпѣашете	мьнѣсте	трыпѣсте	2
	3	мьнѣахж	трыпѣахж	мендша	трыпѣшљ	3

<sup>&</sup>lt;sup>18</sup> The figures here and under the other classes of verbs are intended to show the relative importance of the attested verbs and verbal types in the OCS system. The complete lexicon of OCS contains upwards of 1100 morphological verbs plus about 1400 stems with different prefixes, a total of over 2500 lexical verbs. These numbers are approximate, since there is some disagreement about the derivation of some attested forms, as well as just which manuscripts and fragments are to be used as sources. The general picture, however, is reasonably clear.

116

sg. masc.	neut.	fem.
мьнфлъ	менфло	мьнѣла
трыпѣлъ	трыпѣло	трыпњла
g. masc.	neut.	fem.
менде	ያЪ	мьнѣвъши
тръпѣ	BЪ	трыпѣвъши
sg. masc.	neut.	fem.
мьнжнъ	мрнфно	менжна
трыпѣнъ	трыпено	тръпѣна
мри. <del>р</del> ире	Supine	мьнѣтъ
трыпжные		трьѣтъ
	мыналь трыпаль g. masc. мынан трыпа sg. masc. мынанъ трыпанъ мынанье	мынтала мынтало трыптала трыптало g. masc. neut. мынтавъ трыптавъ sg. masc. neut. мынтанъ мынтано трыптанъ трыптано мынтанъе Supine

**15.21** This group has the u/a marker in the present forms, like the iverbs, with the same iotation in 1st person singular. Since the past passive participle has the consonantal -n-suffix, no truncation takes place, and the t is retained.

15.22 There are only 27 verbs in this group (about 80 including all possible prefixed stems); most are intransitive and express a state rather than an activity. Болѣти 'be sick', Бъдѣти 'be awake', Белѣти 'order', БИДѣти 'see' (with derivatives including ненавидѣти 'hate', давидѣти 'envy', обидѣти 'disrespect, offend, harm'), висѣти 'be hanging', врътѣти 'revolve, turn', Бърѣти 'bubble, boil', горѣти 'burn' (cf. §8.2), гръмѣти 'thunder', дърѣти 'see, look at, view' (with derivatives including дадърѣти 'blame, hold against', прѣдърѣти 'neglect, disdain'), късмѣти 'be late, slow', къшпѣти 'boil', летѣти 'fly', при-льпѣти 'cling to', мръдѣти 'be come abhorrent', мъмѣти 'think, opine' (and сѫмъмѣти ca or соумъмѣти 'shine', скръвѣти 'be grieved, distressed', смръдѣти 'stink', стъдѣти ca 'be ashamed', сѣдѣти 'sit, be sitting', тръпѣти 'suffer, endure', штѧдѣти 'save', -шоумѣти 'sound' and the irregular χотѣти 'wish'.<sup>19</sup>

#### 15.23 Irregularities

15.231 The singular imperative of вид'ети 'see' is виждь (вижди in Euch), cf. §7.2. (Prefixed stems: не обиди Mk 10:19, не завиди Ps 37:1.)

<sup>&</sup>lt;sup>19</sup> In Ps Sin's margin by ps. 127:3 is a gloss, гобъджини 'yielding rich harvest', presumably to the phrase лода плодовита 'fruitful vine'. It perhaps is from a dialect verb \*rобъд-k+, with {z}. The expected root gobb3 (borrowed from Gothic gabigs 'abundant') should have /ž/ before a front vowel, \*gobb3 (z).

Beside regular pres. passive participle видимъ an archaic form видомъ with the sense 'visible' (and иевидомъ 'invisible') is found (Su only).

**15.232** Xothth 'wish, want; be about to' is irregular in that it has the present-marker e (not i, the expected correlate of -e-): {xot- $\check{e}+e$ -} > xotje-(by §6.13 A2), therefore xouttenn, xoutteth, xoutteth, xoutteth. Xoutteth, xoutteth. 1 sing. {xot- $\check{e}+\emptyset$ -Q} > xotjQ > xoujQ > xoutteth and 3 pl {xot- $\check{e}+e$ -tb} > xotmeth as expected. The imperative also has the mutated stem-shape, **b**cxoutth. The present active participle is regular: masc.-neut. xoth, fem. xothurth, etc.

Twice in Supr and once in Vat (J 5:6) хошти occurs instead of 2 sg. хоштеши (for post-OCS parallels see Vaillant §180). Six times in Su the root vowel is  $\overline{o}$  instead of o: хътѣти 114.1, pres. act. part. хъть 532.24, and the present forms cited in §6.41.

# 15.3 Verbs with basic stem in -a+ preceded by a soft consonant other than j.

		Infinitive כאז Prese	ышати 'hear' nt	дрьжати 'hold' Imperative		
Sing.	1	слышж	дрьжж	-	_	1
	2	слъщнан	дрьжиши	слънши	дрьжи	2
	3	слъщитъ	дрьжитъ	су.ріппи	дрьжи	3
Dual	1	слъшивѣ	дрьживѣ	слышив.	дръживѣ	1
	2	слышита	дрьжита	слъщита	дрьжита	2
	3	слышите	дрьжите	-	-	3
Plur.	1	слышимъ	дрьжимъ	слэннимъ	дрьжимъ	1
	2	слъщите	дрьжите	слышите	дрьжите	2
	3	¢ለъነሠሐፕъ	ፈ <mark>р</mark> ьж <mark></mark> атъ	-	-	3
Pres.	act.	part. nom. sg.	masc.	neut.	fem.	
			слыша	•	слышашти	
			дрьжа		дрьжашти	
Pres.	pass	. part. nom. sg	g. masc.	asc. neut.		
			слышимъ	слъішимо	слъншима	
			дрьжимъ	дрьжимо	дрьжима	
		Imper	fect	Aoı	rist	
Sing.	1	слышаахъ	дрьжаахъ	слышахъ	дрьжахъ	1
2	-3	слъщааше	дрьжааше	слъща	дрьжа	2-3
Dual	1	слъшааховѣ	дрьжааховѣ	слъщаховѣ	дрьжаховъ	1
	2	слышаашета	дрьжаашета	слъщаста	дрьжаста	2
	3	слышаашете	дрьжаашете	слышасте	дрьжасте	3
Plur.	1	слышаахомъ	дрьжаахомъ	слышахомъ	дрьжахомъ	1
	2	слышаашете	дрьжаашете	слышасте	дрьжасте	2
	3	слъщаахж	дрьжаахж	слышаша	дрьжаша	3

Resultative part. nom.	sg. masc.	neut.	fem.
	слъщалъ	слъщало	слъщала
	дрьжалъ	дрьжало	дрьжала
Past active part. nom. s	sg. masc.	neut.	fem.
	слънша	1 <b>6</b> Ъ	слышавъши
	дрьжа	BЪ	дрьжавъши
Past passive part. nom.	sg. masc.	neut.	fem.
	слъщанъ	слъщано	слъщана
	дрьжанъ	дрьжано	дрьжана
Verbal substantive	слъщанье	Supine	слышатъ
	дрьжанье		дрьжатъ

**15.31** This type is transparently a variant of the ĕ-verbs. The classifier  $\{-\check{e}+\}$  regularly becomes *a* after a palatal consonant, see §3.5c1. Two verbs with roots ending in *oj* belong here,  $\{stoj-\check{e}+\}$  and  $\{boj-\check{e}+se\}$ , and 13 verbs with stems formerly ending in a velar consonant (with about 40 more forms made with different prefixes): влыштати см 'flash', вогати см 'fear', вѣжати 'flee', движати 'move', дръжати 'hold', клачати 'kneel', кричати 'shout', лежати 'lie, be in a lying position', льштати см 'shine', мльчати 'be silent', мъчати 'push', слъщати 'hear', стогати 'stand' (with prefixed stems like достогати 'befit', настогати 'be present', състогати см 'consist') тъштати см 'hasten, be zealous', -тажати (при-, съ-тажати 'acquire').<sup>20</sup>

15.32 Here can be mentioned the uniquely anomalous съпати 'sleep', which has a hard consonant preceding the *a*, but nonetheless this type of present: 3 pl. съпатъ, 1 sg. съпа́ж; 2 sg. съпиши; imv. съпи; pres. act. part. съпа, съпашти.

#### 15.4 Verbs with basic stems in -j-a+.

Infinitive дѣгати 'do'				дагати 'giv	/e'	
		Pres	ent	Impe	rative	
Sing.	1	дѣж	Данж	-	-	1
	2	дъеши	даеши	дъи	Дан	2
	3	дѣетъ	даетъ	дѣи	Дан	3
Dual	1	дњевњ	даєв в	дѣивѣ	дливѣ	1
	2	дњета	даета	дѣита	Данта	2
	3	дњете	даєте	-	-	3

<sup>20</sup> Historically, the forms were presumably \*blwsk-ě, boj-ě, běg-ě, dvig-ě, dwrg-ě, klęk-ě, krik-ě, leg-ě, lwsk-ě, molk-ě, mok-ě, slyx-ě, stoj-ě, twsk-ě, -tęg-ě.

Plur.	1	дѣемъ	Даемъ	двимъ	Данмъ	1
	2	двете	даете	дѣите	даите	2
	3	<b>д</b> ቈѭтъ	<u></u> ፈልѬፐኄ	-	-	3
Pres.	act.	part. nom. sg	. masc.	neut.	fem.	
			<b>ሏ</b> ቴሎ		дѣѭшти	
			Дам		дажшти	
Pres.	pass	. part. nom. s	sg. masc.	neut.	fem.	
			дњемъ	дњемо	дњема	
			даемъ	даемо	даема	
		Imperf	ect	Aorist		
Sing.	1	дѣпахъ	Дагаахъ	дѣахъ	Дагаахъ	1
2	-3	дѣгааше	Дагааше	дѣа	Дага	2-3
Dual	1	дѣгааховѣ	дагааховъ	дѣаховѣ	дагаховѣ	1
	2	дъгаашета	Дагаашета	дваста	Дагаста	2
	3	дѣмшете	Дагаашете	двасте	Дагасте	3
Plur.	1	дѣгаахомъ	дагаахомъ	двахомъ	дагахомъ	1
	2	дваашете	дагаашете	двасте	дагасте	2
	3	двіаахж	дагаахж	дваша	дагаша	3
Resul	tativ	e part. nom.	sg. masc.	neut.	fem.	
			дѣалъ	двало	дѣгала	
			дагалъ	дапало	Дагала	
Past a	ctive	e part. nom. s	sg. masc.	neut.	fem.	
			дѣмвъ		дѣмвъши	
			Дагавъ		ДАГАВЪШИ	
Past p	assi	ve part. nom	sg. masc.	neut.	fem.	
			дѣанъ	дѣано	дѣгана	
			Даганъ	дагано	Дагана	
Verba	l su	bstantive	дѣанье	Supine	дватъ	
			даганье		дагатъ	

15.41 Variant spellings indicate for some verbs a hesitation between basic stems of this class (e.g. sěj-a+, inf. sěja-ti 'sow') and stems with the zero-classifier, effectively therefore consonantal stems (sěj- $\emptyset$ +, 3pl sějqt $\mathfrak{b}$ , inf. sěti [in this case probably older, historically]). The matter is obscured by a dialectical loss of intervocalic *j*, which may bring about contractions, e.g. sějati **chatu** > sěati **chatu** > a new sěti **chtu**.

**15.42** For variant spellings in the present tense showing loss of stemfinal j and subsequent assimilation of the present-marker (aje > aa > a; ěje > ěa) see §6.5).

120

15.43-15.5

CONJUGATION

15.43 The group includes perhaps thirteen verbs (plus about 35 with different prefixes), some of which have variant forms. Offering fairly certain evidence of -a- are: из-валати 'sculpture', вѣвати 'blow', далати 'give'. калати съ 'rue, repent', лагати 'bark', лагати ambush', на-малати 'indicate', -сталати 'stand', сѣлати 'sift', талати 'melt, thaw', vалати 'expect'.

15.431 The form *worders* 'obtains through mediation' (Cloz) does not fit neatly into this group. The underlying segmentation is unclear; cf. *isxodataj-b* 'mediator, conciliator'.

15.44 The prefixed forms of дѣати 'do' (e.g. съ-дѣати 'do', одѣати 'dress') function both as imperfective and perfective.

**15.441** Certain archaic present forms with the stem  $-d\check{e}\check{z}d$ - are explicitly perfective (e.g. OAEMAEML 'we will dress', BLZAEMAEM' I shall raise'). cf. §16.8.

**15.45** Five poorly-attested verbs are usually listed with *-a-* although no forms in *-aja-* or *-ěja-* are to be found in the canonical mss: grěj-qt<sup>\*</sup> 'warm', rěj-qt<sup>\*</sup> 'push', spěj-qt<sup>\*</sup> 'prosper', soměj-qt<sup>\*</sup> 'dare', and volaj-qt<sup>\*</sup> sę 'be tossed [by waves]'.<sup>21</sup> See §15.92.

15.46 Three verbs (plus 14 prefixed stems) in -ьj-а+ (alternating with -ij-а+, cf. §2.61) have -ėj- in the present system: лыати/лиати "pour", pres. лѣмтъ, imv. лѣи, pres. p. part. лѣемъ; смыати/смиати са "laugh" смѣмтъ са etc., and presumably *\*zbjati/\*zijati* "yawn", for which only the pres. parts. ӡѣѧ, ӡѣмщи are attested.

NB:  $l_{bj}-a+(l_{bj}ati/l_{ij}ati l_{bj}e_{t_{0}})$  does not differ in meaning from  $l_{bj}-\emptyset+(l_{bj}e_{t_{0}})$ , cf. §15.93.

#### 15.5 Verbs with basic stems in -ova+ or -eva+.

Infinitive миловати 'pity'			непьштевати	'suppose'		
		Pres	ent	Imperative		
Sing.	1	милоуіж	непьштоуіж	-	-	1
	2	милоуеши	непьшточеши	милоуи	непьштоуи	2
	3	милоуетъ	непьшточетъ	милоун	непьштоуи	3
Dual	1	милоуевъ	непьшточевѣ	милочивѣ	непьшточивѣ	1
	2	милоуета	непьшточета	милоунта	непьшточита	2
	3	милоуете	непьштоуете	-	-	3
Plur.	1	милоуемъ	непьштоуемъ	милоуимъ	непьшточимъ	1
	2	милоуете	непьштоуете	милоуите	непьшточите	2
	3	милоуіжтъ	непьштоуіжтъ	-	-	3

<sup>&</sup>lt;sup>21</sup> Infinitives like въллати, грѣтати, and рѣтати occur in dictionaries, but they are unjustified for OCS.

Pres. act. part. nom. s	g. masc.	neut.	fem.	
		милоужшти		
	иєпьшто	wa	непьштоужшт	и
Pres. pass. part. nom.	sg. masc.	neut.	fem.	
	милоуемъ	милоуемо	милоуема	
	непьштоуемъ	непьштоуемо	непьшточема	
Imper	fect	Ao	rist	
Sg. 1 миловаахъ	непьштеваахъ	миловахъ	непьштевахъ	1
2-3 миловааше	непьштевааше	милова	непьштева	2-3
Du 1 миловааховѣ	непьштевааховъ	миловаховѣ	непьштеваховѣ	1
2 миловаашета	непьштеваашета	миловаста	непьштеваста	2
3 миловаашете	непьштеваашете	миловасте	непьштевасте	3
Pl. 1 миловаахомъ	непьштеваахомъ	миловахомъ	непьштевахомъ	1
2 миловаашете	непьштеваашете	миловасте	непьштевасте	2
З миловаахж	непьштеваахж	миловаша	непьштеваша	3
Resultative part. nom.	sg. masc.	neut.	fem.	
	миловалъ	миловало	миловала	
	непьштевалъ	непьштевало	непьштевала	
Past active part. nom.	sg. masc.	neut.	fem.	
	миловавт	4	миловавъши	
	непьштеі	вавъ	непьштевавъши	l
Past passive part. nom	i. sg. masc.	neut.	fem.	
	милованъ	миловано	милована	
	непьштеванъ	непьштевано	непьштевана	
Verbal substantive	милованье	Supine	миловатъ	
	непьштеванье		непьштеватъ	

**15.501** The classifier **-ova+/-eva+** truncates to *-uj* before vocalic desinences (but see §9.5 for the imperfect).

15.502 In Mar there are cases where the *uje* of the present is written *-uu*-, e.g. трквоуоутъ for трквоуетъ 'needs', радоуоутъ са for радоуетъ са 'rejoices'; cf. §6.5.

15.51 Ova-verbs are a productive class of about a hundred stems (plus another fifty derived with various prefixes). Many of them are rare, so that this class ranks very low in frequency (ca. 2%) in a given sample of text. Without exception, the verbs are derivatives of other OCS words. For example: AAPOBATH 'give gifts' ( $dar \delta$  'gift'), пользеватн 'profit, use' (polb3a 'use, profit'), вѣровати 'believe' (věra 'belief'), хоуловати 'blaspheme' (xula 'blasphemy'), оубаловати or врачевати 'treat' (balii or vračb 'doctor'), радовати см 'rejoice' ( $rad\delta$ ), миловати 'have mercy' ( $mil\delta$  'de-

serving of mercy'), милосръдовати 'have pity' (milosrьdo 'merciful'). свервповати 'be wild' (svereps 'wild'), урьмьновати 'redden' (črытыль 'red'), коуповати (I) 'buy' (kup-i+ti, P). Many obvious neologisms are intervening substantival formed with an suffix - 6stv-: e.g. 'be apostle' (apostolbstvo апостольствовати an 'apostolate'). пророчыствовати 'prophesy' (proroka 'prophet', proročestvo 'prophesy'), послоушьствовати ог съвъдътельствовати 'witness' (posluxo or sověděteljь 'a witness'), четовьтовластьствовати 'be a tetrarch'. оржженосьствовати 'guard' (\*oroženosьсь 'weapon-carrier, guard'), плодоносьствовати 'be fruitful' (plodonosbn 5 'fruit-bearing, fruitful'). Most of this last type are efforts to render one Greek word by one Slavonic word; the number of doublets attested and the clumsy ad-hoc character of some of the more complex examples are clear indications of the productivity of this suffix and therefore of this verbal class.

15.52 The two verbs in  $-bv-a+(\sim -uj-)$  may be listed here: пльвати 'spit' and блывати 'vomit', with 3 pl. pres. forms плюжтъ and блюжтъ.

**15.53** The verbs *o*-snov-a+ti 'found' and kov-a+ti 'forge' are treated in \$15.642.

15.6	Verbs	with	basic	stems	in -	-a+	preceded	by	a hard	consonant
other	than v									

		Infinitive глаголати 'speak' Present		вадати 'tie' Imperative		
Sing.	1	глаголіж	важж		_	1
	2	глаголеши	важеши	глаголн	важи	2
	3	глагол̂етъ	важетъ	глаголи	важи	3
Dual	1	глаголевѣ	влжевѣ	глаголивѣ	важивѣ	1
	2	глагол̂ета	важета	глагол̂ита	важита	2
	3	глагол̂ете	важете	-	-	3
Plur.	1	глагол̂емъ	важемъ	глаголимъ	важимъ	1
	2	глагол̂ете	важете	глаголите	важите	2
	3	глаголіжтъ	B <b>ሐ</b> жѫ <b>т</b> ъ	-	-	3
Pres. act. part. nom. sg. masc.			neut.	fem.		
глагола				•	глаголжшти	
важа				важѫшти		
Pres. pass. part. nom. sg. masc.			neut.	fem.		
глагол̂емъ		глаголемо	глагол̂ема			
			важемъ	важемо	важема	

123

Sing.         1         глаголаахъ         важаахъ         глаголахъ         вазахъ         1           2-3         глаголааше         важааше         глагола         ваза         2-3           Dual         1         глаголааховъ         важааховъ         глаголаховъ         вазаховъ         1	3
<b>—</b> • • • • • • • <b>—</b> • • • • • • • • • • • • • • • • • • •	3
Dual I глаголаатовъ важаатовъ глаголатовъ ваzатовъ I	
2 глаголаашета важаашета глаголаста вадаста 2	
3 глаголаашете важаашете глаголасте вадасте 3	
Plur. 1 глаголаахомъ важаахомъ глаголахомъ вазахомъ 1	
2 глаголаашете важаашете глаголасте вадасте 2	
3 глаголаахж важаахж глаголаша вазаша 3	
Resultative part. nom. sg. masc. neut. fem.	
глаголалъ глаголало глаголала	
вазалъ вазало вазала	
Past active part. nom. sg. masc. neut. fem.	
глаголавъ глаголавъши	
вадавъ вадавъшн	
Past passive part. nom. sg. masc. neut. fem.	
глаголанъ глаголано глаголана	
вазанъ вазано вазана	
Verbal substantive глаголанье Supine глаголатъ	
ваданьє вадатъ	

15.61 The classifier a becomes j before the vocalic desinences of the present system, effecting iotation of the stem-final consonant. (§6.22).

15.62 The expected -i- of the imperative plural is sometimes replaced by  $\check{e}$  (sometimes so written, contrary to normal spelling rules) or a: глаголате, глаголъте for глаголите 'speak!', покажъте, покажате for покажите (*po-kaz-a*-ti 'show'). See §7.21.

**15.63** This class contains about fifty verbs (plus over 60 prefixed stems), some of them very common. Post-OCS evidence indicates that it was still somewhat productive.

алкати алужтъ 👌	hunger	ЗЫБАТИ ЗЫБÂѬТЪ	toss, agitate	
лакати лачжтъ 🕽	nungei	КАЗАТИ КАЖЖТЪ	show	
вадати важятъ	tie	капати капліжтъ	drip	
глаголати глаголіжтъ	speak	клепати клепліжтъ	indicate	
двизати движжтъ	move	клицати кличжтъ	call, cry out	
въз-дрѣмати -дрѣмл̀жт	ndoze	колѣбати колѣбліжтъ	rock	
дыхати дышять	blow	кжпати кжпліжтъ	bathe	
жадати жаждятъ	thirst	лизати лижжтъ	lick	
Хобати Хобліжтъ	eat (with beak)	лѣгати лѣжѫтъ	lie, recline	

лацати лачятъ	set trap	-сыпати -сыпліжтъ	pour, strew
лобъзати лобъжятъ	kiss	-сазати -сажатъ	touch
ЛЪГАТИ ЛЪЖѪТЪ	lie, say untruth	тесати тешятъ	hew
мачати мажжтъ	anoint	тратати траштжтъ	pursue
измрьмьрати «бріятъ	nibble	тазати тажатъ	question, test
орати ор̀жтъ	plow	халати хапліжтъ	bite
пискати пиштжтъ	pipe	чесати чешятъ	comb
плакати плачжтъ	weep	клеветати клевештжтъ	slander
плакати плачжтъ	rinse, wash	клокотати клокочжтъ	bubble,
плескати плештятъ	clap, slap		gurgle
попасати попашжтъ	gird	кльчьтати кльчьштжтъ	cause (teeth)
ристати риштжтъ	run		to chatter
на-рицати -ричжтъ	name, call	ръпътати ръпъштжтъ	murmur
ръдати ръжятъ	whinny	скрьжьтати скрьжьштжтъ	gnash (teeth)
рѣзати рѣжѫтъ	cut	скръгътати скръгъштжтъ	gnash (teeth)
стенати стећжтъ	groan	трепетати трепештятъ	tremble
страдати стражджтъ	suffer	ШЬПЬТАТИ ШЬПЬШТЖТЪ	whisper
стрѣкати стрѣчжтъ	goad	По-ШТьбьтати	
сълати съл̀іжтъ	send	Поштьбьштжтъ	twitter

**15.641** The chief deviations from these patterns imply stems that base competing forms on -a + vs.  $-\emptyset + vs$ . -aj +, chiefly in the present system.

жадати 'thirst' has pres. жадаетъ in Ps, жадаемъ in KF, жадаа in Su beside regular forms with truncated жажде- жажджшт-.

искати 'seek', regularly иштятъ (reflecting *isk-j*- and predictable from *isk-a*+) but fairly often 1 sg. искя, 3 pl. искятъ, pres. act. part. искяште (as though from *isk-\emptyset*+). The other forms always show mutation (either *isk-j-e*- and iotation, or *isk-e*- and KI palatalization), иште-.

метати 'throw', regularly мештять etc. but also метять etc., along with метаните Ps 125:6, and the like. The imperfect has метаахж As, and мештаахж Su. See Vaillant 308; Koch §67.

15.642 Three verbs have -a- before C-desinences and  $-\emptyset$ + before vocalic desinences: ковати ковятъ 'forge' (impf. kovaaxq Ps 128:3); съсати съсятъ 'suck'; тъкати тъкятъ 'push, weave'. Presumably основати 'found' and -ръвати 'tear, pull' belong here, although the presents \*osnovqtъ and \*rъvqtъ do not happen to be attested in OCS.

**15.643** Eight verbs (plus a dozen prefixed stems) have, or may have, an unpredictable alternation of the root vowel in the present system:

стылати стеляютъ 'spread out'. The participle стелашта Su 332.30 perhaps reflects a late dialect change of nasal vowel after a palatal consonant; cf. reg. стелящте Su 341.12.

имати ємліжтъ {j-ьm-a+ ~ j-em-j-} 'take, get' may have -ьmati after prefixes ending in a consonant: въньмати 'hear', объмати 'gather'. The prefixed forms are in competition with derivatives of im-aj+: възиманятъ 'raise', въниманятъ, обиманятъ. Imperfect Mar сънъмаахя L 5:15 ~ Zo сънимаахя.

The vowel of the present stem is found also in infinitive and other forms (principally in Sav): емати, приемати, подъемати. In Su отъмемьам 331.25, отъмемьжштъ 294.4 the root has an epenthetic и (cf. §3.3101).

дьдати дижджтъ 'build'. Su 204.7 съжиждетъ shows dialect assimilation of root-initial z to the following žd (affirmed by post-OCS examples).

пьсати пишжтъ 'write'. Forms with *pbsa*- are well attested, but regularized forms from newer *pis-a*+ are also found, e.g. написати L 2:1 Zo Mar As ~ напсати Sav.

плѣжжшт- 'crawling, slithering'; vpѣnл̀жште 'drawing (water)'; and въслѣпл̀жшт- 'gushing, spurting up' perhaps imply \*plsz-a+, \*črsp-a+, and \* $v\sigma(z)$ -slsp-a+

трѣжетъ 'tears, rends' may correspond to \*trb3-a+; attested forms with тръzа- (only in Su) may represent \*trb3-aj+.

Note. The patterning of vowels in these verbs is less irregular if we allow the underlying representations {pblz pelz, čbrp čerp, sblp selp, tbrz terz} with the same alternation of  $b \sim e$  as in {stbl stel}, etc. Since no syllable-final r or l may occur in surface forms, we may state that before consonant {br b} metathesize to rb lb, and that {er el} become rë lë. Cf. §3.912.

стръгати строужжтъ 'scrape, flay'. However Su 392.5 остроуга indicates a regular (or regularized) \*strug-a+.

**15.644** Five verbs (plus 17 prefixed stems) have -a- (indicating -a+) before consonantal desinences, but no iotation in pre-vocalic forms (as though based on  $-\emptyset$ +), and unpredictable changes in stem vowel.

бърати бержтъ 'collect, take'; дърати держтъ 'flay, rip', and -пърати -пержтъ 'trample' happen not to have imperfects attested.

жьдати жиджтъ 'wait, await' has (in Su) imperfects based on zbd-a+, zid-, and zbd- (жъдааше, жідѣаҳж, жьдѣаҳж).

гънати женжтъ (prefixed with *iz*, *raz* > изгънати разгънати ~ ижденжтъ ражденжтъ, §3.311) has imperfect женѣахж Su 17.21, 196.15.

15.645 Zъвати (with 5 prefixed stems) has the present tense zoвжтъ. Imperfects zъвааше Su 473.16 and зовѣаше Su 516.6.

		Infinitive	ринжти 'push'	авигнжти 'Ш	iove'	
	Present		Imperative			
Sing.	1	ринж	двигнж	-	_	1
0	2	, ринеши	<b>АВИГИЄШИ</b>	рини	двигни	2
	3	, ринетъ	двигнетъ	, рини	двигни	3
Dual	1	, онневъ	лвигневъ	, оннѣвѣ	авигњвњ	1
	2	оинета	двигнета	, ринѣта	двигнъта	2
	3	ринете	двигнете	-	_	3
Plur.	1	ринемъ	двигнемъ	ринѣмъ	авигнѣмъ	1
	2	ринете	двигнете	ринате	Авигнъте	2
	3	ринжтъ	двигнжтъ	_	<u> </u>	3
Pres.	-	part. nom. sg	••	neut.	fem.	-
		r 2	ринъ		ринжшти	
			двигиъ	1	, Двигнжшти	
Pres.	pass	. part. nom. s	••	neut.	fem.	
	<b>F</b>	· · · · · · · · · · · · · · · · · · ·	риномъ	риномо	ринома	
			двигномъ	двигномо	, Двигнома	
Imperfect			Aoris	st		
Sing.	1	ринњахъ	двигнњахъ	ринжхъ	двигохъ	1
2	2-3	ринћаше	двигињаше	ринж	движе	2-3
Dual	1	ринњаховњ	двигнѣаховѣ	ринжховѣ	двигоховѣ	1
	2	ринѣашета	двигнѣашета	ринжста	двигоста	2
	3	ринѣашете	двигнѣашете	ринжсте	двигосте	3
Plur.	1	ринѣахомъ	двигнњахомъ	ринжхомъ	двигохомъ	1
	2	ринѣашете	двигнѣашете	ринжсте	двигосте	2
	3	ринњахж	двигнњахж	ринжша	двигоша	3
Resultative part. nom. s			sg. masc.	neut.	fem.	
			ринжлъ	ринжло	ринжла	
			двиглъ	двигло	двигла	
Past active part. nom. sg. masc.			neut.	fem.		
ринжвъ				ринжвъши		
двигъ				двигъши		
Past passive part. nom. sg. masc.			neut.	fem.		
			риновенъ	риновено	риновена	
			движенъ	движено	движена	
Verbal substantive			риновенье	Supine	ринжтъ	
			движенье		двигнжтъ	

15.7

15.7 Verbs with the classifier -no.

15.71 This group has two major subdivisions: stems with a vowel preceding the classifier  $-n\rho$ , and stems with a root-final consonant preceding. However, there are some intermediate cases, where the consonant appears in some forms and not in others. Some verbs are poorly attested, so that their classification is not always certain. Although there are only about 60 verbs (plus about 80 prefixed stems) in the class, it is clear that it was productive and used to make new perfective verbs. Some of the older verbs were imperfective.

15.711 The morpheme {nq} usually remains intact before consonantal desinences and is truncated to -n- before vocalic desinences. Before the -en- formant of the past passive participle (and especially its derivative, the verbal substantive), -nq- is replaced by -nov-: rinqti, rinqv5; rinet5, but ot5rinoveni 'pushed away' past pass. part. nom. pl. masc. (Ps 87:6).

**15.712** The common verb **CTANRTb CTATH** 'stand up, take a stand, stop' (with eight prefixed stems, including octan**RTb** octat**H** 'leave, cease', np<sup>4</sup>**c**-**CTANRTb** np<sup>4</sup>**c**-**CTANRTc**-**CTANC**-**CTANRTb** np<sup>4</sup>**c**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTANC**-**CTAN** 

15.72 The following 11 verbs (plus 8 prefixed stems) definitely have a vowel before the -nq of the basic stem and always retain the classifier: -винжти (повинжти са 'be obedient', не обинжти са 'be frank, straightforward'), дочнжти 'blow', динжти 'gape, yawn', на-кънжти 'nod to', поманжти 'beckon', минжти 'pass', поманжти and помѣнжти 'call to mind', въспланжти 'flare up', плинжти and пâюнжти 'spit', ринжти 'push', and и(z)-сочнжти 'draw out'.

15.73 въд-бънжти 'waken', оу-ванжти 'wilt' (see §15.773), гънжти 'bend, fold', канжти 'drip', and въс-пранжти 'get up' are attested only with a vowel before nq, but there is good reason to posit underlying forms with a consonant (and to expect non-present forms without nq), viz. the roots *bъd*, *vęd*, *kap*, and *pręd*. Compare the next paragraph.

15.74 Underlying {u-sъp-(nq+)} 'fall asleep' predicts the infinitive оусънжти and past forms like оусънжуъ, оусънжвъ, where -pn- > -n-(§3.3131), and the forms without nq retain the p: оусъпъ, оусъпъшинуъ.

 $\{top-(nq+)\}\$  'submerge' underlies Su истопижша, истопоша as well as истопж Mt 8:32 Sav ~ оутопж Mar As ~ оутопоша Zo.

<sup>15.75</sup> The aorist присвлде, присвлдж 'was/were scorched' probably had inf. \*prisveneti, while the past passive part. oy-A the end 'take by surprise, ambush' and the aorists  $z_A$ -клепе 'closed' and oy-трыпоста '(the two) became numb' may have had nq forms without the labial

stop (\*закленжти, \*оутрынжти) like оусънжти, or with it (\*заклепнжти, \*оутрыпнжти) like ослыпижти and загывижти, see below.

15.76 Most verbs with a consonant before the classifier nq keep -n- in the present system and -nq- in the infinitive, but allow variants with or without the n/nq in past forms. When the -nq- or -n- disappears, the resultant C-stem behaves like the root-verbs with zero classifier, discussed below in §15.8. For the specific aorist forms, see §10.812. The few verbs attested without *n*-forms beside C-stem forms most probably would be found to have doublets if we had more old manuscripts.

This group comprises 34 verbs (plus over 40 prefixed stems):

-бѣгнжти 'flee'	-крысняти 'be resurrected' (§15.773)
влъснжти 'stammer'	-къкнжти 'sour'
от-врыгнжти 'throw away'	-млькижти 'fall silent'
въкнжти 'learn'	оу-макияти 'become soft'
оу-вадижти 'be ensnared'	-мъкнжти 'push, move'
гасняти 'go out' (§15.773)	-никнжти 'come up'
гонезнжти 'be rid of'	отъ-ригнжти 'eject'
по-градижти 'sink, be submerged'	по-сагнжти 'marry'
гывижти 'perish' (see §15.772)	по-стигнжти 'attain'
двигижти 'move, lift'	съхнжти 'dry up' (see §15.773)
дрьзнжти 'dare'	оу-сѣкнжти 'behead' (see §15.771)
-дъхнжти 'breathe'	и(z)-сакняти 'dry up' (see §15.773)
оу-жасижти 'be terrified'	тлькижти 'knock'
про-забняти 'sprout, grow'	-тъкижти 'hit, bump'
косижти 'touch'	-трыгнжти 'tear'
въс-кликнжти 'shout'	-тагняти 'pull'
кръкнжти 'grunt'	и(z)veznжти (§3.311) 'disappear'

Further, the past active participle поплъдъ са 'having slipped' (Su) suggests \*попльдижти.

15.771 Certain fluctuations within the present system and in the infinitive indicate that the classifier -nq+ was replacing (a) the zero-classifier in infinitive and some past forms on the one hand, and/or (b) the underlying j of present forms that is regularly correlated with the -a+ classifier. The verb oyctknmath 'behead' (oyctknmath [Su], oyctknmaxh [Ev, Ps], cf. oyctknobenne 'beheading' [As]) is attested with the past passive part. oyctknobenne 'beheading' [As]) is attested with the past passive part. oyctknobenne 'beheading' [Su] and the few attested present forms (noctveun, ctvete, noctum inv., ctkru nom sg. masc. pres act part.), correspond to {-sčk-Ø+}, cf. §15.85. Su also has doublet infinitives при-сагижти and присашти, but otherwise OCS has only eight forms from {pri-sęg} and осагъшин.<sup>22</sup> Attested imv. sg. въньзи 'sheath!'J 18:11, 3 sg. aor. воньзе 'pierce' Ps 31:4 presumably represent {-nьz-} and belong with the infinitive вънъзижти (Su 2.23); other prefixes are attested—the root aorist оуньзж Ps 37:3.(= Eu 76a4), and past act. part. възньзъ 'impaling' Mt 27:48, Mk 15:36, J 19:29.<sup>23</sup> The isolated 3 pl. aor. охръмж 'they went lame' Ps 17:46 is usually cited under the phonologically improbable infinitive \*охръмижти (which violates §2.522; cf. §29.813 n. 23).

15.772 The presents formed by underlying j (coordinated paradigmatically with the classifier -a+) alternate in some verbs with *n*-presents. Particularly well attested are forms from {gyb} 'perish': eg. погъщлетъ J 6:12 Mar ~ гъщнетъ Zo As; гъщлижищтее J 6:27 Mar Zo гъщянищтее As.<sup>24</sup>

15.773 Other examples of the *j*-presents—simply irregularities in OCS—are rare:

оу-ванжти 'will' Su 389.14; pres. act. part. не суважда 'unfading' Su 352.23 ~ 3 pl. aor. оуваноша Su 164.19.

гасняти 'go out [fire]': не гашяштин 'not quenched' (e.g. Mk 9:43).

въскрысияти 'be resurrected' has beside many regular forms the isolated 3 sg. inv. въскрыш Ps 73:22.

исъхняти 'be dried up': исъхнетъ J 15:6 As ~ исъшетъ Zo Mar Sav; cf. їсъшятъ Ps 36:2. исакняти 'dry up' has исакнетъ Mt 24:12 (Zo Mar As, Sav 87r) but исаvетъ Sav 47r in a repetition of the verse.

Isolated forms that probably belong here: \*uglbbnqti 'be stuck' 1 sg. pres. фглъвы (for \*uglbbljq); Ps 68:15 ~ aor. 3 pl. фгльвы Ps 9:16, 1 sg. фглевъ Ps 68:3; \*prilb(p)nqti 'cling, cleave to', 3 sg. imv. прильплії Ps 136:6 ~ aor. 3 sg. прильпе Ps 118:25 etc.; unprefixed \*niknqti 'grow, sprout' 3 sg. pres. ничетъ Cloz 12a2.

<sup>&</sup>lt;sup>22</sup> It is unfortunate that lexicographers often feel compelled to invent an infinitive. SJS rightly has both присагняти and присации but cites осагъшии sub theoretical осагняти.

<sup>&</sup>lt;sup>23</sup> The root-shape *niz* is improbable and in any case not in OCS or closely related texts; възнисти, вънисти and оунисти are ghosts from scholarly tradition. It is regrettable that SJS repeats them.

<sup>&</sup>lt;sup>24</sup> There is no evidence for an infinitive тыбати, and потывати is derived from *-gyb-aj*+ and should be in a separate lemma from citations of потыблеть and the like. See Vaillant §205-207.

		Infinitive нести 'carry'		вести 'lead'			
		Presen	t	Impera	erative		
Sing.	1	несж	ведж	-	-	1	
	2	несеши	ведеши	несн	веди	2	
	3	несетъ	ведетъ	неси	веди	3	
Dual	1	несевж	ведевъ	несжвж	ведъвъ	1	
	2	несета	ведета	несѣта	ведѣта	2	
	3	NECETE	ведете	-	-	3	
Plur.	1	несемъ	ведемъ	несфмъ	ведѣмъ	1	
	2	несете	ведете	несъте	ведѣте	2	
	3	не¢ѫтъ	веджтъ	_	-	3	
Pres.	act.	part. nom. sg	. masc.	neut.	fem.		
			несъі		несжшти		
			ведъі		веджшти		
Pres.	pass	. part. nom. s	sg. masc.	neut.	fem.		
	•	1	несомъ	NECOMO	NECOMA		
			ведомъ	ведомо	ведома		
		Imper	rfect	Aorist			
Sing.	1	нестахъ	ведћахъ	несохъ	ведохъ	1	
•	2-3	несташе	веджаше	NECE	веде	2-3	
Dual	1	несњаховћ	ведњаховћ	несоховѣ	ведоховѣ	1	
	2	несвашета		несоста	ведоста	2	
	3	несѣашете		NECOCTE	ведосте	3	
Plur.	1	несњахомъ		несохомъ	ведохомъ	1	
	2	несвашете		NECOCTE	ведосте	2	
	3	несвахж	ведњахж	несоша	ведоша	3	
Resul	tativ	ve part. nom.		neut.	fem.		
		•	неслъ	несло	несла		
			велъ	вело	вела		
Past a	activ	e part. nom.	sg.masc.	neut.	fem.		
		-	несъ		несъши		
ведъ					ведъши		
Past 1	oassi	ve part. nom		neut.	fem.		
1		I	несенъ	NECENO	несена		
			веденъ	ведено	ведена		
Verba	al su	bstantive	несенре	Supine	нестъ		
			веденье	· ·· <b>I</b> · ·· · ·	вестъ		

15.8 Verbs with zero classifier, stems ending in a consonant other than j

15.8

**15.81** This is an unproductive class, but its 56 verbs (plus over 175 prefixed stems) express for the most part everyday activities and are therefore relatively frequent in any sample of text. Eight more verbs (plus nearly 35 prefixed stems), which are irregular or partly so, will be treated here.

It is this class which accounts for the majority of the non-productive aorist forms, including the 3 sing. aor. in  $-t_3$  and the past passive participles in -t. These forms are given in detail in the subdivisions of §10.8 and in §11.321-2; they will not be repeated here.

**15.82** The stem-final consonants may be affected by consonantal desinences:

15.821 Stem-final s remains unchanged: несятъ 'carry', пасятъ 'pasture', съпасятъ 'save', трасятъ 'shake, tremble'.

15.822 z > s before t; ведять ~ inf. вести, supine весть. Here belong ведять 'convey', оу-вадять 'crown' (past pass. part in t, §11.32), -лѣдять 'go', -ньдять 'pierce' (nq-variant §15.771), and \*gryz- 'gnaw' (represented only by гридеть, for \*gryzeta, L 12:33 Sav).

15.823 t d > s before t; t d are deleted before l: веджтъ ~ вести, велъ. Here belong: блюджтъ 'watch', бладжтъ 'talk nonsense', боджтъ 'stab', веджтъ 'lead', бладжтъ 'rule', гнетжтъ 'press', кладжтъ 'put', краджтъ 'steal', -метжтъ 'sweep' (for метжтъ 'throw', see §15.641), матжтъ 'stir, confuse', паджтъ 'fall', плетжтъ 'braid, weave', праджтъ 'spin', растжтъ (inf. расти) 'grow', and the defective граджтъ 'come, go', which has no past forms (inf. грасти, pres. act. part. градъ, граджшти).

15.824 *p b* are deleted before *t*: тепжтъ 'beat' ~ inf. тети; гребжтъ 'bury' ~ грети; and the isolated pres. pass. part.  $z_{ABOMH}$  'being torn' Su 397.27.

15.83 ь + nasal is replaced by e before a consonantal desinence, including the aorist: клынжтъ 'curse' ~ infinitive клати, l-part. клалъ, past passive part. клатъ (§11.32), aor. клахъ (and класъ §10.82), 2–3 sg. клатъ or кла (§10.52). Here also: съ-жылжтъ '(\*squeeze) oppress', жыйжтъ 'reap', -пылжтъ 'stretch; crucify', -тылжтъ 'kill', and -уылжтъ 'begin'.

The common verb {ьm-Ø+} 'take' preposes j if not preceded by a consonant (§3.24);  $j_b$  is written i. Thus имжтъ, prefixed forms изъмжтъ 'take out', съньмжтъ, възьмжтъ 'take' ~ приимжтъ 'receive', подъимжтъ 'lift', and others. The truncated stem is  $\varrho$ : inf. ати (приати, възати), l-part. алъ (приалъ, възалъ), aorist ахъ/асъ (приахъ, възахъ), past passive participle атъ (приатъ, възатъ).

CONJUGATION

Note that the past active participle is MMT (for \*jbm3; cf. BTZEMT). In mss reflecting the replacement of strong b by e (§2.622), this participle may be spelled EMT. Similarly with prefixes: приммъ/приЕMT.

15.841 The four (poorly attested) verbs in -ov replace this sequence by u: плов-жтъ 'travel (by boat)' ~ inf. плоути, l-part. плоулъ, aor. плоухъ. Also словжтъ 'be reputed', на-тровжтъ 'feed', and ровжтъ 'bellow, roar'.

The pres. act. part. ревъи (Hil) and post-OCS evidence imply an alternative verb \*ревжтъ \*рюти.

15.842 Two other verbs end in -v. Живжтъ 'live' (with 6 prefixed stems, including иждивжтъ {iz-živ} 'spend') loses the v before a consonantal desinence (inf. жити, l-part. жилъ, aor. жидъ), except that the 3 sg aor is живе (beside expected житъ and жи). The rare плѣвжтъ 'weed' is attested only in present forms.

15.85 Stems in -k or -g undergo (1) automatic KI-mutation before e and  $\check{e}$ , (2) grammatically specific KAI-mutation before the  $i^2$  and  $e^2$  of the imperative-marker, and (3) a special process (in infinitive and supine) whereby stem-final velar combines with desinential t in  $\check{s}t$ :<sup>25</sup> Thus ctkrжтъ 'cut', мог-жтъ 'be able' ~ 3 sg. ctveтъ, можетъ ~ imv. ctuu ctuttre, мозы мозыте ~ inf. ctuu мощи. In the imperfect,  $\check{e}a$  effects KI-mutation and shifts to aa: ctvaaue, можааше. In the imperative, a root-vowel e is replaced by 6: рекжтъ 'say', жегжтъ 'burn' ~ рьци, жъзи. Some stems have further idiosyncracies.

Here belong: влѣкѫтъ 'drag' (§15.873), жегѫтъ 'burn' (§15.875), дваг- 'sound', -лакѫтъ 'bend', могѫтъ 'be able', небрѣгѫтъ 'not care, disdain', пекѫтъ са 'worry', -прагѫтъ 'join', рекѫтъ 'say', стрѣгѫтъ 'guard', сѣкѫтъ 'cut' (nq-variant §15.771), -сагнѫтъ 'touch' (nq-variant §15.771), and текѫтъ 'run'.

15.86 Verbs in -br have two possible truncated shapes (exceptions in §16.511-12): Before the consonant of the l-participle and the *t* past passive participle br becomes rb: in the aorist, infinitive and supine br becomes  $r\ddot{e}$ : npo-ctpb-Tt 'spread' ~ npoctpbAt, npoctpbt ~ npoctptt, npoctptt.

*Note:* This may be analyzed as underlying  $\{br\} > rb$  before C in the participles, while a second stem  $\{er\}$  is used before the other consonantal suffixes and yields surface re (see note in \$15.643).

<sup>&</sup>lt;sup>25</sup> This surface št is identical with the ψ resulting from iotation of underlying t or st or sk: (plati-ti 'pay', pusti-ti 'let', iska-ti 'seek' ~ 1 sg pres. nsaψm, nsqum, nwym, nwym). The traditional formula is \*tj.

These verbs are -вържтъ 'slide', -жържтъ, 'swallow', мържтъ 'die', по-нържтъ 'submerge', о-пържтъ са 'lean, support self', ра(z)-сквържтъ 'besmirch', and -стържтъ 'spread'.

Spelling note: scribes seem to have been uncertain as how to write the (probably silent) jers in these verbs. They probably pronounced both  $\alpha_{MP}$  and  $\alpha_{MP}$  with a cluster -mr-; they could write or omit the b in either shape, e.g.  $\alpha_{MP}$  are,  $\alpha_{MP}$  by Forms with -rb- were usually spelled -ph-, as were other instances of -rb- between consonants (§2.53).

**15.861** Exceptionally, the past participial suffix  $-v\sigma(\tilde{s})$  is used in **procupies**, Su 311.6, for usual *prostors*.

	'say'	'throw'	'curse'	'spread'
	§15.85	§15.874	<b>§15.83</b>	§15.86
3 pl. pres.	рекжтъ	врьгжтъ	кльнжтъ	простьржтъ
3 sg. pres.	речетъ	врьжетъ	кльн€тъ	простьретъ
2 sg. imv.	ръци	врьзи	кльни	простьри
pres. act. part.	рекъ	врыгъі	кльнъі	простьръі
pres. pass. part.	рекомъ	врьгомъ	кльномъ	простьромъ
3 sg. impf.	речааше	врьжааше	кльнѣаше	простьрѣаше
1 sg. aor.	рекохъ	врьгохъ	клахъ	прострѣхъ
(1 sg. aor.)	(øቴχъ)	(връгъ)	(класъ)	-
2-3 sg. aor.	peve	врьже	кла(тъ)	прострѣ(тъ)
l-part.	реклъ	врьглъ	<b>Кለ</b> ሕለጌ	прострылъ
past act. part.	рекъ	врьгъ	кльнъ	простьръ
past passive part.	реченъ	врьженъ	<b>Кለ</b> ሕፐ <b>ጌ</b>	прострьтъ
substantive	реченье	врьженье	клатье	(оумрьтье)
infinitive	решти	врѣшти	клати	прострѣти
supine	решть	врѣшть	<u> </u>	прострѣтъ

Synopsis of forms:

**15.87** There are unpredictable vowel (or vowel + sonorant) alternations in some roots. In some verbs the alternation is consistent; in others irregular forms occur, usually beside the normal shapes.

15.871 чьтятъ 'count' forms inf., supine, and non-productive aorists from the stem *čit*-: уисти, уистъ, 3 pl. aor. уиса and уиша (2–3 sg. уьте, 3 pl. regular aorist уьтоша; l-part. уълъ), see §10.82.

цвътжтъ 'bloom' has infinitive процвисти Su 300.6, s-aor. процвисм Cloz 13b4 (~ процвътоша Su 450.12).

15.872 -връджтъ 'tie' has inf. -връсти and s-aorist -връсъ (§10.82), with alternating underlying {vъrz/verz}.

почрыпятъ 'draw (water)' {čыр} has the infinitive and supine from -črěp- {čerp}: почрѣти Su 551.4, почрѣтъ J 4:7.

15.873 влѣкжтъ 'drag' {velk} has l-participle regularly from -vlbk-{vblk} and other past participles from either stem-shape: e.g. обльклъ 'dressed' (Su); изблькъ 'having pulled out' Mk 14:47, облькъ 'having dressed' L 23:11, обльчена 'dressed' Mt 11:8 (~ e.g. съблѣкъше Su 103.2; облѣчен-, въблѣчен- Su).

небр'ягжтъ 'neglect, disdain' {berg} has небрьгъше Su 98.8, небръг'ша Su 40.14 {bbrg} ~ небр'ягъш- Su 212.16, 354.24.

15.874 тлыкжтъ 'knock' {tblk} has inf. from {telk}, тлѣшти L 13:25. врыгжтъ 'throw' {vbrg} ~ inf. {verg} -врѣшти, supine выврѣшть Mar L 12:49.

стригжтъ 'shear, tonsure' has inf. постр'вшти (Euch), with an isolated alternation  $i \sim \check{e}$ .

15.875 жегжтъ 'burn' has b not only in the imperative (жьяи, жьяѣте §7.111), but occasionally in other forms: aor. зажьже Mt 22:7, past pass. part. сьжьжена (вжди 'let [it] be burned' Su 19:7), съжъжетъ Sav L 3:17, pres. pass. part. жъгомъимъ Su 476.17. (Details, Koch 347f.)

**15.876 жладе** 3 sg. aor. 'repay, compensate' (Su 494.9), inf. жласти Su 494.4, 5, contrasts with **жлѣдетъ** Su 360.13. The verb may have been obsolete even for the scribe of Su. The shape  $\xi l \ell d$  is probably historically older. (For discussion, see Koch 583–5.)

# 15.9 Verbs with basic stems in -aj+, -ěj+, or -j-Ø+

Infinitive делати {děl-aj+} 'd			do' вити {bьj-Ø+} 'strike'					
Present				Imperative				
Sing.	1	дъланя	быж	(биіж)	-	-	-	1
	2	дѣлаеши	<b>БРЕШИ</b>	(виєши)	дѣлан	бии	(бъи)	2
	3	дѣлаетъ	бь€тъ	(биетъ)	дѣлан	бии	(бри)	3
Dual	1	дѣлаевѣ	<b>бьев</b> ѣ	(виевф)	дъланвъ	биив 🕏	(ернеф)	1
	2	дѣлаета	бьета	(биета)	дѣлаита	бинта	(бьита)	2
	3	дѣлаете	бьете	(биете)	_	-	-	3
Plur.	1	дѣлаемъ	бьемъ	(биємъ)	дѣлаимъ	биимъ	(бримљ)	1
	2	дѣлаете	бьете	(биете)	дѣлаите	бинте	(бьите)	2
	3	двлажтъ	быжтъ	(биіжтъ)	-		-	3
Pres.	act.	part. nom.	sg.	masc.	neut.	fem.		
			дълаж	далажшти				
			66 <b>x</b>	быятши				
Pres. pass. part. nom. sg.		masc.	neut.	fem.				
				дѣемъ	дњемо	дњема		
				бь€мъ	вьемо	бь€ма		

		Imperfect	Aori	Aorist		
Sing. 1	дѣлаахъ	былахъ	(бигаахъ)	дѣлахъ	бихъ	1
2-3	дѣлааше	былаше	(бигааше)	дѣла	би	2-3
Dual 1	дѣлааховѣ	былаховъ	(бигааховѣ)	дѣлаховѣ	биховѣ	1
2	дѣлаашета	былашета	(бигаашета)	дѣласта	биста	2
3	дѣлаашете	былашете	(бигаашете)	дѣласте	бисте	3
Plur. 1	дѣлаахомъ	былахомъ	(бигаахомъ)	дѣлахомч	ь бихомъ	1
2	дѣлаашете	вылашете	(бигаашете)	дѣласте	бисте	2
3	дѣлаахж	былахж	(бигаахж)	дѣлаша	биша	3
Resultati	ve part. nom.	sg. masc.	neut.	fem.		
		дѣлалъ	дѣлало	дѣлал	4	
		билъ	било	била		
Past activ	ve part. nom. s	sg. masc.	neut.	fem.		
дълавъ				дѣлав	ъши	
		бив	L	<b>бив</b> ъц	н	
Past pass	ive part. nom.	sg. masc.	neut.	fem.		
		дѣланъ	дѣлано	дѣлан	IA	
бьенъ (биенъ) бьено (биено) бьена (					(биена)	
Verbal su	ubstantive	дѣланье	Supine	дѣлач	ъ	
		еренре (вне	ире)	битъ		

15.91 This class contains two productive subgroups with the classifiers -aj+ (about 360, plus some 320 prefixed stems) and  $-\check{e}j+$  (about 35, with some 20 prefixed derivatives). Their conjugation is entirely regular. For verbs like  $um-\check{e}j+$  'to know', the forms have  $\check{e}$  everywhere -aj+ forms have a: оумѣты, оумѣты,

There is also an unproductive subgroup of eighteen root-verbs with zero classifier (plus 35 compounds); some of their forms call for special comment.

15.92 The verb  $znaj-\emptyset$ + 'know', is entirely regular: знажтъ знати.

Four poorly attested verbs are to be set up with  $\check{ej}$ - $\emptyset$ +,  $gr\check{ej}$ -qt's 'warm',  $r\check{ej}$ -qt's 'push',  $sp\check{ej}$ -qt's 'prosper', and  $sam\check{ej}$ -qt's 'dare', and one with -aj- $\emptyset$ +, v $\exists aj$ - $\emptyset$ + se 'be tossed [by waves]', cf. §15.45.

15.93 Eight verbs have basic stems in  $-bj-\emptyset+$ . The b is tense (§2.61) and usually is spelled *i*. Before consonant, bj > i: e.g. bbj-qtъ быжтъ or bij-qtъ быжтъ or bij-qtъ быжтъ  $\sim bbj$ -ti, bbj-x-ъ > бити, бихъ. The imperative plural is *biite* and *biěte* бывате; see §7.21.

#### CONJUGATION

выжтъ/вижтъ 'beat, strike', въпыжтъ/въпижтъ 'call, cry out' (§3.25), -выжтъ/-вижтъ 'wind', съ-гныжтъ/-гнижтъ 'rot', лыжтъ/лижтъ 'pour', пыжтъ/пижтъ 'drink', почыжтъ/почижтъ 'rest', and шыжтъ/шижтъ 'sew'.

пыжтъ and -выжтъ have 3 sg. aor. in -to (§10.52), and they and лыжтъ have past pass. part. in -t- (§11.32); шыжтъ has the past pass. part. шьбенъ.

**15.94** The five verbs in  $-\delta j/-yj-\emptyset + 2^6$  and the two in  $-uj-\emptyset +$  have a special replacement of root-final j by v before the past passive participial suffix *-en*-(§11.341). The imperative plural is kryite and kryěte: see §7.21.

крънжтъ 'cover, hide' (ppp -кръвенъ), мънжтъ 'wash' (ppp -мъвенъ), оч-нънжтъ 'be downcast' (subst. очнънье 'weariness'), рънжтъ 'dig', очтънжтъ 'grow fat', ов-очжтъ 'put on footwear' (ppp обочвенъ 'shod'), чочжтъ 'sense, feel'.

## 16. Irregular verbs

Here are included verbs whose conjugations do not fit easily into any of the categories already described. Although the number of verbs is small (20, plus about 80 prefixed stems), most of them are extremely common in any sort of text.

16.1 By far the most frequent and the most irregular verb is 'to be', whose forms are built on three imperfective stems,  $b\check{e}$ -, *jes*-, and *s*-, and two stems which function in both aspects: *bqd*- and *by*-. The perfective forms usually mean 'come into being, come to be, become'.

		Infinitive вълги		Su	Substantive бытье 'being, genesis'			
		Imperfective			Pfctv/Ir	Perfective		
		pres.	imperf.	aorist	future	imperat.	aorist	
Sg.	1	есмь	-	бѣхъ	бѫдѫ	-	быхъ	1
	2	есн	_	6 <b>t</b>	бѫдеши	бѫди	бъктъ (бъі)	2
	3	естъ	бћаше	6 <b>t</b>	Бѫдетъ	бѫди	быстъ (бы)	3
Du.	1	есвъ	_	*вѣховѣ	бѫдевѣ	бѫдѣвѣ	быховѣ	1
	2	еста	-	*бѣста	бждета	бждѣта	выста	2
	3	есте	бълшете	бъсте	бждете	-	бысте	3
<b>P1</b> .	1	€смъ		бѣхомъ	вѫдемъ	бѫдѣмъ	быхомъ	1
	2	есте		*бѣсте	вждете	бѫдѣте	высте	2
	3	(ሕፐጌ	бѣахѫ	бұтч	бѫдѫтъ	БЖДЖ	бънту	3

<sup>&</sup>lt;sup>26</sup> The alternant -3j- is theoretical; these verbs are always spelled with y in OCS (cf. §2.61) except in the truncated stem of past passive participle.

#### CONJUGATION

Pres. act. part. nom. sg.: masc. neut. съ; fem. сжшти Resultative part. nom. sg.: masc. бъллъ, neut. бълло, fem. бълла Past active part. nom. sg.: masc. neut. бълбъ; fem.. бълбъщи

16.101 The negative present has special forms: итсмь, итси, итстъ; итсвъ, итста, итсте; итсмъ, итсте, but normal negation in не сятъ.

**16.1011** Third person forms without desinence are rare:  $\epsilon$ , N<sup>th</sup>, c<sup>Th</sup> (§6.61).

**16.102** The imperfect forms may contract  $\check{e}a$  to  $\check{e}$ ; e.g. **stue**. There is some degree of confusion between the closely related imperfect and imperfective aorist. For meaning and use, see §21.21.

16.103 The perfective aorist быхъ, 2-3 sg. бы (not быстъ!) sometimes replaces the conditional бимь, би, etc., cf. §14.3.

16.104 The participle вжды, вжджшти, etc., means 'future'.

16.11 The stems by- and bqd- are used with prefixes to form perfective stems: by- appears before C-desinences, and bqd- usually before vowel. The 2-3 sg. aorist desinence -sto may be omitted (e.g. npksuctu/npksu 'stayed, remained', cubsuctu ca/ cubsu ca 'took place, happened'), though zasutu 'forget' has only zasu. The only attested past passive participle is zasubenu. The substantive zasubenue 'oblivion' is exceptional beside zasutue; cf. uzsutue 'riddance' (uzsutu 'be superfluous; be rid of').

**16.2** Four other verbs have the same type of present-tense desinences (§5.9, §6.71), with further complications.

16.21 дати дадатъ 'give' is irregular in the present tense, imperative, and 2-3 sg. aorist. Present: дамь, даси, дастъ; \*давѣ, даста, дасте; дамъ, дастє, дадатъ. Imv. sg. даждь (Euch дажди), 1 pl дадимъ, 2 pl. дадитє. Aor. 3 sg. дастъ (beside rarer да). Present act. part. дадъ, даджшти; imperfect дадѣахъ. Other forms are from da-: aor. дахъ; past participles далъ, давъ, данъ; substantive даньє.

16.22 **всти вдатъ** 'eat' is irregular in the present tense, imperative, 2-3 aorist, and it has both an s-aorist and a regular aorist built on the stem *ě*-, see §10.84. Pres.: вмь, вси, встъ; \*ввѣ, вста, всте; вмъ, всте, вдатъ. Imv. sg. вждь, 1 pl. вдимъ, 2 pl. вдите. Aor. всъ/вхъ, встъ, etc. The -sto of 2-3 aorist is omitted only in изъ 'ate up' (Su 138.27, 300.25). Other forms are regular from *ěd*: imperfect вдвахъ, present act. part. вдаъ, вдатшти; past participles влъ (§3.3131b), вдъ вдъши, вдеиъ; subst. вденье.

138

16.23-16.511

CONJUGATION

16.23 вѣдѣти вѣдѧтъ 'know' is irregular in the present system. Pres.: вѣмь ог вѣдѣ, вѣси, вѣстъ; вѣвѣ, вѣста, вѣсте; вѣмъ, вѣсте, вѣдѧтъ. Imv. sg. вѣждь (Euch вѣжди), 1 pl. вѣдимъ, 2 pl. вѣдите. The -sto of 2-3 aor. is omitted only twice (не вѣ Su 382.7, проповѣ L 12:3 As). Pres. act. part. вѣдъи (stem věd-),<sup>27</sup> pass. вѣдомъ. Aorist and past participles regular from vědě-: вѣдѣхъ, вѣдѣ etc.; вѣдѣмъ, вѣдѣвъ, -вѣдѣмъ.

16.24 The verb 'to have' is made from the root \* $\omega m$  with the suffixes - $\check{e}j$ , -a and zero. The present is: имамь, имаши, имать; имавѣ, имата, иматє; имамъ, иматє, иматъ or имѣмтъ. Pres. act. part имъ имъшти or имѣм имѣвшти. The *imėj*-alternatives are chiefly from Supr. Other forms are regular from \* $\omega m\check{e}j$ -: imv. имѣи, имѣимъ, имѣитє; inf. имѣти; imperfect имѣахъ; resultative part. имѣлъ, etc.

Note that  $\mu m \pi \tau b$  is also 'they will take' {bm-@+q-tb}. Forms like  $\mu m a \tau h$ ,  $\mu m a n b$  are from the imperfective a-derivative, see §15.643.

16.3 иджтъ 'go' has three root-shapes: *i*- (possibly *ji*-), *id*- (?*jid*-), and šьd-. Infinitive ити {i+ti}. Imperfect ид тахъ; aor. идохъ/идъ (§10.811). Past active participles: шьдъ (f. шьдъши); шьлъ ({sьd-l-ъ}, §3.3131), f. шьла. Subst. шьстье and шьствье, but prefixed сънитье 'descent'.

The present passive part. underlies the adjective neprěidoms 'impassable'.

16.4  $\ddagger_{A}$ жтъ 'go, ride' forms a root aorist,  $- \ddagger_{A}$ ж (3 pl., cf. §10.811). Alternative stems are  $\check{e}$ - and  $\check{e}$ xaj- and their distribution apparently differed by dialect. Attested are past act. participles  $- \ddagger_{B}$  and  $- \ddagger_{XAB}$  and an imperative  $\ddagger_{XAH}$  beside  $\ddagger_{A}$ . The infinitive was presumably  $= \ddagger_{X}$  and or  $= \ddagger_{T}$ .

Mt 14:34 Mar Zo прѣѣвъше~ As прѣѣхавъше 'having crossed [the lake in a boat]';<sup>28</sup> Mk 6:53 Zo приѣхавъше ~ Mar прѣѣвъше. ѣхан Zo L 5:4 (~ Mar въӡѣдн, As въѣдн).

**16.5** Verbs with *vowel* + *sonorant* in present forms but *sonorant* + *vowel* in others include several subgroups. Unless otherwise noted, the 3 p. present furnishes the stem for the present system while the infinitive shows the stem for all other forms. Imperfects will be noted when they are attested.

16.511 млѣти меліжтъ 'grind, mill'. Impf. мелиаше Su 565.10.29

<sup>&</sup>lt;sup>27</sup> The **B**that cu of Su 305.3 is probably an error for **B**that as in line 5.

<sup>&</sup>lt;sup>28</sup> Note that Sav Mt 14:34 has прил.8въше, cf. §15.841.

<sup>&</sup>lt;sup>29</sup> Positing underlying {mel-ti mel-j-qtb} and the metathesis rule suggested in §15.643, {mel} predictably will yield mlě (before C other than j). An extension of

# 16.512 брати боржтъ 'fight'. Impf. борвахж см Ps 119.7.

Spellings with въра- or бъра- are doubtless purely graphic, reflecting scribes' uncertainty about where to use a jer-letter. Attested substantives are бранье (A sg. бъраник Su 86.13, G sg. бъраник 86.14 'contest, agony') and боренье (G pl. борении Su 486.27 'wars').

16.513 клати коликтъ 'slaughter'. Imv. заколъте L 15:23 Zo Sav (§7.21) ~ regular заколите Mar As. Past pass. part заколенъ (-len-, opposed to -lje- of pres.) and закланъ.

16.521 жрьти жържтъ 'sacrifice' shows traces of a present stem ž*ьгјоt* 5: 1 sg. pres. пожърж Ps 53.8, 115:8, pres. participles жърд Su 115.29, жъремо Su 91.26. The attested substantive is жрътъе (G sg жрътила Su 148.30). This verb is easily confused with -*žьrotъ* -*žrěti* 'swallow' (§15.86).<sup>30</sup>

16.522 тръти тържтъ 'rub' implies *-rj-* in 1 pl. imv. сътъримъ Euch 98a9 (~ сътърѣмъ Su 353.5) 2–3 sg. aor. *-trъ* (сътрь Ps 104:16, отръ J 12:3 Zo Mar As) competes with *-tъre* (отъре J 12:3 Sav; сътъре Su 11:29, 311:16).

16.53 пѣти понятъ 'sing', imperf. попадъ; 3 sg. aor. пѣтъ, past pass. part. пѣтъ; substantive пѣтье and пѣнье.

16.61 сѣсти {sěd-ti} съджтъ 'sit down'. For root-aorist see §10.811. The imv. сѣдѣте (for съдѣте) Mar Mt 26:36, L 24:49, if not a mechanical spelling error, may show either a Macedonian dialectal denasalization or an innovative suppression of the traditional but anomalous alternation.

16.62 лешти {leg-ti} лыгжтъ {leg-qtъ} 'lie down'. Root-aorist, §10.811.

16.7 -рѣсти {rět-ti} -раштятъ {-ręt-j-qtъ} occurs only with prefixes: сърѣсти 'meet', обрѣсти and приобрѣсти 'find'. For root-aorist see §10.811. Imperative сърдштате Su. Imperfect обраштаахъ.

Forms like обрѣтладж Mk 14:35 are from the regular imperfective verb -*rět-aj*. Vat обрѣщетъ Mt 7:8 ~ обръщетъ Sav (~ обрѣтлетъ Zo Mar As).

16.8 -дѣти -дєжджтъ {ded-j-} 'put' occurs only with prefixes, especially одѣти 'dress, clothe' and въздѣти 'lift, raise'; cf. §15.441.

16.91 The s-aorist paceyca ca 'were scattered' Ps 149:7 (2/N) can be analyzed on the basis of post-OCS data as {raz-sup- $\emptyset$ +}, with present -sop-e-.

this rule converts (or ol) to ra la, and underlying {bor-ti bor-j-qtb} and {kol-ti kol-j-qtb} will yield the attested forms.

<sup>&</sup>lt;sup>30</sup> We may posit underlying {žы-ti žы-q-tъ} versus {žer-ti žы-q-tъ} and {tы-ti tы-(j-)q-tъ}.

16.92 аьмъще са 'swelled up' Su 239.27, and надъменъ 'puffed up' Su 117.18 may belong in one paradigm with надоувъши са Su 422.35 'puffed up'.

16.93 The verbs ACQUERTE 'blow' and NEPERTE 'fly' (Su 390.10) are attested only with present forms; their infinitive stems are unknown.

# CHAPTER FIVE

# NOTES ON SYNTAX

#### 17.0 On adjectives

The use of the long and short forms of the adjectives does not have any exact equivalent in English. The juxtaposition of a short-form adjective and a substantive denotes that the combination is presented as a new one:  $v_5 \ pestb \ ognjbnq$  (Mt 13:50) "into a furnace, a fiery one', i.e. 'a fiery furnace'. The long-form adjective presents the quality as one already known and specifically known to belong to the particular substantive which it modifies:  $v_5 \ geonq \ ognjbnqjq$  (Mk 9:47) 'into hell the fiery', i.e. 'the fiery hell'.

Adjectives used as substantives contrast the indefinite short-form with the long-form, which specifies that the substantive has previously been mentioned or is generally known: e.g. Mk 8:22-3 privěsę kъ njemu slěpa ... i imъ slěpaego za rǫkǫ 'they brought a blind man to him, and taking the blind man by the hand'; compare J 9:32 otъ věka něstъ slyšano jako kъto otvrьze oči slěpu roždenu 'since the world began it was not heard of that any man opened the eyes of one born blind', and J 11:37 ne možaaše li sъ otvrьzyi oči slěpuemu sъtvoriti da i sъ ne umьretъ 'could not this man, which opened the eyes of the blind man, have caused that even this man should not have died?'

17.1 An adjective in vocative function normally has the long form (бъсычым пьсе, кръвопивъм дмию Su 115.27 'o mad dog, blood-thirsty serpent!') unless it follows a substantive in the vocative or is itself used as a substantive: fariseju *slěpe* (Mt 23:25) 'o blind Pharisee!'; *bezumene* (L 12:20) 'senseless one!'

17.2 In a series of coordinated substantivized participles (rarely other adjectives also) the long form usually occurs for the first only: (L 6:47, 49) *slyšęi* slovesa moja i *tvorę* ja ... *slyšavyi* i ne *tvorjь* 'he who hears my words and does them ... he who has been hearing and has not been doing'.

#### 18.0 On the use of the cases (not with prepositions)<sup>1</sup>

18.1 The nominative is a "zero-case", specifying only that the speaker's attention is directed to the thing or person represented by it. It is used in naming: both absolute (e.g. in headings; evangelie ots luky 'the Gospel of Luke'; or in exclamations: o velezsloba neprijaznina 'oh the great wickedness of the devil!') and with verbs of naming, sb velii narečets sę (Mt 5:19) 'he shall be called great'; as the subject of the sentence; and sometimes in the predicate after verbs signifying being or becoming, e.g. běašete bo rybarja (Mt 4:18) 'for they [two] were fishermen'; григории поставькить въистъ патриархъ (Su 119.17) 'Gregory was appointed patriarch; vьsěkъ iže sę tvoritъ cěsarjь protivitъ sę kesarevi (J 19:12) 'whoso-ever maketh himself a king speaketh against Caesar'. See also §18.6d, below.

**18.11** The *vocative* specifies direct address. For example: rabe lokavyi (Mt 18:32) 'O thou wicked servant'; iosife synu davydovъ (Mt 1:20) 'Joseph, son of David!'; reko emu otbče sъgrěšixъ (L 15:18) 'I will say to him, father, I have sinned'; ne ostavi mene bože sъpasitelju moi (Ps 26:9) 'do not leave me, o God, my saviour!'; izidi duše nečistyi otъ člověka (Mk 5:8) 'come out of the man, thou unclean spirit!'; idi za mъnojo, sotono 'Get thee behind me, Satan!'; ženo, se synъ tvoi! (J 19:26) 'woman, behold thy son!'; ne boi se Marie (L 1:30) 'Fear not, Mary!'. See also §17.1.

The vocative form is explicit only in singular masculine and feminine nouns; otherwise the nominative form is used. OCS nominative forms in places where a vocative is possible are by no means rare.

**18.2** The accusative functions as the direct object of transitive verbs (unless negated, see §18.3b). Some verbs may take a double accusative ("make, believe, perceive someone [as] something"); e.g. s btvorjo va lovbca člověkomb (Mt 4:19) 'I will make you [two] fishers of men'; simona egože imenova petra (L 6:14) 'Simon, whom he named Peter'; obrěte otrokovico ležešto na odrě i běst ištotašto (Mk 7:30) '(she) found the girl lying on the bed and the devil gone out'; mbněvbša že i vb družině sošto (L 2:44) 'supposing him to be in the company'.

Verbal substantives formed from transitive verbs sometimes may govern an accusative: по прилтии ми отъ ... бога великън даръ (Su 525.15) 'after my receiving the great gift from God'.

Моге detail in Исследования по синтаксису старославянского языка, (ed. J. Kurz), Prague, 1963.

The accusative may express extent of time or space: e.g. bě iona vъ črěvě kitově *tri dыni* i *tri nošti* (Mt 12:40) 'Jonah was in the whale's belly for three days and three nights'; ašte къto poimetъ tę po sile... *popыrište edino*, idi sъ njimь *dъvě* (Mt 5:41) 'if anyone compels thee to go a mile, go with him twain'. In a few expressions it denotes a point in time: ubsjǫtъ i, i *tretы dыпь* vъstanetъ (Mt 7:23) 'they shall kill him, and the third day he shall rise again'; пришьдъшоу кмоу вечер' (Su 275.29) 'when he came [cf. §18.5e] in the evening'.

The accusative is sometimes used with oaths: тако ми великж богъйж артемж (Su 231.1) '[I swear it] by the great goddess Artemis'.

**18.21** The use of the genitive form for an expected accusative with masculine substantives referring to male persons was mentioned in §4.13. The texts are not uniform in their usage, and it is clear that during the OCS period the tendency to develop a new "personal" accusative was spreading even to animals and thus to form an "animate" category. Isolated examples have the regular accusative even for male humans (e.g. prizovi  $m\varrho \xi_b$  tvoi J 4:16 Mar [~ moža tvoego Zo, As] 'call your husband'), and others where normally inanimate nouns are presented as personified (e.g. cero  $\chi \Lambda k Ba$  Maputa  $\rho A \mu$  Su 396.3 'Mary bore this Bread [= Christ]).' Personal names tend to keep the accusative form.<sup>2</sup>

Pronouns, adjectives, and participles referring to male persons regularly (but with numerous exceptions) use the genitive form in accusative function. The personal pronouns me, te, se and i (\*jb), however, are normal accusatives, with *mene*, *tebe*, *sebe*, and *ego* used for emphasis.

In most cases, however, no clear distinction can be drawn between the older emphatic use and the newer animate reference (cf. §4.64). Examples: člověče, kъto mę postavi sodijo li dělitelja nadъ vami (L 12:14) 'man, who made me a judge or a divider over you?'; mene edinogo ostaviste (J 16:32) 'you left me alone'; ne prostbrěste [§9.3] rokъ na mę (L 22:53) 'ye stretched forth no hands against me'; tъgda prizъvavъ i [Mar ~ Sav ego] gospodb ego, glagola...azъ tę [Mar ~ Sav tebe] pomilovaxъ (Mt 18:32) 'then having called him, his master said ... I had pity on thee.'

In Sav and Su, anomalous feminines (§4.413) also use gen. forms for accusative (*matere* 'mother', *dsštere* 'daughter', *neplodsve* 'barren woman'); exceptionally, non-animates appear in this form (*ljubsve* 'love', *crsksve* 'church, temple').

The genitive form for accusative in the masculine plural is exceptional, but examples are found, pronouns and adjectives as well as substantives, chiefly in Sav and Su. E.g., Mt 8:16 Sav H BCtaya Bonnyura Hutanh (~ Zo Mar As vse nedotonye) 'and he healed all that were sick'; **Btat** to CHIP HARMITA (Su 235.17) 'I know that you have *sons*'.

<sup>&</sup>lt;sup>2</sup> Textual disagreement hampers attempts to define usage; e.g., J 16:33, azъ pobědixъ mira M (~ ZoAs vbsego mira) but миръ Sav. The verb otherwise takes only accusative.

18.3 The genitive case has a wide range of functions.

a. It is used as the complement of a number of verbs.

1. Verbs of perception: *bljudqta* 'observe', *zurěti* 'see', *slušajqta* and *poslušajqta* 'hear', and *samotriti* 'look at, see'. *Viděti* 'see' may take either acc. or gen.; *slyšati* 'listen' normally takes acc., rarely gen. *Razuměti* 'understand' normally takes acc., rarely gen. or dat.

2. Verbs denoting striving or attainment: alkati 'hunger for', želějąta 'desire', žadati 'await', žędati 'thirst for', iskati 'seek, look for', prositi and vasprositi 'ask, beg', pytająta 'question, examine', posětiti and prisětiti 'visit', trěbovati 'demand', čajati 'expect', and verbs with the prefix do—doiti doidąta 'reach', dotekąta 'run up to', dovedąta 'lead up to', dozarěti 'perceive', and dožadati 'achieve'. Xotěti 'want' may take either genitive or dative (or, rarely, acc.). Vaprositi 'ask' takes the acc. of person asked and genitive of thing asked for.

3. Verbs denoting sufficiency: *ispluniti* and *napluniti* 'fill (with)', *napoiti* 'give to drink', *nasějati* 'sow', *nasytiti* 'satisfy, satiate', *natruti natrovots* 'feed'.

4. Certain verbs that normally take the accusative are occasionally found with a genitive that perhaps denotes "part of, some of": e.g. vskusiti 'taste' jako že vъkusi arxitriklinъ vina byvašaego otъ vody (J 2:9) 'when the ruler of the feast tasted the wine made from water'. Attestation is far from uniform for *iměti* 'have', \**jъmǫtъ* 'take, receive' (cf. group 2 above): priętъ xlěba Mar (J 21:13) 'he took (?some) bread' (but Zo As acc. xlěbъ; Greek 'the bread'); da života imate Zo As (J 5:40; Mar životъ) 'that you might have life'; imatъ života věčьnaego Zo Mar (J 6:47; As životъ věčnyi) '(he) has everlasting life'.

5. Verbs denoting deprivation and the like: bojati sę 'be afraid of', izbaviti and izbavljati 'rid', izbyti izbędęts 'escape, be freed of', lišiti 'deprive', svoboditi 'free', stradati 'suffer loss of', plakati (sę) 'mourn (loss of)', sramljajęts sę and postyděti sę 'be ashamed of'; běžati and běgajęts 'flee from' and the compounds izběžati, izběgnęti, otsběžati and otsběgnęti; and several other verbs with the prefix ots: otslęčiti 'separate', otsvrsgęts sę 'throw off', otsmětajęts sę 'reject', ostanęts ostati 'leave, let', oslušajęts sę 'disobey'. Otsrekęts sę 'renounce, disclaim' takes gen. or dat. Some verbs take either a genitive or the preposition ots + gen.: otspadęts 'fall away from', o(ts)stępiti and o(ts)stępajęts 'retreat', otsstojati 'be distant from', otsmęts 'take away from'. Razlęčiti 'separate' takes acc. of things separated and gen. or otb + gen. or na + acc. to express 'from'. *Prostiti* and *praštajątb 'forgive'* take acc. of person and gen. of thing. *Prěobiděti* 'insult' takes gen. or acc. *Mustiti* 'avenge' takes a genitive to express cause and a dative to express the object of venge-ance.

6. With impersonal *ne byti* the genitive is normal: e.g. *boga* něstb (ps 13:1; 52:2) 'there is no god'; zanje ne bě ima *města* vb obiteli (L 2:7) 'for there was no room for [the two of] them in the inn'. (Contrast L 14:22: i ešte město estb 'and there is still room'). See also  $\S23.11$ .

**b.** The genitive normally serves as the complement of a negated transitive verb, corresponding to the accusative direct object of a positive verb. For example (see also §23.22):

Nikъtože ne vъlivaet vina nova vъ měxy vetъxy (L 5:37) 'no one pours new wine into old skins'; ne umyeši nogu moeju vъ věkъ (J 13:8) 'you shall never wash my feet'; blodite ne vědošte kъnigъ ni sily božię (Mt 22:21) 'ye do err, not knowing the scriptures nor the power of God'.

With a dependent infinitive: Něsmb dostoinb otbrěšiti *remene* sapogu ego (L 3:16) 'I am not worthy to unloose the strap of his shoes'; bbdělb ubo bi i ne dalb bi podbkopati *domu svoego* (L 12:39) 'he would have watched and would not have let his house be broken into'.

Replacing a double accusative: Ne tvorite *domu* otъca moego *domu kupljъnaego* (J 2:16) 'make not my father's house a house of trading'; jako ne možeši vlasa edinogo běla li črъna sъtvoriti (Mt 5:36) 'for you cannot make one hair white or black.'

c. It is regularly the complement of a supine (cf. 21.5); pride ... viděts groba (Mt 28:1) "(she) came to see the tomb'; izide sěei sějats sěmene svoego (L 8:5) 'a sower went out to sow his seed'. In Su, the acc. is sometimes found in this function.

d. The genitive is often the complement to a substantive, usually indicating possession, quality, or quantity: duxb otbca vašego; 'the spirit of your father'; godpodinb xrama 'the master of the house'; člověkb eterb dobra roda 'a man of good family'; sedmb košbnicb 'seven baskets' (cf. §20); dbšti dbvoju na desęte lětu 'a daughter 12 years old', Aecatopo spatua (Su 279.15) '10 brothers'; mbnožbstvo rybb 'a great quantity of fish', čašejq studeny vody 'with a cup of cold water'. Observe that the possessive genitive is replaced by possessive adjectives if the possessor is represented by a substantive which denotes a person or animal and which is not otherwise modified: tektonovb synb 'son of the carpenter' (tektonb); učenici ioanovi 'the disciples of John' (ioanb). In J 1.11 otb gradbca mariina i marty sestry eę 'from the city of Mary and (of) her sister Martha', the adjective for 'Mary's' is used as is normal, but since Martha is further defined, the adjective cannot be used and the substantival genitive remains. Compare силоня христосовоня и архайтела рафаила (Su 231.7) 'by the power of Christ (adj.) and the archangel Raphael (gen.)'.

The dative is in competition with the genitive in this usage, e.g. šlěm *s spasen bja* 'the helmet of salvation' (Eu 97a5) vs. *spasen bju* (Eu 94a9, 99a10); see also §18.5h.

e. As complement to pronominal and adverbial expressions denoting or implying quantity or number:  $k \pm to i x = who of them'$ ,  $k \pm 2 \pm do vas =$ 'each one of you', koliko xléb = 'how many loaves of bread', vbsi eliko = ix = pride 'all of them who came', malo ix = est = 'there are few of them'. Isolated examples like  $d \pm va u cenik = svoix = 'two of his disciples' are found$ for the more usual type with <math>ot = + genitive.

f. As complement to the adjectives *plbnb* and *isplbnb* (indecl.) 'full': sbsqdb ... plbnb *ocbta* (J 19:29) 'a vessel full of vinegar', **rwkba испльмb** (Su 566.13) 'full of anger'. *Dostoinb* 'worthy (of)' takes the genitive (dostoinb ... *mbzdy svoeę* Mt 10:10 'worthy of his pay') or dative (dostoiny *pokaaniju* L 3:8 Zo Mar Sav [~ *pokaanič* As] 'worthy of repentance').

g. With certain expressions of time: събраша же са мѣсаца мана (Su 201.22) 'they met in May'; кдинож лѣта (Su 227.29) 'once a year'; sedmb kraty *dьne* (PsSin 118.164; Sluck дньмь) 'seven times a day'.

**h.** With comparatives: este lučbši *psticb* (L 12:24) 'you are better than birds'; boljbša *sixo* uzbriši (J 1:51) 'thou shalt see greater things than these'; teče skorěe *petra* (J 20:4) '(he) ran faster than Peter'.

i. The genitive is normal in exclamations: w веды (Su 56.25) 'Oh misfortune!'; w безаконьнааго вьзв'яшения (Su 217.7) 'Oh lawless frenzy!'

**18.4** The *locative* without preposition is very limited.<sup>3</sup> A few expressions of time or place may be interpreted as independent locatives, though they may also be classed as fixed adverbial expressions, for example, *zimě* 'in winter', *polu dbne* 'at noon', and *polu nošti* 'at midnight', *tombčasě* 'at that moment' (e.g. Mt 17:18).

The locative regularly serves as complement to the verbs kosnęti sę and prikosnęti sę touch' (e.g. kъto prikosnę sę rizaxъ moixъ Mk 5:31 'who touched my garments?');<sup>4</sup> to several other verbs with the prefix pri-: priložiti 'add' (also takes dat. or na + acc.), priležati 'take care of' pristanęti pristafi 'take part in', and perhaps others; to naležati 'press

<sup>&</sup>lt;sup>3</sup> Early East Slavic regularly used place-names in the locative case without preposition, and it is highly probable that all 9th-century Slavic shared this usage. OCS offers only a couple of uncertain examples.

<sup>&</sup>lt;sup>4</sup> Under Greek influence, the gen. appears for loc. with *prikosnoti se* twice, while the loc. after *kosnoti se* is less common than acc.

upon' (also takes dat. or na + acc.), napadajots 'attack' (more often takes na + acc), and perhaps a few others.

**18.5** The *dative* case signifies the goal towards which something is directed either in a literal sense or in the more abstract meanings of "intended for" or "for the benefit of", or even "with relation to".

a. Specifically directional examples: se cěsarjь tvoi grędetъ *tebě* (Mt 21:5) 'behold thy King cometh unto thee'; i nese *materi svoei* (Mt 14:11) 'she took (it) to her mother'. The directional meaning is more often expressed by using  $k_{\overline{2}}$  plus dative.

b. Less specifically directional verbs taking a dative complement are numerous. They include various verbs of giving, saying, promising, commanding, scolding, rebuking, annoying, pleasing, liking, believing, serving, helping. A partial list: *oběštajǫts* 'promise', (*po*)*velěti* 'command', *sǫditi* 'judge', *odolějǫts/udolějǫts/udelějǫts* 'conquer', (*vsz)braniti* 'forbid, hinder', *prětiti* 'warn, threaten, rebuke', *prěrekǫts* 'rebuke', *ponositi* 'upbraid', *dosaditi* 'annoy', (*po*)*rǫgajǫts sç* 'make fun of', *zaviděti* 'envy', *rьvьnovati* 'be jealous', *vražьdovati* 'hate', (*u*)*podobiti* 'compare', *ugoditi* 'please' (and *godě byti* 'be pleasing'), *prijajǫts* 'be friendly', *vsnьmǫts* 'heed', *věrovati* (and *věrǫ ęti*) 'believe', *poslědovati* 'follow', *pomagajqts/pomogǫts* 'help', (*po*)*služiti* 'serve', *rabotajǫts* 'work for', *diviti sç* and *čuditi sç* 'wonder at', *povinovati sç* 'obey', *radovati sç* 'rejoice at', *smijati sç* 'laugh at'. (*Xotěti* 'wish', cf. §18.3a2.)

c. Two verbs are found only with the reflexive short dative si: sōžaliti si 'pity' and požaliti si 'be displeased' (не разгичава сд., не пожали си Su 364.2 'he did not become angry, he did not become displeased'). Sotožajoto/sotožiti 'afflict' may take si in the meaning 'despair, be discouraged' (podobaatъ vъsegda moliti sę i ne sotožati si L 18:1 [Zo has se] 'one ought always to pray and not be discouraged').

d. Dative with infinitive (cf. 21.4): dastb imb vlastb čędomb božiemb byti (L 1:12) 'to them he gave power to become the children of God'; чаквште кмоч живоч быти (Su 80.14) 'thinking him to be alive'; glagoljošte vaskrěšenbju ne byti (L 20:27) 'denying that there is any resurrection'; мычкаъ ли кси страхы очбовати сд намъ (Su 176.16) 'did you think that we would become afraid because of threats?'; агъ že glagoljo vamъ ne klęti sę vamъ (Mt 5:34) 'but I say unto you, swear not at all (you are not to swear)'; сътвори ми хыгдинж сѣсти ми вь йен (Su 204.2) 'he will make a hut for me to sit in (it)'; молитвж йем8 сътвори придточ быти отъ йего (Su 547.20) 'he begged him to be received by him'. e. The "dative absolute"—a participial subordinate clause expressing various types of attendant circumstance. For example: m bnogu sq stunarodu i ne imq stems česo ěsti ... isus glagola ... (Mk 8:1) 'the multitude being very great, and having nothing to eat, Jesus said ...'; učę stu emu ljudi vъ crъkъve ... sъstašę sę arxierei (L 20:1) 'as he taught the people in the temple ... the chief priests gathered'; po vъsę dъni sq stu sъ vami vъ crъkъve ne prostъr ste rqkъ na mę (L 22:53) 'when I was daily with you in the temple, ye stretched forth no hands against me'; i abъe este glagolją stu emu vъzglasi kurъ (L 22:60) 'and immediately, while he yet spake, the cock crew'; more že větru veliju dyxają stu vъstaaše (J 6:18) 'And the sea, since a great wind was blowing, was rising.' Normally the dative participle does not refer to the same person or thing as the subject of the main verb, but this rule is occasionally violated.

f. The dative denoting "for the benefit of, with respect to" occasionally presents difficulties in translating into English, for it sometimes verges on the idea of possession and sometimes is so weak as to be superfluous in English. For example, *člověku eteru bogatu* ugobb3i sę njiva (L 12:16) 'the field of (lit. for) a rich man brought forth rich harvest'; съмотри же ми уълод жиство ихъ (Su 443.7) 'consider [for me] their crime!'; къде си иъий ж кси (Su 242.9) 'where are you now [for yourself]?' Cf. the idiom *čьto mыně i tebě* (Mk 5:7) 'what is for me and thee' (i.e., what do we have in common?).

g. The dative serves as the complement of certain adjectives: podobars and točars 'similar', ravars 'equal', ugodars 'pleasing', povinars 'guilty', dostoirs 'worthy of' (also with gen.).

h. The dative complement of substantives is semantically close to the adnominal genitive (§18.3d); dbhbe maštenbju (L 21:22) 'the days of vengeance', xramb molitvě ... vrbtbpb razboinikomb (Mt 21:13) 'house of prayer ... den of thieves', starěišiny ljudbmb (L 19:47) 'elders of the people'; otbpuštenie grěxomb (L 3:3 MarVat ~ grěxovb Zo) 'the forgiveness of sins'; syni světu (J 12:36 Zo ~ съннове св'ята Sav) 'the sons of light'. The possessive meaning is particularly common with the short dative personal pronouns mi, ti and si: e.g. drugth mi pride (L 6:6) 'my friend has come', лице отьцю ми (Mt 18:10 Sav ~ оtьca moego Mar).

Adjectives compete with the genitive or dative adnominal complement. For usual skrbžbt zqbomb 'gnashing of teeth', Su has скръжетъ зжбычын. In Mt 8:28, Sav has бъ землж геръгесиномъ 'into the land of the Gergesenes' vs. As gergesinbskq. (Cf. domu kupljenaago, §18.3b, above.)

i. The dative of price: ne pętъ li ръtiсь věnitъ sę pěnęzema dъvěma (L 12:6) 'are not five sparrows sold for two farthings?'

**18.6** The *instrumental* case signifies tool, agent, means, and manner—various types of attendant circumstance.

Some examples: idq ... korabljemb (Mk 6:32) 'they went by boat', a. вити и жилами соуровами (Su 100.30) 'to beat him with raw thongs'; влъшъвами одолѣвакши (Su 159.20) 'you will conquer by magic'; bodete nenavidimi vьsěmi (Mt 10:22) 'you will be hated by all'; iskušaemъ sotonojo (Mk 1:13) 'being tempted by Satan'; pritzčami glagolati (Mk 12:1) 'to speak by parables'; i tacěmi pritsčami monozěmi glagolaaše imb slovo (Mk 5:33) 'and with many such parables he spake the word unto them'; javi se iněmb obrazomb (Mk 16:12) 'he appeared in another form'; neumsvenama rokama ědetъ (Mk 7:5) 'they eat with unwashed hands'; vъzъpi velьemь glasomь (L 1:42) 'he cried out in a loud voice'; sii ljudie ustьnami čьtętъ mę (Mk 7:6) 'these people honor me with their lips'' кръвь течалше ръками (Su 103.27) 'blood flowed in rivers'; bě poganynji ... rodomb (Mt 7:25) 'she was a pagan by birth'; славенъ съ племенемь, высокъ же саномъ (Su 63.8) 'being famous by descent and high in rank'; sъxoždaaše potemь těmь (L 10:31) 'came down (by) that road'; vъniděte ozokymi vraty (Mt 7:13) 'enter ye in at the strait gate'; četyromi desety i šestijo lětъ sъzьdana стъку si (J 2:20) 'forty and six years was this temple in building'.

The adnominal instrumental člověk *nečistomь duxomь* (Mk 1:23, Zo Mar) 'a man with an unclean spirit' seems out of place. Similarly, **ΜЖЮ правьдивоч и добромъ житинимъ** (Su 294.11) '(to) a man [who is] just and virtuous, ἐναρέτω' seems to function, incongrously, as an adjective in the dative case.

**b.** The adjectives dovolons (dovolons) 'satisfied with' and dložons 'owing, in debt' regularly take the instrumental.

c. A number of verbs normally take the instrumental: vladqts 'rule' (and others with similar meaning) ty vladeši drbžavojq morbskojq (Ps 88:10) 'Thou rulest the power of the sea'; ti obladajqtb zemljejq (Ps 36:9) 'they rule the earth' (cf. MMRUITAAFO BAACTD ... AQUUER H TBAOND Su 157.11 'having power over soul and body'); pekqts sq 'worry about' ne pbcěte sę dušejq vašejq (Mt 6:25) 'take no thought for your soul'; klqti sq (and zaklinajqts sq) 'swear' ni glavojq svoejq klbni sę (Mt 5:36) 'neither' shalt thou swear by thy head'; ženiti 'marry' ženęi sę puštenojq (L 16:18) 'he who marries a divorced woman'; isplaniti 'fill' (usually takes gen.) isplani sę duxomb svętyimb (L 1:14) 'she was filled with the Holy Spirit'; (u)psvajqts 'trust' (also takes na + acc.; or dat., sometimes with  $k_3$ ); BONNT

пъвам своких силонх (Su 105.11) 'a soldier trusting in his own strength'; pokyvajętъ 'shake' pokyvašę glavami svoimi (Ps 108:25) 'they shook their heads'; skrьžыtati zęby 'gnash one's teeth' (e.g. Mk 9:18).

d. The instrumental sometimes occurs in the predicate with verbs denoting being or becoming. Beside normal nominatives such as ědz že ego bě *akridi* i *medz* (Mt 3:4) 'his food was locusts and honey',  $\Lambda \in \Lambda$  **BUCT** BOAA TORNA (Su 78.1) 'the ice became warm water', we find AtBOHT **BO BT EVA** (Su 489.9) 'for Eve was a virgin'; HE **BTAH HUKTOTE HOAOFT** (Su 420.10) 'let no one be a Judas'.

#### 19.0 On the use of the prepositions.

**19.11** Prepositions taking the accusative only are vzz 'in exchange for' (cf. vzskojo 'why?', vzskrai 'on the edge [of]'), *skvozě* or *skozě* 'through', and the compound *podlegz* 'along'.

**19.12** With the genitive only: bez 'without' (cf. §3.311), do 'to, till', iz 'from, out of' (cf. §3.311), ots 'of, from, since, by' (may express agent; e.g. porogant byste ots vlexve (Mt 2:16) 'he was mocked by the wise men'), u 'near, by, from' (u groba 'near the tomb'; prositi česo u/ots kogo 'ask for something [gen.] from someone'). Words which function both as adverbs and as prepositions with genitive are: blize 'near', vrexu 'on top (of)', kromě 'outside', okreste 'around', prěžde 'before, prior to', razvě 'except for', svěnje 'outside of', and edě 'near'.

The words *radi/radыma* and *dělja/děljыma* 'on account of, for the sake of' are postpositions: *sego radi* 'for this reason', мене дѣла 'for my sake'.

The expression vo ... město 'instead of' includes the genitive between its members: vo iroda město 'instead of Herod'.

19.13 With the locative only: *pri* 'near, in the time of'.

19.14 With the dative only:  $k_{\overline{a}}$  'to, toward', and the adverbs *protivo* (*protivo*/*protivu*) 'opposite, according to' and *prěmo* 'opposite'.<sup>5</sup>

<sup>&</sup>lt;sup>5</sup> The use of ka + dat. after verbs of saying is a clear case of Greek influence: almost invariably when the Greek has the dative alone, so does OCS, but when Greek has  $\pi\rho\delta_5$  + acc. (a normal construction), OCS has ka + dat. There is no evidence that this usage was ever part of a spoken Slavic dialect.

Greek interference is obvious or probable in a number of case-usages and in selection of prepositions as well; often the evidence is too slim to allow a clear decision as to what native usage OCS might have preferred. The precise choice and use of prepositions varies with time and place; it is not surprising that OCS manuscripts constantly disagree in details.

19.15 With the instrumental only: meždu 'between'.

19.21 With the accusative and locative cases:

**19.211**  $V_{\mathfrak{F}}$  + accusative means 'into, to',  $v_{\mathfrak{F}}$  + locative means 'in, inside'.  $V_{\mathfrak{F}}$  is used with the accusative in many time expressions and fixed locutions of varying meaning.

**19.212** Na + locative means 'on', also 'concerning'. Na + accusative means 'onto, to, toward, against' (iže něst $\mathfrak{s}$  s $\mathfrak{s}$  m $\mathfrak{s}$ nojo na m $\varrho$  est $\mathfrak{s}$  Mt 12:30 'he who is not with me is against me'); it occurs in a number of fixed expressions.

**19.213** O(ob) with locative normally means 'around, about, concerning', but the relationship it expresses is often tenuous and can be rendered by various other English prepositions, e.g. sily dějots se o njemb (Mt 14:2) 'mighty works do show forth themselves in him'. O(ob) + accusative means 'against', but also 'concerning'. Note the fixed expressions o desnqjq 'at the right (hand)', o šujqjq 'at the left', ob on polo 'on the other side (of)', ob noštb 'during the night', o sebě 'of oneself, by oneself'.

**19.22** With the accusative and instrumental case—the preposition defines a position; the accusative case signals motion directed to the position, while the instrumental case implies rest: nad 'above, over'; pod 'under, beneath'; pr ěd 'in front of' or 'prior to'.

19.23 So takes either genitive or instrumental. So + genitive means 'from, down from, from the surface of' and 'since'; so + instrumental means 'with, accompanying'.

**19.31** Za takes accusative, genitive, or instrumental. Za + instrumental means 'behind, after' (no motion implied), while za + accusative specifies motion to a position behind. Za + acc. frequently means 'because of'; but it has other meanings too, e.g., oko za oko 'an eye for an eye'; etb jo za  $r \rho k \rho$  (Mt 9:25) 'he took her by the hand'. In expressions meaning 'strike on the cheek' za lanit $\rho$  is found beside vb lanit $\rho$  and po lanitama. With the genitive, za expresses cause: page brows ca za страха июденска Su 483.11 'they scattered for fear of the Jews.'

**19.32** Po takes the accusative, locative, or dative. It is rare with the accusative: po  $\check{c}bto$  'why?' (cf. ponje[ $\check{z}e$ ] 'because'); po města 'in divers places', po vuse duni 'daily', po vusě lěta 'every year'; po imena 'by their names' (Euch 67b14). Po + locative means 'after': po tomb 'after this, afterward, then' (cf. also плака по и́муъ Su 38:5 'he wept for them', i.e. at

#### 152

losing them); and occasionally 'in favor of, for', e.g. iže bo něstъ na vy po vasъ estъ (L 9:50) 'for he that is not against you is for you.' Po + dative has varied meanings: po zemlji, po morju, po aeru 'over the/by land, on the/by sea, by air', po vusemu gradu L 8:39 'throughout the whole city', po poti 'on the road', po srědě 'in the middle; amid, among'; po dělomъ 'according to deeds', po vině 'according to guilt', po prědanuju 'according to tradition', po silě 'by force', po rędu 'in order'.

#### 20 On the syntax of the numerals

Edins edino edina (§4.321) 'one' is a pronoun, and agrees with its singular headword in case and gender. (In Su the stem is also KALN-.)

D = va d = ve 'two' and oba obe 'both, the two' are likewise pronouns (cf. \$4.201) and agree with their dual headwords in gender and case.

*True tri* 'three' (\$4.402) and *četyre četyri* 'four' (\$4.4114) are plurals, and agree with their plural headwords in gender and case.

The numerals from five to ten ( $\S4.4$ ) as well as *soto* 'hundred' and *tysosti/tysosti* 'thousand' are substantives which are followed by the genitive plural (cf. \$18.3d).

The teens are expressed by the units + na desete: edino na desete '11', dova (dově) na desete '12', etc. (Note oba na desete 'the twelve'.) The tens are expressed by the units followed by the proper form of deseto: dova deseti (dual), true desete (nom. pl.), petu deseto (gen. pl.), etc. The hundreds similarly use units plus the proper form of soto: dově sotě, tri sota, petu soto, etc.

In principle, the *na desęte* of the teens and the *desęt*- of the tens do not affect the counted substantive: *oběma* zě na desęte *apostoloma* (§18.5h) imena sǫtъ si (Mt 10:2) 'and the names of the twelve apostles are these'. In a number of examples, however, the whole numeral behaves as a substantive requiring the genitive plural: L 9:7 '12 baskets' Zo *koša dъva* na desęte ~ Mar *košъ*;  $Aa \dots$  съценатъ тръмъ десатемъ (§18.5i) мѣдъницъ (Su 331.29) 'so they might appraise [him] at thirty coins'.

Units are added by means of *i* (от ти in Su); səto *i petь deset*ə i tri (sc. rybə, J 21:11); instr. četyrəmi desety i šestəjq lětə (J 2:20) 'in the course of 46 years'. For further details, see Vaillant 157ff.

## 21.0 On the use of the verbal forms

**21.1** The present tense, as opposed to a orist and imperfect, does not specify time. Imperfective presents most usually denote an action viewed as simultaneous with the moment of speech or with a moment in past or

future which is defined by the context; they may also denote actions which are repeated or of general validity. Since the perfective aspect specifies that the completion of the action is envisaged, the present perfective forms cannot denote action in progress. They signify rather action viewed as completed in the future or at any other moment defined by the context. In statements of general validity, the perfective present shows that the action is viewed as completed whenever the situation is suitable. For example, слыньцоу въсходаштоу съкрыктъ са ствив (Su 417.28) 'when the sun rises (cf. §18.5e), the shadow hides itself'; aste k'to bědojo ukradetъ sъnědьno čьto, 40 denъ da pokaetъ se (Euch 103b1) 'if anyone steals something to eat because of need, let him do penance for 40 days (cf. §22.11); ašte li vědělъ ... vъ kyi časъ tatь pridetъ (Mt 24:43) 'if [he] had known in what watch the thief would come'; žena egda raždaets pečals imatъ ..., egda že roditъ otročę къtomu ne pomьnitъ skrъbi (J 16:21) 'a woman when she is in travail [= is giving birth] hath sorrow ... but as soon as she is delivered of [= has given birth to] the child she remembereth no more the anguish'.

On the expression of future time.<sup>6</sup> There is no specific set of 21.11 forms denoting the future. The most frequent means of expressing future action is the present tense (especially of perfective verbs). The future of 'to be' is expressed by bodots. Often the present forms of the verb xotěti 'want' (§15.233)—and rarely the presents of načonoto and vočonoto 'begin'-are used with an infinitive in a sense close to the English future. Rather more common are quasi-futures consisting of the present of *imamb* (§16.24). This construction occasionally means literally "have to", but normally indicates "is to, is destined to"; e.g. агъ brašьno imamь ěsti egože vy ne věste (J 4:32) 'I have meat to eat that ye know not of'; uže ne imamь piti отъ ploda lozьnaego (Mk 14:25) 'I will drink no more of the fruit of the vine'; iže ašte ne priimetъ cěsarьstvuja božuja jako otroče ne imatъ vъniti vъ nje (L 18:17) 'whosoever shall not receive the kingdom of God as a little child shall in no wise enter therein'; koe bodetъ znamense egda imotъ sъkonьčati sę vьsě si (Mk 13:4) 'what shall be the sign when all these things shall be fulfilled?'

Past-tense forms of *xotěti* and *imati* express a relative future—an event regarded as future from the point of view of a past moment: e.g. se že

<sup>&</sup>lt;sup>6</sup> More examples and discussion in Radoslav Večerka, Altkirlchensl. Syntax, II 174-185 (= Monumenta linguae slavicae dialecti veteris, Tom XXIV [XXVII, 2], Freilburg i. Br., 1993).

glagolaše kleplję koeją sъmrьtьją xotěaše umrěti (J 12:33) 'This he said, signifying what death he would die.'

#### 21.2 The past tenses

While the present tense is indifferent as to time, the aorist and imperfect both specify action presented as taking place prior to the moment of utterance. The *imperfect* specifies an action coordinated with a fact or act in the past: this point of reference may or may not be present in the context. The *aorist* has no such specification – it is merely an event in the past. The aorist thus functions as the story-telling device which presents a chain of events, while the imperfect gives the background events or stops to concentrate on an action being performed at a certain moment.

The majority of aorists are of perfective aspect, but imperfective aorists are not uncommon: the action is presented as past, but there is no specification that it was completed. Imperfects are nearly always of imperfective aspect, but perfective imperfects appear when the need arises; a coordinated yet completed action is specified, and this usually means a repeated action.<sup>7</sup> For example... Aute ca chaoyyaaue he hat the kooy hhybroxe dath kmoy, to kothers...dad haue hhumtoyoymoy (Su 207.14) 'if it happened that he had nothing to give him, he would...give his (own) dress to the poor man.'

A few examples of a rist-imperfect usage:

Mk 5:24 ... ide (1) sb njimb i, po njemb idčaše (2) narodb mbnogb i ugnětaaxq i (3) '[Jesus] went with him and after him went a great crowd and they pressed on him'. The determined aorist (1) defines a past moment to which further actions (imperfects, 2, 3) are coordinated.

Mt 26:57ff. (in the story of the arrest and trial of Jesus) poimъše isusa věšę i (1) ... petrъ že iděaše (2) po njemь izdaleče i sěděaše (3) sъ slugami viděti konsčino ... [arxierei]

<sup>&</sup>lt;sup>7</sup> Dostál (*Studie*, pp. 599-600) notes that over 40% of the attested OCS aorists are imperfective, while the 23 perfective imperfects constitute only about 1% of attested imperfects. It is worth emphasizing these figures, for many investigators have assumed that any verb which has an aorist is necessarily perfective, and similarly that an imperfect tense form is proof of the imperfective aspect of the verb in question. It is now clear that in OCS—as well as in Rusian (Early East Slavic) and Old Czech and modern Serbian, Macedonian and Bulgarian—tense and aspect are two independent systems. One should not lose sight of the fact that certain forms (like perfective imperfects) are statistically infrequent because the situations requiring them are uncommon. It is only in a narration of complex events in the past that one can expect to find the full range of the possible past tense forms, including the various combinations of participles; virtually no passages of this type happen to be attested in OCS.

iskaaxq(4) Ibža sbvědětelja na isusa ... i ne obrětq(5) ... isusb že mlbčaaše (6) ... petrb že vbně sěděaše (7) ... i pristopi (8) kb njemu edina rabynji. 'Having seized Jesus, they led him ... And Peter followed at a distance and sat with the servants to see the end... [the chief priests etc.] sought false witness ... but found none ... [they ask questions] ... but Jesus was silent ... [The trial goes on, and the scene shifts:] Now Peter was sitting outside ... and a slave-woman came up to him.' The narration has started with a series of aorists, of which (1) is the last. Two imperfects (2, 3) show the actions of Peter, coordinated to the main flow of action which is centered around Jesus. The third imperfect (4) refocuses on the court and the officials' non-coordinated, unfinished act of searching; the lack of success is summed up by a negated perfective aorist (5). More aorists (not cited) carry on the narration, but through these completed acts by others Jesus, in a coordinated negative action (specified by the imperfect, 6), remains silent. Peter, in the meanwhile, is sitting (coordinated act, imperfect; 7), when another actor appears and completes an action (P aor. 8)

L 1:80 otročę že rastěaše (1) i krěpljaaše sę (2) duxomb, i bě (3) v $\mathfrak{b}$  pustynji do dbne avljenbja svoego k $\mathfrak{b}$  izdrailju 'And the child grew, and waxed strong in spirit, and was in the desert till the day of his showing unto Israel'. The two imperfects (1, 2) denote actions coordinated with the imperfective aorist (3) which states a fact rather than an event.

**21.21** The perfective aorist  $byst_{\delta}$  in principle denotes an event, the appearance of something not present before. The imperfective  $b\check{e}$  ordinarily reports a new fact, without specifically coordinating it to other events. The imperfect  $b\check{e}a\check{s}e$  specifies coordination to some point in past time. The three-way distinction is sometimes blurred, however. Some examples:

I bysta burja větrbna velija (Mk 4:37) 'and a great wind-storm came up'; egda bysta dbnb, priglasi učeniky svoę (L 6:12) 'when day broke, he called his disciples'. On že slyšavb se priskrbbnb bysta, bě bo bogatb 3ělo (L 18:23) 'and when he heard this he became very sorrowful, for he was very rich.'

Vblezb že vb edinb otb korabljicu, iže be simonovb, moli i otb zemlje otbstopiti malo, i sedb učaase is korablja narody (L 5:3) 'and having entered into one of the ships, which was [as a general fact not yet mentioned] Simon's, he prayed (aor.) him that he would thrust out a little from the land, and having sat down, he taught the crowds out of the ship.'

I pomanošę pričęstъnikomъ iže *běaxo* vъ družčemъ korablji (L 5:7) 'and they beckoned to their partners, which *were* [precisely at that time; impf.] in the other boat'.

**21.211** Incidents in the Gospels are frequently introduced by a redundant *bysts* that corresponds to the typically biblical phrase "it came to pass that". The construction is originally Hebrew, literally translated in the Greek Bible. For example:

Bysta že idoštemu po poti, reče edinu ku njemu, ido po tebě jamože koližudo ideši, gospodi (L 9:57) 'and it came to pass, that, as they went (§18.5e) along the way, a certain man said to him, I will follow thee wheresoever thou goest, Lord'; i bysta egda vunide isusu vu domu edinogo ku nga fariseiska vu soboto xlěba ěstu (§21.5) ... (L 14:1) 'and it came to pass, when Jesus went into the house of one of the chief Pharisees to eat bread on the sabbath day ...'

#### 21.3 On participles

The present and past active participles (of both aspects) are freely used in all cases and numbers.<sup>8</sup> In cases other than nominative and accusative, the long-forms are found more often than the short-forms. The present and past passive participles (of both aspects) are used mostly in the nominative short-forms, but other cases occur as well. True participles are clearly verbal in character; they denote an action which is subordinate to another action that is expressed by a finite verb in the context. Often, however, the participial form functions purely as an adjective.

There are instances where variant readings of the same passage have participial constructions equivalent to relative clauses, e.g. Mt 10:40 Sav **nphhemaal bace meterphematere**  $\sim$  Zo *iže vy primeto* mę priemljete 'he who receives you, receives me'. In translating into English, the participles are often best rendered by finite verbs in dependent clauses of various types.

21.31 The present participles, like the present tense, are unmarked as to time, denoting either general verbal action or (most frequently) action coordinated with the time expressed by the context. For example, active participles: mъnoʒi bo pridqtъ vъ moe imę glagoljąšte ... (Mk 13:6) 'for many shall come in my name, saying ...'; běaxq že edini ... glagoljąšte ... (Mk 14:4) 'and there were some that said ...'; passive participles: se estъ tělo moe davaemoe za vy (L 22:19) 'this is my body which is given for you'; isusъ že slyšavъ slovo glagoljemoe (Mk 5:36) 'Jesus, having heard the word which was spoken, said ...'; кгда владъикъ си видѣ vтома (Su 425.17) 'when he saw his Lord being honored'. If the present participle is perfective, it signifies a completed subordinate action; the completion most usually is repeated, or else is in the relative future, but the possibility of a relative past is not excluded. For example: NE Бждѣмъ оубо NE похвальшите такого благодѣтела (Su 494.14) 'let us not fail to praise such

<sup>&</sup>lt;sup>8</sup> These remarks do not apply to the resultative participles, which are found only in the nominative short-forms either (regularly) accompanied by a form of *byti* in the compound tenses defined in §14, or else with omission of the auxiliary in forms of the perfect (§14.1).

In a single passage (Su 386.5-8) the l-participles occur with kuua and mpoor 'would that!; if only!', which also take the conditional (usually with da). Non-OCS evidence seems to indicate that the omission of the auxiliary was normal after these conjunctions.

For a detailed and illuminating discussion of the active participles, see Rudolf Růžička, Das syntaktische System der altslavischen Partizipien und sein Verhältnis zum Griechischen, Berlin, 1963.

a benefactor (lit. 'let us not be [repeatedly] not praising'); и еже аггелы похвалимъ дръжитъ са скврънавъима ржкама (Su 506.19; cf. §22.4) 'and he whom the angels praise [who is praised by the angels] is held by dirty hands'.

The present passive participles, particularly if perfective, may denote the possibility of an act: e.g. vidims 'being seen; visible'; měrims 'being measured; measurable'; especially common with negation, nepobědims 'invincible', nerazorims 'indestructable', and others.

**21.321** The past active participles present an action which started (and usually is completed) before the action of the main verb to which the participle is subordinated. For example, sęděte szde donzdeže žzdz pomolją sę (Mk 14:23) 'Sit ye here while I go (lit. having gone) pray'; žzdz pokaži sę iereovi (L 5:14) 'go (lit. having gone) show yourself to the priest'; žzdzše vz gradz vzzvěstišę vzsě (Mt 8:33) 'they went into the city and told of everything'.

An example contrasting present and past active participles: (L 6:47– 49) slyšei(1) sloves amoja i tvore (2) ja ... podobъnъ estъ člověku ziždoštu(3) xramino ... na kamene; ... slyšavyi(4) i ne tvorjь (5) podobъnъ estъ človeku szzbdavzšu(6) xramino bez osnovanъja ... 'He who heareth my sayings and doeth them ... is like a man who builds a house ... on a rock; ... and he that heareth (=has been hearing) and doeth not (= has not been doing) is like a man that without a foundation built a house'. Here 1 and 2 are present actions, contemporaneous with the moment of utterance, while 3 is a statement of general validity. 4 and 5 show uncompleted (imperfective) actions which started before the principle action, while 6 not only started before, but has been completed.

**21.322** The past passive participle denotes a state produced by an outside agent whose action started prior to the moment denoted by the context. In the vast majority of cases the short nominative forms are used with some form of the verb *byti*; e.g.  $zvans \ ze$  *bysts* i isus (J 2:2) 'and Jesus was also invited'; věruei vъ njъ ne *bqdets osqždens*, a ne věruei juže *osqždenъ estъ* (J 3:18) 'he that believeth on him will not be condemned, but he that believeth not is condemned already'; možaaše bo si xrizma *prodana byti* (Mk 14:5) 'for this ointment might have been sold'; ně u bo *bě vsaždenъ* vъ tъmъnicq ioanъ (J 3:24) 'for John was not yet cast into prison'. The other cases and also the long forms do occur, however, particularly in the dative absolute construction (§18.5e) *obrqčeně byvъši* materi ego (Mt 1:18) 'when his mother was espoused ...'; rači ... prizъrěti na raba tvoego sego padъša grěxy, *poražena* bolěznijq (Euch 30a22)

'deign to look on this Thy servant (who has) fallen because of sins, (who has been) struck down by disease'; за прилтжих отъ ребръ адамовъ женж (Su 482.15) 'for woman, taken from the ribs of Adam'; не срамытанши ли са ... сътворенънуть тобоих зълии (Su 161.22) 'aren't you ashamed of the evils done by you (or, which you have done)?'

A number of words which are formally past passive participles are used as non-verbal adjectives, e.g. prokažens 'leprous', oslabljens 'paralytic', soměrjens 'humble', učens 'learned', izborans 'elect, select'.

21.33 On verbal adverbs. It has been suggested that certain forms spelled -šte represent verbal adverbs (or gerunds), e.g. neže dъvě no3ě imošte vъvrъženu byti vъ geono (Mk 9:45) '[it is better for thee to enter halt into life] than having two feet to be cast into hell'; сладъка ти кстъ въкоущажште нъ горька по въкоуса (Su 350.29) '[beware the pleasure of sin,] it is sweet for you as you are tasting (it) but bitter after the tasting'. The examples are better explained as scribal errors, for dative -štu.<sup>9</sup>

21.4 The *infinitive* is used as the complement of a number of verbs denoting command, desire, will, ability, or various expressions of purpose. The connotation of possibility or duty appears with *ests*: orb cero ... BHATTH KCTB CHAR XPHCTOCOBR (Su 413.16) 'from this is to be seen the power of Christ'; NECTB HAME OVENTH (Su 433.12) 'we are (ought) not to kill'; bystb že (§21.211) umrěti ništuemu i nesenu byti angely na lono avraamlje (L 16:22) 'and it came to pass that the beggar died and was carried by the angels into Abraham's bosom'. (Compare the use of infinitive with *imamb*, §21.11.)

Occasionally an infinitive with the conjunction *jako* (cf. §22.3) expresses result: e.g. i ne otъvěšta emu ni kъ edinomu glagolu *jako diviti sę igemonu* zělo (Mt 27:14) 'and he answered him to never a word, *so that the governor marvelled* greatly'; isplьnišę oba korablja *jako pogrąžati sę ima* (L 5:7) 'they filled both the ships *so that they began to sink*'.<sup>10</sup> Notice the dative as "subject" of the infinitive (cf. §18.5d).

**21.5** The *supine* is used after verbs of motion to specify purpose; e.g. ido *lovits* rybb (J 21:3) 'I am going (in order) to fish'; česo *viděts* izidete (Mt 11:7) 'What did you come out to see?'; i vbsta čistb (L 4:16) 'and he stood up to read'. However, the use of the specifically purposeful supine was apparently not obligatory, and the semantically neutral infinitive could convey the same meaning in the proper context. Thus in Mt 26:55, As

<sup>&</sup>lt;sup>9</sup> For other examples see Vaillant §169, and Jacques Lépissier, in *Studie palaeo-slovenica* (Prague, 1971) 215-20.

<sup>&</sup>lt;sup>10</sup> The OCS infinitive is imperfective; the act of sinking is not presented as completed, and the context of the whole passage implies that it was never completed.

izidoste sъ orožiemь ... ets mene 'are ye come out ... to take me?' is equivalent to Mar Zo Sav eti me—with genitive object of the supine, accusative with infinitive. Cf. мдж Вготоватъ мѣста вамъ (J 14:2 Sav) ~ ugotovati město (Zo Mar As Vat) 'I go to prepare a place for you'.

It is possible that the supine was not part of the dialect of some scribes, and that some infinitives replace an older supine. Thus Mar da ne vzzvratita se vzspęts vzzęta riza svoixa (Mk 13:16) 'let him not go back to get his garments' is "correct", but Zo's phrase vzzęti riza svoixa lacks clear motivation for the genitive.

#### 21.6 On se and "reflexive" verbs.

The form se has two functions; the two often overlap.

**21.61** Se may be the accusative of the reflexive pronoun (§4.6): javljq se emu samt (J 15:21) 'I [myself] will manifest myself to him'; statist se samt in y (L 23:55) 'save thyself and us [two].' In the presence of negation (§18.3b), this accusative se is replaced by the genitive sebe: iny states, all sebe ne možett statistic (Mk 15:31) 'he saved others, but he cannot save himself.'

This true reflexive  $s_{\ell}$  may be replaced by the emphatic or "personal accusative" *sebe*: compare, onto že xotę opravoditi  $s_{\ell}$  sam' reče ... (L 10:29) 'he, wanting to justify himself, said ...'; vy este opravodajoštei *sebe* prědto člověky (L 16:15) 'ye are they which justify yourselves before man'; vozljubiši iskronjaego svoego jako samo *sebe* (Mar ~  $s_{\ell}$  Zo As) Mk 12:31 'thou shalt love thy neighbor as thyself'; sego obrětomo ... glagoljošta *sebe* xrista cěsarja byti (L 23:2) 'we found this man ... saying himself to be Christ the King.'

**21.62** Much more frequent is the  $s_{\ell}$  which has lost the clear meaning and case of the reflexive pronoun and functions more like a particle which imparts some general meaning of intransitivity to the verb.  $S_{\ell}$  may be enclitic and follow the first accented word in the clause, but it usually follows the verb immediately.<sup>11</sup> The  $s_{\ell}$  remains unchanged even in the presence of negation: ašte  $s_{\ell}$  bi ne rodilth (Mt 26:24 Zo Mar Sav; ~ As ašte bi ne rodilth  $s_{\ell}$ ) 'if he had not been born'; ne divi  $s_{\ell}$  (J 3:6) 'do not be surprised'; une est ne ženiti  $s_{\ell}$  (Mt 19:10) 'it is better not to marry'. This fact perhaps allows us to perceive simply intransitivity (and not specific

In East Slavic the particle ся is written as part of the verb, and lexicographers treat ся-verbs as separate entries (ролить 'bear, give birth (to)' ~ ролиться 'be born'). Following this pattern, Russian and Ukrainian scholarly lists of OCS lexical items usually provide two entries. Other Slavs (as well as non-Slavs) include sę-verbs with non-sę verbs; thus SJS puts родити см and рожденъ възти under родити.

reflexivity) in: v $\mathbf{b}$  rizo ne oblačaaše *se* (L 8:27) 'he did not get dressed in clothing.'

In many verbs, the presence or absence of  $s_{\ell}$  indicates intransitivity versus transitivity: azъ umyxъ vaši nozě (J 13:14) 'I washed your feet', but umyxъ  $s_{\ell}$  (J 9:15) 'I washed' (with the context supplying 'my eyes'; this could be taken as reflexive). Some verbs acquire a different meaning with  $s_{\ell}$ : klыnqtъ 'curse' but klыnqtъ sę 'swear, take an oath'. Sometimes a semantic difference is not clear from the available evidence, e.g. plakati ~ plakati sę 'weep; (+ gen. or o + loc.) mourn', although here perhaps the  $s_{\ell}$  adds a note 'for one's own benefit' similar to the dative si (§18.5c). Several verbs never occur without  $s_{\ell}$ : e.g. bojati sę 'fear', postiti sę 'fast', rogajotъ sę 'mock'.

Sometimes it is difficult to distinguish an active reflexive from a passive meaning; in some passages different manuscripts show variants. For example, synt člověčtskt *prědaste sę* (Mt 26:2, Zo) but Mar and As explicitly passive *prědane będete* 'the Son of Man will be betrayed'; ašte ttktmo prikosnę sę rizě ego *sepasena będę* (Mt 9:21, Mar As) but Sav *sepase sę* 'if only I touch his garment, I will be saved'.

The conjunction da plus present (of either aspect) means 'in order 22.1 to', or simply 'to, that': e.g. izide seei da seets (Mt 13:3) 'a sower went forth to sow'; priněse ... děti da rocě vъzložitъ na nje (Mt 19:13) 'they brought children, that he should put his hands on them'; něsmb bo dostoinb da podъ krovъ moi vanideši (L 7:8) 'I am not worthy that you should come under my roof'. The meaning of purpose is made more specific by adding *jako*, see below. The negative may be *da ne* or *eda* (or *da ne kako*, eda kako) 'in order not to, lest'; e.g. moljaaxo i da ne povělitъ imъ vъ bezd $\pm$ no iti (L 8:31) 'they besought him that he should not command them to go out into the deep'; bljudi ubo eda světs iže vs tebě tsma ests (L 11:35) 'take heed therefore that the light which is in thee be not darkness'. The conditional may be used instead of the present: moliše i da bi prěšula отъ prědělъ ixъ (Mt 8:4; Sav has ако да првидетъ) 'they besought him that he would depart out of their coasts'; i drbžaaxo i, da ne bi otaš bla ota  $njix_{\mathfrak{d}}(L 4:42)$  'and they stayed him, that he should not depart from them'.<sup>12</sup>

<sup>&</sup>lt;sup>12</sup> Complete data, with a historical and comparative study, in Herbert Bräuer, Untersuchungen zum Konjunktiv im Altkirchenslavischen und Altrussischen. I. Die Final- und abhängigen Heischesätze (= Veröffentlichungen d. Abteilung f. slav. Spr. u. Lit. des Osteuropa-Inst. a. d. Freien Universität Berlin, Vol. 11), Wiesbaden, 1957.

22.11 In an independent clause, da plus 3rd person present (rarely 1st sg./pl.) may represent an exhortation: da pridets (J 7:37)'let him come'; da svętits sę imę tvoe (Mt 6:9) 'hallowed be Thy name'; samъ o sebě da glagoljets (J 9:21) 'let him speak for himself'; da ne postyždo sę vъ věkъ (Ps 30:2) 'let me never be ashamed'.

22.12 With past tenses, and occasionally with a present, *da* means 'and, and then'.

**22.2** Eda (§22.1) serves also to introduce a question which expects a negative answer: e.g. eda možetъ slěpьcь slěpьca voditi (L 6:39) 'can a blind man lead a blind man?'; eda imatъ xvalq rabu tomu, jako sъtvori povelěnaja? ne mьnjq (L 17:9) 'Doth he thank that servant because he did the things that were commanded him? I trow not.'

22.3 The conjunction *jako* has a very wide range of meaning—'that, so that, for, as, like, since, because'. It is used to introduce both direct and indirect quotations, and sometimes it is difficult to see which is meant. For example: vy glagoljete *jako* vlasvimljaeši zanje rěx*b jako* syn*b* božbi esmb (J 10:36) 'Ye say, Thou blasphemest, because I said, I am the Son of God.' Here the 2nd sg. form shows clearly that the first *jako* introduces a direct quotation, but the statement after the second *jako* could be indirect: 'I said *that I was* ...' Or: Otbvěšta žena i reče emu, ne imamb moža. Glagola ei Isus*b*, dobrě reče *jako* moža ne imamb (J 4:17) 'The woman answered and said to him, I have no husband. Jesus said to her, Thou hast well said, I have no husband.'

22.4 Beside the normal use of the relative pronoun *iže* in an explanatory phrase with a form of *byti* 'to be' (e.g. iže estb otb boga, glagolb božbi poslušaetb J 8:47 'he that is of God heareth God's words'), there are cases where the verb is omitted: e.g. světb iže vb tebě (Mk 6:23) 'the light within you'; w ïwcuht ume otb apumate (Su 447.28) 'about Joseph of Arimathea'; glavy evangelija eže otb luky 'chapters of the Gospel [which is] of Luke'.<sup>13</sup>

The neuter *eže* may be used with an infinitive; e.g. čьto estъ *eže* iz mrъtvyixъ vъskrъsnǫti (Mk 9:10) 'what is [the] rising from the dead?'; *eže* neumъvenami rǫkami ěsti ne skvrъnitъ člověka (Mt 15:20) 'to eat (i.e. the fact of eating) with unwashed hands defileth not a man' нъ дъло кже сътворити дъло (Su 406.8) 'but doing evil [is] evil'.

<sup>&</sup>lt;sup>13</sup> Josef Kurz cites all the examples, *Byzantinoslavica* 7: 336ff.

163

 $E\check{z}e$  is also found as a conjunction because, inasmuch as'. Occasionally  $i\check{z}e$  is found for  $e\check{z}e$ , and vice versa.

## 23. On negation

**23.1** Two morphemes express negation: *ne* and *ni*. *Ne* 'not' stands immediately before the verb or other syntactical unit that it negates. *Ni* expresses denial (the antonym of *ei* 'yes'), or—usually in conjunction with *ne*—'neither, nor, and not, not even'. *Ne* ordinarily occurs only once in a major clause; *ni* may be repeated. A clause containing a negative pronoun or adverb (such as *nikstože* 'no one', *ničstože* 'nothing', *nikyiže* 'no sort of', *niedinsže* 'no, none', *nikoliže* 'never', *niksdeže* 'nowhere', *nikakože* 'in no way') usually has *ne* before the verb; the double negation is presumably normal Slavic syntax. The *ne* is often omitted, however, if *ni* (*ni*-) stands earlier in the clause: this imitates Greek (where, as in English, two negatives signify a positive).<sup>14</sup> Thus *i niktože* ne *dačše emu*. L 15:16, As) 'and no one gave him [anything], Mar *i niktože* ne *dačše emu*. Some examples:

Ne moja volja пъ tvoja da bodetъ (L 22:42) 'not my will, but thine, be done.' Obače ne jakože аzъ xošto, пъ jakože ty (Mt 26:39) 'nevertheless, not as I wish but as you [wish].' Vy čisti este пъ ne vъsi (J 13:10) 'You are clean, but not all.' Ne o vъsěxъ vasъ glagoljo (J 13:18) 'I speak not of you all.' vrěmę moe ne u pride (J 7:5) "My time is not yet come.'

On že reče, ni (Mt 13:29) 'And he said, "No." Ni bratruja bo ego věrovaaxq vu nju (J 7:5) 'for not even his brothers believed in him.' Ni azu tebe osqždajq; idi i otuselě ne sugrěšai ku tomu (J 8:11) 'Neither do I condemn thee; go and sin no more.' Pride Ioanu ni pię ni jady (Mt 11:18 Zo ~ ne... ni Mar) 'John came, neither eating nor drinking.' Něsmu azu xristosu (J 1:27; §16.101) 'I am not the Christ.' Něstu umula děvica (L 8:52) 'the maiden has not died.' Da ne vidętu očima ni razumějqtu subucemu (J 12:40) 'that they should not see with their eyes nor understand with their heart.' Položi e [tělo] vu grobě isěčeně vu nemuže ne bě nikutože nikogdaže položenu (L 23:53) '[he] laid it in a hewn tomb where no one had ever been laid.' Nikomuže ne rabotaxomu nikoliže (J 8:33) 'we have never been in bondage to anyone.'Ni otu edinogože ne može iscělěti

<sup>&</sup>lt;sup>14</sup> Variations in usage are extreme, and many examples are hard to interpret (see note 16 below). Some of the fluctuation must be attributed to dialect variants, both in time and in region. Here only the most general statements are possible. For fuller treatment, see R. Večerka, *Altkirchensl. (altbulgarische) Syntax* I (1989) 33-41, III (1996) 128-140.

(L 8:43 ZoMarSav; As omits ne) 'she could not be healed by any [one of the doctors].'

**23.2** Negation affects case-usage in two types of construction: existential, and transitive (cf. §18.3a6, 18.3b).

23.211 Existential negation combines the impersonal *něsta* (*ne bodeta*, *ne bě*) 'there is no[t] (will not be, was not)' with the genitive: *něsta inogo boga razvě mene* (Su 264.2) 'there is no other god except for me'; ota plati *něsta polazę nikakoęže* (J 6:63) 'from the flesh there is no profit whatsoever'; *něsta člověka* iže živa bodeta i ne sagrěšaeta (Eu 57a10) 'there is no man who will live and not sin'; ne sta nukorome (Su 241.27) 'there was no one there.' *Něsta* with the nominative expresses a negative definition: e.g. ntacta beanko voyao (Su 33.24) 'the miracle is not great'; this construction is normally distinct from *něsta* with genitive, existential negation: e.g. sade tebě *něsta města* (Eu 37a24) 'here for you *there is no* place.'

23.212 The distinction is sometimes blurred. E.g. i něsta sveta razvě tebe gospodi (1Ki 2:2, Sin 2/N) 'there is none holy besides thee, Lord'; něsto bo ničutože taino eže ne avito se (Mk 4:22) 'for there is nothing hid which shall not be manifested'; ивстъ кто милоум, и ивстъ кто милосръдоум (Su 57.9-10) 'there is no one who pities and there is no one who shows mercy'; ивсть никтоже противан са тевъ (Su 232.5) 'there is no one resisting you.' Nikъtože estъ otъ roždenija tvoego iže naricaetъ se imenemь těmь (L 1:61) 'there is no one of your kin who is called by this name.'-The fig-tree which has no fruit because it is not yet the season (cf. Mt 21:18-19) is cursed by Jesus, не вжди къ семоу плодъ отъ тебе (Su 346.19) 'may there be henceforth no fruit from you'; plods could be Gp, but normally it is collective in meaning, and it may be Ns here. Compare Su 350.12 зайе смокъ плода не сътвори 'because the fig-tree has not produced fruit.' Su 345.26 has an unexpected genitive in ne 5th KH BOTEMENE 'it was not its time,' but 351.24 has nominative, NE BT KH BOTMA ILOGT сътворити съмрьтьнъ 'it was not its time to produce deadly fruit.'

**23.22** Transitive negation involves a transitive verb that normally takes an <u>accusative direct object</u>; when the verb is negated (or is subordinate to a negated verb) the object is in the genitive:<sup>15</sup>

<sup>&</sup>lt;sup>15</sup> Indirect objects are not affected by negation; e.g. sodi+ takes dative, whether positive (po zakonu vašemu sodite *emu*, J 18:31, 'judge him according to your law') or negative (otbcb bo ne soditb nikomuže, J 5:22, 'for the father doesn't judge anyone').

Ne ěst ničesože (L 4:2) 'he did not eat anything.' Jako ne věste dbne ni časa (Mt 25:13) 'for you don't know the day or hour.' Ne prěrečet ni vbzbpiet; ne uslyšit nikotože na raspotiix glasa ego (Mt 12:19) 'He shall not wrangle nor cry, neither shall any man hear his voice in the streets.' I prěgrěšixom o vbsemb i zapovědei tvoix ne poslušaxom, ni sbbljusom ni sbtvorixom ěkože zapovědě nam (Daniel 3:29–30, Sin 2/N 26a) 'and we have sinned in everything and not obeyed your commandments and not observed and not done as you commanded us.'<sup>16</sup> Ničuože ix ne vrědit (Mk 16:18) 'nothing will harm them.' I nikomuže ničesože ne rěšę (Mk 16:8) 'and they said nothing to anyone.'<sup>17</sup> Nikogože obidite ni okljevetaite (L 3:14 ZoMar ~ ne obidite AsSav) 'Do not disrespect or slander anyone.'

Ne mogq адъ o sebě tvoriti ničesože (J 5:30) 'I can of myself do nothing.' Ne dostoitъ tebě vъzęti odra svoego (J 5:10) 'It is not proper for you to pick up your bed.' I ne imqtъ česo ěsti, i otъpustiti ixъ ne ědъšь ne xoštq (Mt 15:32) 'and they do not have anything to eat, and I will not send them away hungry (lit. not having eaten).'<sup>18</sup>

23.3 In rhetorical questions, *ne* serves to mark the item that is emphasized: *Ne* azz li vasz dzva na desęte izbzraxz (J 5:70) 'Was it not I who chose you twelve?' *Ne* sz li estz tektonovz synz (Mt 13:55) 'Is not this the carpenter's son?' *Ne* samz li sz otzcz tvoi sztęža tę (Deuteronomy 32:6, Sin 2/N 14b) 'did *not he himself* thy father purchase thee?' *Ne* i mytare li tako tvorętz (Mt 5:47) 'do not even the publicans do so?'

**23.4** Ne serves as a lexical prefix in about 300 stems, bez in about 100. Editors and lexicographers often differ on when to write a space after ne.

<sup>&</sup>lt;sup>16</sup> There are ambiguous or contradictory phrases, e.g. ne sъkryvaite sebě sъkrovištь na zemlji (Mt 6:19 AsSav ~ sъkrovišta ZoMar) 'Do not hide for yourselves treasures on earth'. The form sъkrovišta can be taken as Gs 'a treasure' and explained as a conscious or unconscious change of meaning, or else as an Ap that (probably intentionally) imitates the Greek.

<sup>&</sup>lt;sup>17</sup> The nom. or acc. ničьtože occasionally appears where gen. ničesože is to be expected: Zo nikomuže ničьtože ne гъсі (~ Mar ničesože; Mk 1:44) 'say nothing to anyone.'

<sup>&</sup>lt;sup>18</sup> Note that *ix* is genitive because the infinitive *otspustiti* is governed by the negated auxiliary (§21.11). Compare: i ne obrětsša *ego* ... vsziskajošta *ego* ... obrětete *j* sédęšts ... iskaaxově *tebe* (L 2:45-46, 48) 'and not having found *him* [G with negation] ... looking for *him* [G required by verb] the two of them found *him sitting* [A, direct obj.] ... we [two] sought *you* [G required by verb].' In L 23:2, *sego* obrětoms *razvraštajošta ęzyks* našs 'we found *this* [*man*] *perverting* our nation', the genitive form is dictated by the personal reference (§18.21), cf. obrěte *filipa* (J 1:42) 'he found Philip.'

**23.41** The imperfective *nenaviděti* 'hate' usually takes genitive, while the perfective *voznenaviděti* takes accusative.

23.42 Some examples: něstь nepravbdy vъ njemь (ps 91:16) 'there is no unrighteousness in him'; ne bodi nevěrыль пъ věrыль (J 20:27) 'Do not be without belief but believing'; rabi nedostoini esmь (L 17:10) 'We are unworthy servants.' (Compare něsmь dostoinь, Mt 8:8, 'I am not worthy.') Su 483.12 нъ не не имѣ їс водъ 'but Jesus did not not-have water' is a comprehensible translation of a negated Greek verb 'be in want of, lack for'.

### 23.5 Idiomatic phraseology using ne:

The negated imperative of *mog*- 'be able' + infinitive occasionally appears in the Suprasliensis as a periphrastic prohibition: *ne mozi* mene ostaviti (Su 539.8) '*Do not* leave me' (~ *ne* ostavi mne, ps 37:22 [§4.61]).<sup>19</sup>

The imperative *ne dči(te)* means 'permit': *ne dčite* sixъ iti (J 18:8) 'let these go their way'; не д'влад внанить Sav (Mt 27:49: *ostani* da vidimъ ZoMarAs) 'let us see.'

Prěžde daže ne 'before [a particular moment comes]': sъnidi prěžde daže ne umbretъ otročę moe (J 4:49) 'Come down ere my child die.'

Uže ne 'no longer': Těmže uže něsta dъva пъ plъtь edina (Mk 10:7) 'then they are no more twain, but one flesh.'

Da ne 'in order that not, lest' (cf. 22.1): bljuděte se vraga, da ne nagy s tvorit vy ěko Adama (Eu 97b9) 'watch out for the enemy, lest he make you naked as Adam.'

Ašte ne 'unless': Nikotože ne možetь sъsodъ krěpъkaago, vъšedъ vъ domъ ego, rasxytiti ašte ne prěžde [prьvěe Zo] krěpъkaago sъvęžetъ (Mk 3:27 ZoMar) 'No one can steal the vessels of a strong man, having entered his house, unless beforehand he binds the strong man.'

### 24. Vocabulary and the structure of words

**24.1** The vocabulary of the canonical OCS manuscripts approaches 10,000 lexical items.<sup>20</sup> Some 1500 lexemes are non-Slavic personal or geographic names or derivatives. Foreign stems are adapted to fit OCS inflectional patterns.

<sup>&</sup>lt;sup>19</sup> Su 239.3 не мози не въровати 'do not disbelieve'.

<sup>20</sup> Scholars disagree on just how to write many items, and lexicographers then argue about how to normalize. Thus no one doubts that və and na and sui 'empty, meaningless' and təštə 'empty' are separate, but are there three more "words", vəsue and nasue and vətəšte 'in vain'? SJS allows the first two but "вътъще" is merely a cross-reference to "тъщь". Is mimo an independent adverb 'past, by', or is it a prefix to seven independent verb-stems meaning 'go (carry, run) past'? A different kind of problem is xlqbaję 'begging' (Mk 10:46) in Mar but xlqpaję in Z ~ xlqpati (L 16:3) in both Zo and Mar. Are there two similar stems, or is one merely a spelling error? The rough statistics I provide can only be approximations.

24.11 The exact list is determined by the texts which happen to have survived. The gospels contain short narratives, mostly straightforward, but with cryptic sayings and occult allusions whose meaning depends on traditional religious exegesis. The psalms are poetry, much of it esoteric; the Greek is often obscure, the OCS translation was not particularly good, and the scribes were careless. The prayer-book (Euchologium) is intended for monastic use. The homilies in the Clozianus and Suprasliensis are mostly concerned with explicating details of Christian doctrine, and the saints' lives are placed in societies and geographical locations quite remote from the Slavic world of the ninth century. The OCS translators struggled to reproduce the often elaborate Greek and its richly varied lexicon, but their success was limited—and subsequent copyists failed to transmit the texts accurately.

**24.12** The words we know from this meager set of sources provide a fair sample of basic lexemes and many special terms. Certain everyday items are missing: e.g. nose, daughter-in-law, short, to cough (though post-OCS information guarantees \*noss, \*snaxa, \*kratsks, \*kašeljaj+). Some are not attested in their primary sense, e.g. skovrada 'grill, frying-pan' refers only to instruments of torture (18x). Others are merely implied by derivatives: e.g. govęždina žila 'steer-tendon' (a kind of whip that is contrasted to biče, voštaga, and xrszans) must be based on \*govędo 'steer'. Frequency of individual stems reflects subject-matter: bogs 'god' occurs over 2100 times, božeje 'god's, divine' more than 800x, svęts 'holy' well over 1000x, moli+ 'pray' 500+, molitva 'prayer' 400+—and grěxs 'sin' 300+, běss 'demon' 200+ and dijavols 'devil' 100+.

**24.13** In order to distinguish between words actually attested in canonical OCS (as I define it, \$0.32) and those I believe can safely be attributed to OCS because they are found in specific later copies of certain texts, I will write **\*OCS** (with the asterisk on the label, not the word): **\*OCS** *noss*.

24.2 Meaning is deduced in part from the texts (chiefly Greek) on which the OCS translation is based, and in part from our knowledge of more recent Slavic vocabulary. Problems of interpretation appear at every step.

24.21 Not infrequently a Slavic word apparently has a connotation that the Greek original lacks. Thus the bowl or dish shared by Jesus and his betrayer (Mt 26:23, Mk 14:20) is *solilo*, surely 'salt-container', cf. *solb* 'salt'.

**24.22** Occasionally the meaning that seems obvious for an OCS word or phrase does not fit the Greek text from which it was apparently translated.

A passage in Su, for example, retells an Old Testament parable (Greek II Kingdoms [Hebrew II Samuel] 12:1ff.) about a poor man whose pet ewe-lamb ( $\dot{\alpha}\mu\nu\dot{\alpha}_{5}$ ) is slaughtered by a rich man to make a feast for a visitor. The 'lamb' consistently becomes **TEANUA** (Su 259.29, 260.5, 13), unquestionably the feminine counterpart of **TEANUA** (Su as to why the translator (or an editor) chose to change the species of the animal, but surely lexicographers err if they define *telica* as 'lamb'.<sup>21</sup>

24.23 Sometimes we are faced with a flat error: a translator or scribe has mistakenly omitted, added, or transposed letters or syllables. We must deal with what the scribes wrote, but we should explicitly label any emendation that we have made.

**24.24** A pervasive difficulty is alternate words in the "same" context. Most examples are from the gospels, but parallel texts in the Clozianus and Suprasliensis offer many instructive variants. Some differences surely are stylistic, some are regional, some are historical, and many probably combine these features. Similar variants occur in the translation of the same Greek phrase in different passages. Proper treatment of the OCS material should be in some sort of explanatory dictionary. Here there is room only to sketch general principles.

24.25 Sometimes a Greek borrowing—usually taken as a sign of antiquity—is contrasted with one or more Slavic equivalents. Thus blasphemy  $(\beta\lambda\alpha\sigma\phi\eta\mu\dot{\alpha})$  may be bracebumble or  $\chi\phi\eta\lambda\alpha$  or  $\chi\phi\eta\lambda\mu\mu\alpha$  and the corresponding verb may be vlasvimis-a+ or vlasvimlj-aj+ or translated xul-i+, xulova+, or vasxuljaj+. "Our daily bread" of the Lord's Prayer (Mt 6:11), with the controversial adjective  $\dot{\epsilon}\pi100\sigma105$ , remains  $\chi\Lambda\mu\mu\lambda\mu\mu\mu\mu\mu\mu\mu\mu\mu\mu$  in Vat, but is interpreted as hactabuaaro and in MarSav, hackuphuhu in A (and Sin 2/N), and a phrase beginning hactobuut- that was erased and overwritten by a later cyrillic scribe in Zo.

24.26 Often the contrast is a matter of Slavic derivation; e.g. pastyrjb 'shepherd' MarAsVat ~ pastuxo Sav (Zo uses both); životo 'life' ~ žiznb, konižbniko 'scribe' (usual) ~ konigočijb (rare); raspuncto 'crucify' ~ propuncto. Different roots may supply semantically equivalent forms, e.g. rasxyti+ 'to plunder' ~ razgrabi+, xyštenbje 'plundering' ~ grablenbje, xyštuniko 'robber' ~ grabiteljb.

**24.261** Sometimes a word whose origin is obscure competes with a transparent lexeme; e.g. 'fence' is *xaloga* in L 14:23, but *oplots* in Mk 12:1 (easily associated with o 'around' + *plet*- 'weave' and the type of

<sup>&</sup>lt;sup>21</sup> SJS and SS do just that. SS often, but not always, alerts readers to discrepancies by "!" after the Greek word provided as the source.

fence made by weaving together young branches);  $odr \mathfrak{s}$  'bed, litter' (wide-spread) ~  $lo \check{z}e$  (of more restricted usage; cf.  $lo \check{z}iti se$  'lie down').

Frequently, the competing words are built from native Slavic 24.262 morphemes, and often we can perceive differing semantic nuances. Thus the 'lawyer' of Mk 22:35 is zakononiks in AsSav, but 'teacher of law' zakono-učiteljb in Mar. The general verb 'will stand up' vostajeto of Mt 17:23 in MarSav corresponds to the explicit 'will be resurrected' vaskrusneta in As. In both of these case there are explanations based on Greek words, but it is important to note that within OCS many nearsynonyms change meanings slightly and are redistributed in ways that are not easy to predict. In the dramatic context of Mt 27:23, 'persuaded' is rendered naučišę 'instructed' in As, naustišę 'urged, goaded' in ZoMarVat, and navadiše 'lured, enticed' in Sav. The 'flood' of Mt 24:38 and 39 is potops both times in As, but merely voda 'water' in v. 39 in ZoMarSav. Yet different formations with apparently identical meanings are not unusual. The noun věra 'faith' is opposed to nevěrbje, nevěrbstvo, nevěrustvuje and bezvěruje 'lack of faith'. Religious concepts do not always have stable definitions; 'sin' is always grex- (grexs 'sin', gressmiks, grěšыnica 'sinner [m, f]', grěšылъ 'sinful', bezgrěšылъ 'sinless', grěxovылъ 'pertaining to sin'), but the distinctions between the verbs *pogrěšiti* and prěgrěšiti 'to sin, err' are hard to discern. The 'forgiveness (remission) of sins' varies: in MarSavVat Mt 26:28 vs otsdanbje grěxoms [Vat rotrost] ~ vo otopuštenoje grěxomo Zo ~vo ostavljenoje grěxomo As.<sup>22</sup>

24.3 The external history of OCS presumes ninth-century beginnings that involve possible eastern Bulgarian Slavic dialects near Constantinople (perhaps also in Bythinia, in Asia Minor), the Macedonian dialects near Saloniki, and western dialects (Morava and Pannonia), followed by developments more specifically tied to the Macedono-Bulgarian lands in the tenth and eleventh centuries. In the process, some dialect words or forms became normal or even obligatory (e.g. *absje* 'at once';  $az_{5}$  'I'

<sup>&</sup>lt;sup>22</sup> In Mk 4:1, L 1:77, 3:3, 24:47 otspuštenie and ostavljenie appear with different distributions. The scribe of Zo put both in L 1:77, vs ostavljenie vs otspuštenie grěxs našixs. We may speculate that otsdanse (which otherwise signifies 'repayment, retribution') in this expression is a reminiscence of Old High German vergebnis, while the other two imply sending away or leaving alone, both possible for čφεσις. This Gk noun occurs twice in L 4:18, "preach deliverance to the captives" and "set at liberty them that are bruised"—OCS otspuštense, but otspustits sskrušenyę vs otsradą. Otsrada is possibly a Moravism, but any classification has to be guesswork.

[probably a local eastern Bulgarian shape, although *jazs* was surely universal elsewhere]), while others were gradually replaced (e.g. *ašuts* 'in vain' ~ *vasuje*, *vstsšte; balbji* 'doctor' ~ *vračb*), and still others survived in specific contexts. Thus *rěsnota* 'truth' is well attested in the Psalter, along with its synonym, *istina*, but only the latter appears in gospels (although evidence from post-OCS gospel mss allows us to suspect that *rěsnota* was present in early copies and eliminated by conscious scribal intervention).

24.31 Readers and scribes easily tolerated alternate expressions. Thus "Lo, the hour is coming" of J 16:32 in Sav is both ce rpagets vace (in the lection for the 7th Thursday after Easter, ed. p. 5) and ce hat roghna (for the Thursday before Easter, ed. p. 98). ZoMarAs have se greates godina. We can confidently surmise that rpagets represented a familiar everyday verb for Slavs from what is now western Croatia and Slovenia or from Macedonia, but a bookish equivalent of hat roghts in eastern Bulgaria. The nouns vace and roghna (and also roge), on the other hand, probably had varying meanings in different regions.

The shifting distribution of competing lexemes shows that no 24.32 single manuscript embodies a consistent redaction. The general usage seems to be Macedonian, partly---it is assumed---reflecting the original translation of most of the texts, and partly reflecting the normalizing influences presumably exercised by scribes and editors throughout the tenth century. Yet the major manuscripts have words deemed by modern scholars to be typical of eastern Bulgarian usage, the effect of a standard assumed to have been developed in Preslav and perhaps other cultural centers. It is important to keep in mind that there is no OCS manuscript that exemplifies the hypothetical eastern or Preslav redaction of OCS. The chief works that supposedly were translated in this Bulgarian cultural zone have survived only in post-OCS manuscripts, and to retrieve their original wording is a complex task fraught with difficulties.<sup>23</sup> Much of the scholarship in this field is fragmentary and the assumptions and premises of individual investigators are often so different that results cannot be compared. Here there is space only to remark that there are OCS words generally agreed to be ancient and/or "western" (from the Moravan Mis-

<sup>&</sup>lt;sup>23</sup> Parts of the Suprasliensis are believed to represent Preslav translations. Other probable examples are the *izborniki* of 1073 and 1076 (both copied in Rus') and the *Hexameron* (Шестоднев) by John the Exarch, known from a Serbian ms of 1263 and later ESI copies.

24.32-24.42

sion), and others characterized as "eastern" or Preslavian. The majority of OCS lexemes, including most synonyms, are neutral.

\_\*\_

24.4 Most OCS words are complex (§3): stem + desinence. The stem must contain a root; it may contain one or more affixes.

Stems with more than one root are called *compounds*.

Canonical OCS includes about 2000 roots (some of which surely were felt as foreign) and a store of affixes that can be called native Slavic.

To the prefixes listed in §5.31 may be added *ne* 'not, un-', *bez* 'without', *meždu* 'between', and *protivs* 'against'. Suffixes are more varied; here we will deal chiefly with morphemes—numbering about 60—that seem to form relatively new words.

**24.41** A stem must be verbal, substantival or adjectival. The meaning is arbitrary, although usually it is partly predictable from the sense of the root as supplemented and/or limited by each affix. When a root serves as a stem, we may posit zero-affixes to specify gender and declension. Thus, the root in sal-a+ 'send' serves as a substantive-stem sal-a 'envoy', which presumably is lexically marked as masculine, animate, and personal. The root *ljub* in *ljubi+* 'love' functions as an adjective stem, *ljuba ljuba ljuba* 'dear'. The root zal 'bad' can serve as an adjective (zala zalo zala) or, with a change of declensional marking, a substantive, zalb (i-stem fem., 'wick-edness, evil'). The root čed underlies the neuter substantive čedo 'child' and the feminine collective čedb 'servants, companions, household members, retinue'.<sup>24</sup>

**24.42** Some 620 stems are compounds: e.g.  $\dot{c}ed$ -o-ljub-5 'child-loving' is an adjective—the root  $\dot{c}ed$  joined by {o/e} to ljub.<sup>25</sup> Vodonos5 (m) and vodonos6 (f., i-stem) 'water-pot, vessel for carrying water'; cf. the verbal root in nos-i+ 'carry' and the substantive vod-a 'water'. Zero-affixes are

It is important to keep in mind the practical results of differing decisions about analysis. Adjectives, including passive participles, may function as substantives (slěpð 'blind; a blind man'; zolo 'bad [neut.]; evil, a wicked act or intention'; posolanð 'sent; one sent, envoy'). Are such items to be entered in a dictionary as separate entries, or as subheadings, or can they be assumed to be predictable from the Nsm or basic-stem information? Negated passive participles easily are interpreted as having a sense of potentiality: e.g. ukrot-i+ 'to tame' ~ ukroštenð 'tamed, domesticated' ~ neukroštenð [or ne ukroštenð] 'untamed; untamable, uncontrollable'. Two words or one? Is a theoretical stem cěsar-ova+ sufficient to identify Npm necěsarujemi as 'ungoverned, free from rule' (a sense obtained from the Greek text)?

<sup>&</sup>lt;sup>25</sup> If the first member is a numeral, it may itself consist of two roots: dъvoj-e-nadesęt-e-luč-ьп-oje slъпьсе 'the 12-rayed sun' Su 231.2 (cf. luča 'ray'; §20).

assumed to specify that the compound stem is adjectival or substantival (and provides the declension type). Either root, or both, may be affixed, e.g.  $\check{c}edoljubiv$ <sup>5</sup> 'child-loving',  $(za-kon)-o-(u\check{c}-itelj)-b$  'law-teacher; law-yer', cf. zakon<sup>5</sup> 'law' and u\check{c}itelj<sup>5</sup> 'teacher' (u\check{c}-i+ 'to teach').<sup>26</sup> As a rule, the first root is semantically subordinate to the second.<sup>27</sup>

24.431 Most substantives and adjectives are stems that must be listed in the lexicon even though many are made up of recognizable elements. For instance, danb 'tax, tribute' and dars 'gift' obviously share meaning with the anomalous verb dati 'to give' and suggest a segmentation da-n-b(whereby the -n- is lexically marked for feminine gender and i-stem inflection), and da-r-b (masculine o-stem). Within OCS, however, the morphemes must be dan- (simple decl., fem.) and dar- (hard masc. twofold decl.) and the verbal root {da[d]} (see §16.21); they are part of the knowledge that was, so to speak, recorded in the memory of speakers of OCS. In terms of stem analysis, dar and dan are OCS roots.

**24.432** The substantive *darovanbje* 'donation, donating', on the other hand, is {(dar-ova+n)-bj+e}: the root *dar* is converted into a verb by the *ova* classifier (\$15.5), made a past passive participle by the suffix *n*, and a verbal substantive by the formant *bj* (cf. \$12). This process of formation is productive, given the basic morpheme *dar*; the meaning 'act(s) of donating' is expected for the formation. In grammatical terms, this verbal substantive is part of the morphology of the verb *dar-ova+*, and it is predictable in terms of normal suffixes and productive rules. This kind of derivative easily acquires a new meaning—here 'thing(s) donated'—and lexicographers must decide whether it deserves a separate entry in a dictionary.

**24.44** Analysis of words rests on a knowledge of prefixes, desinences, and the major derivational suffixes. By removing the desinence, one finds the stem. The premise that the root must be of the shape (C)VC (less often CVCVC), makes segmentation of morphemes easier. For example, *na*  $v \sigma z g l a v \sigma n i c$  i on a pillow' (Mk 4:38, Mar) obviously is Ls of  $v \sigma z g l a v \sigma n i c$ , made up of a prefix  $v \sigma z$ , a suffixal sequence *-bnic-* and a root glav,

<sup>&</sup>lt;sup>26</sup> The term compound here does not include prefixed stems like za-kon-, although it is obviously a lexical unit that enters into other complex stems, e.g. vozakoni+ 'establish as law', bezakonova+ 'behave lawlessly'. Another kind of complex stem consists of root plus suffix, e.g. kon-bc-b 'end'; beskonbčbnb 'endless', nedokonbčaj+.'fail to complete'.

<sup>&</sup>lt;sup>27</sup> See also §24.4422 on calques.

NOTES ON SYNTAX

173

approximately "(alongside-head-substantive)-desinence"—a transparent parallel to the Gk ( $\pi\rho\sigma\sigma$ -  $\kappa\epsilon\phi\dot{\alpha}\lambda$ -  $\alpha\iota$ )-  $\sigma\nu$ . In contrast, Zo has *na doxstorě*, surely Ls, but of indeterminable gender. We might take *do* as a prefix, but *xstor* remains isolated; *doxstor*- is synchronically an indivisible unit.<sup>28</sup>

**24.441** In Su both *štuždekromonica* and *jataxulonica* stand for  $\xi$ evodoxeñov 'hospice, guest-house; guest refectory'. The first is transparently *štužd-* 'alien' + *krom-* 'feed' + *onic* (suffix) + *a* (desinence), similar to Gk *ksen* 'alien/guest' + *dox* 'receive' + *ei* (suffix) + *on* (desinence). The second clearly has suffixal -*onic-*, but *jataxul* is a sequence that fits no Slavic patterns. It may include two roots, or it may parallel another OCS word, *gostinica* 'inn', where the root *gost* 'guest' is followed by -*in-ic*, a complex suffix.<sup>29</sup>

24.4421 Many OCS compounds are simplistic calques of Greek words. Roman Palestine had regional officials called "quarter-rulers" in charge of one fourth of a larger region. The title *tetrarch* (τετραάρχης) and the verb *rule as tetrarch* (in Gs participial form, τετραρχοῦντος) occur in the gospels. The adapted loan *tetraarxs* (Zo; *tetrarxs* Mar) remains in Mt 14:1, but is replaced by *četvrstovlastscs* in L 9:7 and *četvrstovlastsniks* in L 3:19; cf. *cetvrst-s* '1/4' and *vlast-s* 'power, dominion'. The verb is *četvrstovlaststv-ujoštu*. Mt 23:23 has the verb "you tithe" ἀπο-δεκατ-οῦτε, rendered in Mar accurately as *otsdeset-sstv-uete*. (A more natural equivalent appears in L 11:42, *desetino daete*, lit. 'you give a tenth'.)

24.4422 Calques are usually subject to native constraints. Thus *ljuboništb* 'loving the poor' is a calque of  $\varphi_1\lambda\phi$ - $\pi\tau\omega\chi$ - $\varphi_5$ , but has the dominant root first and is therefore peculiar; *ništeljubsje* converts  $\varphi_1\lambda\phi$ - $\pi\tau\omega\chi$ - $(\alpha$  to the Slavic pattern. Gk *miso*- 'hate' has no single equivalent; it caused trouble: *nenavidęi člověka dsjavols* 'the devil who hates man' for  $\phi$  µισάνθρωπος διάβολος, *nenavistsniks člověkoms* 'misanthrope'. *nenavistje bratsnje* 'brotherly hate' and *bratoljubsstvije nenavistsnoe* 'hateful brotherly-love' and *bratije neljubsstvo* 'brothers' not-loving' for µισαδελφία.

24.5 The list of clearly productive nominal and adjectival formants is modest. The dominant shape is VC. Front-vowel formants entail KI-palatalization of stem-final velars and c/3. Some important suffixes begin with *j* and therefore trigger iotation (§3.6). A few begin with *t* and interact with a preceding obstruent (§3.3131).

<sup>28</sup> SS imprudently lists it sub Abybroph, guessing at gender and phonology.

<sup>&</sup>lt;sup>29</sup> Space precludes discussion of the hierarchies of cohesion or closeness of the morphemes in various structures. Questions of the immediate constituents of stems and affixes can only be hinted at. I suggest that gostin is base for gostin-ic-q (ZM L 10:34) and gostin-bnic-q (As); the gospod-q of Sav represents a different lexeme, gospoda. The -in- is presumably possessive, as in zvěrins 'beast's', cf. 24.84. In L 10:35, 'host, inkeeper' is Ds gostin-bnik-u in ZMA and gost-bnik-u in Sav. Semantically apart is another derivative, gostin-bc-b 'highway, main street'. There is also a verb, gost-i+ 'be host, receive guests'.

**24.501** Many stems combine adjectival and substantival suffixes: thus tum 'dark' underlies tuma 'darkness', the adjective tum-un-tum 'dark', tumunica 'prison', tum-unic'-un-tum' pertaining to prison', tumunuic'-unic'-un-tum' pertaining to prison', tumunuic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic'-unic

24.51 Suffixes may be simply a lexical component of the stems; they seem to contribute nothing to the meaning: *ot-wc-b* 'father', *slan-wc-e* 'sun', *ov-wc-a* 'sheep', *pěs-ak-a* 'sand', *braš-wn-o* 'food'.

24.52 Stems denoting humans, male and female, often lack any marking, but many contain special suffixes. Rabs m. and raba f. 'servant, slave' (and sluga m. 'servant'), for example, are minimal stems, while rab-ynj-i (§4.18) has a suffix that, with a root indicating person, specifies female. Female reference is often supplied by a suffix that is absent from the masculine equivalent: e.g. proroks 'prophet' ~ proroč-ic-a 'prophetess', sosěd-s m. ~ sosědynji 'neighbor', vladyka m. 'lord' ~ vladyčica 'lady, female ruler', otroks 'boy, child, servant' ~ otrokovica 'girl', vratarjь m. ~ vratarjica f. 'gate-keeper' (cf. vrat-a [n. pl] 'gate, gateway'). Sometimes a masculine suffix is replaced by a feminine one: e.g. star-ьс-ь 'old man; monk' ~ star-ic-a 'old woman'; dvьrьniks m. ~ dvьrьnica f. 'door-keeper' (cf. dvьri [f. i-stem pl.] 'door'), močeniks m. ~ močenica f. 'martyr'.

24.52 Many words denoting male agents are made with five suffixes: -telj-, -arj-, -čbj-, -bnik-, and -bc-.

**24.521** The suffix *-telj-* signifies *human* (or divine) *actor* (about 50 examples). It is added directly to a basic verbal stem:  $vlad-\emptyset+$  'rule' > vlasteljb 'ruler, lord'; *prij-aj+* 'favor' > *prijateljb* 'friend'; svved-e+ 'witness' > svedeteljb 'witness'; služ-i+ 'serve' > služiteljb 'server, servant' (*idoloslužiteljb* 'servant of idols, one who serves idols').

An extended form *-itelj*- is used with some exceptional stems: *sspas*- $\emptyset$ + 'save' ~ *sspasiteljb* 'savior'; *zbd-a*+ (§15.643) 'build' ~ *zižditeljb* 'creator'; *po-da(d)*- (§16.28) 'give' > *podateljb* and *podaditeljb* 'giver'.

<sup>&</sup>lt;sup>30</sup> The Gk has simply Gp 'of murderers'. The translator chose not to use either of the available nouns, *ubbjbca* or *ubojbca* 'murderer' m., a-stem. *Peštera* 'cave' is nearly synonymous with *peštb* (f., i-stem), but *peštb* also means 'stove'. The suffix -*er*- is peculiar to *peštera*; it therefore makes the sense of the stem more precise; see §24.51.

Two feminine equivalents are attested: *služiteljunica* 'servant' (cf. *raba*, *rabynji*) and *roditeljunica* 'parent' (~ *rod-i+* 'beget, bear', *roditelju*'parent').<sup>31</sup>

24.522 The suffix -arj- (8 stems) also denotes a human actor: e.g. rybarjb 'fisherman' ~ ryba 'fish'; gronbčarjb 'potter' ~ \*gronbcb 'pot'. (The loanwords cěsarjb 'king' (L caesar 'Caesar') and oltarjb 'altar' are simple stems in OCS; cěsarjica 'queen' is obviously a new word.)

24.523 Masculine a-stems with the suffix -čbj- (§4.16) denote male persons (and contain non-native roots): e.g. konjigočiji 'scribe' (= konjižoniko, cf. konjigy [pl tant] 'book'), šaročiji 'painter' (=šaropisateljo, cf. šaro 'color').

**24.524** The suffix *-enik*- $\mathfrak{d}$  or *-bnik*- $\mathfrak{d}$  is found with over 165 stems denoting male persons (and about ten with other meanings, see 24.7 below).<sup>32</sup>

-enik- is generally associated with a past passive participial stem (§11.23), e.g. kromljeniko 'nurseling' (cf. kromi+ 'feed, nurture'), blaženiko 'happy man' (cf. blaži+ 'deem happy'), kaženiko 'eunuch' (cf. kazi+ 'spoil'). Note that ljubljeniko 'lover, admirer' has an active sense, while vozljubljeniko 'beloved [man]' fits the typical pattern (cf. voz-ljubi+ 'love').

-bnik- generally occurs with stems that are nominal rather than verbal, but the meaning often implies actor. Thus pobědbnik<sup>5</sup> 'victor' is based on poběda 'victory', while its synonym poběditeljb is associated with the verb pobědi+ 'conquer, prevail' (cf. §24.52 above). A pěnęžbnik<sup>5</sup> 'moneychanger' deals with pěnę3<sup>5</sup> 'money'; a praštbnik<sup>5</sup> 'slinger' is presumably proficient with a \*prašta 'sling'.

**24.525** The suffix -bc-b in about 25 stems based on verbal roots refers to persons: e.g. kupbcb 'merchant' (kupi+ 'buy'), tvorbcb 'maker' (tvori+ 'make, do'), zbrbcb 'priest' (zbr-O- 'sacrifice'), skopbcb 'eunuch' (skopi+ [se] 'castrate [self]'); cbtbcb 'reader', bogocbtbcb 'pious, god-honoring man' (cbt-O+ 'read; honor'); pri-bbl-bc-b 'immigrant; convert' (cf. participle pribbl- 'arrived', §11.222). A-stems with -bc-a (e.g. secbca 'executioner', cf. u-sek-(nq)+, §15.771) may well have been feminine as well as

<sup>&</sup>lt;sup>31</sup> Some synonyms: vladyka ~ vlasteljb, sopaso ~ sopasiteljb, zbdateljb ~ sozbdateljb ~ zižditeljb, davbcb ~ dateljb, samodrbžbcb ~ samodrbžiteljb. There were surely differences in meaning, but we can only speculate what they were.

<sup>&</sup>lt;sup>32</sup> The only evidence for an independent suffix -ik- is Adu zlatika '[two] gold-pieces' (Su 145.25). It is vitiated by the fact that zlatica occurs 11 times in the same meaning in the same text. Compare surebrunks '(silver) coin, money' and mědunica '(bronze) coin'.

masculine; some of them may have had a pejorative sense (as they do today).

**24.5251** Compounds are not always felicitous: Rod-o-tvor-bc-b 'creator [of generation, origin]' in Wisdom 13:5 (Su 534.4) mimics  $\gamma eve\sigma = 00\gamma \phi s;$  English translators overlook the first root. Samodrbžecb (and samodrbžiteljb) 'autocrat' (cf. drbž-ě+ 'hold, rule'—calques involving the Greek root krat) fit OCS patterns well, but samo-vlast-bc-b (like četvrbtovlastbcb) contains the substantive-stem vlast-b 'dominion'.

**24.526** Stems with the suffix *-ěn-* or *-jan-* (usually in plural, Np *-e*) signify a group of human beings, see §4.53. The singulative *-in-* emphasizes the individual: e.g.  $kr \omega t i j ani \sigma = kr \omega t i j ani \sigma$  'Christian': e.g.  $vojin\sigma$  'soldier' ~ voji pl (rarely vojini) 'army'.

24.53 Female persons (or divinities) may be designated by suffixes that also occur with non-personal senses:

-ynj-i (§4.18): e.g. bogynji 'goddess' (cf. bogo 'god'), solunjanynji 'Thessalonian' (soluno 'Saloniki', cf. solunjane 'Thessalonians'); magdalynji 'Magdalene, woman of Magdala'.

-*ic-a* occurs in about 100 stems, only 24 of which denote female humans. Eight in -*n*-*ic*- have corresponding masculines, e.g. *učenica* 'pupil, disciple' ~ *učeniks*; *plěnsnica* 'captive' ~ *plěnsniks*. But while *starica* means 'old woman' (cf. above), *junica* means 'heifer, young cow' (cf. *juns* 'young', *junscs* 'young bull') and *črsnica*—transparently based on the adjectival root *črsn*- 'black'—is (1) 'monastery' or (2) 'mulberry tree'. See also §24.7 below.

**24.6** Young males may be indicated by  $-i\delta t - \delta$ , : e.g. mladeni $\delta t \delta$  'baby boy' (~ mladen $\delta c \delta$  ~ mladet $\delta c c$  neut.), kozbli $\delta t \delta$  'kid' (= kozble, cf. kozbla 'goat'), p $\delta t \delta t \delta t$  'baby bird' (cf. p $\delta t c \delta t$  'bird' [apparently without differentiation of sex]).<sup>33</sup> Less specific is -et-, e.g. agne 'lamb' (~ agnb $c \delta$  or jagnb $c \delta$ ), see §4.414-415.

**24.7** The chief suffixes in nouns that denote objects, acts, and abstractions are: with masculine gender, -bnik-, -bc- (15), -bk-bc (17); with feminine gender, -ynj-i, -ic-a, -bnic-a; -ost/-est-b (45), -ot/et-a, -bb-a (14), -tv-a (10), -in-a (30); with neuter gender, -bc-e (8), -bj-e (45 + 730 -nbj-e), -bstv-o, -išt-e (30), -l-o (25). The suffix -j- (which entails KI or iotation, §3.6) is used in a few masculines and neuters and more than 30 feminines. Some examples:

-bnik-: svěštbnika 'candle-holder' (= svěštilo), cf. světa 'light'; votorbnika 'Tuesday' (votora 'second'); sorebrbnika 'coin' (sorebro 'silver')

<sup>&</sup>lt;sup>33</sup> Su 513.2 младеньци како и птишти corresponds to the Gk 'the young of sparrows'.

- -ьс-ь: diminutive/hypocoristic: odrьсь (odrъ 'bed, pallet'), kovьčežьсь (kovьčegъ 'box, ark'); other: studenьсь 'spring, source' (stud-enъ 'cold').
- -ok-: pribytoko 'profit, gain' (pri-by-ti 'arrive, be successful')
- -ynj-i pustynji 'desert' (pusto 'empty'), grodynji 'arrogance' (grodo 'proud')
- -ic-a: šujica 'left hand' (šujb 'left'), mantijica 'cloak' (\*mantija 'mantle')
- -ыліс-а: košыnica 'basket' (= košы m.); dvыrылісе [plur tantum] 'door'
- -ost-ь: jarostь 'fury' (jarъ 'furious'), dobrostь 'goodness' (= dobrota, cf. dobrъ 'good'), bujestь 'stupidity' (= bujьstvo, cf. bujь 'foolish'),
- -ot-a: pravota 'justness, justice' (= pravostb = pravynji = pravbda), sujeta 'emptiness, vanity' (sujb 'vain')

-ьb-a: svętьba 'consecration' (svęti+ 'consecrate'), alčьba 'hunger, fasting' (alk-a+ 'hunger'), tatьba 'theft' (= tatьbina, cf. tatь m. 'thief')

- -tv-a: žętva 'harvest' (žьnj-Ø+ 'harvest' §15.83), molitva 'prayer' (cf. molьba 'petition'; moli+ 'pray, request')
- -in-a: desętina '1/10' (desętь 'ten'), konьčina 'end' (= konьсь, konьčanьje, sъkonьčanьje), maslina 'olive' (maslo '[olive] oil'), xramina 'house, building, dwelling' (= xramъ), udavljenina 'meat of strangled animal' (u-dav-i+ 'strangle')<sup>34</sup>
- -ьс-е: diminutive/hypocoristic, e.g. čędьce ~ čędo 'child', iměnijьce ~ iměnьe 'property'; shifted meaning, plesno 'sole' ~ plesnьce 'sandal'
- -bj-e: abstracts; veselbje 'joy' (veselb 'joyful'), obilbje 'abundance' (obilb 'abundant'), mil-o-srbd-b 'merciful' (milosrbdbje 'mercy'); noun from phrase, podbgorbje 'foothills' (cf. podb gorami 'below the mountains'), bezumbje 'foolishness' (cf. bez uma 'without mind'); collectives trupbje 'corpses' (trupb, see §4.1022).
- ωstv- ωj-e is semantically equivalent to ωstv-o, e.g. veličωstvo = veličωstvoje 'greatness' (= veličωje, cf. veliku 'great'), cěsarωstvo = cěsarωstvoje 'kingdom' (cěsarju 'king'), otučωstvo 'fatherland' = otučωstvoje (= otučina, cf. otucu 'father').

<sup>&</sup>lt;sup>34</sup> This word is part of the religious code inherited from the Old Testament. Although OCS has no other example, we can be certain that *-ina* 'meat of' could be used with the name of any animal, e.g. *teletina* 'veal', *konjina* 'horse meat'; it is a productive suffix in most modern Slavic dialects. The *Izbornik* of 1073 has *kelpina* probably 'swan meat'.

- -ustv-o: světulustvo 'brightness' = světulosta = světulota (světula 'bright'; světa 'light'); besamrutustvo 'immortality' (= besamrutuje, cf. bez without', samrutu 'death'); episkupustvo 'bishopric, office of bishop' (episkupa 'bishop')
- -išt-e: place, sonsmište 'synagogue, sanhedrin' (son-son-Ø+ 'gather together'), pristanište 'harbor, haven' (pri-sta[n]- 'arrive' §15.712), blędilište 'brothel' (cf. blędsnica 'prostitute')
- -j-b: voždb 'leader, guide' (= vod-i- 'lead'), plačb 'weeping' (plak-a-'weep'),
- -j-e: qže 'rope, bond' (qza 'bond, fetter'; qz-sk-s 'tighty, narrow')
- -j-a: volja 'wish' (vol-i+ 'wish'; vel-ě+ 'command', lžža 'lie' (lžga+ 'lie'), duša 'soul' (duzz 'spirit, breath'), svěšta 'candle, lamp', mežda 'boundary' (see §26.21, n. 8). prědzteča 'forerunner' (tek-Ø+ 'run')
- -l-o: počrьpalo 'dipper' (= počrьpalьnikъ;{čыгр} §15.643); svętilo 'sanctuary'(= svętilište; svętъ 'holy'); světilo 'lamp, light' (světi+ 'illuminate'); kadilo 'incense' (kadi+ 'burn incense')

**24.8** Adjectival root-stems (e.g. *pusts* 'empty', *šujb* 'left', *vesels* 'glad') number about 130. Most stems end in one of ten suffixes: -bn- (about 750), -bsk- (80+), -en- (about 12), -iv- (170+), -it- (about 15), -av- (5); -bnj- (with adverbial bases, 35+); and possessives with -in- (about 12), -j-, -ov- (25+) and -bnj- (with proper names).

24.81 The suffix -wsk- in principle means 'characteristic of' as opposed to the very general -wn- 'pertaining to': gradwscii moži 'the men of/from the town' ~ gradwnye stěny 'the walls [that are part] of the town'. In practice, the two may become synonymous. Thus nebeswska, referring to nebesa 'heavens', is exclusive in Zo and Mar, but only nebeswna is in As, while Sav has both.<sup>35</sup> Both formants are used with borrowed stems, e.g. aera ~ aerwna 'air'(cf. vazduxa ~ vazdušana 'air), manixa 'monk' ~ manišwska (cf. črunuch 'monk' ~ črunuckska).<sup>36</sup> A compound suffix -ovun--evun- seems to have about the same meaning, e.g. duša 'soul' ~ duševuna, duxa 'spirit' ~ duxovuna, věra 'faith' ~ věrovuna 'of faith' and věruna 'faithful'. Similarly, adovwska, adovuna, and adwska all mean 'of hell' (cf. ada 'hell, Hades'). In many instances, wash-forms function as possessives.

<sup>35</sup> SS defines both OCS words as "небесный, nebeský", that is - $\omega$ - has survived in R., -*sk*- in Cz.

<sup>&</sup>lt;sup>36</sup> *Monixo* must represent an adaptation of a Germanic form \**munīx-*, cf. mod. *Mönch*.

24.82-24.84

**24.82** The suffix  $-\check{e}n$ - means 'made of' the material denoted by the base; -it- usually signifies 'characterized by'; -iv- and -av- may be somewhat emphatic. Some examples:

- -čn-: lыněnъ 'linen' (\*lылъ 'linen'); měděnъ 'bronze' (= mědылъ; mědы 'bronze'); trыněnъ 'of thorns' (= trыnovъ; trылъ 'thorn'); moždanъ 'of marrow' (\*mozgъ 'marrow; brain')
- -it-: monogoočito 'many-eyed' (monog- 'many', ok-o 'eye'), znamenito, imenito 'famous' (znamenoje 'mark, sign, signal'; \*jomę 'name'); plodovito 'fruitful' (plodo 'fruit'; cf. also monogoplodono)
- -iv-: nedoživo 'unwell, sick' = nedožono (cf. nedogo 'ailment') pravodivo 'just' = pravodono (cf. pravo 'straight, authentic, just')
- -*liv*-: mlsčalivs 'silent, quiet' (= mlsčalsns, cf. mlsk-e+ 'be silent')
- -bliv-: nerazumblivo 'unintelligent' (= nerazumivo, nerazumičono, nerazumbničono; cf. razumo 'understanding')
- -av-: lǫkavъ 'crooked, foul, deceitful' (= lǫkavъnъ, cf. lǫka 'deceit, trickery'); sědinavъ 'grey-haired' (= sědъ, cf. sědiny f pl 'grey hair')

**24.83** The formant  $-(\check{s}b)nj$ - makes adverbs into adjectives; there are often variants, presumably from regional dialects: *iskrunjujb* or *bližunjujb* 'nearby, closest (one's neighbor)' (*iskru* 'close'; *bliže* 'closer'), *domašunjujb* and *domaštunjujb* 'domestic' (*doma* 'at home'), *nynjašunjujb* and *nynjaštunjujb* 'contemporary' (*nynja* 'now'). These adjectives are usually definite in form.

**24.84** Possessive suffixes are widely used: -j- (-ij- with monosyllabic stems) and -ov- go with twofold o-stems, -in- with most other stems, and -nj- largely with kinship terms:

- -ov- jugova 'of the south wind'(juga 'south, south wind'; južeska 'of the south'), kitova (kita 'whale'), ženixova (ženixa 'bridegroom'), sapasova and sapasiteljeva (sapasa, sapasitelja 'savior'), zmejeva (zmeje 'dragon'); cěsarjeva 'royal, of kings; imperial, of emperors' = cěsarje = cěsarjeska; pilatova and pilašte (pilaštaja 'Pilate's wife';<sup>37</sup> pilata 'Pilate')
- -ij- lbvijb 'of the lions', lbvovö 'of the lion', lbvbskö 'leonine'; kravbjb (\*krava 'cow'); rabijb 'slavish, servile' (= rabbskö, rabö 'slave')

<sup>&</sup>lt;sup>37</sup> The long femine possive adjective from a man's name denotes his wife; this is the only example in OCS, but the usage is securely attested in early Rus'.

NOTES ON SYNTAX	24.84-24.9

180

- -j-: ovenje (ovens 'ram'), oveče (oveca 'sheep'), avelje (avels 'Abel'), igumenje (igumens 'hegumen, abbot'), inorože (inorogs 'unicorn'), solomonje = solomonove (solomone 'Solomon')
- -nj-: gospodьnjь, vladyčьnjь (gospodь, vladyka 'lord'), družьnjьjь (drugs 'friend'), větrьnjь (větrs 'wind')
- -in-: julijanins (julijana 'Juliana'), bogorodičins (bogorodica 'theotokos, mother of god'), zmujinu (zmuja 'serpent'), igulinu (igula 'needle'), neprijazninu (neprijaznu 'enemy'), sotoninu devil's (sotoninusku 'devilish'; sotona 'Satan'); osulętinu (osulę 'kid')

**24.9** A number of important nouns contain the unproductive suffix -t-b (fem. i-stem): vlastb 'rule, domain' (~ oblastb; vlad- $\emptyset$ + 'rule'); slastb 'pleasure, luxury' (sladoko 'sweet'); věstb 'news' (věd-ě-ti 'know'); pověstb 'story, tale' (pověd-ě-ti 'recount'); somrbb 'death' (mbr-qtb 'die')<sup>38</sup>, nenavistb 'hate' (= nenavistbje, nenaviděnbje; nenavidě+ 'to hate'), zavistb 'envy' (= zavida, zavistb; zavidě+ 'envy'), čbstb 'honor' (čbt- 'to regard, honor'); peštb 'stove; cave'; noštb 'night'; pomoštb 'help' (pomog- $\emptyset$ + 'to help'); zabytb 'oblivion (= zabytbje; zaby-ti 'forget').

<sup>&</sup>lt;sup>38</sup> The prefix so here is not 'with; down from' (IE \*sun, §29.815) but IE h<sub>1</sub>su- 'good' (cf. Gk \*esu- > eu- εὐ, e.g. in εὐαγγέλιον 'good message, gospel'). It survives in a few Sl stems, including \*so-dorw- 'healthy', OCS sodravo; \*so-ręt-j-a 'encounter; fate', OCS soręšta 'meeting' (cf. §16.7, 45.11), \*so-čęst- 'good part', \*OCS sočęstoje 'happiness'.

### CHAPTER SIX

# A SKETCH HISTORY: FROM LATE INDO-EUROPEAN TO LATE COMMON SLAVIC

## INDO-EUROPEAN AND SLAVIC

25.0 Indo-European is the name given to a large genetic family that includes most of the languages of Europe and extends across Iran and Afghanistan to the northern half of the Indian subcontinent. The earliest documentation is for Anatolian, in central Turkey, where Hittite writings of perhaps 1700-1200 BCE have survived. Indo-Iranian is known from Sanskrit texts (the oldest probably composed from c1500 BCE but written down much later), and from Avestan and Old Persian of somewhat later date. Armenian has been written since the fifth century CE. The oldest surviving Greek, from Crete and the Mycenean mainland, dates to about 1200 BCE, and from about 800 there is a continuous record. Old Latin, with some closely related dialects, is datable to the sixth century BCE. Celtic was widely spoken in Europe during the first millennium BCE. while Germanic is not known before about 250 CE.<sup>1</sup> The common ancestral language must be assigned a date no later than 3500 BCE-more likely considerably earlier.

**25.1** Slavic is not documented before the activities of Cyril and Methodius (cf. §0.1), starting after 860. At that time, Slavic groups were surely in contact with speakers of varieties of several IE subdivisions: Baltic in the northwest, Germanic in the west (roughly in a broad frontier zone that

For an authoritative sketch, see Calvert Watkins, "Proto-Indo-European", in Ramat, Anna Giacalone, and Paolo Ramat, *The Indo-European Languages*, London-NY, 1998, pp. 25-73. In that same volume (pp. 415-53) Henning Andersen provides a more traditional account of Slavic. See also Baldur Panzer, *Die slavischen* Sprachen in Gegenwart und Geschichte; Sprachstrukturen und Verwandtschaft. (= Heidelberger Publikationen zur Slavistik A. Linguistische Reihe, Band 3.) Frankfurt-Bern-NY-Paris 1991.

#### A SKETCH HISTORY

extended from the Danish peninsula nearly to the Adriatic), Romance along the Adriatic littoral and in many internal regions (probably in mountains from Croatia through Bosnia into Serbia and Macedonia eastward through Bulgaria to Rumania), Albanian (in the extreme southwest present-day Montenegro, Macedonia and Albania), and Greek. Information about the inhabitants of most of the Balkans and eastern Europe from 600 until about 1100 is far too skimpy to provide a history of the movements of groups speaking these different language-types.<sup>2</sup>

**25.2** The hypothetical homeland of the Indo-Europeans, in the current view of many specialists, was north of the Black and Caspian Seas, in an area from which it was easy to move eastward over the steppes, westward into central Europe, or southward to Greece. Roughly the same general area, perhaps extending westward from the middle Dniepr to the headwaters of the Bug and Dniestr, is favored by many scholars as the "cradle" of the Slavs. There is no tangible evidence whatsoever for these theories, attractive and plausible as they may seem; they are working hypotheses, based on complex assumptions and corollaries that must be accepted on faith.

**25.3** The very existence of Slavs is uncertain until near 600 CE, when a new wave of invaders appeared from the east and north and devastated southeastern and central Europe. Contemporary observers call some of these hitherto unknown intruders Slavs—a wholly new designation. This fits the linguistic evidence; the sixth-century Slavs are a nascent ethnos with a newly consolidated language.

25.4 OCS permits us to posit a Late Common Slavic dialect continuum that existed c800-c1100 (*LCoS*). As we shall see, internal reconstruction and outside comparisons (chiefly with Baltic) imply a Middle Common Slavic system that is virtually without dialects (*MCoS*), and an Early Common Slavic, whose origin can be no earlier than about 300 C.E. (*ECoS*). A Pre-Slavic and a Pre-Baltic *état de langue* may be posited as subdivisions of a variegated dialect continuum of late Indo-European that

<sup>&</sup>lt;sup>2</sup> The oldest recorded (very brief) texts of Albanian date from the mid-fifteenth century; its connection with remnants of earlier Balkan languages such as Illyrian to the northwest or Dacian and/or Mysian to the east must remain in the realm of speculation. In any case, Albanian must be deemed a special branch of IE. The division of continental Balkan Romance into western Arumanian (still spoken in parts of Macedonia and Greece) and eastern (Daco-)Rumanian, the dialects underlying standard Rumanian, appears to be no older than the tenth century. Dalmatian, on the Adriatic coast, survived till the 1890s.

might be called **Pre-Balto-Slavic** (*PBS*). This leaves us with some 4000 years between the epoch when a community spoke Indo-European and the demonstrable appearance of Slavic. This is the temporal distance between the oldest Latin and the variegated dialects of contemporary Rumanian, Portuguese, French, and the other Romance languages. Surely the linguistic systems that eventually evolved into Slavic underwent many metamorphoses, but we must admit that we lack the evidence to validate any theories. What is clear is that the reconstructible Slavic of the mid-ninth century indeed is Indo-European, distinct from its nearest cousins, the Baltic dialects of Lithuanian and Latvian.<sup>3</sup>

25.5 For the last nine or ten centuries the Slavs have been settled in the same general areas they now occupy (discounting the eastward expansion of Russia). The western frontiers with Germans have moved repeatedly, so that the Sorbs have been isolated from their Slavic neighbors to the south and east, and the northwesternmost group, the Polabians, have disappeared. The Pomeranians of the Baltic coast for the most part became Germanized, and the surviving communities speak highly Polonized dialects called Kashubian. Today's West Slavs have five standard languages: Upper and Lower Sorbian, Polish, Czech, and Slovak. The East Slavs have three standard languages: Ukrainian, Belarusian and Russian, which are fairly closely related. From the 11th to 14th centuries, the northern rim dialects (Pskov-Novgorod) were notably deviant in many features from their more southerly neighbors, but these communities were dispersed in the 15th century. The North Slavs (West and East) have long been separated from their southern Slavic kin by a broad zone occupied by speakers of German (in Austria), Hungarian, and Rumanian (including the Moldavians who now have abandoned attempts at maintaining a separate standard language).

**25.6** The Eastern South Slavs have two standard languages, Bulgarian and Macedonian. The Western South Slavs recently had two standards, Slovenian and Serbo-Croatian. On purely linguistic grounds (phonology, morphology, syntax, derivational processes) the dialects of Croatia, Bosnia, Montenegro, and Serbia can be viewed as one—I will do so and will use the term Serbo-Croatian, with apologies to anyone who might

<sup>&</sup>lt;sup>3</sup> Apart from isolated words and a few sentences, Baltic is not documented before c1550. Old Prussian is known essentially from Protestant catechisms translated from German; it died out not long after 1600. Lithuanian and Latvian represent two branches of a dialect continuum called East Baltic.

#### A SKETCH HISTORY

consider it inappropriate. As of early 2001 there are certainly two standards, Croatian and Serbian, and strenuous efforts to establish a third, Bosnian. The geographical positions are approximately as in the chart (non-Slavic neighbors printed in italics).



**25.7** I hold that the Late Common Slavic of c1000 CE had four regional variants or macrodialects: NorthWest, SouthWest, SouthEast, NorthEast. Rapid local differentiation took place as social conditions changed radically. In the NorthWest, change was particularly significant; by c1150 important features of the Polish and Czech-Slovak regional dialects were clearly different from each other and both were sharply set apart from East Slavic, which was still a single language, with the weakest of local variations. I call the common (North-) East Slavic language (up to the first half of the 14th century) Rusian. Despite its physical distance from the SouthEast or Bulgarian dialects, it still had much in common with them. The boundaries separating the SE and SW (eventually Macedonian vs. Serbian) were relatively weak. The differentiation into regional dialects and then standard languages is extremely complex; this table is a grossly simplified outline:

SE	NE	

1000	NW	SW	SE	NE
1100	Cz-Slk ~ Pol	kaj~ča~ čto	west ~ east	Rusian
1200	Cz-Slk ~ Pol	west kaj ~ east kaj	west ~ east	Rusian
		~ ča~ čto		
1500	Cz ~ Slk ~ Pol	Sln ~ SC	Mac ~ Bg	R/BR ~ Ukr
1700	Cz ~ Sik ~ Pol	SIn ~ SC	Mac ~ Bg	R ~ BR ~ Ukr

Documentation is uneven. For Rus', a standard based on OCS was used until after 1600; despite regional and temporal variants that indicate some local developments, the standard spelling systems obscure important details. Texts in Czech have survived from before 1300; by 1400 there was a flourishing literature. Polish texts begin from the mid-14th century, and show heavy Czech influence. Slovak writing is uncommon until about 1450, when long documents begin to appear. The earliest datable written Slavic, the Freising texts, represent Alpine dialects (cf. §1.04), yet Slovene is essentially unwritten until the Protestant movement; from about 1550 on there is a steady stream of translations and original works. Slovenia and northern Croatia shared many important features, constituting what may be called the "kaj-dialect": later on, the kaj-zone in Croatia was attracted more to its eastern neighbor, the "što-dialect". The northeastern Adriatic coast and adjacent islands developed a "ča-dialect" that was the basis of inscriptions and documents from the late 12th century on. Documents from Bosnia and the nearby Adriatic coast of the same early period, as well as from Serbia, provide fundamental (if incomplete) information about the što-dialect. (The names of SC dialects are based on variant reflexes of IE  $*k^{wi}d$  'what?' [OCS  $*\delta to$ ], see §38.21.)

\*

26.0 A comprehensive history of a language compares a particular linguistic system with one or more prior stages of the same system and attempts to account for the differences. Morphophonemic alternations play an important role in OCS, and their genesis and evolution are part of the history of the phonology. Synchronic alternations may imply historical phonetic changes. Observing that klonots 'they swear' has an infinitive kleti, and that the genitive semene 'of a seed' goes with a NA seme, we formulate a generalization that a sequence front vowel + nasal before vowel is replaced by e before consonant (cf. §3.313, 4.415). The background knowledge that a nasal consonant between a vowel and a consonant often (in languages all over the world) becomes non-consonantal and merges with the vowel allows us to hypothesize that at an older stage of the language the forms \*klonti and \*semen could have existed. This latter form looks startlingly like the Latin word for 'seed': sēmen. This internal reconstruction of a hypothetical earlier form is confirmed by external comparison with the Latin cognate (and patterns of relationship established by many other examples): \*sēmen (with a zero NA desinence) 'seed' is indeed to be posited for early Slavic as well as for some earlier stages of Indo-European. A root \*klin is thoroughly plausible for early Slavic, but its exact meaning and its affinities with words in other languages is not so clear.

**26.01** Internal reconstruction and careful comparisons of thousands of words with all sorts of Indo-European sources, most importantly Baltic, have established rules that allow us to convert attested OCS words to earlier shapes—first of all Middle Common Slavic.

**26.1** Let us define some phonological terms. The dialects we need to consider have consonant systems determined by four points of articulation: labial, dental, palatal, and velar.<sup>4</sup> Labial means "articulated with the lips (or upper lip and lower front teeth)" or, in terms of distinctive features, /+labial/ ( $p \ b \ f \ w$ ). Velar is "articulated with the back of the tongue" /-labial -coronal/ ( $k \ g \ x \ \gamma$ ). Dental is "articulated with the tip of the tongue" /+coronal + anterior/ ( $t \ d \ s \ n \ l \ r$ ). Palatal is "articulated with the middle part of the tongue" /+coronal -anterior/ ( $k \ g \ s \ z \ n \ l \ r$ ). The sibilants may further be "hushing" /+high +back/ ( $s \ z$ ) as opposed to the "hissing" dentals and palatals ( $s \ z \ ; \ s \ z$ ). In addition, we distinguish three sets of affricates, hissing dental  $c \ z \ \sim$  hushing  $\check{c} \ \check{z} \ \sim$  (hissing) palatal  $\acute{c} \ \check{z}$ .<sup>5</sup>

**26.11** Note that this inventory involves *simple segments*. OCS provides no reason for assuming *compound segments* of the type well known from Russian "palatalized consonants" which are produced by means of two simultaneous articulations (see *IJSLP* 41: 52). The appearance of this sort of "palatalization" surely developed as the jer-shift ran its course; it belongs to the nascent regional dialects of North Late Common Slavic (and perhaps eastern Bulgaria).

**26.12** The term *palatalization* is used in confusing ways by Slavists.<sup>6</sup> Here we attempt to use it (1) as a synonym of the synchronic descriptive term *substitutive softening*, and (2) as the historical process (or processes) whereby a consonant changed articulation from non-palatal (velar or dental) to palatal, e.g. t or k is replaced by k or č. It may also include a further step whereby the palatal articulation shifts to non-palatal (usually dental); thus the historical development "k > c" may be referred to as palatalization. This kind of shift of articulation from non-palatal to palatal usually takes place in the environment of a front vowel, particularly a high front vowel, or the front glide j.

Evidence from languages of many types demonstrates that palatalization of non-labial stops starts with a shift of articulation from velar (k/g)

<sup>&</sup>lt;sup>4</sup> The retroflex or domal consonants of the Indian subcontinent, the glottalized consonants of the Caucasus, and the uvulars or pharyngeals of Semitic are irrelevant to our discussion.

<sup>&</sup>lt;sup>5</sup> Labial and velar affricates (*pf kx*) are foreign to most Slavic, and the distinction between tongue-tip dental versus alveolar articulation is phonologically irrevelant.

<sup>&</sup>lt;sup>6</sup> Terms like "distinctive feature of palatality" or "palatal correlation" are meaningless without explicit definition in articulatory and/or acoustic terms.

or tongue-tip (t/d) to mid-dorsal (i.e. palatal) position, k/t > k, g/d > g. Palatal stops are prone to develop a sibilant offglide, thereby becoming affricates; the articulation may shift either toward "hissing" s or toward "hushing" š, so the affricates are c [t<sup>s</sup>] or č [t<sup>s</sup>] (voiced z ž 3 š). The affricates may then lose the initial closure and become sibilants, s or š, z or ž. A sibilant that precedes the stop adapts to the articulation of the stop; an affricate may become a simple stop after a sibilant.

$$\begin{aligned} k/t &> \mathbf{k} > \begin{cases} c > s \\ \dot{c} > \dot{s} \\ \dot{c} > \dot{s} \end{cases} & sk/st > \mathbf{\hat{s}}\mathbf{k} > \dot{s}\dot{c} > \begin{cases} sc > st \\ \dot{s}\dot{s} > \dot{s} \\ \ddot{s}\dot{c} > s\dot{s} \end{cases} \\ g/d > \mathbf{\hat{g}} > \begin{cases} 3 > z \\ \dot{3} > \dot{z} \\ \ddot{3} > \dot{z} \end{cases} & zg/zd > \mathbf{\hat{z}}\mathbf{\hat{g}} > \mathbf{\hat{z}}\mathbf{\hat{3}} > \begin{bmatrix} z_3 > zd \\ \dot{z}\dot{z} > \dot{z} \\ \ddot{z}\ddot{3} > \ddot{z} \end{bmatrix} \end{aligned}$$

The intermediate stages may last for generations, but observations of change in progress have shown that k and  $\check{c}$  or c (or even  $\check{s}$  or s) may be variants in the speech of individuals—in effect the whole process takes place at once. The voiced affricates 3,  $\check{z}$ , and  $\check{z}$  readily become continuants  $(z, \check{z}, \check{z})$ , while their voiceless counterparts tend to remain  $(c, \acute{c}, \check{c})$ . The three-way contrast  $c \sim \check{c} \sim \check{c}$  may persist, but it seems to be unstable, and  $\acute{c}$  easily merges with  $\check{c}$ .

**26.13** In *iotation*, triggered by a post-consonantal *j*, the iod ordinarily fuses immediately with the palatal consonant: kj/tj > k (which may immediately become *c* or *č*). This process too is found in diverse languages all over the world.<sup>7</sup> A less common but nonetheless widespread alternate is a geminate stop, kk/gg. Depending on the language, the geminate may simplify and participate in the processes outlined above, or it may remain. A final possibility is that the *j* becomes a spirant *s* or *š* and fuses with the stop as an affricate.

**26.21** Iotation in OCS is productive; it operates in conjugation and word-formation. The underlying j is a theoretical morphophoneme that never appears in surface forms; its presence is deduced from consonantal alternations. Thus, for example, infinitive and past passive participial stems contrast according to formulas CV ~ CjV: nositi ~ nošens 'carry', svętiti ~ svęštens 'sanctify', čistiti ~ čištens 'cleanse', soditi ~ soždens 'judge', prigvozditi ~ prigvoždens 'nail'. The participial stems correspond

<sup>&</sup>lt;sup>7</sup> Compare the informal and/or rapid pronunciation of English *can't you, won't you, did you, would you*; trisyllabic *Indian* versus disyllabic "Injun".

to underlying {nos-j-en-, swęt-j-en-, čist-j-en-, -gwozd-j-en-}—in traditional notation \*sj, \*tj, \*stj, and \*zdj. The standard examples for \*tj/\*djare the derived nouns {swět-j-a} 'light, candle, lamp' and {med-j-a} 'median, frontier, boundary-marker': OCS **свѣща**, межда.<sup>8</sup>

**26.22** A minor morphophonemic rule of OCS converts {gt kt} to št, §15.85. The historical sequence was approximately \*kt (> \*kt) > \*jt followed by the same assimilative processes that took place with the more frequent \*tj combination. See §26.13.

**26.23** The two-unit *št* and *žd* that represent \*tj/\*dj are distinctive for SE LCoS. It is only in Bulgaria and Macedonia that the *j* did not fuse with the preceding *t/d*, but produced geminates \*kk/\*gg that became \*sk/\*zg and (to fit the constraint that the first of two obstruents must be a sibilant, §2.522) presumably \*sc/zg. For other MCoS dialects we may posit \*k/\*gg. Only SE LCoS, therefore, has the same reflexes for \*tj, \*stj, and \*skj (which always = \*sk before front vowel).

**26.3** The consonant inventory of MCoS is nearly identical with that of OCS ( $\S2.12$ ); the cover symbols "\**tj* \**dj*" mean SE dialect \**kk/\*gg* versus \**k/\*g* in other regions.

pbtdczčž\**tj*\*djkgmnlrwj sz šž x njljrj

**26.4** Rusian surely had  $\check{c}/\check{z}$  for \*tj/\*dj, but the written language retained the OCS spellings in most instances, so that citable ER examples are rare. Early Western South Slavic (= SouthWestern LCoS), the ancestor of Slovene and Serbo-Croatian, had palatal stops  $*\check{k}/*\check{g}$ , except that in the west  $*\check{g} > *j$  (cf. Sln and Cr. dial. meja < \*medja). Early Czech seems to have had  $*\check{c}/*\check{z}$ . Polish and Slovak presumably had  $*\check{c}/*\check{z}$ .

The majority of examples are from inflection or word-formation:

OCS	svę <b>šte</b> nъ	roždenъ	ljuš <b>t</b> e	xu <b>žd</b> e	gra <b>žd</b> aninъ
MCoS	*swę <b>tj</b> enъ	*ro <b>dj</b> enъ	*lju <b>tj</b> e	*xu <b>dj</b> e	*gor <b>dj</b> aninъ
ER	*svęčenъ	roženъ	*ljuče	xuže	*gorožaninъ

<sup>&</sup>lt;sup>8</sup> These two words are to be found in nearly all of the hundreds of Slavic dialects that have been even partially described, but formal etymological identity does not guarantee that the meaning will be the same. OCS mežda occurs only once in the canonical mss, Su 397.12, where it renders a Greek word meaning 'lane, sidestreet'. The preposition meždu 'between' is etymologically \*medju, a dual locative "on the two boundaries". It (or an alternate Ls \*medji) is often attested in dialects where the base noun is unknown.

EWSS	*svę <b>k</b> enъ	*rogenъ	*lju <b>k</b> e	*xuģe	*gra <b>ģ</b> aninъ
ECz	*svećenъ	*roźenъ	*ljuće	*xu <b>ź</b> e	*graźaninъ
	'sanctified'	'born'	'fiercer'	'worse'	'townsman'

Compare svet-i+ 'sanctify', rod-i+ 'bear', ljuts 'fierce', xuds 'thin, poor', \*gords 'town'.

26.41 MCoS \*tj is also posited where \*kt is indicated by morphological or comparative evidence: \*petji 'to bake' {pek-ti} > OCS peštь, ER pečti; \*motji 'to be able' {mog-ti} > mošti/moči; \*notji 'night' [Latin Gs noctis] > nošti/nočti; \*dotji 'daughter' [< \*dukti < \*dugtër] > OCS došti, ER doči.

**26.51** A salient difference between hypothetical Middle Common Slavic and attested OCS and Rusian (SE and NE LCoS) lies not in the phonological inventory, but in the distribution of units—in particular the liquids (l, r) in contact with the lax vowels ( $b \ b \ e \ o$ ) and the tense low vowels  $\check{e}$  and a. Rusian consistently writes "neutral" jers before r and l(vbrxb 'crest', gbrdb 'proud', xblmb 'hill', and vblkb [for MCoS \*wblkb] 'wolf'), cf. §2.631. This direct evidence for four *liquid diphthongs* with jers as the first member is supplemented by indirect evidence for four more with e or o. If "t" stands for any consonant, MCoS \*tert, \*telt, \*tort, \*tolt > OCS trět, tlět, trat, tlat. The lax vowel has become tense, and the order of elements has been reversed. In Rusian, there is neither tensing nor metathesis. Instead, the vowel is repeated after the liquid: teret, torot, and tolot (for \*el \*bl had previously backed to \*ol \*d). This East Sl process is called *pleophony* or *polnoglasie*. For example:

'bank' 'booty' 'city' 'hunger' 'crest' 'proud' 'wolf' 'hill' OCS/LCoS b**rě**gъ plěnъ grad<sub>b</sub> g**la**dъ g**rъ**dъ vlьkъ х]ътъ v**гь**хъ Middle CoS \*bergъ \*pelnъ \*gordъ \*g**ol**dъ \*w**ьг**хъ \*gъrdъ \*wьlkъ \*хъlmъ LCoS/Rusian beregs polons gorods golods **vьг**хъ gъrdъ vъlkъ х**ъl**mъ The Czech-Slovak reflexes of the *\*tert* diphthongs agree with those of OCS and all South Slavic (although there are complications with wordinitial diphthongs, see §30.35).

**26.511** LCoS initial clusters with ll, dl, vl, vr, ml, mr, srr, zr, cl, cr, zl, zr, and  $\delta l$  reflect the metathesis of MCoS liquid diphthongs: e.g.,

	'interpreter'	'palm'	'rule'	'rope'	'milk'	'dark'	'stink'
OCS	tlъkъ	dlanь	vlastь	vгьvь	mlěko	mrakъ	smradъ
MCoS	*tъlkъ	*dolnь	*wolstь	*wьгwь	*melko	*morkъ	*smordъ
	'member'	'worm'	'through'	'yellow'	'staff'	'foal'	'helmet'
OCS	*člěnъ	čгьvь	črěsъ	*žlьtъ	žгьdь	žrěbę	šlěтъ
MCoS	*čelnъ	*čьгvь	*čersъ	*žьltъ	*žьrdь	žerbę	*šelmъ

**26.52** A second pervasive difference is that Rusian has lost the nasal vowels that are so well attested in OCS. The earliest mss, obviously copied from OCS models, generally reproduce the proper nasal-vowel letters

correctly, but numerous errors demonstrate that \*q had been replaced by u, and \*q by a newly distinctive low front  $\ddot{a}$ . In spelling,  $\pi = o_i$  and  $i\pi = i_i$ , while  $\alpha$  (or m in blocked position) stands for older \*q and for \*a (after  $j \check{c} \check{s} \check{z} c$ ):

MCoS/OC EarlyRusia	• •	moją moju	męso mäso	tę tä	moję mojä	otročę otročä
written	ржкж 'hand [As]'	моыт 'my [Asf]'	MACO 'meat'	та 'thee [As]	-	отроча, отроча sf]' 'boy [NAs'

NB: Ap *moje* was South Slavic, in principle different from North Slavic *mojě* (§29.814, 36.52), but Rusian scribes resolutely tried to adhere to the SSI norm, and spellings like mot are extremely rare.

# From Indo-European to Common Slavic

**27.1** The early Indo-European phonological system may be assumed to have had 17 obstruents, 4 sonorants, 5 vowels (occuring both long and short), and 3 consonants called *laryngeals* (conventionally symbolized as  $h_1$ ,  $h_2$ ,  $h_3$ ). Stops could be voiceless, voiced, or voiced aspirated, with 5 kinds of articulation (labial, dental, palatal, velar, and labio-velar).<sup>9</sup> The short high vowels *i* and *u* could follow other vowels to form diphthongs, while between vowels they could function as glides (*j* and *w*).<sup>10</sup> The sonorants could function as syllabic (*m* n l r).

	obstrue	ents			sonorant	laryngeal	vowels
-	d K lh gh	-	k" g" g"h	S	mnlr	h <sub>1</sub> h <sub>2</sub> h <sub>3</sub>	īiēeāaōoūu

27.2 Baltic and Slavic do not reflect all these distinctions. In terms of Slavic evidence, the IE labio-velar consonants have fallen together with the velars, and the aspirated stops with the unaspirated voiced stops. We posit, therefore, a post-IE system where eight stops stand for the original fifteen. The four-position series (labial, dental, palatal, velar) is posited

<sup>&</sup>lt;sup>9</sup> Specialists wrangle about almost every item in this statement, in particular the phonological nature of the three-way opposition in stops and the number and definition of /-anterior/ or dorsal articulations formerly called *gutturals* and recently dubbed *tectals* ("prevelar, velar, postvelar"--among other terms). I generally rely on the works of Helmut Rix and Manfred Mayrhofer, moderated by advice from my Harvard colleagues Jay Jasanoff and Calvert Watkins.

<sup>&</sup>lt;sup>10</sup> Indo-Europeanists often write the labial glide (or semi-vowel) "u" and the iod "i" or "y". Since the symbol "y" in Slavic studies refers to the high mid to back [i]—written "ы" in Russian and "ъ" in OCS, I follow Slavistic tradition and write j for the glide: OCS възма 'neck' is transliterated vyja, but its hypothetical MCoS equivalent is \*wyja. Similarly, OCS отмъвъ 'having washed' = umyvō, older \*umywō.

also for Indo-Iranian (as opposed to Germanic, Greek and Latin, where reflexes of labio-velars remain distinct, but those of palatals and velars are indistinguishable).<sup>11</sup> The laryngeals had played a significant role in many early dialects, affecting the quality and quantity of vowels in many morphemes. They were gradually lost in most regions; their effects had become lexicalized (and perhaps sometimes morphologized) long before a distinctively Slavic system emerged.

27.3 Baltic and Slavic reflect an innovation that is shared by Indo-Aryan and Iranian: s after r, k, i or u became dorsal š. This variant eventually resulted in a new phoneme. IE s was non-labial but unspecified as to place of articulation. In the new system, /+anterior/ s became distinctively opposed to /-anterior/ š. The "ruki-rule" operated under somewhat differing conditions in various regions; in Pre-Slavic it does not affect s before consonant. In Slavic, š replaced every desinential s (unless a consonant followed); in Baltic, desinential š reverted to s.

**27.31** Since the timing of the replacement of desinential s by  $\check{s}$  is unknown and the  $\check{s}$  disappeared before Slavic writing began, the hypothetical  $\check{s}$  will not be indicated in the derivations offered in the rest of this book.

**27.41** The simpler late Indo-European system that is reflected by Baltic and Slavic therefore has the following phonemes:

obstruents	sonorants	vowels
pbtdKģkgsš	mnlrjw	īiēeãaōoūu

Long and short diphthongs are assumed (*ei*  $\bar{e}i$  *eu*  $\bar{e}u$  *ai*  $\bar{a}i$  *au*  $\bar{a}u$ ). Further, short high vowels were inserted before the syllabic liquids and nasals:  $m_n$ ,  $l_r > im$  um in un il ul ir ur. Scholars have not discovered the conditions that determined whether *i* or *u* was selected. What is important is that these combinations still serve as diphthongs (called liquid or nasal diphthongs) in Lithuanian.<sup>12</sup> It is assumed that early Slavic had parallel series of diphthongs (cf. §26.51).

<sup>&</sup>lt;sup>11</sup> The traditional label for dialects that had this palatal series is satem, recalling the sibilant reflex of \*k in \*kmt-om '100' in Avestan (early Iranian) satem, as opposed to the Latin reflex kentum (confusingly spelled with c and, as a result of very late Latin or post-Latin regressive palatalizations, pronounced [s], [t<sup>s</sup>], [θ] in various modern Romance dialects). Germanic, Latin, and Greek are kentum-languages. Luvian (and presumably Common Anatolian) preserved the five-position series.

<sup>&</sup>lt;sup>12</sup> The liquid or nasal functions as the second member of the syllabic nucleus (as do o or e), and the accent may fall on either member, e.g. úoga 'berry' ~ sesuõ 'sister'; pienas 'milk' ~ miēstas 'city'; várna 'raven', vardas 'name'; ginti 'to defend' ~ giñti 'to chase'.

**27.42** Pre-Balto-Slavic words could begin in any vowel or consonant (except  $\check{s}$ ), or consonant clusters limited to s + stop + sonorant. Closed syllables were common, and word-internal consonant clusters of considerably more variety occured.

Words could end in t or d, in n, s, ns, nts and rarely in r. The final syllable is ordinarily a desinence—a grammatically meaningful morpheme—or part of a desinence; sometimes a change in the phonemes of a desinence is to be interpreted as morphological substitution rather than phonetic evolution. Yet we must also assume special developments that can be defined in phonetic terms (including the word boundary, symbolized #).

28. The Slavic vowel system evolved in steps that are not always easy to correlate with the changes in the consonantal system.

**28.1** The IE non-high back vowels  $a/\bar{a}$  and  $o/\bar{o}$  merged completely in a vowel-pair I will write  $a/\bar{a}$ . The resulting Early Common Slavic system is fully defined by the features  $\pm high$ ,  $\pm back$ , and  $\pm long$ :<sup>13</sup>

	+long	-long	-long	+long		oral	diphth	ongs	
+high	ī			ū					
+high		i	u		+	ei	ai	eu	au
-high		е	а						
-high	ē			ā					
	-back	-back	+back	+back					
Also nas	al and li	iquid di	phthon	gs: im	in um	un il	ir ul u	r	

em en am an el er al ar

(In Pre-Baltic, short o > a, but long  $\bar{o}$  remained generally distinct from  $\bar{a}$ .)

Contrast the Middle Common Slavic system which eventually resulted:

tense	i	у	u						īn	ūn
lax		ьъ		+	ьг	Ы	ъr	ъl		
lax		e o			er	el	or	ol		
tense	ĕ	а							ēn	ān
+nasal	ę	Q								
	-back	+bacl	ĸ							

<sup>&</sup>lt;sup>13</sup> Let me stress that the presence or absence of *rounding* is not distinctive for this system, nor was it distinctive for the /+high -low/ $\bar{u}/u$  or the /-high -low/ $\bar{o}/o$  of the older system. See *IJSLP* 41: 19.

Only u is distinctively /+round/; i and y are distinctively /minus-round/.

It appears that the first element of a diphthong could be long or short in ECoS, and the possibility remains for MCoS. Since it seems that the first member of the nasal diphthongs usually lengthened, the table shows this notation. The final emergence of unitary nasal vowels, \*e < \*in/\*in, \*en/\*en and \*e < \*an/\*an, may well have occurred after the \*y and \*uwere fully established, cf. §29.8202.

**28.2** Late Common Slavic had, in principle, only open syllables. Its Pre-Balto-Slavic predecessor, however, allowed syllables that ended in a non-syllabic high vowel or resonant (i.e. diphthongs like *ei au en er*) or obstruents (including word-boundary #). The syllable is a surface unit of phonotactic organization: underlying morphemes are affected by the rules of syllabic structure. A morpheme like \**sup* 'sleep' or \**poi* 'sing' would automatically (so to speak) admit a syllabic boundary before its final unit if followed by a vowel (\**sup-ā-tei* 'to sleep' = \**su.pā.tei*; \**poi-e-te* 'you sing' = *po.je.te*) but remained intact before a consonant (\**sup-no-s* = \**sup.nos* [noun, Nsm], \**poi-tei* 'to sing [inf.]' = *poi.tei*). See §29.8.

29. The first crucial steps that set Slavic irrevocably apart from its Baltic cousins involve the /-anterior/ consonants, i.e. the palatals (/+coronal/ k g) and velars (/-coronal/ k g).

**29.1** The IE palatal stops (k g) became hissing affricates, c 3, then s and z.<sup>14</sup> There are now three continuants: s, z, and š. This process is surely very old.

**29.2** New palatals split off from the velars and became hissing affricates  $(c_3)$ , attested in OCS. The environment was very specific: the conditioning factor stood before the k g, so the process is progressive. A velar after a non-diphthongal high front vowel  $(i \bar{i})$  [optionally followed by a nasal (n)], and before a low back a or  $\bar{a}$ , became palatal. This progressive palatalization is called BdC:

k g > K g | /+ syllabic -high -back/ (/+cons +nasal/) \_\_ /+syl -high +back/

<sup>&</sup>lt;sup>14</sup> Pre-Baltic k g > hushing  $c \tilde{J}$ , then  $\tilde{S}$  and  $\tilde{z}$  – which survive in modern Lithuanian. The system from which modern Baltic dialects are derived has these two obstruents and unchanged t/d p/b k/g s m n l r j w, along with  $\bar{l}/i \bar{e}/e \bar{a}/a \bar{o} \bar{u}/u$ . Unlike Slavic, it does not, in its early stages, undergo mutative palatalization of velars. Iod-palatalization of dentals is far in the future, and quite without connection to phonetically comparable Slavic processes. The distance from the PBS phonological system to modern Lithuanian dialect systems is remarkably short.

That is, *ika inka* (etc.) > *ika inka* (etc.). The diphthongs ai and ei (etc.) do not trigger palatalization; see §29.921.

29.3 Results of the ruki-rule ( $\S27.3$ ), the affrication of IE k g, and BdC:<sup>15</sup>

IE \*s either remains or shifts to  $\check{s}$  (which later becomes x before back vowel):

\*sūnus > OCS synъ, Li sūnùs
\*rosa 'dew' > OCS rosa, Li rasà
\*wetusos 'old' > OCS vetъхъ, > Li vētušas;
\*moisos 'bladder'(?) > OCS měxъ 'wineskin', > Li maĩšas 'sack'
\*wirsus 'top, crest' > \*wiršuš > OCS vъrхъ; Li viršùs (with reversion of desinential š to s).

- The IE palatals \*k \*g diverged at the outset (§29.1): \*koino- 'hay' > OCS sčno, Li šiënas; \*prok- 'ask' > OCS prositi, Li prašýti \*geim- 'winter' > OCS zima, Li žiemà \*wegh- 'convey, go' > OCS vezo 'I convey', Li vežù.
- IE velar \*k \*g remain except in the BdC environment :
  \*kou- 'hew, strike' > OCS kovati 'beat, forge', Li káuti 'beat'
  \*tek- 'run' > OCS tekç 'I run', Li tekiù
  \*nogutis '(finger)nail' > OCS nogoto, Li nagùtis
  in contrast to \*woinikos 'wreath, crown' > OCS věnoco, Li vainikas
  Germanic \*kuningaz 'king' > OCS konezo, Li kùnigas 'priest'.

**29.4** Still another pair of palatals began to appear when a front vowel ( $\bar{e} \ \bar{i} \ i$ ) or j followed  $k \ g$ ; the velars became hushing affricates, attested in OCS as the voiceless  $\check{c}$  but voiced  $\check{z}$ . This is the *First Regressive Palatalization* (KI):<sup>16</sup>

k g > K g | \_\_\_ /-back -consonantal/.

That is, ki gi ke ge kj gj etc. > ki gi ke ge kj gj etc. > či  $\exists i$  ( $\exists i$ ) če  $\exists e$  ( $\exists e$ ) etc.

<sup>&</sup>lt;sup>15</sup> Word-final \*-us and \*-os yield -o (or -b) by later rules: see §29.7.

<sup>&</sup>lt;sup>16</sup> The First Regressive Palatalization is called KI in this book, while the Second Regressive Palatalization is KAI (a much later process that affected most of Slavdom by c1000, but did not reach the peripheral dialects of Pskov-Novgorod). The Progressive Palatalization (BdC) is assumed by many Slavists to be a late process and therefore called the Third or Second Palatalization. ("BdC" recalls the Polish scholar who first pointed out some of the difficulties of explaining the data, Jan Baudouin de Courtenay). The sequence I propose here is BdC-KI-KAI. Most recent handbooks assume either KI-BdC-KAI or KI-KAI-BdC, often remarking that BdC and KAI are two phases of a single process.

29.41-29.7

**29.41** The BdC and KI environments for the palatals are mutually exclusive, and theoretically the affrication could be viewed as immediate:

```
K/g > c/3 before /-high +back/ OR > č/š before /-back -cons/
> č/š otherwise > c/3 otherwise
```

**29.5** It is probably at this time that sibilants + j underwent iotation: sj and šj both yielded  $\check{s}$ , and  $zj > \check{z}$ . We assume further that  $\check{s}$  before back vowels became x—a new phoneme.

**29.51** Post-consonantal *j* disappears in the process of iotation. We therefore assume it as part of KI and sibilant-iotation. Iotation of *n l r* produced unit palatals that will be written here with digraphs, *nj lj rj* (cf. §1.22). There is no evidence to locate the appearance of new palatal sonorants in chronological sequence. They surely were in place in MCoS. Iod after labial and dental stops (*pj bj mj*; *tj dj*) remains until late MCoS.

**29.6** The way these early rules interact is important. Here are stages of the history of OCS \**pusati* 'to write', \**pusati* 'to pound', \**sucati* 'to piss', \**pušeno* 'millet', and \**sučits* 'pisses [3s pres]':

IE saten	1	ruki	c 3 > s z	BdC	KI		> OCS
si <b>k</b> ātei	>	_	_	si <b>k</b> ātei >	si <b>c</b> atei	>	*sьcati
pi <b>k</b> ātei	>	picātei >	pisātei >	-	-		*рьsati
pisātei	>	pišātei >	-		pixātei	>	*рьхаti
piseno	>	pišeno >	-	_	pišena	>	*рьšепо
sik-ī-(?)	- >	-	-	si <b>k-</b> ī-t- >	sičīt-	>	*ѕьčітъ

The ruki-rule must operate before the deaffrication of the satem c/3 produces an s that can stand after i, so that the three-way contrast  $ik \sim ik \sim is$  or  $ik \sim ic \sim is$  is maintained as it becomes  $ik \sim is \sim is$  and later  $*bc \sim *bs \sim *b\delta/*bx$ .<sup>17</sup>

**29.7** Two vowel processes interacted to produce a new morphophonemic situation: Vowel Raising (VR) changed the shape of many desinences, and Vowel Adjustment (VA) imposed new constraints on possible CV sequences. A preliminary condition is that a vowel is long before \*nC, see §29.814.

<sup>&</sup>lt;sup>17</sup> No form of the verb \*sbcati is written, as far as I know, in any Slavic text before c1650, but the comparative evidence suffices to establish it as a parallel to sbp-a+ 'sleep', \$15.32. See Vaillant 1966 405. The i/ę present is assured, but the shape of the 3s desinence for this early period is uncertain.

NOTE: Vowel Raising concerns syllables that are grammatical morphemes and therefore subject to morphological influences; see §36, esp. 36.41-42.

**29.71** Vowel Raising was limited to final closed syllables; it shifted the low vowels \*a and  $*\bar{a}$  (and possibly \*e) to \*u and  $*\bar{u}$  (\*i). The final syllable could end in a nasal, a sibilant, or both. In present active participles, \*nts is posited.

VR1 a low short vowel raises before n# or s#:-an# -as# > -un -usNØ1 \*n is lost after a high vowel and before  $(s)\#:^{18}$  $-\bar{i}ns\# -\bar{u}ns\# > -\bar{i}s -\bar{u}s$ VR2 a long vowel raises before ns#: $-\bar{a}ns > -\bar{u}ns$ 

Final  $*\bar{a}n$  is not raised, and therefore contrasts with  $*\bar{u}n(s)$  and  $*\bar{i}n(s)$  at this stage. Then the nasal in  $*\bar{u}n$  is lost and  $*\bar{u}$  yields y, while  $*\bar{a}n$  and  $*\bar{i}n$  become unit nasal vowels,  $\rho$  and  $\rho$ . See examples in §32.

**29.72** Vowel Adjustment is neutralization of the back ~ front opposition. It applies to a vowel in position after the glide \**j* or the stops \*k \* g resulting from BdC (or the affricates that replaced them). VA had two phases, probably separated in time. In the first, both the /+back/ high vowels ( $\bar{u} u$ ) and the short /+back/ low vowel (*a*) fronted (to  $\bar{i} i e$ ); in the second, the long front low vowel ( $\bar{e}$ ) backed to  $\bar{a}$ :

after  $k \notin j$  VA<sup>1</sup>  $\bar{\mathbf{u}} \mathbf{u} \mathbf{a} > \bar{\mathbf{i}} i \mathbf{e}$  : VA<sup>2</sup>  $\bar{\mathbf{e}} > \bar{\mathbf{a}}$ /+coronal -back/ /+back/ /-back/ /-back -high +long/ /+back/

VA<sup>1</sup> in effect is a progressive assimilation, while VA<sup>2</sup> is dissimilation. VA<sup>2</sup> applies to the output of KI, so that  $k\bar{e} > c\bar{e} > c\bar{a}$ , yielding OCS  $c\bar{a}$ . VA<sup>1</sup> applies to diphthongs (ai \*au > ei \*eu).

**29.73** The syllables  $k\bar{u} ku g\bar{u} gu$  resisted palatalization in all environments. The syllables  $k\bar{a} ka g\bar{a} ga$  became  $c\bar{a} ca 3 \bar{a} 3$  only if they were preceded by  $\bar{i}(n) i(n)$ . But VR produced  $c\bar{u} cu 3 \bar{u} 3$ . These syllables then became  $c\bar{i} ci 3\bar{i} 3$  by VA<sup>1</sup> (as ca 3a became  $c\bar{e} 3e$ ). For example: *atrak-as* ~ *atik-as* Ns 'boy' ~ 'father'; BdC > *atrakas* ~ *atikas*; VR > *atrakus* ~ *atikus*, VA > *atrakus* ~ *atikis*, eventually OCS *otrok* ~ *otscb*.

	earliest S	BdC	VR	VA		OCS
Ns 'boy'	atrak-as	> atrakas	> atrak <b>u</b> s	> atrakus :	>	otrok <b></b>
Ns 'father'	atik-as	> ati <b>k</b> as	> atik <b>u</b> s	> atik <b>i</b> s 🛛	>	otъcь

<sup>&</sup>lt;sup>18</sup> This rule is required for Ap of *i*-stems, e.g. PBS \*gostins 'guests' > \*gastins > OCS gosti, see §38.51.

In this way certain desinence-initial vowels at an early date became variables defined by the stem-final consonant. BdC, VR and VA established the morphophonemic basis for the twofold declensions of LCoS.

\*

**29.8** In early Slavic times, syllable-final obstruents disappeared or adapted to function as the onset of the next syllable; \*sup-n-as (with reinterpretation of morphemic boundaries) > OCS son-o, not obviously related to the infinitive sop-a+. A PBS diphthong in a closed syllable becomes a LCoS monophthong, but before a vowel it becomes vowel plus heterosyllabic glide; \*poi-tei > pěti 'to sing', \*poi-e-te > pojete 'you sing'. (Cf. §28.2.)

Some of the processes survived as generative rules in OCS. The infinitives \*grebtei 'dig' and \* $p\bar{a}ktei$  'herd, pasture', for example, have become greti (gre.ti) and pasti (pa.sti)—in underlying form {greb- $\emptyset$ +ti}, {pas- $\emptyset$ +ti}. See §§3.31, esp. 3.3131.

The phonetic deletion of syllable-final obstruents is assumed to be an early process, but there is no decisive evidence.

**29.81** Syllable-final nasals before C/# were part of a nasal diphthong, \*klintī 'to swear'~ klinānt-. Before a vowel or \*j, the nasal was consonantal and the syllabic boundary was adjusted: {klon-i+ti} > klo.ni.ti 'to incline' but past passive participle \*klon-i-en- > \*klon.je.nto > klo.nje.nto (where nj is a unit palatal); {lom-i+} \*lo.mi.ti 'break' ~ \*lo.mje.nto > lo.mlje.nto (§3.521).

**29.811** Nasal diphthongs became unit nasal vowels (\*in \*en \*im \*em > \*e; \*un \*on \*um \*om > \*e), e.g. kleti klungto. The front nasal of LCoS seems to have been phonetically a tense non-low vowel [e] or [j], the back nasal a tense non-low [ $\psi$ , q] or else low [a].<sup>19</sup>

**29.812** Stem-final alternations of nasal vowel before C/# but vowel + heterosyllabic nasal before V became normal by the end of MCoS: \*wermen- $\emptyset$  'time' Gs \*wer-men-es > OCS vrě.me vrě.me.ne (§4.414).<sup>20</sup>

<sup>&</sup>lt;sup>19</sup> In terms of patterning, \*q behaves like \*u (cf. §2.11), but in terms of later development, \*q became unrounded nasal *schwa* in Poland (later front q if short, q if long), non-nasal *schwa* in Bulgaria, but rounded [u q] elsewhere.

<sup>&</sup>lt;sup>20</sup> The IE neuter nominative of this suffix surely was \*-mēn or zero-grade \*-mn. This does not guarantee that the same shape survived into early Slavic; for my purposes \*-men# > -mę suffices.

**29.813** Nasal diphthong before nasal consonant apparently varied by dialect. OCS has both *pomenoti* and *poměnoti* 'call to mind, remember' (cf. *po-mbn-ě+* and *po-min-aj+* 'remember, have in mind' and *pametb* 'memory'), \*-*min-n-* or \*-*men-n-.*<sup>21</sup> OCS adjectives like *drěvěnz* 'wooden' and *měděnz* 'of brass' are regularly spelled -AN- in Rusian texts, as are ethnic or regional designations like OCS *Izmailitěne*, Rus' HZMAMANTANE 'Ishmaelites'. Clearly these two suffixes originally had \*-*en-n-*;<sup>22</sup> South Slavic regularly developed \*-*ěn-* while North Slavic had a nasal vowel: \*-*gn-* (which became open front \**än* by 1000).<sup>23</sup>

**29.8131** OCS *ime* 'name' surely was \**jume* in some dialects, < older \**in-men* < \**n-men* < IE \**h*<sub>1</sub>*nh*<sub>3</sub>*.men*, zero-grade of the root \**h*<sub>1</sub>*neh*<sub>3</sub> (cf. Gk ovoµ $\alpha$ , La *nomen*). The \**n* of the root disappeared before the nasal \**m*.

**29.814** The hypothetical early distinction of short vs. long initial elements of nasal diphthongs is hard to reconstruct; it seems safe to assume that all ECoS vowels are long before \*nC or \*nC#, but may be short before \*n#. The regular relationships of IE final closed syllables and OCS are these:

PBS	in#	un#	on#	ōns# āns#	īns#	ūns#
LCoS	ь	ъ	ъ/ь	y/ę	i	У

The first two represent As i-stems and u-stems, e.g. gostb 'guest' and syna 'son'; the last two are Ap of the same (gosti, syny). The middle two illustrate Asm and Apmf of the twofold declension (e.g. stola, strojb: stoly, strojg: ženy, strujg ~ Ns stola 'seat', strojb 'order', žena 'woman', struja 'stream').

In the first two and last two, the consonants are simply lost. The third requires a near-final stage with \*-u/\*-i. The contrasting back nasal versus front oral vowel of the fourth implies earlier  $*-\bar{u}n/*-\bar{i}n$ .

Loss of the original nasal consonants took place in stages:  $VR^1$  raises on# to un#;  $N\emptyset^1$  deletes the nasal after a high vowel in a word-final syl-

<sup>&</sup>lt;sup>21</sup> Cloz and Sav have only é, Zo and Mar strongly prefer it. Euch has only e (17x), Su has 24 e ~ 1 é. Ps Sin varies by scribe: B 4 é, 2 e; A 12 e; G 18 é: E 1 é, 8 e. The evidence of modern dialects is conflicting; in all probability the stems have been re-formed.

<sup>&</sup>lt;sup>22</sup> Kaměns 'of stone' (cf. kamens 'stone', §4.412) is an inherited word, derived from the stem \*kamen- (§4.412) by the archaic formant \*n: kamen-n-. OCS has a competing lexeme with the productive suffix \*sn: kamensns.

<sup>&</sup>lt;sup>23</sup> Ps 17:46 oxrom q 'they became lame' surely represents the reduced grade of xrom-'lame' but there is no guarantee that the verb had forms with -nq.

lable:  $in\# \bar{\imath}ns\# > i\# \bar{\imath}s\#$ ,  $un\# \bar{u}ns\# > u\# \bar{u}$ . Then VR<sup>2</sup> converts  $\bar{a}ns\#$  to  $\bar{u}ns\#$ , which is split by VA<sup>1</sup> into  $\bar{\imath}ns\#$  and  $\bar{u}ns\#$ , whereupon NØ<sup>2</sup> deletes the nasal after the high back vowel, leaving nasal \*- $\bar{\imath}n(s)$  opposed to \*- $\bar{u}(s)$ ,<sup>24</sup> i.e. the morpheme {y/ę}. In effect, NØ<sup>1</sup> applies only to i-stem and u-stem Ap (e.g. gosti, syny); NØ<sup>2</sup> accounts for the oral/nasal contrast in {y/ę}.

Exceptions seem to be determined by morphological analogy, e.g. the twofold NAs neuter  $\{o/e\} \sim IE *-om$ , see \$38.42.

OCS -e in these {y/e} desinences represents the South Slavic forms; North Slavic had - $\check{e}$  (although Rus' scribes resolutely continued to spell in the OCS fashion). The northern "soft" variant {y/ $\check{e}$ } is called  $\check{e}$  tertium or  $\check{e}^3$  as opposed to  $\check{e}$  primum (=  $\check{e}^1 < *\bar{e}$ ) and  $\check{e}$  secundum (=  $\check{e}^2 < *ai < IE *oi$ , \*ai). See §38.52.

**29.8141** Final  $*\bar{a}n$  is to be posited for three important desinences reflected in OCS by non-alternating  $\{q\}$ : accusative singular feminine two-fold (e.g.  $\check{s}ujq \; rqkq$  'left hand'), first person singular and third person plural present (e.g. pojq, pojqts 'I/they sing'; nesq, nesqts 'I/they carry').

As f was \*- $\bar{a}n$  in PBS, from somewhat older \*- $\bar{a}$ -m. It was not subject to Vowel Raising or Vowel Adjustment; the evolution to q is straightforward.

The OCS -q of 1st person singular present goes back to PBS \*- $\bar{o}$ , which became ECoS \*- $\bar{a}$ . Later on, a nasal (probably \**m*) was added. Thus 'I sing' was \**poi*- $\bar{o}$ , yielding ECoS \**pajā*, then \**pajām*, OCS *pojq*.

The IE third person plural was \*-onti,<sup>25</sup> from which we derive ECoS \*- $\bar{a}nti$  (e.g. \*nek-onti, \*poi-onti > \* $nes\bar{a}nti$ ,  $paj\bar{a}nti$  > OCS nesqts, pojqts).<sup>26</sup>

Desinential -e in Nsm poje (Gs pojexta) 'singing', on the other hand, is the "soft" variant of the morphophoneme {y/e} and goes back to \*poi-on-t-s. ECoS \*pajants > \*pajunts > \*pajin > poje vs. \*pajantja > \*pojexta(cf. §29.814, §30 C ##40-45). North LCoS had {a/e}, see §39.421.

**29.8142** The -*q* of 3p root-aorist (*padq* 'they fell'; *mogq* 'they were able') never occurs after palatal C. We posit PBS \*-ont, ECoS \*-*ānt*. (This same desinence serves in the imper-

<sup>&</sup>lt;sup>24</sup> Hypothetical  $*\bar{u}n$  is extremely rare except in these desinences; it may well be that it always lost nasality and yielded LCoS y.

<sup>&</sup>lt;sup>25</sup> For most LCoS dialects, the third person terminal desinence was \*-tb; OCS -to is exceptional. See §43.13.

<sup>&</sup>lt;sup>26</sup> Without the assumption that \*onC > ECoS ānC, \*nekonti (and most stems) would yield desinential -qt-, but \*poionti would be subject to VR and yield \*pajenti and †pajet-. One might hypothesize that the less frequent "soft" variant of this alternating desinence was eliminated in favor of non-alternating -qt-.

fect 3p - ea-x-q.) The alternative 3p aorist -e (jese or jese 'they took', rese or rekose 'they said') seems to reflect IE \*-nt.

The OCS verbal classifier \*no derives from IE \*nou: in tauto-29.8143 syllabic position, \*nowC > \*nowC, whereby the whole syllable became nasal. Thus \*ri-now-ti > \*rinonti > rinoti (inf, 'to push') but the past passive participle \*ri-now-en- > rinovens (§15.711).

In similar fashion, nasalization sporadically appeared in roots with initial nasal, but not in all dialects: e.g. nud-/nod- 'annoy', mud-/mod-'delay, be slow', see §2.71.

The preposition  $k_{\overline{\sigma}}$  'to' and the prefix-prepositions  $s_{\overline{\sigma}}$  'with, 29.815 from' and we 'in' represent older \*kun, \*sun, and \*un. When followed by the third person pronominal stem \*j, the \*nj becomes a unit palatal. From forms like ko njemu 'to him' and so njimi 'with them', the nj eventually was abstracted as a suppletive form of the root to be used with most or all other prepositions (§4.25). As a prefix, \*sun before a consonant could become so-: sosědo 'neighbor' (sěd 'sit, settle'), soporjo 'disputant, adversary'  $(pbr-\acute{e}+ \text{'dispute'});^{27}$  sombněnbje ~ sumbněnbje 'doubt'. Before a vowel—chiefly the roots \*i/\*id 'go', \*im/\*em 'take', and \*ed 'eat'—the nasal remains: soniti 'descend', voniti 'enter', soněsti 'eat up'. Lexemes with these variant prefixes belong to a list of inherited words; these formations are no longer productive.

Early \*un (< IE \*n, §27.41) 'in' retained nasality and back 29.8151 quality in otrb 'within' and otroba 'entrails', while \*en or \*in survives in \**jetra* neut. pl. 'innards' (not OCS, but clearly CoS; sg. *jetro* usually means 'liver'). The nouns odols and odolsje 'valley' imply \*un-dol- (with \*dhel 'hollow'). The denasalized form \*ws is to be seen in vstors 'second' (dial. \*wsters, §41.82).

Similarly, the zero-grade of the negational particle \*ne occurs in orodz 'fool' (\**n*-rod- > \**un*-rod-, meaning approximately "degenerate"; rod means both 'kin' and 'generate'), otblz 'leaky' (of a vessel; tblo 'bottom').

The IE root  $*h_2enh_1$  'breathe, breath' (cf.  $ave\mu os$  'wind', La 29.8152 animus 'spirit') in o-grade underlies LCoS \*w-on-j-a 'fragrance' and  $\rho$ -x-aj+ 'to smell'. The prothetic w is unexpected, and the suffix x is isolated; see §35.13 (g).

<sup>&</sup>lt;sup>27</sup> Attested \*sasěda, and \*saparja are \*OCS; they occur in post-OCS copies of surely OCS texts.

**29.8201** The evolution of the oral diphthongs brings about an essential restructuring of the vowel system and inaugurates Middle Common Slavic. The diphthong  $*ei > *\bar{i}$  (or \*bj, see §29.92), and  $*ai > *\bar{e}$  (written  $\check{e}$  in the new system). This amounts on the surface to a redistribution of extant phonemes, but in terms of underlying morphophonemes, the new  $\check{e}$  is distinct: it has special distributional restrictions (§3.5c2). Therefore we distinguish  $\check{e}^2$  (< \*ai) from  $\check{e}^j$  (< \* $\bar{e}\bar{i}$ ) in Middle and Late Common Slavic.

**29.8202** Tautosyllabic \**ou* and \**eu* became a high back round monophthong \* $\bar{u}$  that remained distinct from the inherited long high back vowel that up to now we have written \* $\bar{u}$ . The opposition we now write \**y* vs. \**u* (< \* $\bar{u}$  vs. \**au*/\**eu*) requires reevaluation of distinctive-feature marking, cf. §28.1. For examples, see §29.95.

**29.821** Early CoS  $*\bar{u}$  was defined /+high +back/, distinct from /+high -back/ $\bar{\imath}$ ; the phonetic degree of rounding was irrelevant. Middle Common Slavic is defined by the new three-way opposition  $*i \sim *y \sim *u$  (< ECoS  $*\bar{\imath} \sim *\bar{u} \sim *au$ ). Rounding has become distinctive for the new high long back vowel, if not for the inherited short back vowels. The inherited long vowels had presumably been redundantly tense; in MCoS they retain their articulation but the defining feature is now /+tense/.

The nasal diphthongs of ECoS are tentatively symbolized iN and  $\bar{a}N$  here: the development to late MCoS is not altogether clear.

ECoS	ī ei	i	ū	u	eu au	ai ē	e	ā	a	in	ān
{MCoS}	i	Ь	У	Ъ	u	$\check{e}^2 \check{e}^1$	e	a	0	ę	ð
/MCoS/	i	Ь	У	ъ	u	ě	e	a	0	ę	9
high	+	+	+	+	+	_	-	-	_		
back	-	-	+	+	+	-	-	+	+	_	+
round	-	-	-	(-)	+		-	-	(+)		
tense	+	-		+	+	+	-	+	-	+	+

The writing systems of OCS and most early Slavic do not mark accent (or length or possible rising or falling pitch or intonations). Evidence from modern dialects (often bolstered by comparative data from Baltic and even more remote IE dialects) guarantees that the tense vowels could be long or short; it is probable that the new lax vowels occasionally lengthened.

It is notable that Common Slavic (and Baltic) segmental phonology in sharp contrast to early Indo-European, Germanic, and Latin and its Romance daughter-languages—shows little or no effect from accentual factors. **29.822** This slightly asymmetrical 11-vowel array (cf. the display on p. 192, above) is a system from which we can derive all subsequent Slavic dialects. It is documented (except for the liquid diphthongs) by Old Church Slavonic.<sup>28</sup> It is a new and significant stage, deserving a special label; we call it Middle Common Slavic.

\*

**29.91** The diphthongs became monophthongs before C or # but remained, with a modified syllabic boundary, before a vowel. Thus \*-auC > \*uC, \*auV > \*owV; \*aiC > \*eC, \*aiV > \*ojV. OCS reflects this in part by systematic morphophonemic alternations (esp. nontruncated -ov- $a \sim$  truncated -uj-: milovati ~ milujqto §15.501, 29.951 below, and in part by isolated relics ( $peti \sim pojqto$  'sing' §16.53).

**29.92** The diphthong \*ei before vowel yields OCS bj, e.g. IE treies 'three [masc. nom.]' > trbje (which is spelled trbe and trie, §2.61). This implies raising, \*e > \*i.

**29.921** The progressive palatalization (BdC, §29.2) is triggered by a high front vowel before the velar stop that shifts articulation, yet the diphthongs \**ei* and \**ai* apparently block the rule. We assume a slight lowering of the \**i* in \**ai* (as in Latin, Avestan, and mod. German) > [ae], but a metathesis of elements in \**ei*, [ie]. Therefore the masculine derivational suffix \*-*ineik*- with Gs desinence would be \*-*iniek*- $\bar{a}$ , yielding OCS -*bnika*. The corresponding feminine \*-*inik*- $\bar{a}$  (with Ns desinence) would undergo BdC, yielding -*bnica*.

**29.93** Long  $*\bar{u}$  followed by a vowel functions like \*uw (short V + glide). Thus  $*leub-\bar{u}$  'love [Ns]' but \*leub-uw-i '[As]' >  $ljuby \sim ljubwb$ . This is reinterpreted in OCS morphophonemic terms as {ljub- $bw-\emptyset$ } ~ {ljub-bw-b}, cf. §4.413. For complications in verbal stems, see §44.35.

**29.94** Long  $*\overline{\imath}$  before vowel yields OCS \*bj and its functional equivalent \*ij. The distinctive feature /tense/ is neutralized before \*j (§2.61). Later morphophonemic adjustments were made in different regions. Separating older \*ej from \*ij or  $*\overline{\imath}$  is difficult.

**29.95** Heterosyllabic \*eu (\*ew) merged with \*au (\*aw); e.g. \*neu- $\bar{a}$ 'new [Nsf]' > \*naw $\bar{a}$  > nova. Tautosyllabic \*eu early became \*jau, then MCoS \*(j)u; tautosyllabic \*au > u. E.g. \*seuj- $\bar{a}$  'left [Nsf]' > \*sjauj $\bar{a}$  > šuja.

<sup>&</sup>lt;sup>28</sup> While seven of the vowels imply a certain immediate ancestor, four are somewhat opaque: *i* may go back to *ei* or  $\bar{i}$ ; e < ai or  $\bar{i}$ ; e < any front vowel + n/m, and e < any back vowel + m/n.

**29.951** By LCoS, the monophthongization of \**ow* before \**j* was completed; the alternation of full-stem {-ova-} and truncated {u-je} is a fact of the lexicon. Although the forms do not happen to occur in OCS, we confidently assume that *lovi*+ 'to hunt' generated 1s pres \**lovljo*, past passive participle \**lovljens*, and substantive \**lovljensje*. The borrowed name 'Jacob' is \**ijakovs*, and the usual possessive adjective is \**ijakovljs* 'belonging to Jacob'.

**30.1** OCS words may begin with a vowel, but there are severe constraints ( $\S2.51$ ). Initial *o*, *u* and *q* offer only isolated problems (see  $\S36.5$ ). Initial *a* is limited to

(1) the conjunctions a 'and, but' (and derivatives like acě 'though' and ali 'but, whether') and ašte 'if';

(2) recent borrowings (e.g. apostols 'apostle'), and

(3) words that are also attested (sometimes within OCS, more often in post-OCS or modern dialects) with ja (e.g. aviti/javiti 'reveal, make plain'). It appears that all Slavdom but eastern Bulgaria has ja in morphemes of this last group; the variants with a- are LCoS dialect forms. (See §36.2)

**30.2** Initial high back y and  $\mathfrak{s}$  do not occur. Instead, the glide w is preposed as a syllabic onset. The process survives in unprefixed *vspiti* ~ prefixed *vsz-spiti* 'cry out' (§3.25). The prefix *vy*- (typical of Northern and westernmost SW LCoS; marginal in OCS), corresponds to Germanic  $*\bar{u}t$ -, Eng. out (< IE  $*\bar{u}d$ ):  $*\bar{u}$ -> \*y-> \*wy-.<sup>29</sup> The exact shape underlying ws 'in' is uncertain; we posit ECoS \*un > \*wsn-> ws. See §29.8151.

**30.21** The sequence \*bw (only at the boundary joining the prefix ob- with root-initial \*w) either did not develop or was simplified, cf. oblaks 'cloud' {ob-wolk-} vs. vlač-i+ 'drag'; obyknq+ 'become accustomed' vs. navyknq+ 'learn' < MCoS \* $\bar{u}k$ - (root \*ouk: nauči+ 'teach'). See §3.312. Alternate forms probably coexisted for a long time.

**30.31** Initial \**i* seems certain for the ubiquitous conjunction and particle *i* 'and, even, indeed', but otherwise front vowels preposed a iod: \**ji*, \**jb*, *je*, *je* and \**jě* (automatically > \**ja* by the phonotactic constraint that originated with VA<sup>2</sup>, cf. §3.5c).

**30.311** It is possible that OCS *ese* 'lo, see!', perhaps *ei* 'yes', and *eda* 'lest' had initial [e]. Common loanwords such as *episkops* (or *episkups*) 'bishop' and *evangelije* (and derivatives) very probably were pronounced with an initial glide by many or most early Slavic Christians. Unfortunately the glagolitic alphabet has no device for specifying j and cyrillic orthographies are inconsistent.

<sup>&</sup>lt;sup>29</sup> In OCS the name David is rigorously spelled *Davyds*, which does not correspond to Gk  $\Delta \alpha \beta \delta$ . The Muslim equivalent is *Daūd*. If we assume a borrowing of this form,  $\bar{u} > wy$  explains the attested spelling.

**30.312** The negated present  $n \check{e}smb$  'I am not' (§16.101) implies that contraction occurred before prothesis: \**ne* esmi > \**n* $\bar{e}smi$ ; later on \**es*-> \**jes*-.

**30.32** Prothesis of iod before front vowel is late. It appears only to mark the beginning of a syllable in forms like ATH ~ HARTE 'take' (§3.24) and ZAATH ZAHMRTE 'hire' as opposed to BEZATH ~ BESEMRTE, BEZEMÂIRTE or the suppletive stems in HTH HARTE 'go' ~ HZHTH HZHARTE 'go out'. If *j* had been part of the underlying stems, \*zj should have yielded \*z.

**30.33** Only one stem appears to combine prefix-final consonant with root-initial *j*: oštut-i+ 'sense, feel, perceive' (Rusian očut-i+) < \*ot-jut- < \*at-*jaut-*, cf. Li *atjausti* 'to sympathize' (*jaut-* 'feel'). Some words clearly reflect an inherited iod: *igo* 'yoke' goes back to IE \**jugom* (> \**jbgo* by VA<sup>1</sup>, cf. Cz *jho*).

Su възъмривъ and възъмривъ 'becoming enraged', have a jer to separate prefix from root {vъz-jar-}; implying restructuring at this boundary after the jer-shift. See §3.3111.

**30.34** The exact provenance of certain other stems is obscure, in part surely because of different late dialect developments. OCS \**igsla* 'needle' could represent \**jbgsl*- with a strong jer in the initial syllable (therefore Cz. *jehla*, Gp *jehel*) or \**igsl*- with a jer only in the pre-desinential syllable (like R.  $\mu$ гла́, Gp  $\mu$ гол, SC *igla*, Gp ~ *igálā*).

The verb 'have' is persistently *ima*- in South Slavic, but *jma*- or *ma*- in Czech-Slovak. Similarly, East and SSI имя, *ime* 'name' but OCz *jmě*, mod. *jméno* (cf. §4.414).

The suppletive stems *i-ti* ~ *id-qt* 'go' perhaps contrasted \**ei*- to \**id*-; these stems shifted about in manifold ways in later dialects. SC, for example, has  $i\dot{c}i \sim idu$  (as though from \**ij-ti* \**id-u*) but  $po\dot{c}i \sim p\hat{o}du$  (as though from \**po-j-ti* \**po-jd-u*).

The prefix *iz* 'from' retains *i*- in South Slavic, but in Czech-Slovak and Polish older  $*j_{bz} > *j_{z-} > z$  (alternating with *s*).<sup>30</sup>

The stem igr-aj+ 'to leap, play' (and igra 'amusement, entertainment; game', by chance not OCS) also have *i*- in SSI, but \**j*<sub>b</sub>- in the northwest, including Ukrainian.

ECoS \**irg-āj-e*- can plausibly be interpreted as containing a metathesized zero-grade of IE \* $h_i$ ergh 'be sexually excited, mount' (Av *rghāyāte* 'be sexually excited', Gk ὀρχέομαι 'dance, mime'). [Lunt, 1977.]

<sup>&</sup>lt;sup>30</sup> Moreover, the merger includes the preposition \*s5 'with; down from'—not only in West Slavic but also in most Ukrainian and Belarusian dialects.

**30.35** Word-initial liquid diphthongs were eliminated by metathesis (#VRT > #RVT), but with Late CoS local variation. The modern isoglosses generally divide North Slavic (including Rusian and Czech) from South Slavic. For internal liquid diphthongs, Czech agrees with SSI in that *ToRT* > *TRaT* (cf. §26.51). In initial position, certain morphemes need special marking. We posit #aRT (as opposed to #oRT) as ancestor of *ra*- in all dialects: \**ar-dl-o* > OCS/Rusian*ralo* 'plow', Cz*rádlo* (cf. *orati* 'to plow'), Gk ἄροτρον (< IE \**arə-tron* < \**h*<sub>2</sub>*erh*<sub>3</sub>*-tr-*; \**arm-* > OCS *ramo* 'shoulder', Cz *rámě* (arch., now *rameno*), IE \**arəm-* (cf. Engl *arm*).<sup>31</sup> The more common formula \**oRT* corresponds to North Slavic *roT-*: OCS *rab-* 'slave' ~ Rusian, Cz *rob-* (IE \**h*<sub>3</sub>*orbho-* 'bereft', Gk ὄρφανος 'orphan'); prefix *raz-* ~ Cz, Rusian *roz-*.<sup>32</sup>

**30.351** OCS has *rob*- once in Zo and, with other derivatives, more than 30 times in Su. It survives in modern SC. It probably represents an old borrowing from a northern dialect, but perhaps the 9th-century isoglosses were more complex than those of modern times.

**30.36** Similarly for marked #alT, \*alk-om- > OCS, Rusian lakoms 'hungry, greedy', Cz lakomý 'miserly'; unmarked #olT, \*olkstb 'elbow' (cf. Li alkūnė) > OCS lakstb ~ Rusian lokstb, Cz loket.

**30.361** Yet the root *alk*- (cf. Li *álkti* 'be hungry') is attested in OCS, *alkati* beside *lakati* (and *vъzalkati vъzlakati*) 'hunger; fast'. Both shapes continue to be used in post-OCS manuscripts. A second OCS doublet is *aldii* 'boat' and *ladii*. Rusian has *lodъja*, Cz *lod*'.<sup>33</sup>

31. When \*ai became  $\check{e}$ , the sequences  $*k\check{e}$ ,  $*g\check{e}$ , and  $*x\check{e}$  were created, but they did not last. OCS and most early recordings of other regional speech affirm that the velars had mutated, by a process known as the Second Regressive Palatalization (here called KAI). Only the region of Pskov and Novgorod was not affected by KAI, and  $k\check{e}$ ,  $g\check{e}$ , and  $x\check{e}$  persisted.

OCS has  $c\check{e}$ ,  $3\check{e}$  and  $s\check{e}$ : e.g. Ls otroc $\check{e}$ ,  $bo3\check{e}$ ,  $dus\check{e}$  from otrok- $\mathfrak{s}$  'boy', bog- $\mathfrak{s}$  'god', dus- $\mathfrak{s}$  'spirit'. The written s in  $s\check{e}$  (<  $x\check{e}$ ) presumably was a hissing palatal for a time, but it soon fell together with the s of other

<sup>&</sup>lt;sup>31</sup> The precise nature of the marking of such syllables (ancient laryngeal? "acute" intonation?) is hotly debated.

<sup>&</sup>lt;sup>32</sup> Rusian scribes generally held to SSI spelling in such words. Pa6, pa6ora and many other words are simply loans that have displaced congruent East Slavic forms.

<sup>&</sup>lt;sup>33</sup> Su 232.30 кдина отъ мънии implies альнии for expected лании (< \*alnbjb) 'one of the roes', cf. R ланъ.

origins in South Slavic and most of Rusian; NorthWest LCoS normally has  $š\check{e}$  (*duš* $\check{e}$ ), although there are exceptions and morphological complications we will not deal with.<sup>34</sup>

Foreign words with k before front vowel were adapted by shifting k to c, e.g. ocut 'vinegar', cf. La *acetum*; *ceta* 'cent [a coin]'. KAI does not affect Greek loans.

\*

**32.** Here are examples of typical relationships between OCS words (or rather their presumed late MCoS equivalents) and their hypothetical Early Common Slavic shapes. Some mutually exclusive processes have been lumped together in this compressed scheme.

Line **1** : the effect of the progressive palatalization, BdC (§29.2). Line **2** : (a) VA<sup>1</sup> ( $\bar{u} \ u \ a > \bar{i} \ i \ e \ after \ j, \ c, \ f$ ) and

(b) an unrelated special process—"breaking" of eu to jeu (§29.95).

Line 3: (a) KI ( $k g > \check{c} \check{z}$ ), (b) iotation of sibilants (sj and  $\check{s}j > \check{s}$ ),

(c) the appearance of  $x < \check{s}$  before back vowels, and (d) the loss of j after palatal.

Line  $\boldsymbol{4}$ : VA<sup>2</sup> ( $\bar{e} > \bar{a}$  after  $j \check{c} \check{z} \check{s}$ ).

Line  $5: ei > \overline{i}$  and  $ai > \overline{e}$ .

Line  $\boldsymbol{6}$ : (a)  $\bar{u}$  > distinctively /-round +tense/ y; (b) au  $eu > \bar{u}$ ; and (c) new symbolization of 9 oral vowels (i y u, b 5, e o,  $\check{e}$  a).

Line 7 : KAI (k g x > c f s).

Examples 1-16 illustrate the velar stops, while 17-31 deal with  $s \, \dot{s} \, \ddot{s}$  and x. 8, 16, 29 and 30 (Nsm -*as*) are subject to Vowel Raising between **1** and **2**; the *s* drops, and the final result is - $\overline{s}$  for the first three. 30 is subject to VA<sup>1</sup>, so -u > -i, and the final result is - $\overline{b}$ . These steps are included in 37-39.

- atrak-ā > OCS otroka 'boy' [Gs]'
   kunīng-ā > OCS konęza 'prince [Gs]'
   ak-a > OCS oko 'eye'
   līk-a > lice 'face [Nsn]'
   awik-a > owbce 'sheep [Vs]'
- 4. awik-ai > owbci '[DLs]'
- 5. atrak-ai > otrocě '[Ds]'

- 5a. raik-ai > rěcě 'river [DLs]'
- 6. awik-ā > оwьса '[Nsf]"
- 6a. *raik-ā > rěka* '[Nsf]'
- 7. atik-e > otьče 'father [Vs]'
- 7a. kunīng-e > koneže 'prince [Vs]'
- 8.  $k\bar{e}s$ - $as > \check{c}as$ <sup>\*</sup> 'time, hour'
- 8a. gēb-ā > žaba 'frog'

<sup>&</sup>lt;sup>34</sup> The only roots affected are \*xoid- 'gray-haired' (OCS sěds) and its near-synonym \*xoir- 'gray', which is absent from OCS, barely attested in pre-modern East Slavic, and rare in medieval Bulgarian/Macedonian, while šěr- is well documented in Old Czech. Fourteenth-c. Novgorod merchants used the term  $\chi$  top 'coarse undyed cloth'. See §35.11.

- keist-as > čisto 'pure [Nsm]'
- atrak-j-a > otroče 'boy's [poss. aj. NAsn]'
- 11. atik-j-a > otbče 'father's [poss. aj. NAsn]'
- 12.  $lauk-j-\bar{a} > luča$  'beam, ray'
- 13. atik-j-ai > otuči '[poss. aj. DLfs]
- 14. keud-es-a > čudesa 'miracles [NAp]'
- 15. atik-j-au > otьču '[poss. aj. Dsmn]'
- 16. kaup-as > kupъ 'heap'
- 17. sauš-a > suxo 'dry [NAsn]
- 18. dauš-ai > dusě 'spirit [Lsm]'
- 19. dauš-e > duše 'spirit [Vsm]'

- 20. dauš-ā > duxa 'spirit [Gs]'
- 21. slūš-ē -tei > slyšati 'to hear''
- 22. dauš-j-ā > duša 'soul [Nsf]'
- 23. dauš-j-a > duše 'soul [Vsf]'
- 24.  $bas \bar{a} > bosa$  'barefoot [Nsf]'
- 25. bas-e > bose 'barefoot [Vsm]'
- 26. nās-j-a > naše 'our [NAsn]'
- 27. kes-j-e-te > češete 'comb [2p pres]
- 28. nās-j-ā > naša 'our [Nsf]'
- 29. sauš-as > sux b 'dry [Nsm]'
- 30. seuj-a > šujь 'left [Nsm]'
- 31. kos-j-ou > košu 'basket [Dsm]'

<b>(A)</b>	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	1
	kā	ka	ika	ikai	kai	ikā	ke	kē	kei	kje	kja	kjā	kjai	keu	kjau	kau	
1		•	ć	ć	•	ć	•	•	•	•	•	•	•	•			1
2		•	će	ćei	•		•	•	•		kje	•	kjei		kjeu		2
3		•	•	•	•		č	č	č	č	č	č	č	č	č		3
4		•		•	•			čā			•						4
5		•		•	ē		•	•	ĩ	če	če	čā	či	•			5
6			•	i	ĕ	•		a	i		•	a	•	u	u	u	6
7		•		•	ćě									•			7
	ka	ko	ьсе	ьсі	cě	ьса	če	ča	či	če	če	ča	či	ču	ču	ku	Γ
	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	1

<b>(B)</b>	17	18	19	20	21	22	23	24	25	26	27	30	29	30	31	]
	ša	šai	še	šā	šē	šjā	šja	sa	se	sja	sje	sā	sau	seu	sjau	1
2	•	•	•	•	•	•	šje	•		sje		•		sjeu	e	2
3	xa	xai	•	xā		•	še	•		š	Š			š	e	3
4		•			šā	•		•		•	•	•		•		4
5		ē		•	•	•	.	•	•	še	•		•	•		5
6	0	ě		а	a	а		0		•	•	a	u	u	u	6
7		śě		•			•	•		•	•		•			7
	xo	śě	še	xa	ša	ša	še	SO	se	še	še	sa	su	šu	šu	
i	17	18	19	20	21	22	23	24	25	26	27	30	29	30	31	1

Here are illustrations of the chief complications in final closed syllables.

- 32. rānk-ān > rǫkǫ 'hand [Asf]'
- 33. awik-ān > owьcq 'sheep [Asf]'
- 33a. stig-ān > stb3q 'path [Asf]'
- 34. atrak-an > otroks 'boy [Asm]'
- 35. atik-an > otьcь 'father [Asm]'
- 36. mēsīnk-an > měsęсь 'month [Asm]'
- 37. atrak-as > otroks 'boy [Nsm]'
- 38. atik-as > otьcь 'father [Nsm]'

- 39. mēsīnk-as > měsęсь 'month [Nsm]'
- 39a. kunīng-as > kъnęзь 'prince [Nsm]'
- 40. plāk-j-ant-s>plačę 'weeping [Nsm]'
- 41. rek-ant-s > reky 'saying [Nsm]'
- 42.  $dau \dot{s} j \bar{a}n > du \dot{s}q$  'soul [Asf]'
- 43. dauš-j-āns > dušę 'souls [Apf]'
- dauš-āns > duxy 'spirits [Apm]'
- 45. slūš-ī-nt-s > slyšę 'hearing [Nsm]'

(in# iš#) un# uš/us#>(i) u

onts ons > āns

 $\bar{a}ns > \bar{u}ns$ 

 $\bar{u}n > \bar{u}$ 

 $VR^{1}$  a low short vowel raises before n# or s/s -an# -as# > -un -us

 $N\emptyset$  word-final *n* or *s/š* drops after *i* or *u* 

VL short vowel lengthens before n(t)s#

 $VR^2$  long vowel raises before n(t)s#

UN *n* drops after  $\bar{u}$ 

 $\tilde{v}$  front vowel +  $n > \varrho$ ; back vowel +  $n > \varrho$ 

Loss of final  $s/\check{s}$  takes place after  $VR^2$ : it is not specified in the table.

( <b>C</b> )	32	33	34	35	36	37	38	39	40	41	42	43	44	45	]
	kān	ikān	kan	ikan	inkan	kas	ikas	īnkas	kjants	känts	šjān	sjāns	šāns	šints	
VR1	•	<b>·</b>	un	un	un	us	us	us							VRI
VL	•			•	•		•		ā	ā					VL
NØ	•		u	u	u	u	u	u							NØ
VR2									ū	ū		ū	ū		VR2
BdC		ć		ć	ć		ć	ć							BdC
VA			•	i	i		i	i	ĩ		1.	ī			VA
=	kān	icãn	ku	icu	Inci	ku	ici	īnci	kjīn	kūn	šjān	šjīn	šūn	šjīn	=
KI			•	•			•		č		š	š	x	š	KI
UN		1.		•						kū			ũ		UN
v	kǫ	ьсо	kъ	ьсь	ęсь	kъ	ьсь	ęсь	čę	ky	šq	šę	ху	šę	v
	32	33	34	35	36	37	38	39	40	41	42	43	44	45	1

(D) Here are the same examples, rearranged to show how the OCS syllables correspond to their ancestral bases.

OCS	ka	ko	ьса	ьсе	cě	ьсі	če	či	ča	ču	ku
ECoS	kā	ka	ikā	ika	kai	ikai	ke kje kja	kei kjai	kjā kē	keu kjau	au
	1	2	6	4	5	3	7, 10, 11	9, 13	12, 8	14, 15	16

OCS	xo	xa	śě	še	ša	so	se	sa	su	šu
ECoS	ša	šā	sai	še šja sja sje	šē sjā	sa	se	sā	sau	seu sjau
	17	20	18	19, 23, 26, 27	21, 22	24	25	28	29	30, 31

OCS	kǫ#	cǫ#	kъ#	сь#	čę#	ky#	šę#	xy#	še <del>#</del>
ECoS	kān	ikān	kan kas	ikan=īnkan, ikas≕īnkas	kjānts	kānts	šjān	šāns	sjāns šīnts
· · · · ·	32	33	34, 37	35=36, 38=39	40	41	42	44	43, 45

The voiced velar g of PBS becomes MCoS 3 where k > c, and  $\check{z}$  where  $k > \check{c}$ , cf. examples 1b, 7b, 8a, 33a, 39a.

**33.** OCS words on the whole allow us to reconstruct earlier shapes if we follow certain procedures. The word needs to be analyzed in terms of stem and desinence, root and affixes; and possible morphophonemic clues are required. Morphophonemic and phonotactic characteristics of theoretical

## 208

Pre-BaltoSlavic must also be considered. Chief among them is the assumption that most roots have the shape (C)VC, where V is normally \*ebut may be any vowel or diphthong. In conformity with the processes called *ablaut* or *apophony* (cf. §3.9), the \*e in some formations becomes \*o, and in still others  $*\bar{e}$  or  $*\bar{o}$ . In earliest Slavic terms, after  $*\bar{o} *o$  have merged with  $*\bar{a} *a$ , these variations involve the specifications *front* or *back* and *short* or *long*. The theoretical zero-grade, with no *e* or *o* at all, is represented in Slavic by a "reduced" grade, with *i* or *u* before nasal or liquid.

Pre-Balto-Sl Earliest CoS	pbtdk	ruents ģ k g s š ) z k g s š	sonorants mnlrjw mnlrjw	vowels Īiēeāaōoūu Īiēeāa ūu
Dip	վ	e-grade	o-grade	zero / reduced grade
oral		ei eu	ai au	i u
nasa		en em	an am	in im un um
liqu		er el	ar al	ir il ur ul

ECoS \*s is ambiguous; it could go back to satem-IE \*k, or to ancient \*s.

**34.** The evolution of early Indo-European to the late dialect continuum we dub PBS depends on data from many languages with extremely different histories. Here only a few examples will be cited in order to hint at the striking correspondences that establish the plausibility of a common linguistic ancestor of English, Latin, Greek, and Slavic.

**34.1** Let us start with a small store of standard etymologies of MCoS words as examples of the usual relationships back to early Indo-European. Although our focus is generally on the simplified array of Pre-Balto-Slavic, these examples include some of the evidence for the earlier voiced aspirates (*bh* etc.), the labiovelar stops ( $k^w g^w g^w h$ ), and the laryngeals ( $h_1 h_2 h_3$  [or *H*, in unclear cases]). Greek, Latin, and Germanic have undergone consonant changes that must be taken into account; here are the chief equivalents:

IE	р	b	bh	t	d	dh	K	ģ	ģh	k	g	gh	k"	g"	g*h
Gk	р	b	ph	t	d	th	k	g	kh	k	g	kh	p/t	b/d	ph/th
Lat	р	b	f/b	t	d	f/d	с	g	h	с	g	h	qu	u/gu	f/u
Gm	f/b	f/b	b	þ/d	t	d	h/g	g	h/g	h	g	g	hw	kw	w
														g/j	
Slav	p	b	b	t	d	d	s	Z	Z	k+	g*	g+	k+	g+	<b>g</b> *

The symbols  $k^+$  and  $g^+$  in the last line are intended to include c 3 and č ž, see the summary tables in §32D, page 208 above.

**34.2** The Slavic in the alphabetical list below is OCS except that asterisks mark unchanged MCoS liquid diphthongs, \*tj, or words not attested in canonical OCS. The Pre-Balto-Slavic etymon is given first, followed by the early IE form (usually only the root, marked by  $\sqrt{}$ ), with parallels from other languages. The symbol "H" indicates a laryngeal whose exact definition remains uncertain.

Abbreviations: Av(estan), ER - Early Rusian, Gmc - Germanic, Go(thic), G(ree)k, La(tin), Li(thuanian), ME - Middle English, OE - Old English, OHG - Old High German, Sa(nskrit)

- berǫ 'I take' < \*ber-, √bher 'carry': Sa bhárāmi, La ferō, Gk φέρω, Go baira
- \*berza 'birch' (R berëza, Cz březa) < \*berz-, √bherHg-: Li béržas, Gm \*birkjōn-, OHG birihha, Sa bhūrjás
- \*běgǫ 'I run, flee' < \*běg-, √bheg<sup>w</sup>: Li bègu, Gk φέβοµαι 'I flee, fear', φόβος 'fear'
- bljudǫ 'I observe' < \*beud-, √bheudh: Sa bódhāmi 'I notice', Homeric πεύθομαι 'learn (by enquiry)'
- \*blaxa 'flea'< \*blu-s-ā: Li blusà; cf. \*plou+k- in Gmc \*flauhaz, G Floh; \*plus-/pusl- La \*puslex > pūlex, Gk \*psul-ya > ψύλλα
- boljbjb 'bigger, better' < \*bol-, √bel: Sa bálīyān 'stronger', Gk βέλτερος 'better', La dēbilis 'deprived of strength, weak'
- \*boršьno (OCS brašьno 'food', SC bräšno 'flour') < \*bors-in- √bhars 'barley': cf. La farīna < \*bhars-īn-
- bratrs 'brother' < \*brātr- < \*bhréh<sub>2</sub>ter-: Sa bhrấtā, Gk φράτηρ, La frāter, Go brōþar
- byxz, bysta, byšę < \*bū-s-om, -tā, -ņt 'I was, you two were, they were'; √bhuH
- cěna 'price' < \*kainā < \*k<sup>w</sup>oi-neh<sub>2</sub>- : Li kainà 'price', Gk  $\pi \sigma_{1}v\eta$  'fine, penalty'
- -cěstiti 'cleanse < \*kaist-: ? Saketús 'brightness', German heiter 'gay, clear'; Li skaistùs 'bright; fresh; untouched,
- čisto 'pure, clean' < \*keist-: | innocent'
- desętь 'ten' < \*dekmt- : Li dēšimt, Sa daśāt '[quantity of] ten', La decem, Gk δέκας, δέκα
- $d\check{e}$ -l-o 'deed' < \* $d\bar{e}$   $\sqrt{dheh_1}$  | Li déti 'to do, put', Sa ádhām 'I put',

 $d\check{e}$ -ti 'to do' Gk έθηκα 'I (have) put', La faciō, fēcī 'put, have put', OE dōn 'to do'

- dome 'house' < \*domus : Sa dámas, Gk δόμος, La domus
- dymē 'smoke' < \*dhūmas < \*dhuh<sub>2</sub>-mó-,  $\sqrt{dhweh_2}$ : Sa dhūmás, La fūmus, cf. Gk θυμός '(onrush of) courage'

- \*dətji 'daughter' (OCS dəšti, ER dəči) < \*duktēr < \*dhugh2tér- : Li dukté, Sa duhitá, Gk θυγάτηρ
- ěd-ęts 'eat' <  $*\bar{e}d$ -,  $\sqrt{h_1}$ ed: Li édmi; Sa ádmi, La edō, Gk ἕδομαι 'I will eat'
- \*gojb (SC goj 'health, peace', OCz hoj 'abundance') < \* $g^{w}oj$ -,  $\sqrt{*}g^{w}eih_3$ 'live' [compare  $\tilde{z}iws$ ]
- gorěti 'burn' < \*gor- √g<sup>w</sup>her : cf. Sa ghŗņóti 'burns', háras 'flame'; cf. also požarz, žeravz
- gostb 'guest' < \*ghostis 'stranger' : Go gasts 'guest', La hostis 'enemy'
- \*govędo 'steer' < \*gou-en-d-; √g<sup>w</sup>ou-/g<sup>w</sup>ōu : cf. Latv gùovs 'cow', Sa gáus, La bōs bovis, Gk βοῦς [Gs βοῆς], OE cū
- goniti gənati 'drive' < |\*gen-/gan-/-gun-  $\sqrt{g^{when}}$ , cf. Li ganýti 'to herd'; ženǫtə Sa hánti 'strikes' 3p ghnánti; Gk  $\theta \epsilon i v \omega$  'strike' < \*t<sup>h</sup>enįō,  $\phi \delta v \circ \varsigma$  'murder' < \*p<sup>h</sup>onos;
- jesmь 'I am' < \*esmi √h<sub>1</sub>es: Sa ásmi, Li esmì, Gk єї́µı (< \*esmi)
- junz 'young' < \*jaun-  $\sqrt{h_2}$ ieu-H(o)n- : Li jáunas; Sa yúvan- 'young; youth'; La juvenis; jūnior, Go juggs [spelled gg = phonetic  $\eta$ g]
- kolo 'wheel' < \*kal-  $\sqrt{k^{w}el}$  : OPrussian kelan, ONorse huel, Gk πόλος 'axis of sphere, pole'
- kowati kową 'forge' < \*kau- : Li káuju káuti; OHG houwan; La cūdō
- \*kry 'blood' < \*krū  $\sqrt{\text{kreuh}_2}$ : Sa kravíş- 'raw meat', Gk κρέας 'meat', La cruor 'meat', crūdus 'raw'; Gmc \*hrawa- > E. raw
- lajq 'bark' <  $l\bar{a}$   $\sqrt{l\bar{a}}$   $\sqrt{l\bar{a}}$   $\sqrt{l\bar{a}}$   $\sqrt{l\bar{a}}$  laid 'l bark', Sa  $r\bar{a}yati$  'barks'; La  $l\bar{a}$ -mentum 'lament'
- lěwъ 'left' < \*laiwas : La laevus, Gk λαιός < λαιϊός
- ljub<sup>5</sup> 'pleasing' < \*leub-, √leubh- : Sa lúbhyati 'desires'
- luna 'moon' < \*lauk-s-nā, √leuk: OPrussian louxnos (pl) 'stars', La lūna 'moon'
- lzgati 'lie' < \*lug- √leugh: cf. Go liugan, OE lēogan
- medz 'honey' < \*médhu- : Li medùs, Sa mádhu, Gk  $\mu \epsilon \theta v$  'intoxicating drink' (cf. E mead)
- měxa 'skin, bag' < \*maisas : Li maĩšas 'big sack', Sa meşa- 'ram, hide'
- *mъzda* 'reward, just due' < \**mizd-*, \**misdhó-*: Go *mizdo*, Av *miždəm*, Gk μισθός 'pay'
- měsęсь 'month, moon' < \*mēs-ņ-k- : Sa mậs (Gs mấsas), OPers māhyā (Loc sg)
- *męso* 'meat' < \**mems*  $\sqrt{*mems}$  'flesh': OPrussian *mensā*, Sa *māmsám*, Go *mimz*
- mętą 'I disturb, mix' < \*menth<sub>2</sub>- : Sa mánthati 'mixes, disturbs'
- \*mozgь < \*mazg- : Sa majján- 'marrow', Av mazgəm 'marrow, brain', OHG marg, OE mearg > E marrow

- \*maxo 'moss' < \*musas : Li mūsaĩ 'mold'] OHG mos 'moss', La muscus \*myšb 'mouse' < \*mūsi-s : La, OHG mūs, Gk µũs, Sa mūş-
- nebo 'sky' < \*neb- √nebh: Sa nábhas 'mist, cloud', Gk νεφέλη 'cloud', La nebula 'cloud', Hittite nepiš 'heaven'
- \*noso 'nose' < \*nasas : nās- > La nāsus 'nose'
- $o\check{c}i$  '[two] eyes (Ndu)' < \* $ak\bar{\iota}$  < \* $h_3ok^{w}$ - $ih_1$ ,  $\sqrt{h_3ek^{w}}$ ; Li aki
- ostrows 'island' < \*ab-srau- 'flow around' [-sr- > -str-] : compare struja ostrs 'sharp' \*asras < \* $akros \sqrt{h_2ek}$ : Gk äkpos 'topmost, extreme'
- оwьca 'sheep' < \*owikā, < \* $h_2$ ówi-: La ovis, Gk  $\mathring{o}(F)$ ış
- \*pelnъ 'booty' (OCS plěnъ, R pólon) Li pelnas; cf. OHG fâli < \*fēl-ja-'for sale' (?)
- pěna 'foam' < \*poin- < \*(s)poim-n-: Sa phénas; Li spáinė 'foam [on waves]', La spūma; OHG feim
- \*pыnъ 'full' < \*piln- < \*piln<sub>1</sub>-no-: Li pilnas, Go fulls (< \*fulnaz)

роčьjetъ 'rests' < \*-kei- √k<sup>w</sup>eiH: La quies

pokojь 'rest' < \*-kai-as

- \*požara 'conflagration' < \*po-žēr-,  $\sqrt{g^{w}}her$  (cf. gorěti, žerava): cf. Sa gharmás 'heat, glow'; La formus = G $\theta \epsilon \rho \mu \phi \varsigma$  = OHG warm 'warm, hot'
- роть m. 'road' < \*pant- : Sa pánthās 'way', La pons pontis 'bridge' rosa 'dew' < \*rasā : Li rasà, Sa rasā 'dampness'; La rōs rōris 'dew' sĕmę sĕmene 'seed' < \*sē-men  $\sqrt{seh_1}$  'sow': La sēmen, OHG sāmo smějǫ sę 'laugh' < \*smēj- $\sqrt{smei-}$ : Latvian smeju; Sa smáyate '(he) smiles' \*sьrdьce 'heart' < \*krd-i-k-a  $\sqrt{k}$ erd: Li širdìs, Gk карбіа, La cor cordis sněgъ 'snow' < \*snaigas,  $\sqrt{sneig}$ 'h: Li sniẽgas, Go snaiws solь 'salt' < \*sal-i-s: La sal, cf. Go salt
- sta (3s aor) 'stopped, stood' < \*stā,  $\sqrt{\text{steh}_2}$ : Sa ásthāt, Gk  $\xi\sigma\tau\eta$
- *stojati* 'to stand' < earliest Slavic \**staj-ē-tei* < possible earlier \**sta-* <  $*sth_2$  (zero-grade of  $\sqrt{steh_2}$ )
- stignǫti 'arrive, reach' 3s aor stiže < \*steig- √steigh: Sa stighnoti 'climbs', Go steiga 'I climb', Gk στείχω 'I go'
- struja 'stream, current' < \*sraujā √sreu: East Li sraujà 'stream', sraũjas 'rapid', cf. Gk βέïω 'flow'
- synō 'son' < \*sūnus, √seuH 'give birth': Li sūnùs, Go sunus
- \*šujь 'left' < \*seu-jo- : Sa savyás
- topiti 'heat' teplo 'hot' < \*top-/\*tep- : Sa tápati 'heats', La tepeō; tepidus 'warm'
- truje 'three (m)' < \*trejes: Gk tres, La tres, Li trỹs
- ture 'aurochs' < \*tauras: Li taũras; Gk ταῦρος, La taurus 'bull'
- uxo 'ear' (Gs ušese, du uši) < \*ous-  $\sqrt{h_2}$ eus-: Li ausis, Go auso, La auris wesna 'spring' < \*wes- : Li vasarà 'summer', Sa vasantás 'spring'

vey		y' < *weg- √wegh-: Li vežù, Sa váhati 'goes', La vehō 'con-
		f' < *wlkos, *wlk <sup>w</sup> os : Li <i>vilkas,</i> Sa vrkás
		' < *wrsu- vwer: Li viršůs 'top', Sa vársisthas 'highest', La
	-	< "isu" ( $<$ "-rs-)
		e' < *wikos $\sqrt{\text{weik}}$ : Sa viś- 'tribe'; La vīcus 'village', OE wīc,
		'village'
		er' < $*\bar{u}dr\bar{a}$ , $\sqrt{wed}$ 'water': Li $\hat{u}dra$ ; Gk ὕδρα 'water snake'
zima'	winter	r' < *geim-ā, √gheim: Li žiemà; La hiems, Gk χειμών
znati '	to kn	ow' < *ýnō-, √ýneh <sub>3</sub> : Li žinóti; La gnōscō 'recognize', Gk
רוע	νώσκ	ω 'know', Go kunnan 'know'
		< * gombos, √gembh: Li žambas 'sharp edge', Sa jámbhas
•		Gk γόμφος 'nail', E comb
•	•	chase': see goniti
		crane' (Bg žérav, SC ž $\ddot{e}rav$ [cf. R ž $uravl'$ ]) < * $ger-aw$ , $\sqrt{gerh_2}$ :
		, Gk γέρανος, OE <i>cran,</i> La grūs grŭis
	-	owing' $< *ger$ , $\sqrt{g}$ wher: see <i>požar</i> $\delta$ and <i>gorěti</i>
žiwo 'l	living'	* $gtw$ - < * $g^{w}$ ih <sub>3</sub> -wo-, $\sqrt{g^{w}}$ eih <sub>3</sub> : Li gývas, Sa j $tv$ ás, Go qius, La
vīv	us; Gl	k βίος 'life'
zělo 'v	very' •	< *ghoil- : Li gailùs 'strong, bitter'; OHG geil 'exuberant'
IE pho	neme -	reflected in Slavic by the following words:
-		
р	-	teplъ, topiti, *pelnъ, pěna, potь
р b	- -	teplъ, topiti, *pelnъ, pěna, potь boljьji
	- - -	teplъ, topiti, *pelnъ, pěna, pǫtь boljьji berǫ, *berza, *-běgǫ, bljudǫ, *boršьno, bratrъ, bysta, byšę, byxъ,
b	- - -	teplъ, topiti, *pelnъ, pěna, pǫtь boljьji berǫ, *berza, *-běgǫ, bljudǫ, *boršьno, bratrъ, bysta, byšę, byxъ, ljubъ, nebo
b	-	teplъ, topiti, *pelnъ, pěna, pǫtь boljьji berǫ, *berza, *-běgǫ, bljudǫ, *boršьno, bratrъ, bysta, byšę, byxъ, ljubъ, nebo bratrъ, čistъ, *dъtji, gostь, mętǫ, pǫtь, sta, stignǫ, teplъ, topiti, trъje,
b (bh) t	-	teplъ, topiti, *pelnъ, pěna, pǫtь boljьji berǫ, *berza, *-běgǫ, bljudǫ, *boršьno, bratrъ, bysta, byšę, byxъ, ljubъ, nebo bratrъ, čistъ, *dъtji, gostь, mętǫ, pǫtь, sta, stignǫ, teplъ, topiti, trъje, turъ
b (bh) t d	-	teplъ, topiti, *pelnъ, pěna, pqtь boljьji berq, *berza, *-běgq, bljudq, *boršьno, bratrъ, bysta, byšę, byxъ, ljubъ, nebo bratrъ, čistъ, *dъtji, gostь, mętq, pqtь, sta, stignq, teplъ, topiti, trъje, turъ cěditi, cěstiti, čistъ, desętь, domъ, ědętъ, *sьrdьce, *wydra
b (bh) t d (dh)	- - - -	teplъ, topiti, *pelnъ, pěna, pqtь boljьji berq, *berza, *-běgq, bljudq, *boršьno, bratrъ, bysta, byšę, byxъ, ljubъ, nebo bratrъ, čistъ, *dъtji, gostь, mętq, pqtь, sta, stignq, teplъ, topiti, trъje, turъ cěditi, cěstiti, čistъ, desętь, domъ, ědętъ, *sьrdьce, *wydra bljudq, dělo, děti, dymъ, *dъtji, mьzda
b (bh) t d (dh) K	- - - -	teplъ, topiti, *pelnъ, pěna, potь boljъji bero, *berza, *-běgo, bljudo, *boršьno, bratrъ, bysta, byše, byxъ, ljubъ, nebo bratrъ, čistъ, *dъtji, gostь, męto, potь, sta, stigno, teplъ, topiti, trъje, turъ cěditi, cěstiti, čistъ, desętь, domъ, ědętъ, *sьrdьce, *wydra bljudo, dělo, děti, dymъ, *dъtji, mьzda desętъ, *sьrdьce, wьsь
b (bh) t d (dh) K ý	- - - -	teplъ, topiti, *pelnъ, pěna, potь boljъji bero, *berza, *-běgo, bljudo, *boršьno, bratrъ, bysta, byšę, byxъ, ljubъ, nebo bratrъ, čistъ, *dъtji, gostь, męto, potь, sta, stigno, teplъ, topiti, trъje, turъ cěditi, cěstiti, čistъ, desętь, domъ, ědętъ, *sьrdьce, *wydra bljudo, dělo, děti, dymъ, *dъtji, mьzda desętъ, *sьrdьce, wьsь *berza, znati, zobъ
b (bh) t d (dh) k ý (ýh)	-	teplъ, topiti, *pelnъ, pěna, potb boljsji bero, *berza, *-běgo, bljudo, *boršьno, bratrъ, bysta, byše, byxъ, ljubъ, nebo bratrъ, čistъ, *dъtji, gostь, meto, potь, sta, stigno, teplъ, topiti, trъje, turъ cěditi, cěstiti, čistъ, desętь, domъ, ědętъ, *sьrdьce, *wydra bljudo, dělo, děti, dymъ, *dъtji, mьzda desętъ, *sъrdьce, wьsъ *berza, znati, zobъ wezo, zima
b (bh) t d (dh) k ģ (ģh) k	- - - -	teplъ, topiti, *pelnъ, pěna, pqtь boljъji berq, *berza, *-běgq, bljudq, *boršьno, bratrъ, bysta, byšę, byxъ, ljubъ, nebo bratrъ, čistъ, *dъtji, gostь, mętq, pqtь, sta, stignq, teplъ, topiti, trъje, turъ cčditi, cěstiti, čistъ, desętь, domъ, ědętъ, *sьrdьce, *wydra bljudq, dělo, děti, dymъ, *dъtji, mьzda desętъ, *sьrdьce, wьsъ *berza, znati, zqbъ wezq, zima cčditi, cěstiti, čistъ, kopati, kowq, měsęcь, owьca, *sьrdьce
b (bh) t (dh) k g (gh) k g	- - - -	teplъ, topiti, *pelnъ, pěna, potь boljъji bero, *berza, *-běgo, bljudo, *boršьno, bratrъ, bysta, byšę, byxъ, ljubъ, nebo bratrъ, čistъ, *dъtji, gostь, męto, potь, sta, stigno, teplъ, topiti, trъje, turъ cčditi, cčstiti, čistъ, desętь, domъ, čdętъ, *sьrdьce, *wydra bljudo, dělo, děti, dymъ, *dъtji, mьzda desętъ, *sьrdьce, wьsъ *berza, znati, zobъ wezo, zima cčditi, cčstiti, čistъ, kopati, kowo, měsęcь, owьca, *sьrdьce *dъtji, *golsъ, polьza, lъgati, mozgъ, *žerawъ
b (bh) t (dh) k g (gh) k g (gh)	- - - -	teplъ, topiti, *pelnъ, pěna, pqtь boljъji berq, *berza, *-běgq, bljudq, *boršьno, bratrъ, bysta, byšę, byxъ, ljubъ, nebo bratrъ, čistъ, *dъtji, gostь, mętq, pqtь, sta, stignq, teplъ, topiti, trъje, turъ cčditi, cčstiti, čistъ, desętь, domъ, čdętъ, *sьrdьce, *wydra bljudq, dělo, děti, dymъ, *dъtji, mьzda desętъ, *sьrdьce, wьsь *berza, znati, zqbъ wezq, zima cčditi, cčstiti, čistъ, kopati, kowq, měsęcъ, owьca, *sьrdьce *dъtji, *golsъ, polь3a, lъgati, mozgъ, *žerawъ stignq, stiže, gostь, zělo
b (bh) t d (dh) k ý (gh) k g (gh) (k <sup>w</sup> )	-	teplъ, topiti, *pelnъ, pěna, pqtь boljъji berq, *berza, *-běgq, bljudq, *boršьno, bratrъ, bysta, byšę, byxъ, ljubъ, nebo bratrъ, čistъ, *dъtji, gostь, mętq, pqtь, sta, stignq, teplъ, topiti, trъje, turъ cčditi, cčstiti, čistъ, desętь, domъ, čdętъ, *sьrdьce, *wydra bljudq, dělo, děti, dymъ, *dъtji, mьzda desętъ, *sьrdьce, wьsь *berza, znati, zqbъ wezq, zima cčditi, cčstiti, čistъ, kopati, kowq, měsęcъ, owьca, *sьrdьce *dъtji, *golsъ, polь3a, lъgati, mozgъ, *žerawъ stignq, stiže, gostь, zělo cčna, kolo, oči, *wьlkъ
b (bh) t d (dh) k g (gh) k g (gh) (k <sup>w</sup> ) (g <sup>w</sup> )	-	teplъ, topiti, *pelnъ, pěna, pqtь boljъji berq, *berza, *-běgq, bljudq, *boršьno, bratrъ, bysta, byšę, byxъ, ljubъ, nebo bratrъ, čistъ, *dъtji, gostь, mętq, pqtь, sta, stignq, teplъ, topiti, trъje, turъ cčditi, cčstiti, čistъ, desętь, domъ, čdętъ, *sьrdьce, *wydra bljudq, dělo, děti, dymъ, *dъtji, mьzda desętъ, *sьrdьce, wьsь *berza, znati, zqbъ wezq, zima cčditi, cčstiti, čistъ, kopati, kowq, měsęcь, owьca, *sьrdьce *dъtji, *golsъ, polь3a, lъgati, mozgъ, *žerawъ stignq, stiže, gostь, 3člo cčna, kolo, oči, *wьlkъ *-běgq, *gojъ, gowędo, žiwъ
b (bh) t d (dh) k ý (gh) k g (gh) (k <sup>w</sup> )	-	teplъ, topiti, *pelnъ, pěna, pqtь boljъji berq, *berza, *-běgq, bljudq, *boršьno, bratrъ, bysta, byšę, byxъ, ljubъ, nebo bratrъ, čistъ, *dъtji, gostь, mętq, pqtь, sta, stignq, teplъ, topiti, trъje, turъ cčditi, cčstiti, čistъ, desętь, domъ, čdętъ, *sьrdьce, *wydra bljudq, dělo, děti, dymъ, *dъtji, mьzda desętъ, *sьrdьce, wьsь *berza, znati, zqbъ wezq, zima cčditi, cčstiti, čistъ, kopati, kowq, měsęcъ, owьca, *sьrdьce *dъtji, *golsъ, polь3a, lъgati, mozgъ, *žerawъ stignq, stiže, gostь, zělo cčna, kolo, oči, *wьlkъ
b (bh) t d (dh) k ý (gh) k g (gh) (k <sup>*</sup> ) (g <sup>*</sup> ) (g <sup>*</sup> h)	-	teplъ, topiti, *pelnъ, pěna, potb boljsji bero, *berza, *-běgo, bljudo, *boršьno, bratrъ, bysta, byše, byxъ, ljubъ, nebo bratrъ, čistъ, *dъtji, gostь, męto, potь, sta, stigno, teplъ, topiti, trъje, turъ cčditi, cčstiti, čistъ, desętь, domъ, čdętъ, *sьrdьce, *wydra bljudo, dčlo, dčti, dymъ, *dъtji, mьzda desętъ, *sъrdъce, wьsъ *berza, znati, zobъ wezo, zima cčditi, cčstiti, čistъ, kopati, kowo, měsęcъ, owьca, *sъrdъce *dъtji, *golsъ, polь3a, lъgati, mozgъ, *žerawъ stigno, stiže, gostь, 3člo cčna, kolo, oči, *wьlkъ *-běgo, *gojъ, gowędo, žiwъ goniti, gorěti, gъnati, požarъ, sněgъ, ženo, žerawъ
b (bh) t d (dh) k ý (gh) k g (gh) (k <sup>*</sup> ) (g <sup>*</sup> ) (g <sup>*</sup> h)	-	teplъ, topiti, *pelnъ, pěna, potb boljsji bero, *berza, *-běgo, bljudo, *boršьno, bratrъ, bysta, byše, byxъ, ljubъ, nebo bratrъ, čistъ, *dъtji, gostь, męto, potь, sta, stigno, teplъ, topiti, trъje, turъ cčditi, cčstiti, čistъ, desętь, domъ, čdętъ, *sьrdьce, *wydra bljudo, dčlo, dčti, dymъ, *dъtji, mьzda desętъ, *sъrdъce, wъsъ *berza, znati, zobъ wezo, zima cčditi, cčstiti, čistъ, kopati, kowo, měsęcъ, owъca, *sъrdъce *dъtji, *golsъ, polь3a, lъgati, mozgъ, *žerawъ stigno, stiže, gostь, 3člo cčna, kolo, oči, *wьlkъ *-běgo, *gojъ, gowędo, žiwъ goniti, gorčti, gъnati, požarъ, sněgъ, ženo, žerawъ *blъxa, *boršьno, bysta, byxъ, byšę, *golsъ, gostь, jesmь, męso,
b (bh) t d (dh) k ý (gh) k g (gh) (k <sup>*</sup> ) (g <sup>*</sup> ) (g <sup>*</sup> h)	-	tepls, topiti, *pelns, pěna, pots boljsji bero, *berza, *-běgo, bljudo, *boršsno, bratrs, bysta, byše, byxs, ljubs, nebo bratrs, čists, *dstji, gosts, męto, pots, sta, stigno, tepls, topiti, trsje, turs cěditi, cěstiti, čists, desęts, doms, ědęts, *ssrdsce, *wydra bljudo, dělo, děti, dyms, *dstji, mszda desęts, *ssrdsce, wsss *berza, znati, zobs wezo, zima cěditi, cěstiti, čists, kopati, kowo, měsęcs, owsca, *ssrdsce *dstji, *golss, polsza, lsgati, mozgs, *žeraws stigno, stiže, gosts, zělo cěna, kolo, oči, *wslks *-běgo, *gojs, gowędo, žiws goniti, gorěti, gsnati, požars, sněgs, ženo, žeraws *blsxa, *boršsno, bysta, byxs, byšę, *golss, gosts, jesms, męso, měsecs, měxs, msxs, rosa, sols, sěmę, smějo sę, sněgs, sta, stignoti, stojati, struja, syns, šujs, uxo, *wsrxs mozgs, mszda
b (bh) t d (dh) k ý (gh) k g (gh) (k*) (g*) (g*h) s	-	tepls, topiti, *pelns, pěna, pots boljsji bero, *berza, *-běgo, bljudo, *boršsno, bratrs, bysta, byše, byxs, ljubs, nebo bratrs,čists, *dstji, gosts, męto, pots, sta, stigno, tepls, topiti, trsje, turs cěditi, cěstiti, čists, desęts, doms, ědęts, *ssrdsce, *wydra bljudo, dělo, děti, dyms, *dstji, mszda desęts, *ssrdsce, wsss *berza, znati, zobs wezo, zima cěditi, cěstiti, čists, kopati, kowo, měsęcs, owsca, *ssrdsce *dstji, *golss, polsza, lsgati, mozgs, *žeraws stigno, stiže, gosts, zělo cěna, kolo, oči, *wslks *-běgo, *gojs, gowędo, žiws goniti, gorěti, gsnati, požars, sněgs, ženo, žeraws *blsxa, *boršsno, bysta, byxs, byše, *golss, gosts, jesms, męso, měsecs, měxs, msxs, rosa, sols, sěme, smějo se, sněgs, sta, stignoti, stojati, struja, syns, šujs, uxo, *wsrxs

214		A SKETCH HISTORY 34.2–35.1	
n	-	cěna, gъnati, junъ, luna, mętǫ, nebo, *nosъ, ǫzъkъ, pěna, pǫtь, sěmę, sněgъ, synъ, znati	•
r	-	berq, *berza, *boršьno, bratrъ, gorěti, požarъ, rosa, struja,	,
		*srьdьce, trьje, turь, *wьrxъ, wydra, žerawь, žerawъ	
1	-	bljudǫ, *golsъ, kolo, lajǫ, lĕvъ, ljubъ, luna, polьʒa, lъgati, *pelnъ, solь, *wьlkъ, ʒělo	,
w	-	lėwъ, owbca, wesna, wezą, *wblkъ, *wbrxъ, wbsb, žiwъ	
j	-	роčьjetъ, pokojь, smějo sę, trьje, junъ	
( <b>h</b> <sub>1</sub> )	-	dělo, děti, jesmь/sqt- <sup>v</sup> /, *рыпъ	
( <b>h</b> <sub>2</sub> )	-	bratrъ, byxъ, *dъtji, dymъ, *kry, ostrъ, [stojati,] uxo	
( <b>h</b> <sub>3</sub> )	-	*gojь, oči, znati, žiwъ	
ą	-	kopati, nosъ, solь	
ā	-	bratrъ, laję, owъca, rosa	
ai	-	lěwъ, zělo	
au	-	turъ, kowati, junъ, ostrowъ, struja	
ě	-	berq, *berza, desętь, jesmь, medъ, nebo, *pelnъ, teplъ, wesna,	,
		wezo, *žerawъ, *žerawь	
ē	-	*-běgo, dělo, děti, ědętъ, měsęсь, pěna, požarъ, sěmę, smějo sę, sněgъ, stojati	,
ei	-	čistъ, stigno, truje, zima	
eu	-	bljudq, ljubъ, nowъ, šujь	
ĭ	-	jesmь, gostь, mьzda, owьca, *sьrdьce, vьsь	
ī	-	oči, žiwъ	
ŏ	-	*boršьno, domъ, *golsъ, goniti, gorěti, gostь, kolo, mozgъ, nosъ,	,
		оči, оwьca, rosa, topiti	
oi	-	cěditi, cěna, cěstiti, *gojь [?], měxъ	
ou	-	gowędo, junъ, luna, ostrowъ, struja, uxo	
ō	-	znati	
ų	-	*blъxa, *dъtji, lъgati, тъхъ	
ũ	-	dymъ, myšь, synъ, wydra	
[ə	-	stojati $<$ *sth <sub>2</sub> - (the laryngeal yields a vowel, $\vartheta$ : this "schwa" symbol is widely used in older scholarly literature)	

**35.** The general outlines of phonological development and the morphophonemic structure of inflectional patterns are clear enough, but the individual histories of many morphemes and words are full of puzzles.

**35.1** A striking difference between the Indo-European and OCS phonemic inventories is that IE had a single continuant obstruent, s (with allophonic [z] before voiced stops), while OCS has two voiced continuants, z and  $\check{z}$ , and at least three voiceless ones  $(s, \check{s}, and x)$ ; a possible fourth  $(\hat{s})$  may be posited for the pronominal stems sb 'this' and wbsb 'all' and/ or in such declensional forms as Ls *dusě* and Np *dusi* (from *duxb* 'spirit'). Late IE \*s split into s vs.  $\check{s}$  by the ruki-rule (§25.3). New and specifically Slavic morpheme-shapes with s appeared as the IE palatal stops  $\check{k}$  and  $\check{g}$ became s (merging with and indistinguishable from the extant s) and z (a new phoneme of transparent origin), cf. §27.1. At a later date, sj and šj merged in š which could be followed only by  $\bar{i}$ , i, e,  $\bar{e}$  (§27.4); š in any other position backed to x (see examples 17–31 on page 207). By VR<sup>2</sup> (§27.72),  $\bar{s}\bar{e} > \bar{s}\bar{a}$  and the contrasts  $s\bar{a} \sim s\bar{e} \sim \bar{s}\bar{a} \sim x\bar{a}$  (and the exclusion of  $\bar{s}\bar{e}$  and  $x\bar{e}$ ) are established in lexical entries and phonotactic rules. Schematically:

**35.11** Initial x and š are well documented in LCoS. Where did they come from? The ruki-rule has been invoked to explain the variants xod and šbd 'go'. They reflect the IE root \*sed: the o-grade \*sod-o- (cf. Gk  $\delta\delta\delta\varsigma$  'way, journey', whence E odometer) and an anomalous zero-grade \*sid, after prefixes like \*per-, \*prei-. Thus \*per-sod-os > \*peršadas, which provided the stem for a denominative verb, yielding OCS prěxodz 'crossing, transition' and prěxod-i+ 'cross'; \*prei-sid-us- > \*prei-šid-uš- > OCS prištdz 'having arrived'.<sup>35</sup> The numeral šestb 'six' may reflect \*kseks-ti-s > \*kšekstiš > \*šesti; in any case, Baltic also has initial š. Two roots have š from \*sj: \*sjū-tei > \*sjītei > \*šiti 'to sew' (§15.93), and \*seu-j-os > \*sjaujas> šujb 'left' (cf. §29.95). Šum- 'sound (of water, wind)' and šbpbt- 'whisper' seem to be onomatopoetic.

35.12 Many roots or stems beginning in x or  $\check{s}$  are plausibly explained as borrowings. New loans from Greek are clear—including some where x is followed by a front vowel (xerowims 'cherubim'; xitons 'chiton, tunic, undergarment)'. Germanic contributed xlebs 'bread' (< \*hlaib-, cf. Go hlaifs, G Laib, 'loaf'), and šlěmz 'helmet' (< \*xelm-, < Gmc \*helmaz). Šarā 'color' (šaropisatelja 'painter') is from Turkic, presumably Proto-Bulgarian, the language of the founders of the Bulgarian state. Iranian origin is called on to explain several roots (e.g. xwala 'praise'), but the resemblances between the Slavic and Iranian words, tempting though some of them may be, do not fit into systematic patterns that establish acceptable evidence. Another line of investigation is to posit \*ks as a result of \*sk or \*sg, with "mobile s" (coopted from a desinence-final consonant of a preceding word). Proposals are imaginative but, in my opinion, unconvincing. Most roots with initial x remain unexplained—as do some with š, e.g. šir 'wide, broad' (OCS široka), šija 'neck' (marginal

<sup>&</sup>lt;sup>35</sup> Just why these particular allomorphs achieved independent status (while *s*-variants predominated in all other roots) remains unexplained.

in OCS but wide-spread in later dialects),  $\xi t$  'rage, behave violently, insolently'. Proposals that appeal to a putative IE \*x or \*kh are no longer taken seriously.<sup>36</sup>

35.13 Stem-final x and š are often hard to account for.

(a1) An s after LCoS i, y, u, b, or  $\mathfrak{s}$  may represent IE \*k (e.g.  $p\mathfrak{s}ati$  'to write' §27.6) or a consonant cluster with a stop other than \*k or \*g (e.g. \*ks) \*osb 'axis' < ECoS \*assis < IE \*aks-i- [cf. Li ašsis]), wys-ok- $\mathfrak{s}$  'high' < \* $\overline{u}ps$ -, cf. Gk  $\psi\eta\lambda\delta\mathfrak{s}$  'high, tall'). Kys-nq+ 'ferment' and kysěl $\mathfrak{s}$  'sour' go back to \* $k\overline{u}p$ -s-, cf. OCS kyp-ě+ 'to boil'.

Thus the stative  $wis \cdot \check{e} + ibe$  hanging' and causative  $w\check{e}s \cdot i + ibang'$  (perfective  $ob \cdot \check{e}s \cdot i + ibang'$  terative  $ob \cdot \check{e}s \cdot aj + ibang'$  suggest  $weik \cdot ibang' + woik$  or else  $weiC \cdot s + woiC \cdot s$ ; attempts to find suitable cognates in any other language have not been successful.

(a2) Or the stem may be borrowed: e.g. kusi+ 'taste, try, tempt'; \*kausis from Germanic, cf. Go kausjan (< IE \*geus- [> E choose]).

(b) Stem-final - $\check{es}$ - may go back to \*aiCs. It has been argued that  $\check{bess}$  'demon' and  $\check{beda}$  'danger, catastrophe' and *bojati se* 'be afraid' (cf. Li *baisùs* 'frightening, scary', *baidýti* 'to frighten') demonstrate a late IE \*boid-s-.

(c) The verbs was-krws-nq+, was-krws-aj+ 'rise again' and was-kres-i+ 'resurrect' imply zero-grade \*kris vs. o-grade \*krois (§42.14). The persistent use of this stem in OCS to denote the fundamental Christian conception of rising from the dead (rather than the more literal wa(z)-sta-, wa(z)-staw-aj-, wa(z)-staw-i 'stand up, cause to rise' that occasionally appear) surely indicates that the earliest translators found it an effective translation. Post-OCS data associate \*kres/\*krws with the summer solstice and pre-Christian ritual festivities (particularly bonfires); the exact nature of these practices and beliefs remains unknown, but a connection with rebirth and new life seems highly plausible. Li has a formally suitable verbal root kreip/kraip 'turn, change direction'. Some scholars believe that this sense can be accomodated with hypothetical PBS \*krips/\*kroipsand a meaning involving new life.

(d) The adjective pěšb 'on foot, walking' (e.g. po njemb ido pěši 'they followed him on foot' Mt 14:13) or 'walker, pedestrian; foot-soldier'

216

<sup>&</sup>lt;sup>36</sup> A recalcitrant problem is R *soxá* 'wooden plow' ~ Li *šakà*, Sa *śaknā* 'branch, forked stick' and many other apparent cognates.

surely is based on IE *ped/pod* 'foot', a root known in most IE languages (cf. La *pēs*, Gs *pedis*, Gk  $\pi o \tilde{v}_5 \pi o \delta \delta_5$ ). We can invent a suitable phonetic ancestor, \**pēd-s-jo-*, but the *s*-element has no obvious parallel.

Cz pěchota 'infantry' is known from 1487. It appears in Polish about 1600, and then in a Ruthenian text in 1609. It seems therefore to be a fairly modern formation, but easily travelled from one Slavic language to others. It was taken into Slovene in the 19th century.

(e) The OCS noun *směxs* 'laughter' is obviously related to *smbj-a+ směj-qts* 'laugh' and the IE root \**smei*, but it is uncertain whether the x represents an old \**s* (\**smoi-so-?*, \**smōi-so-?*) or a newer and independent \**x*. An iterative stem \*-*smis-aj+* 'mock' appears to be a recent formation with the formant \**sā*. Similarly, *pospěxs* 'capability' and *uspěxs* 'success' go with *spěj-Ø+* (§15.45), IE \**speH* 'thrive, prosper' (cf. Gmc \**spō-di-z*, E *speed*; La *spēs* 'hope').

(f)  $Gr\check{e}x\mathfrak{s}$  'sin' strongly points to  $gr\check{e}j-\emptyset+$  'to warm', but the Christian sense is apparently adapted from a native meaning 'miss the mark, fail to do correctly'. Perhaps we should posit \*groi-so-, suggested by early  $gr\check{e}za$  'confusion' (early Rus', possible OCS) and Li graīžas 'rim' < \*groi-go-, cf. Latv greizs 'oblique, slanting'.

(g) The x in jaxa- 'ride' (\$16.4) and ob-qx-aj+ 'to smell, have sense of smell, perceive by smelling' (a synonym of ob-(w)onj-aj+) must be considered new suffixal elements, perhaps no older than LCoS.

(h) Isolated suffixes (e.g. in *ženixs* 'bridegroom', cf. *žena* 'woman, wife', *ženi*+ 'marry'; *pustošs* 'trivial matter, trifle', cf. *pusts* 'empty; *junoša* 'young man', cf. *juns* 'young') provide no helpful information. We may surmise that some of them are affective, usually hypocoristic but sometimes pejorative. Later Slavic dialects provide a wealth of affective x/s suffixes.

(i) The problematic sibilant in the pronominal stems cb ce cu 'this' and Bbcb Bbce Bbce's 'all' is treated in §40.11-40.123.

**35.14** Slavic luna 'moon' is (like, but entirely independent of, Latin luna) derived from \*louk-snā- (cf. luča 'ray' < \*lauk-j-ā,  $\mathbb{E}$  \*leuk 'light' [> Gmc \*leuk-to- > E light]). The sequence \*ksn > \*kšn > n, while in OPrussian ksn remains: louxnos 'stars'. OCS čыпъ 'black' corresponds to Sa krsná- 'black', OPr kirsnan. Expected ECoS \*kirsna- > \*kiršna- > MCoS čыnъ. Such etymologies are opaque; they require information not available in the Slavic forms.

36 Constraints on initial vowels belong to dialects of Late Common Slavic. In many regions the constraints changed over time; unfortunately, lack of documentation makes it hard to follow many important details. It appears that LCoS generally disallowed #a-, and added prothetic *j* regardless of etymology. Yet certain eastern Bulgarian dialects preferred #a- to #ja-, and removed etymological j at least in some words. The OCS and later medieval texts are inconsistent, and more modern sources present further complications that cannot be discussed here.

**36.1** The two ECoS roots that began in  $*\bar{e}$  are assumed to have preposed *j*: then  $*j\bar{e} > *j\bar{a}$  in SE LCoS: OCS *jadb* 'food', *jasti* 'to eat' (§16.22). The spelled " $\check{e}$ " in forms of  $\check{e}xati \check{e}dqta$  'to ride' (§16.4) is justified as standing for  $j\check{e}/ja$  by modern forms like Uk їхати їдуть. On the other hand, SC and SIn have only *jahati* (with a new j-present *jaše*). The etymology is surely  $*j\bar{a}$ - (Li *jóju jóti*, Sa yāti 'goes').

**36.2** (a) For *az*<sup>5</sup> and *jaz*<sup>5</sup> 'I', see §40.311-12.

(b) OCS avi+/avljaj+ 'reveal, make clear' ( $av\check{e}$  'openly, in the open, manifestly, clearly') are somewhat less frequent than javi+, javljaj+, or javljenbje, but no j is indicated in obavi+ obavljaj+ 'show, reveal', obavljenbje 'appearance, apparition' (or the rare obavaj+ 'charm, work magic', obavbnik's 'charmer, magician'). The PBS root  $*\bar{a}w$ - 'clear, mani-fest, reveal' fits with \* Sa  $\bar{a}vis$  'obvious', and, more remotely, La *audēre* 'hear'. The modern Slavic languages have only jav-.

(c) Unsuffixed \*aje/\*jaje 'egg' is less common in Slavdom than \*ajbce (OCS Ap aica)/\*jajbce: \*aj-e suggests IE \* $\bar{o}y$ -on, a form that is comparable to Gk  $\phi$ ov; speculation about relationships with \*awi- 'bird' remain inconclusive. Czech and Slovak have initial \*w: \*waj- > OCz, Slk vajce, mod. Cz vejce. Upper Sorbian has wejo and jejo. The \*waj- probably is the result of late and local dissimilation from \*jaj-.

(d) The root \*agn- 'lamb' (OCS'agnę, agnьcь/jagnьcь) presupposes long \* $\bar{a}$ , with LCoS prothesis of *j*. The initial vowel is short in Lat agnus, Gk  $\dot{\alpha}\mu\nu\dot{\alpha}\varsigma$ ; evidence from other languages suggests \* $g^{w}$  and \* $g^{w}h$ .

(e) The *apple*, though not attested early, surely was \**ablvko* in some parts of Bulgaria, and \**jablvko* elsewhere in LCoS. Late IE \**abel-* or \**abl-* (cf. Gmc \**apalaz*; Li *óbuolas, obuolỹs* 'apple') became a u-stem, to which a regularizing (diminutive?) \*-*ko-* was added. The *j-* is present in all the standard languages, but the gender varies: R *jábloko*, Uk *jábluko*, Mac, Sln *jabolko*, Cz, Slk *jablko*, P *jablko*; Bg *jabvlka*, SC *jabuka*; BR *jáblyk*.

(f) Agoda 'berry; strawberry' (OCS Mt 12:35 Sav 'fruit'), with Li ioga, establishes ECoS  $*\bar{a}g$ -ad- $\bar{a}$  and PBS  $*\bar{o}g$ -. It goes with IE  $*h_3eg$ -, which in zero-grade underlies Gmc *ak-ran*-, E. *acorn*. The *ja*- is now in all the Slavic languages.

A SKETCH HISTORY

(g) The phrase Aπst covct ABOPOSt (Ds) translates ἐπί δένδρου πλατάνου ξηροῦ 'to a dry X tree'; avorovā is an adjective based on \*avorā, now represented by forms with initial *ja*- in Sl dialects. It is the sycamore, or maple, or plane-tree; the Greek here has πλάτανος 'plane-tree'. Whatever this passage may have meant, \*avorā is probably a borrowing from Old Bavarian \*āhor (OHG ahorn) 'maple'; the initial *j*- is added in LCoS.

(i) Not mentioned in early texts is \*jasenb 'ash-tree', which has been seen as based on IE \* $\bar{a}s$ - with an \*n suffix \* $\bar{a}sen$ - (cf. La ornus). Gmc used a \*k suffix: \*askiz > G Esche.

(j) The noun pojass 'belt' and verb pojasa+ 'gird' retain the \*j- of IE \*yōs (cf. Li júosta 'belt'; Gk  $\zeta \omega v \eta$  'girdle' < yōs-nā). The adjective jars 'furious' may come from IE \*yōr (< IE \*yeH-r-), cf. Gk  $\zeta \omega \rho \phi \varsigma$  'sheer, undiluted', used of wine; a semantic association of "strong, forceful, heady" and "furious" is assumed.

(k) Other words beginning in *ja*- in many or most Slavic languages have no plausible etymologies. Among them are OCS *jasnъ* 'bright, clear'; *jama* 'ditch, trench, pit'; *jarъmъ* 'yoke' (or neuter \**jarъmo*, post-OCS); *jazwa* 'wound' and *jazwina* 'cave, burrow, lair'; along with later-attested \**jalъ* and \**jalovъ* 'barren, sterile, unable to reproduce'; \**jarębъ* 'partridge'; \**jastrębъ* 'hawk'.

**36.3** Prefixed stems or compounds offer some contrasting forms.

(a) paqcina and pajqcina 'spider-web' imply paqks/pajqks 'spider': -qks (cf. R  $\pi ay\kappa$ , SC pauk) is compared to IE  $h_2onkos$  'hook' (La uncus 'hook', Gk  $\delta y \kappa os$  'barbed hook', Sa  $a\pi kas$  'hook, bend' and the shape of a spider's legs.<sup>37</sup> The *j* must be an ECoS addition, attested in farflung dialects: e.g. P pajak, Sln pajak, Mac pajak, dial. pajek. Uk  $\pi aBy\kappa$  and Cz pavouk illustrate later prothetic w/v before *u* in certain regions.

(b) rqkowqtb and rqkojqtb 'sheaf' are analyzed as "amount one can grasp with both arms"; they seem to show older and newer compounding. The first is ECoS \*ronk-au+im-t-i(š), GL dual (OCS rqku) plus a noun-stem with the root. The second is a normal LCoS compound: rqk-o- is the usual combining form, and j is expected as prothesis to older bm or newer q.

<sup>&</sup>lt;sup>37</sup> The pa- is an old lengthened variant of po, used in nominal formations; paguba 'ruin, destruction' (cf. pogubi+ 'destroy'), pamets 'memory, memorial' (<\*pāmin-ti-s, cf pomeně+ 'remember'), and pažits 'meadow, pasture, feeding-ground' (cf. živ- 'live'), see Vasmer sub na-.

**36.41** OCS blagovonjunt and blagoxanunt 'sweet-smelling, fragrant' are synonymous, as are the verbs obonjaj + and obxaj + 'smell, perceive by smelling'. The roots wonj and qx surely go back to ECoS \*an and IE \* $h_2enh_1$  (Sa aniti 'breathes'; Gk aveuos 'wind'; La \*anamos > animus 'spirit'). The reasons for prothetic \*w in one and final \*x in the other are unknown.

**36.42** OCS szązz and szyązz share the meaning 'bond, fetter; union'. The root is qz- < PBS \*ang < IE \* $h_2$ engh 'tight, constricted' (La angere 'draw tight' [> E anxious], angustus 'narrow [> E anguish]; Gmc \*angaz, E anger). Otherwise qz appears in qza 'bond, fetter', qze {qz-j-0} 'chain, rope; snare', qzskz 'narrow, tight' (comp. qze), qzika m/f 'relative' (< ECoS (anz-j-ei, qzikz stroo 'kin, relatives'. The phrase zzreti podz obqzomz means 'look askance, with suspicion'. Prothesis is unexpected. (Prothetic j in R coio3, a much later borrowing from Bg Slavonic, is also unexplained.)

**36.43** The synonyms  $qz \omega n k \overline{s}$  and  $q \overline{z} \omega n k \overline{s}$  'prisoner' ( $qz i l i \overline{s} t e$  'prison') compete with  $t \omega n \omega n i \overline{c} \omega n k \overline{s}$  ( $t \omega n \omega n i \overline{c} \alpha$  'prison';  $t \omega n \alpha$  'darkness') and  $s \overline{\omega} v \overline{c} z \omega n \overline{b}$  ( $s \overline{\omega} v \overline{c} z a +$  'tie up'). The root  $w \overline{c} z$  'twist, tie, entwine, ensnare; wreathe, crown' (cf.  $uv \overline{c} z - \emptyset +$  'crown' §15.822,  $ob \overline{c} z a n \overline{c}$  'diadem') suggests ECoS \*wenz, and possibly older \*weng. More remote ties with the ancestor of  $-v \overline{b} / -v \overline{i} -$  'wind' (§15.93) and \*wens 'wreath' (cf.  $v \overline{e} n \overline{b} c \overline{b}$  'wreath', Li vainikas) seem possible. A connection with qz- 'narrow, tight' is improbable.

**36.44** The word meaning *mustache* (and perhaps *upper lip*) was \*qsa (not OCS), usually used in the plural. If OPrus *wanso* '(beginning) facial hair (on youth)' is a cognate (and not a borrowing from Slavic), initial *w*-was deleted in part of Bulgaria and perhaps Rus', cf. R ycbí.

OCS gosěnica 'caterpillar' seems to have a LCoS parallel \*osěnica, perhaps \*osěn- or \*osen- meaning 'fuzzy, hairy'. The g is puzzling.

**36.45** \*OCS osa 'wasp' surely represents PBS \*wopsā (Li vapsvà; metathesized form Gmc wosp-, La vespa); the initial \*w has been lost. Contrast OCS voska 'wax' (cf. Li vāškas; OHG wahs).

OCS\**język*<sup>5</sup> 'tongue', by comparison with OPrus *insuwis*, Go *tuggō*, OLa *dingua*, suggests ECoS \**inzū* < PBS \**ngū*. The initial consonant of IE \**dngū* has been lost, and later prothetic \**j*- was added; the complexities of  $\bar{u}$ -stem declension have been obviated by the addition of stem-final \**k*-.<sup>38</sup>

220

<sup>&</sup>lt;sup>38</sup> Latin changed initial d to l in several words, possibly indicating borrowing from a related dialect. Li *liežùvis* has modified the first syllable seemingly in accord with the verb *liēžti* 'lick' (cf. OCS *liza*+ < IE \**leigh*-).

36.46-37

**36.46** OCS root-initial vr and vl imply ECoS syllables with liquid diphtongs (§26.511), e.g. vlbk $\mathfrak{z} < *wblk\mathfrak{z}$  'wolf', vrag $\mathfrak{z} < *worg\mathfrak{z}$  'enemy'. Older initial \*wr- lost the glide: \*wrughyo- 'rye' > \*rugj-a-s > \*r\mathfrak{z}\mathfrak{b}. It is possible, however, that the early Slavs borrowed Gmc \*rugi- (cf. E. rye). There are no plausible examples of initial OCS *l*- from hypothetical PBS \*wl-.

**36.5** Where South and West Slavic have initial *je-*, East Slavic often has o-: e.g. *jedina* ~ odina 'one', *jedava* ~ odava 'hardly', *jelenb* ~ olenb 'deer', *jesenb* ~ osenb 'autumn', *jezero* ~ ozero 'lake.

Henning Andersen (1996) has examined the full list of these words in a broad linguistic and geographical framework of Baltic and Slavic. He offers stimulating suggestions as to how language shift (Baltic to Slavic dialect) and eroding dialect boundaries combined over time to produce distributions of these words that could not have resulted from internal development in a settled and homogeneous community.

**36.6** South and West Slavic initial *ju*- correspond to Rusian *u*- in three roots: *jun*- $\mathfrak{s}$  'young', *jug*- $\mathfrak{s}$  'south', and *juxa* 'broth, soup'- only two of which have solid etymologies. (*Juns* is cognate with Li *jáunas*, E *young*, and the root in the La comparative *jūnior*. *Juxa* goes with Sa *yus* 'broth', OPr *juse* 'meat broth', La *jūs* 'broth, soup' [whence E *juice*]. No satisfactory source has been found for *jug* $\mathfrak{s}$ .) OCS *uže* and *juže* 'already' are interchangeable, while *ne u* 'not yet' is clearly the norm, *ne ju* being rare. OCS *utro* 'morning' (with derivatives) is normal, but *jutro* occurs. In more modern times, *utro* is eastern (ESI, Bg, Mac) and *jutro* (in appropriate phonetic shapes) is western (SC, Sln, WSI). Matters are complicated by OCS *zaustra* 'in the morning, the next morning' (and mod. Mac. dial. *zastra* 'tomorrow morning'), plus OP *justrzenka* 'morning star' and other forms that suggest \**ustr-* < \**usr-*. The etymology is disputed.

## Morphology

37 OCS morphology is unmistakably Indo-European, but the inherited elements have been rearranged. Though the declensional system is conservative (and is relatively close to Baltic) and exhibits familiar patterns, the detailed evidence of the oldest texts shows that a fundamental reorganization is still under way. Post-OCS regional data affirm that our earliest texts illustrate transitional phases of a system undergoing rapid change. Conjugation in OCS has been radically reshaped (and differs substantially from Baltic). The outlines of the verbal system are simplified IE. The materials are for the most part recognizable morphemes or parts of morphemes, yet the combinations are new. **37.1** The basic form classes of the IE nominal system remain in OCS: substantives and adjectives; demonstrative, relative and interrogative pronouns; and personal pronouns. The framework of three genders (masculine, neuter, feminine) and three numbers (singular, dual, plural) is intact, but OCS has lost one IE case—the *ablative*, which has merged with the genitive. (The Baltic system generally agrees with Slavic. Greek very early merged genitive and ablative, while the dative absorbed the old locative and instrumental.)<sup>39</sup> Like IE, Slavic had special vocative forms only for masculine and feminine singular.

**37.21** Late Common Slavic declensional stems always end in a consonant. Desinences are vocalic, with the outstanding exception of the nominative (and sometimes accusative) singular zero-marker that characterizes active participles and a list of special stems (including one productive formant  $\{qt\}$ ,  $\{4,414\}$ ). From the lexical form of a stem, including its inherent gender, the array of desinences is predictable; morphophonemic alternations in the stem-final C (or cluster) are determined by mutation rules (KI, KAI). In the exceptional cases where a terminal zero-desinence is to be posited, alternations may affect the VC that is the end of the inflectional stem. The vowels within a stem otherwise remain constant in all paradigmatic forms.

**37.22** Indo-European had a very different system, where a paradigm could include vowel alternations (apophony) in root, derivational suffix, and desinence. There were several types of stems that ended in consonants (C-stems), some consisting only of a root and others with consonantal derivational suffixes. The final C or V of a stem could combine with desinence-initial C or V, often with phonotactic changes. Desinences often ended in a consonant. The details of some of the paradigms are disputed, and citable forms clearly show local modifications in the behavior of individual words.

**37.31** Some important nouns were *heteroclitic*, utilizing different stemsuffixes within a paradigm. For example, the "r/n" or "l/n" stems contrasted NAV singular to other forms by a liquid-suffix  $(-r-/-er-/-\bar{e}r-; -l-)$ opposed to a nasal-suffix (-en-/-on-/-n-), e.g. Hittite 'water', Ns *wat-ar*, Gs *wet-en-as*.

**37.311** The hypothetical root \*wed 'wet, water' is represented by a bewildering range of variant forms; it may have had Ns \*wed- $\bar{o}r$  (Gk \*ud $\bar{o}r$  ὑδωρ) but Gs \*ud-n-és (Sa udnás).

<sup>&</sup>lt;sup>39</sup> Such mergers are called *syncretism*. In Latin, the old locative and instrumental functions were taken over by the ablative.

Greek augmented the n-suffix with \*t (and \*nt > at),  $\delta\delta\alpha\tau\sigma\sigma$ . Lithuanian has a root with a nasal diphthong and a stem-suffix -en- that is modified before a zero desinence: Ns vanduõ, Gs vandeñs. English wet goes back to  $*w\bar{e}d$ -o-, winter (the wet season) to \*we-n-d-. Latin unda 'wave' (cf. E inundate, undulate) is from \*u-n-d-ā-. Slavic vydra is cognate with E otter, and vědro 'bucket' presumably has this same ancient root. The noun for 'water' had surely become a regular a-stem, voda, by MCoS times.<sup>40</sup>

37.312 Hypothetical IE \*seh<sub>2</sub>wol 'sun' justifies the shape \*sāwel-jo- as ancestor of older Greek  $\hbar(\lambda)$ os, classical  $\hbar(\lambda)$ os, and the variants \*swōl- for La sōl, \*sowl- for Li saúlė, and \*swen-, \*sun- for Gmc \*sunnōn > Ger Sonne, E sun. Slavic evidence establishes MCoS \*sblnbce, from which we imply an early \*sln plus a suffix \*-iko-. The end result is a regular Slavic noun {sbln-bc-e} which is of little help in establishing IE relationships, but is itself indubitably of IE origin. Similarly žena 'woman' (implying ECoS \*gen-ā) is a regularized noun ultimately derived from IE \*g\*en, the root underlying E queen, as well as the Gk morpheme gun in gynecology, polygyny and many other learned words.

**37.32** Slavic has adapted its few remaining consonant-stems to the i-stem paradigm (see the table on page 224 for the IE desinences). Moreover, i-stem feminines took on certain distinctive desinences, while i-stem masculines tend to adopt twofold desinences.

These remnants of the IE consonant-stems are therefore called *the* anomalous type of simple declension in this book (\$4.1-.414).

For example, the OCS masculine NA  $zv \check{e}rb$  'wild animal' corresponds to IE As  $*g^{h}w\bar{e}r-m$ : the syllabic nasal had become \*-*im*, indistinguishable from the i-stem desinence. Similarly Ap  $zv\check{e}r-i$  could reflect \*-*ns* or i-stem \*-*ins*. The lexeme eventually is redefined as an i-stem. Within OCS, masculine i-stems are being reclassified as regular soft twofold nouns (cf. §4.5).

**37.4** The vocalic stems were defined by formants that were subject to ablaut (apophony). The formants \*-i (with others like \*-ti, \*-ni) and \*-u are zero-grade; e-grade \*-ei (\*-tei, \*-nei) and \*-eu also appear.

37.5 The major paradigms for masculines and neuters contain the suffix known as the *thematic vowel*, symbolized by  $*^{o}/_{e}$ : it usually appeared as \*o, but in certain morphemic contexts as \*e. It may mark a root as a noun (e.g. Gk \*tok-o-s 'birth' ~ \*tek 'beget, give birth to' [< IE \*tek]; cf. Sl tek- $\emptyset$ + 'flow, run' [< IE  $tek^{w}$ ] ~ PBS \*tok-os 'flow')<sup>41</sup> or it may be the final element of a formant (e.g. \*sup-no- 'sleep'), OCS tok- $\mathfrak{d}$ ,  $s\mathfrak{d}n$ - $\mathfrak{d}$ . Closely related is the suffix  $*\mathfrak{d}$ , which provides the feminine forms of adjectives and, in Slavic, a wide variety of substantives.<sup>42</sup>

<sup>&</sup>lt;sup>40</sup> Heteroclisis was apparently absent from LCoS, but it reappears in new forms in later dialects, e.g. the R type τελёнοκ 'calf' (with -eH plus -κ) vs. pl. τελяτα (with ят < *ęt*).

<sup>&</sup>lt;sup>41</sup> Some scholars consider the Gk and Slavic to represent a single IE root.

<sup>&</sup>lt;sup>42</sup> The variable e vs. o represents IE grades of apophony (ablaut); it also appears in the formant  $j'_o$ . The Slavic morphophoneme {o/e} is a new entity that is governed

38 Any attempt to trace the history of desinences has to reconcile internal Slavic reconstruction with heterogeneous IE comparative evidence that has serious gaps. The reduction in number of formal paradigms has involved complex processes of analogy. Sometimes the pieces of evidence do not fit together. The behavior of word-final syllables may require special reference to the word-boundary (#), see §24.52.

**38.1** The inflection of pronouns in IE differed in certain critical respects from that of substantives. Alternate stems are combined in paradigms that vary considerably from one dialect to another.

This table summarizes evidence that may be pertinent for Slavic. It contains controversial items that cannot be treated here. In any case, many more variants are required to account for all of the early IE languages.

Note that the -m- of Instr sg and pl and Dat pl is reflected by Slavic, Baltic and Germanic, as opposed to -bh- of all other groups.

The IE ablative was formally like the genitive except in the singular of o-stems; there the ablative desinence took over genitive function in pre-Baltic and Slavic.

	C-stems	i-st.	<b>u</b> -st.	o-st.	ā-st.	pronom.
Ns	s, Ø	is	us	os (om)	ā	o   od id   ā
As	ŵ	im	um	om	ām	om l ām
Voc sg	Ø	ei	eu	e	ă	-
Gs   Ab	es/os/s	eis, ois	eus, ous	os(j)o l õd	ās	osjo I esjās
Ls	i, Ø	eyi, ē(i)	ewi, ēu	oi	āi	
Ds	ei	eyei (eyai)	ewei (ewai)	ōi < o-ei	āi	osmōi   esjāi
Is	mi	imi	umi, ū	ō	āmi	ajā
Np	es	eyes	ewes	ŌS	ās	oi
Ар	ņs	ins	uns	ons	ās āns?	
Gp	ōm	iōm, yōm	uõm, wōm	ōm	?	oisōm
Lp	su	isu	usu	oisu	āsu	oisu
Dp	mos	imos	umos	omos	āmos	omos
Ip	mis	imis	umis	ōis	āmis	ōis
NAdu	he <sub>1</sub> ?			ō		ōloī

by specifically Slavic phonotactic environments. The IE feminine \*- $\bar{a}$  in the older language is \*- $eh_2$ , that is, the e-grade of the thematic vowel plus a laryngeal consonant that "colors" the vowel (e > a) and lengthens it ( $a > \bar{a}$ ) before disappearing. The formant  $je_{a}$  had a fem. Ns with the thematic vowel ( $*jeh_2$ ) and one without it (\*- $ih_2$ ); the glide becomes *i* and is lengthened, yielding - $\bar{i}$ . (Cf. §4.18.)

A SKETCH HISTORY

**38.2** IE u-stems included neuter nouns, e.g. \**medhu-* 'honey' (cf. Sa *madhu*). Early Slavic has only masculine u-stems (including *medv*).<sup>43</sup> OCS demonstrates that o-stems have adopted some u-desinences, while the u-paradigm has disappeared; its remnants are preserved in the form of certain lexemes marked to take special optional desinences, see §4.145.<sup>44</sup> IE u-stem adjectives have been adapted to the twofold paradigm by adding suffixal \**ko*: e.g. \**slād-u-ko-* 'sweet', is reinterpreted as \**slad-vk-z, sladvko, sladvka*.

**38.3** IE desinences utilize very few consonants, see §27.42. It would seem that final stops were eliminated fairly early, while the final \*s remained nearly to the historical period (§25.3). Some nasals disappeared, and some survived into OCS in the form of nasal vowels, cf. §29.814. Their history depends in part on their status in special morphemes.

**38.41** IE had separate forms for nominative and accusative (in both singular and plural), except in neuters. IE Ns ~ As of u-stems and i-stems \*-us ~ \*-um and \*-is ~ \*-im became Slavic NA \*-b and -b (synb 'son', gostb 'guest'). IE thematic \*-os ~ \*-om yields twofold -b and -b (e.g. stolb 'seat' ~ \*strojb 'order') for which a more complicated history must be constructed:. a vowel-raising rule (§27.71) operating before #, followed by the vowel-adjustment or fronting rule (§27.72): \*-os \*-om VR<sup>1</sup> > \*-uš \*-um > \*-u ~ \*-i > -b.

IE \*-os > - $\mathfrak{s}$  also in (1) \*-mos of dative plural desinences (gostuma, synama, stoloma, strojema, ženama) and (2) \*-mos of the first person plural verbal desinence -(nesema, nosima 'we carry', něsoma 'we carried', cf. §5.9, 10.602).<sup>45</sup>

IE Lp \*-su > -šu by the ruki-rule (§27.3) in \*-isu, \*-usu, and \*-oisu, and by analogy in  $*\bar{a}su$ , yielding OCS - $\delta x_{5}$ ,  $-\delta x_{5}$ ,

<sup>&</sup>lt;sup>43</sup> Li *medùs* 'honey' is also masculine, but Li and Latv have completely eliminated the neuter gender.

<sup>&</sup>lt;sup>44</sup> In many Slavic dialects, u-stem desinences have acquired special distribution and varied significance. These innovations do not provide evidence that allows us to identify lexemes as LCoS "u-stems".

<sup>&</sup>lt;sup>45</sup> Post jer-shift regional developments introduce a new vowel to 1p desinences: -mo, -me, -my. They surely are innovations, serving to keep such forms as damb 'I give' and damb 'we give' from merging; the plural is marked by adding a distinctive vowel.

<sup>&</sup>lt;sup>46</sup> The vowel e in locative plurals like donexs (and Dp. donexs) is unexpected. Although I have left -exs and -ems in the table in §4.41, I strongly suspect that the 9th century texts regularly had -bxs and -bms.

**38.42** The twofold NA neuter -o/-e (stado 'herd, flock', polje 'field') corresponds to IE \*-om. Why didn't \*-om yield OCS -z? There are two traditional explanations. (1) The IE demonstrative pronoun had \*to for Ns masculine, and \*tod for NAs neuter. The masculine added \*s (from other paradigms) and -os > -z, while the neuter desinence developed regularly as \*-o: the contrast NAsm \*t-z ~ NAsn \*t-o was taken over by o-stem nouns. (2) Neuter s-stems had o-grade \*os before a zero NA desinence, and e-grade \*es elsewhere: NA \*neb-os-Ø vs. Gs \*neb-es-es 'sky'. The final C was lost, yielding \*nebo ~ \*nebese; the NA was reinterpreted as \*neb-Ø-o and this new desinential \*-o was extended to other neuters.<sup>47</sup>

The pronominal forms surely were critical. The influence of the two inherited s-stems (*nebo*, *slovo* 'report, word') have sufficed to introduce the *es* into some forms of other neuters (§4.55), yet this untypical alternation seems unlikely as a model for the class of neuters as a whole. Thus I posit that NAsn \*-*om* was early replaced by PBS \*-*o* or ECoS \*-*a*, then by VR > OCS {o/e}. IE \**yugom* 'yoke'  $\rightarrow$  \**jugo* or \**juga* > OCS \**jbgo*.

**38.43** OCS kamy 'stone' and plamy 'flame' are archaic Nsm, though they and the forms kamens/plamens function also as accusative. Descriptively, they contrast -y to stem-final -en-.. IE consonantal stem in \*-en- could have a Nsm desinence \*-s, or zero with quantitative or qualitative ablaut (\*- $\bar{e}n$ - $\emptyset$ , \*-on- $\emptyset$ ). A hypothetical \*kāmons (\*polmons) for ECoS will produce the desired kamy (plamy), but the combination of elements is implausible for earlier IE.<sup>48</sup>

**38.44** The -i of Ns *mati* 'mother' and  $ds\delta ti$  'daughter' (§4.413) descriptively is correlated with -er, but historically the ancestor is surely  $-\bar{e}r$ , cf. the Gk cognates  $\mu\eta\tau\eta\rho$ ,  $\theta\nu\gamma\dot{\alpha}\tau\eta\rho$  (and the masculine  $\pi\alpha\tau\eta\rho$  'father'). The IE suffix \*-ter- was subject to apophony within the declension: compare the Gk forms Ns  $\mu\eta\tau\eta\rho$  As  $\mu\eta\tau\epsilon\rho\alpha$ , Gs  $\mu\eta\tau\rho\dot{\alpha}s$ . Why \*- $\bar{e}r$  raised to \*- $\bar{\iota}(r)$  is unexplained. OCS  $ds\delta ter$ - and Rusian  $ds\dot{c}er$ - (with *mater*-) allow us to posit earlier \*dukter- (§26.41)-; forms from other languages establish IE \*dhugh\_2tér-.<sup>49</sup>

**38.51** The IE Ap of u-stems and i-stems, \*-uns and \*-ins, became \*- $\bar{u}ns$  \*- $\bar{i}ns$  (§29.814), and deleted the nasal; \*- $\bar{u}$  \*- $\bar{i}$  > -y -i (syny, gosti). This denasalization (NØ<sup>1</sup>) affected originally high vowels, and it was chrono-

<sup>&</sup>lt;sup>47</sup> The generalization of \*š or \*x to desinences may be called on to mark the genitive \*-eš as different from the formant-final sibilant of NA \*slowos; this does not strengthen the argument.

<sup>&</sup>lt;sup>48</sup> MCoS \*kāmen- 'stone' is cognate with Li akmuõ, Gs akmeñs. The etymology is disputed. OCS korenь 'root' appears as kore in a few post-OCS examples; I believe it to be a local innovation. Similarly, R пла́мя (Gs пла́мени) is a borrowing from Slavonic with regularized Ns neuter desinence, like OCS vrěme. The root is \*pol.

<sup>&</sup>lt;sup>49</sup> The parallel IE stem \*bréh<sub>2</sub>ter- (with zero-grade suffix \*tr) > OCS bratro, which lost the r within the OCS period in SSI. In Czech dialects, bratr still exists; brat is the Pan-Slavic norm.

logically prior to VR<sup>2</sup>. C-stem Ap \*-*ns*, behaved like i-stem \*-*ins* > -*i* (*dwni*, *materi*).

**38.52** IE Ap o-stem and a-stem \*-ons/\*- $\bar{o}ns$  and \*- $\bar{a}ns$ , raised to \*- $\bar{u}ns$ , split to \*- $\bar{u}ns \sim *-\bar{n}ns$ . Nasality was lost (by NØ<sup>2</sup>) in the back variant, \*- $\bar{u} \sim *-\bar{n}n > -y \sim -e$ , i.e. {y/e}. The North Slavic reflex, however, is  $\check{e}$  (§29.814). We may speculate that the +high nasal diphthong [ $\bar{n}N$ ] lowered to [-high -low]  $\bar{e}N$  and nasality was lost in these morphemes (and in \*-*en-n*, §29.813), while other front nasal syllabics lowered maximally to [ $\ddot{a}N$ ] before nasality disappeared. The Rusian  $\check{e}$  of all origins presumably was lower than /i/ but higher than /e/ and the / $\ddot{a}$ / that is the normal reflex of ECoS \*e, e.g. \* $s\bar{e}$ -men- $\emptyset$  'seed' > SS1 sěme, NS1 sěmä; Gs \* $s\bar{e}$ -men-es > all LCoS sěmene.

**38.61** The consonantal Np desinence \*-es survives as OCS -e in personal nouns with the suffixes -tel-, -arj-, or -ěn-/-jan- (see §4.52-53). It also is usual in the Npm of comparatives and of present and past active participles: e.g. vęštuše 'bigger'; nesǫšte, nosęšte 'carrying', nesuše 'having carried' (see §4.19).

**38.62** Twofold masculine Np -i (morphophonemically  $\{i^2\}$ ,  $\{3.5c2\}$  goes back to the IE pronominal \*-*oi*. Just why it did not become  $\check{e}$  (per  $\{29.8201\}$ ) is unknown.

**38.7** OCS relics of u-stems have -ove (Np), -ovi (Ds), and -u (Gs, Ls) contrasting with IE\*-ewes, \*-ewei, and \*-eus. This has been interpreted (contrary to §29.8202) as a regular phonological adaptation of \*e to \*o before \*u/\*w, cf. novs 'new' but IE \*new-os. Or perhaps it may be a substitution of o-grade for e-grade vocalism in the desinences (cf. IE Gs \*-ous). The allomorphs -ewe and -ewi are Slavic adaptations used with soft twofold stems, such as zmijeve, zmijevi 'dragon', moževi 'man [Ds]', vračevi (apparently with historical \*j, <\*mong-j-, \*wrāk-j- [\*work-j-?]).

**38.8** OCS i-stems have departed farther from the IE shapes that closely parallel IE u-desinences. OCS i-masculines evolved the appropriate phonological Np desinence -bje < \*-ejes (cf. §29.92), but the feminines have -i (gostbje 'guests' ~ kosti 'bones'). Gs -i matches IE \*-eis, and Ds -i may be from IE C-stem \*-ei. Locative -i brings the i-stems into conformity with the pattern that feminine dative and locative singular share the same form.

**38.9** Slavic vocatives correspond closely to the IE forms, except that the twofold "hard" -e (from the e-grade of the IE thematic vowel [§37.5] is coupled with -u (implying \*-ou, although IE u-stem has \*-eu).

**39.1** Formally distinct nominative and accusative forms were maintained in the singular of a-stems (e.g. žena 'woman', vojevoda 'general' ~ ženq, vojevodq), in masculine present active participles (Nsm nesy, nosę 'carrying' ~ Asm nesqštb, nosęštb), in masculine plurals (e.g. Np stoli 'seats', gostbje 'guests', synove 'sons', dbne 'days' ~ Ap stoly, gosti, syny, dbni), and in some remnants of C-paradigms (§4.413, e.g. Ns ljuby 'love', mati 'mother' ~ As ljubzvb, materb; for kamy 'stone' see above, §38.43)

**39.2** OCS a-stems have final -q in As and Inst sg -ojq/-ejq (e.g.  $\check{z}enq$  $\check{z}enojq$  'woman'; strujq strujejq 'stream'; sqdbjq sqdbjejq 'judge'~ Ns  $\check{z}ena$ , struja, sqdbji [m]). The IE As was  $*-\bar{a}m$  and development to LCoS is straightforward. The IE instrumental may reflect the pronominal  $*-aj\bar{a}$ plus a reduced form of the \*-mi found in other paradigms:  $*-aj\bar{a}m$ .

**39.31** IE instrumental singular \*-*mi* spread from the C-, i-, and u-stems to the o-stems (e.g. gostome 'guest', syname 'son'; stolome 'seat', strojeme 'order'). Feminine i-stems, however, and a-stems have terminal -jq: kostojq, ženojq, strujejq. This seems to be a blend of the fem. pronominal desinence \*- $aj\bar{a}$  with the nasal from other instrumental desinences.

**39.32** North LCoS and perhaps some regional southern dialects extended Is -omb and -omb to twofold stems (*stolomb, strojomb*). The evidence is indecisive because the jer-shift eliminated the distinctions in most dialects and later redistribution of individual desinences differed by locality.

**39.41** OCS final -e of forms like NAs *vrěme* 'time' and *otroče* 'boy' represents a truncated stem-final suffix (cf. Gs *vrěm-en-e*, *otroč-et-e*), §4.414. Similarly in the Nsm of the present active participle of verbs with *i/e* presents (§8.11), e.g. *puste* 'letting go' ~Nsf *pustešti*; *slyše* 'hearing' ~ *slyšęšti*. They seem to reflect older non-alternating front nasal diphthongs.

**39.42** OCS {y/ę} occurs also as the OCS participial Nsm ending, *nes*y vs. Nsf *nes-qšt-i* 'carrying' and *poj-ę poj-qšt-i* 'singing' (verbs that have -onto in 3p pres., §5.602). The participial stem ends in {-qšt-}; and the Nsmn form is anomalous (§8.1). IE morphology suggests a masculineneuter consonantal stem \*-o-nt-, opposed to a feminine \*-o-nt-yā-: Nsm \*-ont-s would be expected. Slavic has generalized the feminine stem with \*j to all genders, with normal soft twofold desinences except for Nsf -*i* (< IE -yh<sub>2</sub>) and Nsmn \*-s. MCoS {-ontj-} > OCS {-qšt-} and Rusian {-uč-}. The environment for the vowel-raising rule (VR) is to be written \*-n(t)s#.

The participial -nt- is used also in verbs with i/e-presents; the front vowel quality may rest on a syllabic \*-nt- or an  $-\overline{t}$ - added before the

nasal,\*-*īnt*-, e.g. OCS Msmn nosę, stoję ~ Nsf nosęšti, stojęšti, from {nos-i+} 'carry' and {stoj-ě+}, 3s pres stojita, 3p stojęta.

**39.421** The North Slavic Nsm participial ending was -a for hard stems (*nesa nesuči* 'carrying'); -onts# yields a. In theory  $-\check{e}^3$  developed in soft stems (\*pojě pojuči 'singing'), as in §38.52, but in fact Rusian spellings are universally of the type non or nota, match or match (§26.52). Scribes do not violate the grammar of OCS. OCz evidence for  $-\check{e}^3$  in this ending is stronger.

**39.5** The OCS genitive plural -5 is often counted a regular development from an IE variant \*-on. The equation is tempting but unfounded, for the argument is essentially circular: the chief "proof" for the IE short-vowel variant is precisely the Slavic -5. IE evidence establishes only  $*-\overline{o}m$  (or something more complex); we must admit that the uniform Slavic \*-u (or \*-i by VA<sup>1</sup>) is unexplained.<sup>50</sup>

Accusative plurals ~ Ns pres. active participles ~ NA i-stems and u-stems										
	VL	NØ۱	VR <sup>2</sup>	VA <sup>1</sup>	Nز					
NApf strauj-an	s# > strauj-āns	;> -	> straujūns	> straujīns	> struję S	SI				
NApf strauj-an	s# > strauj-āns	; > -	> -	> straujēns:	' > strujě N	ISI				
NApf raik-ans#	> raik-āns	> •	> raikūns	> -	> rěky al	11				
Apm stroj-ons	≠ > straj-āns	> -	> strajūns	> strajīns	> stroję S	SI				
Apm stroj-ons	≠ > straj-āns	> -	> ?	> strajēns?	> strojě N	<b>IS</b> I				
Apm stol-ons#	> stal-āns	> •	> stalūns	> -	> stoly al	11				
Nsm nes-onts#	> nes-ānts	> -	> nesūns	> -	> nesy S	SI				
Nsm nes-onts#	> nes-ānts	> -	> -	> -	> nesa N	151				
Nsm poj-onts#	> paj-ānts	> -	> ?	> pajēns?	> pojě? N	151				
Ap sün-uns	> sūn-ūns	> sūnūs	> -	> -	> syny al	n				
Ap gost-ins	> gast-īns	> gastīš	> -	> -	> gosti a	all				
Ns sūnuš	> -	> •	> -	> -	>synъ a	all				
Ns gostiš	> •	> -	> -	> -	> gostı a	all				

**39.61** LCoS instrumental plural \*-mi (synomi, gastomi, kostomi, ženami, strujami) implies an older long vowel or diphthong (\*- $\bar{i}$  or \*-ei), and Baltic allows us to posit PBS \*-m $\bar{i}$ s. Yet IE otherwise points to \*-mis (with short \*i). The long vowel we attribute to PBS remains unexplained.

**39.62** OCS twofold non-feminine instrumental plurals (*stoly, stroji, městy, polji*) imply ECoS  $*\bar{u}$  (or  $*-\bar{i}$  by VA<sup>1</sup>). IE had  $*-\bar{o}is$ , which should

<sup>&</sup>lt;sup>50</sup> The desinential jer in Gp in pre-SC LCoS also requires explanation: despite its eminently "weak" position at the end of a word (§2.621, 2.6241), Gp \*-ъ survives into most dialects as -a with special accentual properties. In medieval mss it is regularly written "ьь", е.g. женьь, градьь.

yield ECoS \*- $\bar{a}i\check{s}$ . We may include this with other cases of raising  $*\bar{a} > *\bar{u}$ in a final closed syllable, noting that the unique diphthong  $*\bar{u}i$  discards \*i. It is far better simply to admit that this desinence resists explanation. The twofold non-feminine dative singular \*-u (stolu, stroju, městu, polju) is also unexplained. It implies \*-au (and IE \*-ou, \*-au), but IE offers strong evidence only for  $*\bar{o}i$  (which—like DLs fem. a-stems—should yield LCoS \*- $\check{e}$  via ECoS \*- $\bar{a}i$ ).

**39.71** The IE comparative suffix was apophonic \*(i)yes \*(i)yos \*is. By what was probably a series of adaptations, non-alternating \*-jijiš- emerged; the feminine had the suffix  $*-y\bar{a}$ - (IE  $*-ih_2$ -, §37.5, n. 42), and the *j* was generalized to all forms: \*jijiš-j- > OCS {jbj-bš}, except for neuter NAs (§4.71). The newer and productive formant began with  $*\bar{e}$ ,  $*-\bar{e}jibs$ - > OCS { $\check{e}j-bš$ }.

**39.72** The past active participle formant  $\{(w)-b\check{s}\}, \S11.11$ , goes back to the zero-grade form of a suffix \**wes* ~ \**us*: \**us*- was used for masculilne and neuter, and \**us*-yā for feminine. In the pre-history of Slavic, the stem \**uš* was generalized, and \**w* was added if the stem was vocalic.

40 The OCS pronominal declension (§4.2) has much in common with the twofold nominal declension. Some desinences belonged originally to one or the other type (see table on p. 224 above). The g in Gsmn t-ogo, moj-ego seems to replace an older s, preserved in the interrogative pronoun česo 'of which' (§4.24). The two-syllable Dsmn t-omu, moj-emu lack the \*s of IE forms, but share the final \*- $\bar{o}i$  that yields OCS Ds -u (cf. §39.62). The -oj- of feminine singular GLDI and dual GL is perhaps based on IE pronominal Is \*- $aj\bar{a}$ .

**40.11** The pronoun sb 'this' takes the "soft" desinences that otherwise are called for by stems ending in j (\*mojb 'my', našb [< \*nās-j-] 'our'); it has an optional variant stem \*sij-, §4.22. Its IE equivalent used suppletive stems: \*ki- and\*kjo-. This would yield ECoSI \*siš for Nsm and \*sj-+"soft" desinence in most forms, then MCoS \*si vs. \*šego.<sup>51</sup> The evolution cannot be reconstructed, but it would seem that the alternation of the root, \*s ~ \*š, was obviated by substituting \*sij- for \*sj or incipient \*š. The end result was a root consisting of a single consonant plus soft desinences; morphophonemically this {s-} requires a mark for this paradigmatic oddity. I am inclined to believe that this is a separate phoneme, \*ś. In any case, the same mark is needed for the pronoun wbsb 'all', whose origin is obscure.

<sup>&</sup>lt;sup>51</sup> Li Nsm šis, Nsf ši; most other forms reflect \*šj- (e.g. Gsm šiõ, Gsf šiõs).

**40.121** OCS www {wwws} takes soft desinences unless the basic variant begins with  $\check{e}$  (§4.21), e.g. GLp www. (like  $t\check{e}x$ ) vs. Nsm www. NAsn www. and  $t\check{e}mb \sim simb$ ). The stem itself varied by region. South Slavic and most of Rus' had \*www. NorthWest Late Common Slavic had \*www. (with the same distribution of desinence allomorphs), while the NorthEast periphery (Novgorod) had \*www. a regular hard stem like ta. This distribution matches the KAI-reflexes of \*xai (e.g. ECoS \*xaid- 'gray-haired' > \*xed- > WSI \*šed-, OCS [and general] sed-, Novg xed- [in place-names], §29).<sup>52</sup>

There are two questions: what is the historical origin of the stem-final consonant? why do desinences beginning with LCoS \* $\check{e}$  appear? By the rules I have suggested, an IE \*wisos would become \*wisos by the rukirule and eventually\*wwx, i.e. the Novgorod forms, but with \*x mutated to \* $\check{s}$  or \* $\acute{s}/*s$  before the  $\check{e}$ -desinences (like \*wws $\check{e}x$ ) by KAI.<sup>53</sup> An IE \*wikos would yield \*wwsz, where the \* $s\check{e}$  of \* $wws\check{e}x$  (etc.) would appear by simple juxtaposition. The former might be related to a root \*weis/wais 'engender, breed, multiply' that is well attested in Lithuanian. The latter perhaps could be the \*weik/\*woik 'dwell' that underlies OCS vwsb 'village'; cf Savis- 'settlement, community; visva- 'all, every, whole'.<sup>54</sup> Could the two have been blended?

**40.122** W<sub>655</sub> is the sole example traditionally cited to illustrate the progressive palatalization of velar \*x to \*s(>\*s), while the s $\check{e}$ -forms arise from KAI. Most accounts are vague about the details and chronology of the processes I call VA (§29.72). My proposals place BdC at a time before the appearance of \*x. Evidence that allows resolution of these questions simply is not available. This word is not solid evidence for any theory.

**40.123** This sort of enigma is not surprising in a pronoun. Furthermore, puzzles concerning words denoting 'all' and/or 'whole' have engendered a vast literature because of local variability (e.g., Gk \*pant-, \*hol-; Lat. *omnis, totus*; Gmnc *all-, hail-*) combined with etymological obscurity.<sup>55</sup>

<sup>&</sup>lt;sup>52</sup> Note that North Slavic permitted \*ě after \*š. Another word for 'gray' is \*sěr- (not OCS, marginally attested in ESI), WSI šěr-; it implies early \*xair-, which is explicable as a loan from Gmc \*haira- (cf. E hoar, hoary), < IE \*kei-. IE \*koi-r- would yield sěr- but not šěr-. See note ### to §31.</p>

<sup>&</sup>lt;sup>53</sup> KAI did not reach the Pskov-Novgorod periphery of Slavdom (§29); the progressive palatalization (BdC), however, had the same effects as in the rest of Rus'.

<sup>&</sup>lt;sup>54</sup> Lithuanian visas 'all, whole' is no help, because it should have  $\check{s}$  either by the rukirule or from \* $\check{k}$ . Li vi $\check{e}\check{s}as$  'public, communal' surely is from IE wei $\check{k}$ .

<sup>&</sup>lt;sup>55</sup> In German, ganz—etymology unknown—has spread from the extreme southeast over most of the territory since about 800 (replacing heil < IE \*kailo- > OCS cěl3).

**40.21** IE interrogative forms  $k^*o(s)$  and  $k^*id$  lie behind OCS  $k_{\delta}(to)$ 'who' and  $\check{c}_{\delta}(to)$  'what'. The personal kato seems to been universal, but  $\check{c}_{\delta}$ without reinforcement, and  $k_{\delta j \delta}$  may be ascribed to LCoS dialects. (The modern reflexes are  $\check{c}_a$  and kaj, and these words serve as symbols of the  $\check{c}akavski$  dialect of Croatia, and the old macrodialect underlying both Croatian kajkavski and Slovenian, §25.7.)

40.22 The IE relative \*yo- survived as \*j- (see §4.25 and 29.815).

**40.23** The IE apophonic suffix *-tero-*, denoting *alternative*, appears in *\*j-e-ter-* $\mathfrak{s}$  'a certain', *\*k-o-tor-* $\mathfrak{s}$ -*j* $\mathfrak{s}$  'which' (with LCoS variants *\*kotersjs* [OCS, but rare], and *\*kstersjs* > Cz který), *\*někotorsjs* 'a certain', and *\*nikotorsjs* 'no, none'. Jeters is obsolescent in OCS, but lives on for generations as a bookish archaism.<sup>56</sup>

**40.3** The personal pronouns show irregular allomorphy between the nominative (always stressed, syntactically emphatic) and other cases. It is probable that accusative and dative had both tonic and enclitic forms. LCoS forms, though quite different from their equivalents in the older languages, demonstrate complex idiosyncratic rearrangements of old materials. Here we will note only a few details.

**40.311** The IE and Slavic 1st sg nominative 'I' is completely distinct from the other declensional forms, whose stem begins in \*m. The OCS form azz was native only to the Slavs of (eastern) Bulgaria; in all the rest of Slavdom the form was \*jazz. It is true that canonical OCS, with about 850 examples spelled "correctly", offers just one example of  $\check{e}zz$  (Mk 11:29 Mar). Yet this fact must be placed in its cultural context: the name of the first letter in both alphabets was azz, and to write the pronoun with "a" and not "ta" was a symbol of medieval Slavonic literacy.

When the weak - $\delta$  dropped, the word-final /z/ in all LCoS dialects must have been subject to complex allophonic variation as it adapted to the following word-initial phonemes, indeed approximately the range noted for prefix-final /z/ of OCS (§3.111). The modern shape is *ja* for most regions, but standard Bulgarian has {*az*}, Macedonian *jas* (dial. *jaska*, *jaze*, *jazeka*), Slovenian "*jaz*" (usually pronounced [jest]); Old Czech had *jáz*, but the *z* was lost during the 14th century. Rusian documents (as opposed to church texts) use *jaz* freely.

**40.312** OCS azz allows us to posit early  $*\bar{a}zu < *\bar{a}g$ - or  $*\bar{o}g$ -. Gk  $\epsilon\gamma\omega$ , La ego, Sa aham suggest IE  $*egh_2om$ . The final -z is compatible with IE

<sup>&</sup>lt;sup>56</sup> IE \*-tero- is perhaps in wators 'second', see §41.82.

-on, but the initial long back vowel has not been satisfactorily explained. There is no Slavic evidence whatsoever for positing IE  $*\bar{e}$ - here, tempting though the supposition might be.

40.32 The accusative me (< IE  $*m\bar{e}\cdot m$ ) could be tonic or enclitic (as could te and se). Gen mene corresponds to an IE variant (cf. Av mana). The other case-forms, with mon- or mon- and o-stem pronominal desinences, probably are late formations, subject to considerable regional variation.

**40.33** The 2s N ty continues older  $t\bar{u}$  (cf. La  $t\bar{u}$ ); the stem teb-/tobshows redistribution of inherited materials. The OCS LD tebě is SSI; North SI had tobě. So also the parallel reflexive LD sebě/sobě.

**40.34** IE had forms with \*w, \*n, and \*y in 1p and 2p, e.g. La NA  $n\bar{os}$  'we' and  $v\bar{os}$  'you'. The \*w was generalized to Sl 1st person Ndu  $*w\bar{e}$ , but otherwise serves as 2nd person dual and plural marker ( $vy vas\bar{o}$ , etc.). The \*n persisted in dual and plural, except for the nominatives  $v\bar{e}$  and my, opposed to Acc na and ny. 1p plural ny is found in the Kiev Folia (7x Nom, 35x Acc), and Hbl begins to replace N Mbl in Macedonia and Bulgaria after about 1200. Standard Macedonian and Bulgarian now have Hue.<sup>57</sup>

**40.4** The possessives  $na\breve{s}b$  'our' and  $va\breve{s}b$  'your' show old NA  $*n\bar{o}s$  and  $*w\bar{o}s$  with thematic  $*-ye'/_o$ - and  $*-y\bar{a}$ - (or in Slavic terms, -j-): ECoS  $*n\bar{a}s$ *j-as*,  $*w\bar{a}s$ -*j-as*. Reflexive \*svojb, 1s \*mojb, and 2s \*tvojb have the same suffix with old stems \*mo-, two-, and \*swo-.

**41.1** The cardinal number 'one' has syntactic gender and forms of the pronominal declension, §4.201: *jedino*, *jedino*, *jedina*, etc. (In the plural it signifies 'some, only'.) An alternate form *jeduno* is attested in Supr and reflected in the modern languages.<sup>58</sup> The (*j*)ed- is of obscure origin (some scholars suggest a deictic particle); *-in-* is supposed to represent *\*ein-*, the e-grade of the *\*oi-no-* underlying Germanic *\*ains* (cf. Gk oĭvŋ '1 on dice').

**41.2** Two has dual pronominal forms: NA  $*d \omega a m < IE *d u \bar{o}$ , a variant form, cf. δύω, La *duo*, Go *twai*. Oba, obě 'the two, both' is related to Li abù, Sa ubhāu 'both' (cf. Gk ἄμφω, La ambō 'both').

<sup>&</sup>lt;sup>57</sup> OCS has a clear formal distinction between nominative and accusative (*azo ty vě my vs. mę tę na ny*); KF's NA *ny* violates that status. It could be an archaism or a regional innovation.

<sup>&</sup>lt;sup>58</sup> ESI has the stem odin in Nsm only, otherwise odn-: Nsn odnó, Nsf odná. Bg has edín ednó edná. The other languages have {ъ} in Nsm, i.e. the vowel-zero alternation, e.g. SC jedan jedno jedna.

**41.3** Three has plural forms, Nm troje < IE \*treies (Sa trayas, Gk τρεῖς), GL troxo (Sa Loc trišu).

**41.4** Four is also plural: četyre Nm implies ECoS \*ketūres, and related forms with četvor- or četver- in others imply \*ketwor-, \*ketwer-, along with MCoS \*četvbrtb 'quarter' < \*ketwirt- as regularized reflexes of IE \*k<sup>w</sup>etwer- \*k<sup>w</sup>etwor- \*k<sup>w</sup>etwr-ti- and other shapes.<sup>59</sup> NW monosyllabic stems, e.g. Cz čtyr- čtver- (and the like), point to \*čbt-, perhaps a reflex of IE zero-grade \*k<sup>w</sup>tūr-.

**41.5** The numerals petb '5', šestb '6', sedmb (ESl semb) '7', osmb '8', devetb '9', and desetb '10' are i-stem nouns (§20), closely related to the ostem forms that serve as ordinals. The IE cardinals first apparently gave way to stems built with the ordinal suffixes -to- or  $-h_2o$ -, then shifted to istems.<sup>60</sup> As in other languages, adjacent numerals seem to have affected each other: the stem-final *m* is expected in '7' but not '8'; initial *d* is original in '10' but not '9'. The voicing of *d* in NW and SSl sedmb is unexpected, and dm is a unique internal cluster; loss of *d* in Rusian semb seems more regular.

	IE cardinal	ordinal	new cardinal	OCS card.	ordinal
5	*penk <sup>w</sup> e	*penk <sup>w</sup> -to	*pen(k)-ti-s	рętь	pętъ
6	*kseks	*kseks-to	*kše(k)s-to-	šestь	šestъ
7	*septm	*septm-h <sub>2</sub> o-	*sebdm-i	se(d)ть	se(d)mъ
8	*okteh3 *oktō	*okt+mo	*ok(t)-mi-	оѕть	osmъ
9	*h₂newn-	*newn-to-	*newn-ti-	devętь	devętz
10	*dekm	*dekm-to-	*dekm-ti-	desętb	desętъ

**41.6** IE \**dkmtom* '100' (usually shortened to \**kmtom*, whence Li *šim̃tas*, Av satəm, La centum) is replaced by  $s \delta to$ , which keeps the neuter gender but has  $\delta$  where we expect a nasal vowel.<sup>61</sup>

**41.7** OCS tysesti or tysesti are soft a-stem feminines, analyzed as an ECoS compound,  $t\bar{u}$ -sint-j- $\bar{i}$  or  $t\bar{u}$ -sunt-j- $\bar{i}$  (parallel to Gmc thus-hundi-, E thousand). The first element is from IE tuh- (zero-grade of teuh-swell'), and the second is tuh-om '100', therefore 'swollen hundred'.

234

<sup>&</sup>lt;sup>59</sup> OCS četvrědbnevbnö '4-day' and četvrěnogö '4-legged [creature]' show MCoS \*četver- as the combining form in compounds, cf. §26.51.

<sup>&</sup>lt;sup>60</sup> The data of even the ancient dialects shows extreme variability in numerals; the hypothetical IE forms given here are chosen from many possibilities as most compatible with attested Slavic.

<sup>&</sup>lt;sup>61</sup> Rumanian borrowed the word as *sută*, obviously before the jer-shift, perhaps as early as 1000.

41.7-42

LCoS seems to have had the q variant in SW and NW dialects; except for Sln *tisoč*, SSl now uses *hiljada* (in varying form), borrowed from Gk  $\chi i \lambda i \Delta \alpha$ .<sup>62</sup>

**41.81** The ordinal \**pbrws* 'first' represents \**pr-wo-s*, from the IE root \**perh*<sub>2</sub> that underlies Li *pirmas*, Gk  $\pi\rho\tilde{\omega}\tau\sigma\sigma$ , Latin *prīmus*, and E *first*, foremost.

**41.82** Votors 'second' and votoruniks 'Tuesday' (\*wster- in pre-Czecho-Slovak) suggest ECoS \*untar-, with the apophonic suffix \*-tero-, denoting alternative (as in jeters, §40.23). Li añtras and añtaras, Go anþar 'other, second' point to \*an-, a deictic particle. Apparently \*anclosed to \*un- (cf. the preposition \*un, §29.815); the loss of nasalization remains puzzling.

**41.821** (The numeral survives only in R BTOPÓH; otherwise the reflexes of \*drugs 'other, another' have taken over as 'second'. 'Tuesday' generally maintains the old root, but often reflects the suffix \*-sks (i.e. a vowel-zero alternation, /-rVk ~ -rkV/: SC utorak, Sln torek (also vtorek), (Cz úterý and úterek), Slk utorok, P wtorek, Uk vivtórok, BR awtórak; R, Bg, Mac vtornik. Notice that the ws - lost its weak jer, leaving initial \*wt-. In SC the glide became a vowel u- (in all words). In P, R, Bg, and Mac the glide became a voiceless consonant /f/. In Cz and Slk a vowel is exceptional in this root; [ft] is normal for other words. In Sln, the glide persisted and though initial [wt] is common, in this word the glide was lost. In Uk and BR [wt] occurs in many words, the glide was reinforced, [v'iwt] in standard Uk, [awt] in BR. This special treatment seems to be defined by the lack of an underlying boundary; {wtor-Vk-} has a different fate from {wV-t-} where the morpheme is the prefix 'in, into').

**41.83** Tretbjb (\$2.61) 'third' comes from \*t*r*-tijo-s (cf. Sa t*r*tiyas), with an unexpected *e*.

41.84 \*Četvortojo 'fourth' corresponds to Li ketvirtas.

'Thursday' is četvrstökö in OCS and, in appropriate form, most mod. Sl languages; četvorgö is attested very early in Rus' (cf. Li ketvérgis '4-year-old').

41.85 'Friday' is OCS *petъkъ* and its descendants in all but ESI today. Патьюща is known in ESI from 1056.

## Conjugation

42 The Indo-European verb had stems expressing voice (active, middle), state (the perfect), and tense (present, aorist), and desinences ex-

<sup>&</sup>lt;sup>62</sup> Cz tisic shows unusual complications: \*tyśęc (with palatalization of s before a front vowel) was apparently blended with \*tysęc to produce \*tuśúc, and both syllables were fronted, yielding mod. /Kisīc/. This markedly Cz form was borrowed into Slk.

pressing number and person. These stems were essentially independent. Thus, for example, the IE root \**bheudh* 'be aware, make aware' provides contrasting autonomous present tenses that can be approximated by Slavic derivatives that constitute elements in the paradigms of different verbs.

*bheud- <sup>c</sup> / <sub>o</sub> - 'observe': Sa bódhati 'awakes, becomes aware'; πεύθεται 'learns'	1	OCS <i>bljudet</i> & 'observes' {bljud-Ø+}
*bhundh-'/ <sub>o</sub> - 'become aware': Li bundù 'I wake up' πυνθάνομαι (with a secondary n-suffix) 'ask, enquire, learn'	2	*bъdnetъ 'wake up, become awake' {bъd-nǫ+}
*bhoudh-ey <sup>e</sup> / <sub>o</sub> - causative: Sa bodháyati 'he wakes [someone]'	I -	budits {bud-i+} 'wakes [someone]'
*bhudh-y <sup>e</sup> / <sub>o</sub> - 'be awake': Sa budhyáte [pass.] 'is awake'	4	bəditə 'is awake, vigilant' {bъd-ё+}

The phonetic change of \*eu > \*ju and the shift of \*j to \*lj after a labial (§3.71) in (1) disguises the relationship of the first root-shape (e-grade or basic) to zero-grade \*bad and o-grade \*bud; in fact *bljud-e-to* is the cognate of Sa *bodh-a-ti* except for the final vowel. (2) shows an *infixed n* in IE in contrast to a Slavic *suffixal* nasal that continues the same general meaning of inception (see §45 below). The formation of the causative (3) in Slavic is perhaps a reflection of the IE, while the *i* of the stative (4) is new.

The distance between early IE and OCS verbal systems is too great for us to reconstruct the many intervening stages that would make the genesis of the Slavic clearer. The materials of LCoS conjugation are mostly IE, but there are many shifts in use and significance. The morphological paradigms expressing the middle (or medio-passive) voice disappeared without trace, and categories variously labelled subjunctive, conjunctive, and optative survive only in the Slavic imperative.

43 Reference to present ("the here and now") as opposed to past underlies the formal distinctions of present and preterite in OCS (see §5.6). The origins of the vowel-morphemes  $\check{e}$ , a, and i that play a major role as verbal classifiers are not clear. The stative  $\check{e}$  of  $b \cdot d - \check{e} +$  'be awake' recalls IE  $\bar{e}$ morphemes associated with stativeness and aorist paradigms. The a and the i remain without satisfactory explanations; for  $\rho$ , see §49.1ff. below.

**43.1** Let us look first at the present desinences and then at stem-formation.

### A SKETCH HISTORY

IE person-number desinences used with the present indicative ("primary endings") differed somewhat from those used with a orist and imperfect ("secondary endings"). This contrast survives under rather different form in OCS (§5.9). Here are the chief variants pertinent for Slavic:

**43.11** All desinences began with a consonant except  $1 \le -\bar{o}$ ; the original 1s desinence was  $\ast -h_2$ . The  $\ast -o -h_2$  of thematic forms  $> \ast -\bar{o}h_2$ , later  $\ast -\bar{o}$ ; the resulting Slavic  $\ast -\bar{a}$  was augmented, probably by the primary  $\ast -mi$ , that then lost its vowel, yielding  $\ast -\bar{a}m > LCoS - q$ .

**43.12** OCS 2s -si (esi, dasi, ěsi, věsi) implies older \*-sī or \*-sei; no satisfactory explanation has been found. Normal OCS -ši requires \*-šī or \*-šei: the consonant surely represents generalization of \*š from the rukirule to all prevocalic \*s in desinences (§27.3); the long vowel remains mysterious. The constant presence of -ši in OCS surely proves that it existed in the dialect of the original translators and had the full approval of early scribes. It is, then, to be posited for a part of 9th-century Bulgaro-Macedonian regional dialects. Yet the tenacious spelling contradicts all other evidence, which points to \*-šb as LCoS (and its eventual adoption in all of SE Slavic).

**43.13** OCS 3 person singular and plural has \*-ta. This is a regional peculiarity of the standardized language; \*-tb (= inherited \*-ti) is to be posited for most LCoS dialects. The ta may have arisen in part through the influence of the demonstrative ta 'this, that' which in some dialects functioned as subject pronoun of the third person. The dialectal zero-desinence (§6.61) was surely relatively new and arose in different localities in varying environments.

**43.1311** The -t (Hecët, Hecýt; Hócut, Hócut, Carries/carry') of modern standard Russian is a peculiarity of central Great Russian dialect that appeared after 1350 in Muscovite documents. It spread slowly over the whole Great Russian area. Most Belarusian and Ukrainian dialects retained the palatalized consonant /c'/ or /t'/ that go back to \*-tb; many dialects have zero in some paradigms.

**43.14** OCS 1du \*-we was probably general LCoS, but it is replaced by -va (probably under the influence of the numeral dwa) in most regions later on. The Slavic dual desinences are not easily explicable from the hypothetical IE.

# 43.15 OCS 1p -m<sup>3</sup> can be derived from \*-mos, see §38.41.

**43.16** Five verbs have -mb in 1s present (four of which have -si in 2s); they are the remnants of an ancient group of presents consisting of root plus person-number desinence. Their anomalies are typical of irregularities that are tolerated in basic every-day words. At the same time their histories illustrate the complex ways allomorphs influence each other.

**43.171** The OCS morphemes {jes} and {s} underlying the present forms \*jesmb, \*jesi {jes-si}, \*jestb, sqtb (§16.1) are unmotivated in the system; they are simply suppletive. They derive from allomorphs es/s that survive elsewhere with different distribution (Sa, with a for \*e [and \*o], ásmi, ási, ásti ~ 1p smas, 2p stha, 3p sánti; Latin 1s sum, 1p sumus, 3p sunt ~ 2s es, 3s est, 2p estis). The IE root \* $h_1$ es and rules governing apophony show that this present was once quite normal: if accented, the root was \* $h_1$ es (e-grade); if unaccented, then zero-grade \* $h_1$ s: \* $h_1$ ésmi, \* $h_1$ ésti vs. 3p \* $h_1$ sénti. The expected 3p \*-nt- \*-int- (cf. dadęts, vědęts) has been replaced in Slavic by the more neutral \*-ont-: sqts.

Modern dialects have rearranged these elements in many ways. SC has a set of accented forms, and an enclitic set. Polish retains 3p (without a final t) and builds new forms for first and second person with *jest* as stem.<sup>63</sup>

OCS	јеѕть	jesi	jestъ, je	jesmъ	jeste	sqtz
SC (stressed)	jesam	jesi	jest(e), jề	jesmo	jeste	<b>je</b> su
SC (enclitic)	sam	si	je	smo	ste	su
Polish	jes <b>te</b> m	jes <b>te</b> ś	jest	jes <b>teś</b> my	jes <b>teś</b> cie	są

**43.172** OCS *dati* 'to give', with its present forms, requires two underlying root-shapes,  $\{da\}$  for the infinitive, and  $\{dad\}$  for 3p and the forms with *-st-*:

{da-ti}	{da(d)-mь}	{da(d)-si}	{dad-tъ}	{da(d)-mъ}	{dad-te}	{dad-ętъ}
dati	damь	dasi	dastъ	damъ	daste	dadętъ

Further, the imperfective *dajati* requires underlying {daj}.<sup>64</sup> Older Slavic had  $*d\bar{a}$ ,  $*d\bar{a}d$  from  $*d\bar{o}$ ,  $*d\bar{o}d$  (cf. Li inf. *dúoti*, old 1s *dúomi*, mod. *dúodu*), ultimately based on the IE root  $*deh_3$ . The Gk 1s  $\delta(\delta\omega\mu)$  and its Sa equivalent *dádāmi* exemplify an IE formative device called reduplication: preposing a syllable made up of the root-initial consonant followed by a short vowel. In Pre-Balto-Slavic the vocalism has changed:  $*d\bar{o}d$ -mi.

<sup>&</sup>lt;sup>63</sup> The form *sqchmy* for 1p is found in a major P writer, P. Skarga, early 1600s.

<sup>&</sup>lt;sup>64</sup> The substantive *daw-bcb* 'giver' requires still another shape, {daw}.

**43.173** A second Slavic complex of stems includes a somewhat different reduplication. IE  $*d^heh_1$  'set, put' underlies the OCS  $d\check{e}$  in a series of words meaning 'do, put' (e.g.  $d\check{e}_j - a + \$15.44$ ,  $d\check{e}_l - aj +$ ). There are also prefixed presents with  $\{-\text{ded}-j-\}$  ( $o-de\check{z}dqta$  'put on, dress',  $va-de\check{z}dqta$  'put in',  $vaz-de\check{z}dqta$  'lift, raise') that seem to be equivalent to forms based on  $\{d\check{e}_j\}$ . The short vowel e is unexpected;. perhaps it represents the archaic reduplication syllable,  $*d^he-d^heh_1$ . An old alternate not attested in OCS is  $\{d\check{e}-nq\}$ , e.g. zadeneš 'you set, assign' (Freising), mod R. odénut 'they will dress'.

**43.174** OCS *ěd-jad-* 'to eat' has an unexpected long vowel (as does Li *ed-*), IE \**ed-*. See also §36.1.

**43.175** OCS ved-e-i 'to know' shows the intersection of perfect—reference to past from point of view of present—and stative; the root \*weid 'see' and a form meaning 'I have seen = I know', comparable to 1s perfect in Gk ( $roi\delta\alpha$ ) and Sa (*veda*). IE \* $\bar{e}$  was a morpheme marking perfect; as a non-present marker in Slavic it usually denoted state. The desinence of 1s vede is isolated, historically and synchronically. Though it was generally eliminated from western OCS (including the Gospels and Psalter), where vemb is almost exclusive, vede survived in Old Czech and early Rusian.

**43.176** OCS *im-ě-ti* 'to have' surely had the root  $\omega n$  with prothetic *j* (\**j* $\omega n$ -, §2.22). The perfect or stative \* $\bar{e}$  in non-present forms and the \* $\bar{a}$  in the present produce the meaning 'have taken = possess'. The 2s desinence is normal - $\bar{s}i$ , and the 3p -qta probably replaces an older \*-qta. Still newer is \**j* $\omega n \bar{e}jqta$ . The root is IE \*em; Slavic  $\omega n$  represents the zero-grade, IE \*m.

44 Slavic has two regular types of present, e/q and i/e (§6.11). The e/q present-marker is descended from IE \*\*/<sub>o</sub> (called the *thematic vowel*, cf. §37.5). In IE, \*o appeared before sonorant or laryngeal, \*e before obstruent; in 3 pl this meant \*o-nt-, whence OCS q. Slavic generalized \*e to all other persons. The present-sign \*ī corresponds in part to a short *i* in some Baltic forms, and in part to the IE causative \*- $ey^{e}/_{o}$ -; the history may involve the loss of intervocalic *j*, contraction, and analogical reshapings. The third person plural - $et_{5}$  presumably goes back to \*īnt-.

44.11 The i/e presents are paradigmatically linked in Slavic with nonpresents made with non-present suffixes  $\check{e}$  or i. **44.12** OCS ĕ-verbs (§15.2, 15.3) are generally stative and intransitive. Roots had zero-vocalism in principle, but there are innovations. The class is not productive.<sup>65</sup>

**44.13** OCS i-verbs include iteratives (e.g. nos-i+ti nosets 'carry'), causatives (e.g. bud-i+ti budets 'wake [someone] up', cf. §42), and denominatives (*gost-i+ti gostets* 'be host' cf. *gosts* 'guest'). The last two types were not clearly opposed, and both were productive.

**44.14** The various types of IE presents included complex rules for apophony, in particular the distribution of zero-grade and e-grade allomorphs. Remnants of this kind of rule are still visible in OCS stem-formation, but they are fully lexicalized. The details of allomorphy were perturbed by monophthongization, and OCS manifests a strong tendency to re-shape the forms.

eēleg-ě sěd-	ě ei	-	-	wis-ě	eu	-	er	-	-
o ō log-i sad-									
Ø	i	lьp-ě	swы-ě	-	u	bъd-ě	r	mrьz-ě	wrыt-ĕ

The table illustrates verbs which in late IE did not permit zero-vocalism  $(le\check{z}-\check{e}+$  'to be lying',  $s\check{e}d-\check{e}+$  'be sitting') and verbs with diphthongs which allowed zero-grade. Thus  $lsp-\check{e}+$  'be clinging' corresponds to  $l\check{e}p-i+$  'cause to adhere' (older  $*lip \sim *loip$ ) as  $swst-\check{e}+$  'be shining' to  $sw\check{e}t-i+$  'cause to shine';  $wis-\check{e}+$  'be hanging', on the other hand is surely new, replacing unattested  $*wss-\check{e}+$ . Within OCS,  $sw\check{e}t-\check{e}+$  is found instead of  $swst-\check{e}+$  (e.g. J 5:35 ZM  $svste \sim As \ sv\check{e}te$ ). For  $bsd-\check{e}+ \sim bud-i+$  see §42 above. MCoS  $*msrz-\check{e}+$  presumably meant 'be freezing, chilly' but OCS  $mrsz\check{e}+$  has only the figurative sense 'be repulsive, revolting'; \*morz-i+ is attested as 'congeal, make solid' (while the noun mrazs means 'frost' and 'ice'). MCoS  $*wsrt-\check{e}+$  and wort-i+ might well have been transitive; the few OCS and early post-OCS examples all have se. The former means 'twirl, go around'; the latter is 'return; turn around'.

**44.151** The apophonic alternations are not productive in LCoS, and the meanings of related stems tend to diverge. Older \**skend-/\*skond-* 'in-sufficiency' underlies transitive šted-e+ 'be sparing of'; causative *o-sked-*

<sup>&</sup>lt;sup>65</sup> The influence of vid-ě+ 'see' and zы-ě+ 'see, look at, observe' perhaps is responsible for the shift of sъ-motr-i+ 'look at' to R смотре́ть (about 1350), and ględ-aj+ 'see, look over' to R глядеть (about 1600).

*i*+ 'diminish, make lesser' is matched by (o)skqd-ej+ 'become lesser, diminish'.

44.152 Many roots do not change at all, e.g. MCoS \* $skarb-\check{e}+$  'be sorrowful' ~ \*skarb-i+ 'afflict' (like skrabb 'affliction').

**44.2** The e/q presents occur with stems that have no overt verb-forming suffix and with stems that have any suffix but *i* or *e*. This includes a large number of heterogeneous verbs, many of which require special morphological information in their lexical definition (e.g. §15.64, §15.8). IE \* $e/_e$  was a component of suffixes used exclusively or primarily in the present. Thus  $*d^o/_e$  appears in the irregular presents *id-qt* and *jadqt* 'go' (inf. *iti*, *jaxati*), IE roots \*ei/i and \**ja*.

**44.21** The traditional classifications of OCS verbs posit *je-stems* and *ne-stems*. Some je-stems go back to IE  $*y^{o}/_{e}$  verbs, and the ne-stems are comparable to an IE  $*n^{o}/_{e}$  group, but the rearrangement of various kinds of presents into Slavic paradigms that join them with a orist-infinitive stems blurs the historical relationships.

**44.31** Early IE roots like  $*deh_3$  'give' and  $*d^heh_1$  'put' conformed to the general constraint that a root should end in a non-syllabic element (including laryngeals, liquids, nasals, and \*i/\*j and \*u/\*w). They evolved into  $*d\bar{a}$ - and  $*d\bar{e}$ -, which do not fit the canonical shape of early Slavic roots. When, in these new circumstances, they were combined with a vocalic suffix, a \*j was affixed, creating root-variants like those underlying the imperfective or iterative daj-a-ti and  $d\check{e}j$ -a-ti. Presents like dajets may have originated as  $*y^o/_e$  forms at an early date or as  $*o'_e$  later on:  $*d\bar{a}$ -je-t- or  $*d\bar{a}j$ -e-t-.

**44.32** OCS *znajets* 'knows' (§15.92) presumably shows  $zn\bar{a} + y^{o}/_{e}$ , since the non-present has no  $\bar{a}$  (inf. *znati*). The IE root is  $zn\bar{o} < zn\bar{o} < zn\bar{o}$ .

OCS  $kaj \cdot a +$  'blame' and  $\check{c}aj \cdot a +$  'expect' are believed to derive from  $*k^{w}\bar{o}i$  and  $*k^{w}\bar{e}i$ , lengthened-grade forms of  $*k^{w}ei$  'pay, compensate'. This implies relatively old  $*^{o}/_{e}$  in the present tense.

**44.33** Other verbs listed in §15.43 and 15.45 fit these patterns: laj-a+ 'bark, scold' (<\*la, cf. La lamentum 'lament'), laj-a+ 'lie in ambush' (<\* $leh_2$ , cf. La latere 'to lie hidden' < extended root, zero grade \* $lh_2dh-\bar{e}$ ), taj-a+ 'melt' (<\* $teh_2-$ , cf. Gmc  $\bar{p}aw-$ , E thaw). Vaj-a+ 'sculpt' and maja+ 'beckon' have no plausible etymologies. Sej-a+ 'sow', sej-a+ 'winnow', and \*vej-a+ go back to \*se (a. La semen 'seed'; b. Gk  $\eta\theta\omega\omega$  'sift, strain'), and \*we (which is from \* $h_2weh_1$ , Sa vati 'he blows'; E weather, wind). Spěj- 'ripen, be successful' is from  $sp\bar{e}$  (speH: La  $sp\bar{e}s$  'hope'), rěj- 'push' is usually associated with rei- 'run, flow' (cf. rino §45.3). Soměj- 'dare' is obscure. Vol-aj- (possibly swol-aj) 'toss' is surely related to swolna 'wave' and wal-i+ 'to roll', from swel- 'turn, roll' (cf. La volvere 'roll', E revolve).

*Gr-ěj*- 'to warm, heat' represents a zero-grade root, gr-,  $< *g^{w}her$ , related to gor- $\check{e}$  'burn' (cf.  $*\check{z}ar\mathfrak{z}$  'glowing coals; conflagration' < ECoS  $*g\bar{e}r$ -).

**44.34** The root-alternation in OCS zijqta zejati 'yawn' (§15.46) shows older  $*gh\bar{e}(-i)$ - (whence \*zej-a+), with a zero-grade variant  $*gh\bar{n}$ , which may be posited as the base for an  $*o'_e$  present, eventually  $*z\bar{i}-e- > *zbj-e-$ . This set of processes is explicable in terms of laryngeals, but the precise environments for each root-shape are disputed. Lijqta/lbjqta 'pour' fits an IE root \*lei (for  $*lei-o'_e-nt-$  would yield lbjqta). While  $*l\bar{e}i-$  could well be an old lengthened-grade present, the combination with  $l\bar{e}j-a-$  seems to be an innovation. The same is true of OCS smej-a- with relation to IE \*smei.

44.35 OCS verbs with zero-classifier preceded by j (§15.93) show the same set of problems.

OCS kryjęts 'cover, hide' probably goes back to  $*kr\bar{u}$ -je-, and  $\check{s}ij$ ęts 'sew' to  $*sj\bar{u}$ -je-, while bijęts 'strike' is rather  $*b_{bj}$ -e- or \*bij-e- (from older  $*b^{h}eiH^{-o}/_{e}$ -), and pijęts 'drink' is  $*p_{bj}$ -e-  $<*p\bar{i}$ -e- (a root-form derived, some will argue, by metathesis from zero-grade  $*ph_{3}i$ -, from extended root  $*peh_{3}(-i)$ -).

So also myjqta 'wash' (cf.. IE \*meu 'damp'), -nyjqta 'be despondent' (cf. IE \*nāu 'death; be exhausted'), ryjqta 'dig' (cf. IE \*reu(H) 'ruin; knock down; dig up'), -ujqta 'put on, take off footwear' (cf. IE \*eu 'wear') čuj-qta 'feel, sense' (cf. IE \*keuH 'pay attention, perceive [preternaturally])'.

In the verbs with a back vowel in the root, the j of the present becomes w before the past passive participial suffix *-en-*: e.g. *umyvens, obuvens*. The precise historical sequence of changes is not clear.

**44.351** The dissyllabic stem \*(w) = pij- 'cry out' stands apart. It seems to be based on a noun \*(w) = pi that survives as *vep* in OCz. The \*ij/ej, then, looks like a derivational suffix.

**44.36** IE  $*y^{e}/_{o}$  presents were often associated with  $*\bar{a}$  aorist stem (in LCoS terms, non-present). Slavic has two paradigmatic relationships: the verb-formant *j* alternates with *a*, or it is added to *a*. Thus  $*m\bar{a}z$ - $\bar{a}$ -tei 'to anoint' > maz-a-ti ~  $*m\bar{a}z$ -j-e-te 'you anoint' > mažete, or else  $*m\bar{a}z$ - $\bar{a}j$ -e-te > mazajete 'you (repeatedly) anoint'.

44.36-44.8

243

In this book, the paradigmatic relationships maza-  $\sim$  maže and maza-  $\sim$  mazaje- are presented in terms of morphonemic {a} alternating under specific conditions with {j} (§6.22), or else {aj} being truncated to /a/ under other conditions (§13.2c). This type of description has proved to be an effective pedagogical device for students of OCS and other Slavic languages, and it provides useful units for comparative and historical purposes. This is not the place to debate its possible relationship to linguistic models in the brains of individuals who use these languages.

**44.361** The  $*\bar{a}$  suffix added to a velar stem could trigger BdC (§29.2): \*lenk/\*lnk 'bend' formed \*link- $\bar{a}$ -> \*lec-a- (§15.63). Similarly,  $*\bar{a}j$  in iterative formations could establish the BdC formula. By LCoS, the productive rules for iterative or imperfective formation included "mutate rootfinal C if possible" and the specification k g > c z (regardless of the vocalism of the root). This process has continued, with numerous local variations, to produce ever new secondary imperfectives. OCS attests -strig ~ strizaj 'shave',  $-\check{z}bg ~ \check{z}izaj$  'burn' and -wyk ~ -wycaj 'learn' (<  $*\bar{u}k$ -), see §5.712ab.

44.371 The  $*\bar{a}$  non-present marker exceptionally goes with the i/e present only in *szp-a*- 'sleep' and \**szc-a*- 'piss' (see note to §29.6).

44.372 OCS xot- $\check{e}$ -ti 'wish, want; be about to' (§15.233) is unique and inexplicable. It has a je-present except for the 3p xotets, which might be an old athematic (like dadets) with \*-nt-. The alternative stem xst, rare in OCS, is normal in the non-present system in SC (inf. htjeti, 3s pres hoće), and all forms in Cz (inf. chtít, 1s pres chcí or chcu, 3p chtějí, past chtěl) and Polish (inf. chcieć, 1s pres chcę).<sup>66</sup> Some scholars have linked the root-forms xst and xot to xyt-i+ and xvat-aj+ 'rob, grab, steal, carry off', but the origin of the initial x and the exact apophonic relationships have so far resisted explanation. Compare kys/kvas in vskysnq+ 'ferment' and kvast 'yeast', presumably  $*k\bar{u}t$ -s-  $*kw\bar{a}t$ -s- < \*kwHt- \*kwoHt- (cf. Sa kváthati 'boils, cooks').<sup>67</sup>

**44.38** The *a*-paradigm continued to be marginally productive, chiefly with onomatapoetic words for sounds like *gurgle* and *twitter*, which vary extremely from dialect to dialect.<sup>68</sup> Individual verbs may have doublet

<sup>&</sup>lt;sup>66</sup> The formula \*xstj-e->\*xke- produces a wide variety of phonetic variations in the dialects of several languages; none of this material helps explain the ECoS forms.

<sup>&</sup>lt;sup>67</sup> It is doubtful that La  $c\bar{a}seus$  'cheese' belongs here. (E cheese goes back to the La word.)

<sup>&</sup>lt;sup>68</sup> SC may nativize a foreign stem with a suffix -isa-~ present iše-, e.g. anatèmisati 3s pres anatèmisé 'anathematize' infòrmisati infòrmiše 'inform'. The origin is the modern Gk aorist form of the -iz-, e.g. ἀναθεματίζω 'I curse' ~ ἀναθεμάτισα 'I cursed'.

present forms, cf.  $\dot{z}eda + \sim \dot{z}edaj +$  'thirst' §15.64. The anomalous verbs that have apophonic alternations, usually contrasting present to infinitive/ aorist stems, will be treated below, §50.1.

**44.39** Verbs with aj have been continually productive in most dialects, especially for creating secondary imperfectives (see §5.71–.7112).<sup>69</sup> This function corresponds to the IE iterative sense of the suffix. Note that when -aj- is added to a stem ending in j, the stem-final glide usually becomes w: see §5.712e.

**44.41** Verbs with  $\check{e}j$  often denote a change in status, e.g.  $bl\check{e}d\check{e}j$ + I ( $obl\check{e}d\check{e}j$ + P) 'become pale, fade' ( $bl\check{e}ds$  'pale'),  $omrstv\check{e}j$ + 'become numb' and  $umrstv\check{e}j$ + 'mortify'(mrstvs 'dead');  $zapust\check{e}j$ + and  $opust\check{e}j$ + 'become deserted' (pusts 'empty');  $raslab\check{e}j$ + and  $oslab\check{e}j$  'weaken, become paralyzed' (slabs 'weak');  $vszb\check{e}sbn\check{e}j$ + 'become mad, delirious' ( $b\check{e}ss$  'demon',  $b\check{e}sbns$  'possessed, mad'). Some are stative, e.g.  $gov\check{e}j$ + 'be respectful, pious';  $gon\check{e}j$ + 'suffice'.  $Pit\check{e}j$ + 'feed' is transitive (and is replaced in innovative OCS by pit-aj+); pitoms 'fatted' implies that this was once a C-verb (§8.13).

**44.42** The prefixes o- and u- are particularly common to make perfectives correlated with  $\check{e}$ -stems and  $\check{e}j$ -stems denoting change; they also serve with i-verbs to specify causation: e.g.  $umr \imath vii + kill$ , put to death; mortify'; opusti + devastate, make empty'; oslabi + make weak, paralyze'. Occasionally an i-verb may be used with se as an "anticausative"; thus  $omr \imath z \check{e}j + be$  abhorrent' is semantically almost identical with omrazi + se 'make self abhorrent, become abhorrent'.

**45.1** IE had several formations using n (\**neu*, \**nou*, \**nu*, \**nū*) and generally signifying the beginning or inception of action.

**45.11** The \*n could be inserted (infixed) before the final consonant of a root; Slavic provides only four examples, sed-qts 'sit down' (§16.61), legqts 'lie down' (§16.62), -reštqts 'encounter' (§16.7), and bqdqts 'will be' (§16.1). On the premise that infixation goes with zero-grade, we posit PBS shapes \*sind (~ \*sed, though the non-present stem in Slavic has lengthened-grade \*sēd), \*ling (~ \*leg), \*rint-j (~ ?\*rēt; the etymology is uncertain), and \*bund (~ \*beud < \*bheudh 'be, grow').

45.12 OCS has partly regularized the distribution by placing n in the present system, nq in infinitive and past forms, and now in the past passive

<sup>&</sup>lt;sup>69</sup> In South and West Slavic the *j* of most forms has disappeared as -aje- contracted to  $-\bar{a}$ -. These complex developments cannot be discussed here.

participle (§15.7). The nasal vowel in nq is a relatively recent development, see §21.8143. The distribution of nq in infinitive and preterite forms varied in LCoS dialects (§15.76).<sup>70</sup>

**45.2** OCS *stanets* 'will stand, take a stance; come to a halt' goes with non-nasal perfective forms *stati*, *stals*, aor *sta* in contrast to imperfective \**stajets* 'continues to stand, be' (with inf. \**stajati*).<sup>71</sup>

	root	stative		iterative		causative
lie	*leg	leg-ě i	lęg-e	lěg-a lěže-	lěganьje	lož-i
sit	*sed	sěd-ě i	sęd-e	sěd-aj-e-	-sědanьje	sad-i
stand	*stā	stoj-ě i	stan-e	*staj-a -e		staw-i

**45.21** An archaism preserved only in Rusian mss is the anomalous  $*kr \omega jets kriti$  'to buy'; it apparently had a j-present (and imv. крыни, крыните), but otherwise followed the pattern of  $b \omega jets$ , §44.35. It continued in active use in northern Rus', but two forms (past pass part оукрикнааго 'bought [Gsn]'; verbal substantive по критии 'after the purchase, ransom') in 12th-c. copies of originally OCS texts imply it was known in Bulgaria in the 10th c. The Sa cognate kr n n i 'buys' has a nasal infix; the IE root is  $*k^w reih_2$ - (Gk  $\pi p i \alpha \mu \alpha_1$ , OIr cren[a]im 'I buy').

**45.3** The no-verbs with vowels before the suffix represent old diphthongs:

\*weiH- turn, twist' > wi-nq, \*dheuH- 'rise as vapor, smoke', PBS ograde \*dou- > du-nq-; \* $gh\bar{e}(-i-)$ - 'gape' (§44.34 above) > zi-nq;

IE ?  $(s)keu > ECoS *k\bar{u}(w)$  'beckon' (cf. nakynǫ, pokywaj 'nod, wag [head]';

ECoS \* $m\bar{a}j$ - 'gesture' (cf. Li m oju 'beckon' < IE \*mei- 'go, change'; La meāre 'go, pass' [E permeate]); měno/meno (cf. men-e+ 'think'; see §29.813;

\*pol-n- > SSl planq-, cf. pol-ě+ (o-grade) 'burn', pal-i+ (ō-grade) seems to be connected with \*pel- 'dust'; pli-nq 'spit' (cf.pljbva- \$15.52) is phonetically difficult (cf. Li spiáuju, IE \*spyeu-); rinq 'run' < \*rei-'flow, run' \*rei-wo- > La rīvus 'stream' (> E rivulet) ~ rěj- 'push'; \*sousovaat < \*sowajets 'surges [boils over]' Li šaúju 'shoot; put, place [bread

<sup>&</sup>lt;sup>70</sup> Russian retains a vowel alternation but no nasality in сесть ся́дут 'sit down' and лечь ля́гут 'lie down'; SC has re-formed the stems with a nasal suffix in the present, sjesti 3p sjednū and leći legū.

<sup>&</sup>lt;sup>71</sup> Attested OCS "stajati" is an error for the noun *staja* 'stable', but the forms are assured by post-OCS data.

into oven]' (< IE kou ?). Zero-grade roots are common with nq but by no means exclusive: \*sup- 'sleep'< swep (like sp-a+); \*wyk- < ECoS \* $\bar{u}k$ - 'become accustomed, learn' is a Slavic "lengthened reduced grade" replacing \*uk- (cf. IE \*euk- 'become accustomed').

46 The imperative markers  $\check{e}$  and i (§7.1) must go back to an ECoS diphthong, \**ai*. Its origins seem to lie in IE optative forms; a detailed history of analogies and restructuring must be speculative. The singular *i* is particularly difficult to explain.<sup>72</sup> The irregular singular forms  $da\check{z}db$   $\check{e}\check{z}db$   $v\check{e}\check{z}db$  and  $vi\check{z}db$  (§7.2) manifestly reflect \*-*d*-*ji* (affirmed by KF-*dazb*, with the Czech reflexes of \**dj*): these forms remain enigmatic.

47 The imperfect tense is obscure because the philological evidence is contradictory and the forms and paradigms are remote from imperfects in other languages. The LCoS forms are surely relatively new, and they are already evolving into diverse local forms. It seems plausible that ECoS imperfects were a combination of verbal stem (usually infinitive/aorist) plus an auxiliary based on the root 'to be'—very probably a periphrastic construction by origin. Descriptively, the distinctive suffix has two vowels,  $-\check{e}a$ -: historically, the segmentation may have been something else. What is apparent is that the formant x is followed by the desinences of the root aorist.<sup>73</sup>

**1**s 2 sg 3 sg 1 du 2 du 3 du 1 pl 2 pl3 pl ā-š-u ā-š-es ā-š-et ā-š-awē ā-š-etā ā-š-ete ā-x-anu ā-š-ete ā-š-ant **ECoS** OCS aše aše axowě ašeta ašete ахотъ ašete ахъ axo

**48.1** The aorist is relatively faithful to the IE heritage. Except for first person singular - $\mathfrak{s}$ , the desinences of the root aorist consist of the thematic vowel  $*^{o}/_{e}$  (whereby \*e appears before obstruent, \*o otherwise) plus the past person-number desinences (see §5.9, 10.601).

**48.2** The newer paradigms are marked by the IE s, with special innovations. The person-number suffixes are preceded by -e-before a terminal consonant (or, at a later stage, zero) or -nt; the 3p pl desinence is therefore -s-ent. The original \*s remained dental after non-velar consonants, the root-final consonant was deleted, with concomitant length in the root if

<sup>&</sup>lt;sup>72</sup> Development of \*-ois# to OCS-i is posited also for the twofold Np masc desinence, see §39.62.

<sup>&</sup>lt;sup>73</sup> The imperfects in modern SC, Mac, and Bg (in the south) and Upper Sorbian (in the northwest) vary greatly by region. The imperfect was lost in Czech during the 1400s, and in Rus' probably after about 1250.

possible (\$10.6022).<sup>74</sup> After velars, *s* became *š* before front vowel, *x* before back vowel. In both types, the second and third persons singular lack the preterite marker. During the early historical period, the rules change and the x-desinences no longer cause lengthening and truncation of the root, but insert *o* after the root.

**48.3** The original s-desinences had shifted to the x-type in i-verbs, and in ECoS surely had spread to all verbs with an overt classifier.

	1 sg	2 sg	3 sg	3 du	l pl	2 pl	3 pl
IE	-om	-es	-et	-ete	-omos	-ete	-ont
root-aor	-u	-es	-et	-ete	-amu	-ete	-ant
ECoS	pād+u	pād+es	pād+et	pād+ete	pād+amu	pād+ete	pād+ant
OCS	padъ	pade	pade	padete	padomъ	padete	padq
s-aor	-su	-es	-et	-ste	-samu	-ste	-sent
ECoS	wēd+su	wed-es	wed-et	wēd-ste	wēd-samu	wēd-ste	wēd-sent
OCS	wěsъ	wede	wede	wěste	wěsomъ	wěste	wěsę
x-aor	-xu	-es	-et	-ste	-xamu	-ste	-š <b>en</b> t
ECoS	rēk+xu	rek-es	rek-et	rĕk-ste	rēk-xamu	rēk-ste	rēk-sent
OCS	rěxъ	reče	reče	rěste	rěsomъ	rěste	rěšę
ox-aor	rek-xu	rek-s	rek-t	rek-xte	rek-xamu	rek-xte	rek-xent
OCS	rekoxъ	reče	reče	rekoste	rekoxomъ	rekoste	rekošę
	nos-i-хъ	nos-i	nos-i	nos-i-ste	nos-i-xomъ	nos-i-ste	nos-i-šę

49 The infinitive and supine are believed to be frozen case-forms of ancient verbal substantives. The -ti of the infinitive probably represents the dative of a \*-ti- derivative, \* $t\bar{e}i$  or \*tei, while the  $-t\delta$  of the supine is from an accusative of a u-stem \*-tum.

**50.1** In IE, apophonic alternations were widely used within the inflection of individual verbs (e.g. the present tense of 'to be', §43.171, above). The LCoS list (probably not much shorter than the MCoS list) is already restricted to something under 40 verbs: the 4 with nasal infix (§45.11) presents, and the little groups with varying distribution of normal-grade in some forms and reduced grade in others. The roots involved are chiefly those with former diphthongs:  $*ei \sim *i$ ,  $*eu \sim *u$ ,  $*er \sim *r$ ,  $*el \sim *l$ ,  $*em \sim *m$ ,  $*en \sim *n$ , whereby the syllabic sonorants added a short \*i (>\*ir, \*il, \*im, \*in). In theory, a sonorant should not be syllabic between consonant and vowel (trt > trt but tra remains); in fact, the presence of hypothetical reduced-grade roots before vowel implies that the distribution of root-shapes is not original. The infinitive trbit < \*tbrit < \*trtēi fulfils expecta-

<sup>&</sup>lt;sup>74</sup> The long vowel probably reflects an inherited apophonic characteristic, rather than a new compensatory lengthening connected with the loss of the consonant.

A SKETCH HISTORY

tions, although in OCS terms the verb is exceptional (§16.522); infinitives like *b*srati and *s*tslati imply a complex history.

The tables provide samples of the alternating stems; for details see §15.643-645, §15.86, §15.871-.875, §16.511-.522, §16.91.

inf/aor pres	werz wirz	keit kit	telk tilk	werg wirg	straig streig	-soup -sup	stil-ā stelj	kirp-ā kerp
OCS	врѣсти	чисти	тлѣфи	врѣщи	стрѣци	-*соути	стьлати	чрыпати
	врьзжтъ ореп	vьтѫтъ count	тлькжтъ knock	<del>врыгжтъ</del> throw	стригжтъ cut hair	*-съпѫтъ strew	стеліжтъ spread	чрѣпл́иктъ dip
inf	welk	berg	mer	im-ā	zid-ā	pis-ā	slip-ã	
pres	wilk	birg	mir	emj	zeidj	peisj	sloipj	
OCS	влѣщи	брѣфн	мрѣтн	имати	3ьда <b>т</b> и	пьсати	сльпати	
	влькжтъ drag	врыгжтъ care for	мбржтъ die	емлять take	зижджтъ build	пишѫтъ write	слѣпл̀ият spurt	Ъ
inf	tirg-ā	bir-ā	sir-ā	strug-ā	gid-ā	zuw-ā	gun-ā	gin
pres	tergj	ber	ser	strougj	geid	zow	gen	ginj
OCS	тьрзати	върати	*сърати	стръгати	жьдатн	<b>ЗЪВ</b> АТН	гънати	жати
	трвжжтъ tear off	бержтъ gather	*сержтъ defecate	строужжтъ scrape	жнджтъ await	<b>20вжтъ</b> call	женѫтъ chase	жыйжтъ harvest

**50.2** The verb \**swati* \**serqts* 'shit' is easily reconstructable from modern dialects; it belongs with *bwa-ti* in \$15.644. \**Jebati* or \**jeti* \**jebqts* 'fuck' (from IE \**yebh-;* cf. Sa *yábhati* 'fucks') has alternate forms that go either in \$15.642 (with *sws-a-ti*) or \$15.824.

**50.3** The vowel-change in the imperative of *rek*- 'say', *pek*- 'cook', *tek*- 'run', *žeg*- 'burn' (§7.111) is unexplained. The root \**geg* (OCS *žeg/žig* §15.875) is unique; it seems to represent a modification of \**deg* < \**dheg*<sup>wh</sup> cf. Li *degù dègti* 'burn'; Sa *dahati* 'burns'). The root *tek* comes from IE *tek*<sup>w</sup> 'run, flow' (cf. Li *tekù* 'I run'; Sa *tákti* 'hastens'), *rek* has no clear cognates, and *pek* (the meaning 'cook' is by chance not attested in OCS) goes back to IE \**pek*<sup>w</sup> (cf. Sa *pácati*, Av *pačaiti* 'cooks, bakes'; Li, with metathesis, *kepù* 'I bake'; Latin *coquō* [< \**k*<sup>w</sup>*ek*<sup>w</sup>ō] 'I cook').

# **On Slavic Accent**

51 The Late Common Slavic accentual system must be hypothesized on the basis of fairly modern evidence. Written texts (including modern standard languages) ordinarily fail to note vowel length, pitch, or stress, but a few medieval East and South Slavic manuscripts do have systems of diacritic symbols that indicate prosodic features accurately. This material, together with detailed information from modern dialects, can be com-

pared to data from modern Baltic dialects and from ancient IE languages to reconstruct IE, Baltic, and Slavic systems. The data is enormously complex and scholars disagree on just what constitutes evidence and how questions are to be formulated. Here I can merely sketch some fundamental points.<sup>75</sup>

52 Accent is an underlying property of morphemes; stress is a phonetic manifestation. Every morpheme is either accented or unaccented.

Root morphemes belong to one of three classes: A. accented (on any syllable); **B**. post-accenting (i.e. accent falls on following syllable); **C**. unaccented. Late Common Slavic vowels could be underlyingly long or short; the exact distribution seems to have varied regionally, and even greater diversity was created after the jer-shift. If more than one accented syllable occurs on an underlying word, stress is assigned to the first (left-most). If there is no accented syllable, stress is assigned to the first syllable.<sup>76</sup>

A (accent fi	ixed on stem)	B (accent fe	ollows stem)	C (stem unaccented)		
p'org-ъ	por'og	korlj'-ъ	kor'ol'	gord-ъ g'orod		
p'org-a	por'oga	korlj'-a	korol''a	gord-a g'oroda		
w'orn-ěx'ъ	wor'oněx	korlj'-ix'ъ	korol''ix	gord-ěx'ъ gorod'ěx		
k'orw-a	kor'ova	sux'-j-'a	s'uša	golw-'a golov'a		
k'orw-q	kor'ovu	sux'-j-q	s'ušu	golw-q g'olovu		
l'ěz-q	l'ězu	nos'-i-q	noš'u	nes-q n'esu		
lěz-e-t'ь	l'ězeť	nos'-i-i-t'ь	n'osit'	nes-e-t'ь nes'et'		
l'ěz-e-t'e	l'ězete	nos'-i-i-t'e	n'osite	nes-e-t'e neset'e		

The table illustrates salient points of principle on the basis of hypothetical underlying LCoS forms and approximations of early East Slavic words. The symbol (') stands before accented vowels.

In A, the masculine noun 'threshold' (Nsm, Gs, Lp), the feminine 'cow' (Ns, As; for pleophony see \$26.51), and present tense forms of 'to clamber' (1s, 3s, 2p) illustrate that the stem-accent always generates phonetic stem-stress. The desinence of Lp was later replaced by the *-ax* of feminine stems.

In B, the masculine noun 'king' shows that in the same case-forms the stress is always post-stem (except that in Ns there is no post-stem vowel and the stress automatically must fall on the final stem-vowel. The femi-

51-52

<sup>&</sup>lt;sup>75</sup> The analyses provided in traditional accounts (e.g. Meillet-Vaillant, 1934, Vaillant GC 1. 1950) have been fundamentally revised on the basis of the work of Christian Stang (1957), Paul Garde (1976), Vladimir Dybo (1981) and others. See particularly Morris Halle, in *Language* 1997: 275-313.

<sup>&</sup>lt;sup>76</sup> Tone is not a distinctive feature in Slavic; it is phonetically important in SC and Slovenian, but not in underlying forms.

nine noun 'drought' shows a fixed stem-stress that is interpreted as a poststem accent that has been retracted. The present of the verb nosi+ 'carry' indicates that the post-accentuation yields stress on the desinence in 1s, but retraction back to the root in the other forms.

In C, the nouns 'city' and 'head' have stress on the first syllable of the stem unless the desinence is accented. If preceded by a preposition or other clitic that falls within the definition of phonological word (which varies by dialect), the initial stress moves as far left as possible. Theoretically, z'a-gorod, n'e\_za\_gorod, 'i\_ne\_za\_gorod; 'i\_ne\_na\_golovu. Similarly, the 1s pres with prefix would be 'prinesu, n'e\_prinesu, 'i\_ne\_prinesu.

53 These fundamentals are still valid—with numerous provisos—in modern SC and East Slavic.<sup>77</sup> Slovene (with its many dialects) adds many complexities, among them a strong contrast between long and short vowels that results in phonetic changes or disappearance of short unstressed vowels. West Slavic (except for the Kashubian dialects) eliminated the lexical accents underlying types A and B, and Czech and Slovak have automatic initial stress, accompanied by lexical long vs. short vowels and morphophonemic alternations involving length. Polish and East Slovak now have automatic stress on the penult. Standard Macedonian has antepenultimate stress, although some dialects have fixed stress on the penult and others have more complex placement involving the final three syllables of words. Bulgarian has several systems that are modifications of the old LCoS situation.

# On the Slavic lexicon.

54 The Late Common Slavic lexicon includes a solid framework of Indo-European words. Some are derived by specifically Slavic processes, but many—as we have seen—deviate in major or minor ways. There is no evidence of early borrowing from non-Indo-European groups, indeed nothing solidly datable before about 600, precisely when Byzantine and European observers are remarking on a Slavic presence.

**55.1** Slavic and Baltic have particularly intertwined vocabularies. Intimate words like *head* and *hand* (\*golwa, OCS glava ~ Li galvà; rqka ~ Li rankà) are unmatched elsewhere.<sup>78</sup> Germanic is lexically close to both

<sup>&</sup>lt;sup>77</sup> Halle argues cogently that these fundamentals were valid for IE and for Lithuanian. Note that *tone* is a distinctive characteristic of underlying vowels in Baltic.

<sup>&</sup>lt;sup>78</sup> Neither word has a clear outside etymology, but Li has a verb *rinkti* 'to collect' that furnishes motivation for *ronkā*. \*OCS *ročka* 'jar, fig. womb [of the Theotokos]' implies *rok-j-ak-a*, and the correlated verb.

Baltic and Slavic, although the phonology is often very different. Yet Slavic maintains its distinct individuality.

**55.2** The OCS word for man, human is  $\check{c}lov\check{e}k_{\delta}$ , with an ESI variant,  $\check{c}elov\check{e}k_{\delta}$ . A front vowel is needed to account for  $\check{c}$ , so unattested  $\check{c}_{\delta}l$ - is proposed as a putative allegro-form; in any case, all modern forms except for ESI are explicable from  $\check{c}lov\check{e}k_{\delta}$ . Etymologists tend to see  $\check{c}el$ - as meaning 'kin' (cf.  $\check{c}eljad_{\delta}$  f. coll. 'servants; members of household'). SI  $\check{v}\check{e}k_{\delta}$  'age, eternity' does not fit. Perhaps a \*woiko- cognate to Li vaĩkas 'boy' is involved. Man, husband is mož<sub>b</sub>, presumably \*man (cf. Gmc \*manw-, E man) with a root-extension \*-gj- (or -zj-?).

**55.3** The oldest form of the Slavic ethnonym is Np GASETANN (note desinential -*i*, not -*e*!),<sup>79</sup> almost certainly involving a soft twofold stem \**slověnj*-. The root is surely *slow*- 'be known, renowned' < IE \**kleu*-'hear'; the suffix is possibly adjectival and perhaps ECoS \**slow-ēn-as* 'characterized by, participating in fame' was used as a name. \**Slověnji* would then be 'the band (soldiers, clan) led by Slověn'.

**56.1** One requirement for a successful etymology is that differences in meaning be plausibly explicable. We know that meanings shift, but at the same time phonetic changes may cause formerly distinct words to fall together (e.g. E *rite, right, write, wright*). There is no question that OCS *ponos*<sup>5</sup> 'reproach' (related to *ponosi*+ 'revile') is historically "the same" as both R *ponós* 'diarrhea' (cf. *ponosit*' 'to revile') and SC *pónos* 'pride'. OCS *krasun*<sup>5</sup> means 'fine, beautiful' and *zivot*<sup>5</sup> is 'life'. Mod. Cz *krásný život* means 'beautiful life'; R *krásnyj živót* is 'red belly' (in ordinary language). R dictionaries may distinguish *život*<sup>1</sup> 'stomach' from *život*<sup>2</sup> '(archaic) life', but etymologists class them as descendants of a single older word. This is often difficult in seeking prehistoric etymologies.

**56.2** Ukrainian Шанець Gs Шанця 'trench', Pszaniec, -ńca 'trench (obsolete)', Slk šiance pl 'moat', SC šanac, -nca, Np šanci or šančevi 'trench (military), ditch', Mac šanec, and SC ušančiti P, ušančivati ušančujem I 'entrench', Mac se ušanči 'entrench self' would seem to indicate LCoS \*šanьcь and \*u-šanьči+. We have the evidence for another explanation: this is German Schanze and verschanzen, adapted long ago to the Slavic of the Hapsburg lands (and extended into Macedonia). The vowel-zero alternation and the automatic replacement of c by č when followed by a front-vowel morpheme are part of the rules of the individual dialects—they are "Pan-Slavic". Without the background information, this true etymology would be impossible to reach.

<sup>&</sup>lt;sup>79</sup> This philological fact needs to be taken into account in any etymological proposal. Forms like GAOBERNE, GAOBERNE do not appear until the mid-1300s. See Lunt, *IJSLP* 39-40 (1996): 281-2.

### A SKETCH HISTORY

57 The satem/kentum distinction (see note to §27.2) is far from absolute. The goose must have been known to all Slavs as \*gqsb, with a kentum-reflex for IE \*ghans-, like OHG gans, rather than a satem-reflex, like Li žqnsis. Perhaps the Slavs obtained a special, somehow preferable, variety from Gmc groups, along with the slightly changed name. The IE verb \*melg- that means 'rub, massage, caress' in Indo-Aryan was specialized in the west to mean 'to milk',  $d\mu e\lambda \gamma \omega$ , La mulgēre, OE melcan, Li mélžti mélžiu. \*OCS \*mlbzqtb (with SC müsti múzē, Sln molsti molze) shows ECoS \*mblz-ti, \*mblz-qtb, while \*melz-iw-o 'colostrum, foremilk' is illustrated by Slk mledzivo (with secondary dz) and Uk dial molozyvo. Surely then ECoS had \*melz/\*milz, with the satem z.<sup>80</sup> The universal word for milk, however, is ECoS \*melko (OCS mleko, R moloko). The voiceless k can only come from a Gmc form. Why this basic item bears an imported name has not been explained.

**58** It is tempting to speculate about differences that existed in the "homeland" as opposed to innovations that arose after the migration to new lands. It appears, however, that the Slavs who settled south and west of the Danube somehow kept in touch with their cousins to the north. Documentation is sparse and late for a large part of the everyday vocabulary. In theory, similarity of words in apparently related languages is to be explained by (1) genetic relationship, (2) borrowing, or (3) coincidence. In practice, borrowing from dialect to dialect obscures the difference between 1 and 2, and words can pass back and forth among various dialects. Items of commerce and warfare are particularly likely to employ this kind of word.

59 OCS korabljb 'boat' looks very like Gk  $\kappa \alpha \rho \alpha \beta \iota ov$  (with *o* for unstressed Gk *a*, and *a* for stressed *a*), but the stop *b* (rather than the spirant *v*) is improbable for a direct borrowing. Surely the Slavs learned this stem from an intermediary language.

**60** The productive suffix -arjb (§24.522) is by origin La -arius, but surely came to Slavic from various sources. Thus Go  $m\bar{o}ta$  'toll' and  $m\bar{o}tareis$  'taxgatherer' become OCS myto and \*mytarjb (the long  $\bar{o}$  was perceived as ECoS  $*\bar{u}$ ). Another adaptation takes Balkan Romance

<sup>&</sup>lt;sup>80</sup> The verb is not recorded for North Slavic or, as far as I can discover, for Bulgaria; the sense is expressed by \*doj-i+ 'nurse; suckle' < \*dheh(i) 'suck' (cf. Sa dhāya-'nourishing'; La fēmina 'woman [<she who suckles]'. Compare also \*doi-ten > dětę 'child', \*doi-w-ā > děva 'girl, virgin'.

60-63.1

\*pastore 'shepherd' to OCS \*pastyrjb, and then (by substituting a possibly affective suffix) to pasture (§24.26).<sup>81</sup>

**61** Gothic is recorded Germanic of a Balkan region (or more accurately of one individual writer, Wulfila, who died c383), so scholars try to derive Gmc words in Slavic from this language. Yet Slavs were interacting with Frankish military units and Saxons and Bavarians (including Catholic missionaries) with ever-increasing frequency after c750.<sup>82</sup> German peddlers and water-borne traders may well have introduced some words very early, and continued contacts may have resulted in slight modifications in the phonetics of words already borrowed. Latin-based terminology, with some Romance and Germanic (esp. Old Bavarian) details, was known to the Slavs who initiated the Cyrillo-Methodian mission that established OCS.

62 The stems of borrowed words are subject to KI and BdC for the early period, and borrowings from Germanic that have suffixal -*ing*- are assimilated to Sl - $e_3$ -b format. There is a short period when *velar* + *front vowel* in a stem conforms to KAI (§31), e.g. *ocbt*<sup>5</sup> 'vinegar', from a Romance form based on La *acētum*. The wave of Greek stems taken into OCS illustrates a new set of rules: velar stops before front vowels remain in spelling, but very probably were pronounced as palatals (as were the Gk models, see §2.4121).

**63.1** The word *church* and its Slavic equivalents are agreed to have something to do with the Gk adjective  $\kappa \cup \rho \mid \alpha \kappa \circ \varsigma'$  the Lord's [temple; flock, people]', but just how the items are related is controversial. The stem [kiriak-] or [kirjak-] would not ordinarily lose the *a*. The shape actually spelled in Rusian mss, *cwrky* (G *cwrkwe, -vi*) fits some forms: OCS *crwky*, R *cerkov'*, P *cerkiew*, SC, Mac *crkva*.<sup>83</sup> One sets up \**kirkū* and its putative ancestor OBavarian \**kirkō*. Yet the Kiev Folia have Gs *cirъkъve* and an adjective *cirkъnaĕ*, (generally interpreted as misspelled \**cirъkъvenaě*).<sup>84</sup>

<sup>&</sup>lt;sup>81</sup> The Slavic and Latin roots pas- represent IE \*pah,s 'protect, feed'.

<sup>&</sup>lt;sup>82</sup> German *Grenze* 'frontier, boundary' is from SI *granica*. Though it surely was borrowed during Carolingian times, before 800, it is first recorded in the 13th century in the Polish-German region.

<sup>&</sup>lt;sup>83</sup> Regularization of feminines in -y, -vv-e etc is easily achieved by combining the v with normal feminine a-stem desinences: roughly -k-y -k-ov-b > -kv-a -kv-u. Bulgarian has dialectal crskva (now being advocated as standard), but more usual is čerkva, apparently an older hypercorrect artificial form.

<sup>&</sup>lt;sup>84</sup> Mod. Cz *cirkev* means 'body of believers, ecclesia'; the usual word for a Catholic church is *kostel*, < La *castellum* 'small fort, castle'.

The data of Old Czech, Slovene, and Croatian are complex; suffice it to say that scholars have posited "Arian Go  $*kirik\bar{o}$ ", \*cbrbky, \*cirbky, and other variants.

Surely competing missionaries spoke several languages and dialects and freely introduced new terms into the speech of their Slavic clients. It is highly possible that different (or at least variant) names were used for the houses of worship of different missions, as a village in Scotland might have both a *church* (Catholic) and a *kirk* (Protestant).

The near-universal Slavic kupi+ and kupova+ 'to buy' owe their 63.2 root \*kaup- to Gmc sources, who learned it from La caupo 'shopkeeper'.85 A rival verb, the anomalous kriti \*kronjo 'buy' is implied for 10th-c Bulgaria as well as early Rus' (§45.21), but there is no way to reconstruct possible dialect distribution. The \*OCS noun \*usereze 'earring' reflects \*ausering-, from a form similar to Go \*ausihriggs, but no evidence is available about the time and place of borrowing.<sup>86</sup> The variety of terms for 'profit, interest' offers a hint of dialect differences: the usual lixva is derived from a hypothetical Go deverbal noun from *leihuan* 'loan, lend'; vozvitb and vozvitbje belong presumably to voz+vbj- 'go after, obtain' (n. 89, page 256); name is exclusive to early Novgorod, and is unexplained. Repayment is expressed (only in Su) by žlěd-, based on Gmc \*geld- (cf. Go fragildan, usgildan 'recompense'). Another loan where a velar was affected by KI is OCS \*šlěms 'helmet' (ESl šeloms) < \*xelms < Gmc \*helmaz. OCS mečь m 'sword' requires older \*mekj-, with an alternate shape to account for Old SC mbčb (Gs mča). Scholars cite Go \*mēkeis, while readily admitting that the phonetic relationships are imprecise. One guess is that both Gmc and SI inherited the term from a third group. Sek-yr-a 'axe' has a lax root vowel (though sek- 'cut' is always tense) and a unique suffix -yr-; a synonym is sěčivo, with the expected root and a rare suffix. Latin secūris has the same meaning and gender. We may surmise that Slavs obtained the implement, with its name, through trade or plunder-from speakers of Romance or intermediaries-and regularized the declension.

63.3 The usual word for 'money, coin' is pěnę3ь (Gmc penning [a silver denarius]). In Mt 22:19, for νόμισμα 'coin', Sav writes *cκudazu* and Mar sklęzs (I normalize \*sklęzu). In the first, \*penn- yielded pěn-

<sup>&</sup>lt;sup>85</sup> The source of La caupō is unknown. SI could have borrowed from Go \*kaupjan or OHG kouffen. Note that OE cēap 'trade', in contexts implying "good price", shifted meaning: cheap.

<sup>&</sup>lt;sup>86</sup> Contacts with Gmc peddlars may well have introduced some words very early; as Gmc shapes of words change, so might the SI pronunciation. (A connection with later R ссрыги is possible, but the direct source appears to be Turkic.)

(\$29.813) and *-ing* was equated with *-e3-b*. The second is Gmc \**skilling*-(first coined c550); I interpret the first syllable as [\$Kb] with automatic palatalization of *sk* before a front vowel—a dialect form corresponding to the normal *sc/st* spellings (\$3.311, n). A puzzling word *ceta* means 'coin, money' and implies \**kent*- or \**kint*- but the source is elusive.

63.4 The name of the city of Rome, OCS  $Rim_{\delta}$  (Cz  $\check{R}im$ ) is presumably masculine because most Slavic town-names are.  $R\bar{\delta}ma$  would be expected to yield \*rym-: the front vowel is explained from an Alpine Romance \* $\ddot{u}$ . The ethnonym židove 'Jews' (singular židovin $\delta$ ) surely is based on La judaeus, from some dialect (not yet identified) with \* $\ddot{z}\ddot{u}d$ -, perceived by Slavs as žid-.The fundamental symbol of Christianity, the cross, is called  $kr \delta st \delta$  in OCS, but western Catholics had \* $kri\check{z}b$  (Cr. dial  $kri\check{z}$ , Gs  $kr\bar{z}\check{a}$ ; Cz  $k\check{r}i\check{z}$ ). The first is probably from OHG krist, while the second goes back to late La  $cr\bar{u}ce$  [krūče], via a hypothetical Alpine dialect with a form \* $kr\bar{o}\check{z}$ .

Latin *lactūca* 'lettuce' is known to Slavs as *loškíka* (Macedonian dialect), SC *ločika*, Sln *ločika*, Cz *locika*. It surely is a 7th-8th century borrowing from a word perceived as *\*loktyka*, and immediately adopted as *\*lotjyka*. It then developed in accord with local Slavic dialect phonology.

Dalmatian toponyms often begin with *sut*-, reflecting La *sanctus* 'saint' borrowed surely as \**sqt*-. St. George occurs as *Sućuraj*, implying \**sqt jurbjb* and the general name derived both from late La Giorgius [gorgo-] and Gk  $\Gamma \epsilon \omega \rho \gamma \log$  [yoryos], with the closed o Slavs perceived as u (cf. *Solunb* for *Thessaloniki*). The *tj* developed according to local rules.

**64.1** There are no titles that clearly indicate social ranking. Two terms are built on the root wold 'rule'; vladyka (common, with a unique formant -yk-) and vlastelb (rare, but with a productive formant, cf. §24.521); neither is used as a title. Starějbšina m. 'elder' (obviously built on the comparative of starb 'old') translates a series of general names for leader. Starosta is only North Slavic; it apparently referred to a minor local dignitary, not necessarily elderly.

64.2 The supreme authority is gospodb 'lord, κύριος, δεσπότης, dominus [OHG trohtîn]', widely used to refer to divinity, and amply attested for earthly leaders. The word is a compound, IE \*ghos(-ti)- 'guest' and \*poti- 'master; power', reminiscent of La hospes (stem hospit-) 'host, guest, stranger' house-master'; Li viẽšpats 'lord', where vieš- is from IE \*weik- 'house, home, community, clan', related to Gk rolkos 'house'). The voiced d (for voiceless t) is difficult.<sup>87</sup> A frequent substitute is

<sup>&</sup>lt;sup>87</sup> Some scholars see IE \*poti- in the word spelled potspězě (Ls, Cloz), podspěg-,

gospodina, with a singulative suffix, often in the phrase gospodina domu (xraminy, xrama) 'master of the house' (which may render oikoδεσπό-  $\tau\eta\varsigma$ ).<sup>88</sup> \*OCS further attests gosudarb, and a collective, gospoda. Gospoda may also mean 'inn' (as in Sav; cf Cz hospoda, P gospoda). Two feminine derivatives, gospožda (< \*-pod-j-a, possibly possessive?), and gospodyni occur.

**64.21** In SW dialects, *gospoda* could be shortened to *gozda*, and Magyar borrowed it as *gazda*. SC took this form back with the meaning 'master, boss, chief, owner, landlord' and created a whole family of derivatives (e.g. *gazdarica* 'female boss, etc.; *gazdovati* 'to manage'). In Slovak, *gazda* is a farmer, *gazdiná* 'housewife', *gazdovstvo* 'agriculture'

64.3 Czech pán 'lord' and paní 'lady', are often linked by scholars with the župans, the apparent ruler of a župa, known from Greek, Latin and Slavic texts as an administrative regional unit. In OCS, župani appear in one translated text as unspecified highly-placed functionaries. The feminine pani seems to be connected with  $\pi \dot{\sigma} \tau v \iota \alpha$  'mistress of the household, wife' (Sa patnī), although the long vowel in the initial syllable of hypothetical \*pātinyā- needs explanation. Masculine pán is seen by some as derived from this feminine. Others propose \*geupānas > župans and zerograde \*gupānas > \*gspans > \*hpán > pán. A relationship to OHG \*gawi, G Gau 'region' has also been proposed. Another early medieval ruler, chiefly in Croatia, was ban (\*bans?), perhaps from Avar \*bojan 'rich man', perhaps from Iranian ban 'keeper, guard'.

64.4 Another important personage is entitled vojevoda (m), made of voj- (voji mp 'soldiers, army', vojuna 'war') and vod- (vodi+ 'to lead'), a compound that corresponds to OHG herizogo (heri 'army'; ziohan 'lead') and Gk  $\sigma\tau\rho\alpha\tau\eta\gamma$ ós (stratos 'army'; agō 'lead').<sup>89</sup> To judge from the range of terms it translates, OCS vojevoda designates not only a general, but a leader or chief in other spheres. As new societies emerge, the title acquires specific local meaning.

podsběg- '[wife] who has been put aside', conjecturing "wife who fled"; what is clear is only that the word was unfamiliar to the scribes.

Boms 'house, household; members of household' reflects an IE u-stem (like La domus), and has survived into most modern dialects. MCoS \*xorms and \*xormina are of unknown origin; puzzlingly, they share the same range of meaning in OCS. The modern sense 'temple' is a later specialization, still used in most of Slavdom.

<sup>89</sup> The root voj- belongs to IE wei- 'to go after something', cf. Li vejù výti 'chase', Av vayeiti 'chases'; La venāri 'hunt'. The root wed/wod is from IE \*wedh 'lead, lead home; marry' (cf. Li vedù 'I lead'; Av vādayeiti 'leads'.

64.51-64.54

64.51 OCS gospel texts reproduced the Gk hierarchy, \*kesars 'emperor, Caesar,  $\kappa\alpha\tilde{\alpha}\sigma\alpha\rho' \sim c\check{e}sarjb$  'king, rex,  $\beta\alpha\sigma\lambda\epsilon\dot{\nu}s' \sim ksne3b$  'prince, princeps,  $\check{\alpha}\rho\chi\omega\nu'$ . The first is strictly a book word, doubtless reproducing the Greek pronunciation [Késar], with a palatal  $\check{k}$ . In the few passages where it is appropriate, it is modified within OCS to make it closer to and finally identical with c $\check{e}sarjb$ .

**64.52** Kone36 is clearly a general-purpose title denoting 'chief, head man'; it must reflect \*kuning-, corresponding nicely to Gmc \*kuningaz, OHG kuning. It must have been borrowed early in independent Slavic linguistic development, while the progressive palatalization (BdC, §29.2) was still operative, possibly in the 4th century.

64.53 OCS cěsarjb implies earlier \*kaisārjas, which predicts the shapes that in fact exist among Roman Catholic Slavs, viz. northern Croats and Slovenes (kajkavski, Sln césar, Gs cesárja), Czechs (Cz císař), Slovaks (cisár), and Poles (cesarz). The OCS noun is the same as the usual adjective cěsarjb 'belonging to the king'; an unambiguous possessive cěsarjeva is also attested. The source seems to be a Latin adjectival form, caesareus [kaisārjus], or a later shape \*kēsārju, probably via Germanic. In OCS, the stem is usually abbreviated, u'ph, uph, but it always has t when written in full; this form surely was "correct" for OCS. The early Rus' scribes occasionally wrote uhcaph, providing the form that underlies uaph.<sup>90</sup> This surely represents a second source, with a short vowel in the first syllable; the suggestions that  $\check{e}$  was "shortened" or "reduced" to b in an unstressed syllable, or else in an allegro form of a frequently-used title or term or address, are inapplicable.

64.54 The Frankish leaders facing 9th-century Slavs were "Karl's men" in the sense that they were deputies of Charlemagne (Karl the Great, 742?– 814) and his descendants. The name \**Karlъ* surely formed a possessive adjective \**karljъ* to refer to these petty rulers. This, I maintain, is the word that was adopted to refer to Slavic kings in Catholic regions: SC *krâlj* Gs *králja*, Cz *král krále*, P *król króla*, R, Uk корóлъ короля́. The possessive is \**koroljevъ*.

<sup>90</sup> Uk цари́ця has hard c at the beginning, from cs < сы but palatalized c' in the final syllable.</p>

### VERB INDEX

Roman alphabetical sequence: a b c č d e ě f g h i ь j k l m n o p r s š t u ъ v x z ž з ž

e 6.61, 16.1011 bas- aor 10.82 berots 15.644 emati 15.643 bě- 16.1 emljota 15.643 -běgnoti 10.812 ems 11.13 bijqtь 15.93, 44.35 esmb 16.1, 43.171 bimь biše 14.3 bьrati 15.644 ědots ěxa- 10.811, 16.4, 44.2 bljъvati 15.52 ěsti ědeto 10.84, 16.22, 44.174, 46 bljudqts 10.82, 42 bodots 10.82 eti 15.83 bojati se 15.31 gasnoti 15.773 borjots brati 16.512 u-glubnoti 10.812, 15.773 bod- 16.11 bodots 10.82, 45.11 gobbzěti 15.22 note brati 16.512 gorěti 8.2, 44.33 byti 16.1, 43.171 grebots 10.82 za-byti gręząta 10.811 ganati 15.644 cvisti cvbtots 10.82, 15.842 gybnoti 10.812, 15.772 i(z)-čazaj- 5.712b idots 10.811, 16.3, 44.2 imati 15.643, 50.1 *i(z)-čeznoti* **10.812** imoto 10.82, 15.83 čisti 10.84, 15.842 po-črupota -črěti 15.841 imota/imějota 16.24, 43.176 črěpljota 15.643, 50.1 iskati 15.641 ištez- 10.812, 15.76 čujęts; 15.94, 44.35 -čыnotъ -četi 10.52, 10.83 ištota 15.641 čыots 10.84, 15.842 iti 16.3, 30.34 \*ьт 15.83, 16.24 dad- dati 16.21, 43.172, 46 -deždota 15.441, 16.8, 43.173 ja- see ěderots 15.644 u-děb- 15.75 klati 16.513 děja 15.44, 43.173 -klep 15.75 dovыěj- 6.42 kleti klunota 10.83 dušątъ 16.93 koljęts 16.513 kovati kovota 15.642 na-duj 16.92 dumotu 16.92 kradots 10.811 dvignǫti 10.812 voz-kros-ng 10.812, 15.773

#### VERB INDEX

krviota 15.94. 44.35 kwsněti 9.212 vzs-kysnąti 10.812 leg- lešti 10.811, 16.62 lějots 15.46, 44.34 lězota 10.811, 45.11 legota 10.811, 16.62 lek 10.83, 44.361 lijati 15.46, 44.34 pri-lspnoti 15.773 meljots mlěti 16.511 metati 15.641 mes- aor 10.84 meštota 15.641 metots 10.84, 15.823 -mlsknoti 10.812 mlěti meljots 16.511 mogoto 7.7, 10.811, 15.85 mviota 15.94 тыгоръ 11.13, 15.86 nebrěgota 15.5, 15.873 nesota 10.82 ně 16.1011 něsmb 6.73, 16.101, 30.312 něs- aor 10.82 ničet 15.773 -niknoti 10.812 -nozoto 10.811, 15.822 u-nyjots 15.94 oblěk- oblsk- 15.873 osnov- 15.55 padots 10.811 pekots 15.85 -perots 'tread' 15.644 perots 'fly' 16.93 pěti 16.53 vzs-pěvajotz 5.712e pišota 15.643, 50.1 рырть 11.32, 15.93, 44.35 -pыnots -peti 10.82 -pыrati 15.644 рызаti 15.643, 50.1 plěvots 15.842 plěžota plazati 15.643 plěvots 15.842 pljьvati pljujotъ 15.52 plovota pluti 15.841

pojota 16.53, 29.91 rasuse 16.91 rekota 10.83, 15.85, 50.3 revota 15.841 гъvati 16.642 -rět-, -rěsti 10.82, 10.90, 16.7 -reštota 16.7, 45.11 rovota 15.841 sěd- sěsti 10.811, 16.61, 45.11 sějati/sěti 15.41 u-sěknoti 15.771 sěkota 10.83 sedqta 16.61, 45.11 -segno 15.771 setb 6.8 i(z)-seknoti 15.773 -skvыrotъ 15.86 va(z)-slěpljota 15.643 slovots 15.841 smijati smějota 15.48, 44.34 o-snov- 15.642 stanots 15.712, 45.2 steljot 3 15.643 stulati 15.643 stojati 15.31, 15.47, 45.2 po-strěšti 15.874 strigota 15.874 stružota stragati 15.643 u-sznoti 15.74 səpati səpetə 15.32 szsati 15.642 i(z)-swnoti 15.773 sv. sošti 16.1 pri-sved- 10.812, 15.75 šud-, šul- 11.222, 16.3, 35.11 *šьjot*ъ 15.93 tekots těxs 10.83, 15.85, 50.3 tepots 15.824 tыnots 15.83 tl bkots 15.874 -topnoti 10.812, 15.74 trыр 15.2 trыi 16.522 trěžota 15.643 tresoto 10.811, 10.84 na-trovots 10.24, 15.841 tskati 15.642 po-tsknoti 10.812

ob-ujots 15.94

vedqtə 10.82 věděti 16.23, 43.175, 46 věs- aor 10.82 u-venqti 15.773 u-vezqtə 11.32 viděti 8.11, 15.231, 46 vijqtə 15.93 vijqtə 15.93 vijqtə 15.93 vijk-, vičk- 10.83, 15.873, 50.1 -vrěsti 10.82, 15.872, 50.1 vrěšti 13.31, 15.874, 50.1 vrbgqtə 10.811, 11.221, 15.874, 50.1 -vrbzqtə 10.82, 15.872, 50.1 na-vycajqtə 5.712b, 44.361 na-vyknqti 10.812, 30.21

xotěti 6.41, 15.232, 44.372 o-xr5mo 10.811, 15.771

zějota 15.46, 44.34 pro-zębnęti 10.812 zębomi 15.824 ziždots 15.643, 50.1 zudati 15.643, 50.1 zovota 15.645, 50.1 zovati 15.645, 50.1 u-žasnoti 10.812 žegota 15.85, 15.875 ženąto 15.644, 50.1 žeda- žedaj 15.641 židots 15.644, 50.1 života 10.52, 15.842 žbg- 15.875, 50.3 žытотъ - 15.83 žыпјотъ 13.2d, 15.83, 50.1 žrěti 15.86 žladots žlědots 15.876 žrыti 16.521

# SUBJECT INDEX

Bold-face numbers refer to paragraphs

a ~ ja 2.21, 30.1, 36-36.2 abbreviations 1.34 ablative 37.1 ablaut: see apophony accent 51-53 accusative (see case) gen. for acc. 4.13, 18.21 active participle: see verb usage, verbs adjectives 4.021-0211, 4.1021 long (definite) 4.3 usage 17.0-17.2 vocative 17.1 series 17.2 adverbs 4.8 affricates 2.121-122, 26.1 aj-verbs 15.9, 44.36 alphabets 1. alternations, vowel 3.5, 3.9, 33, 37.5 animate acc. 18.21 anomalous simple nominal declension 4.41, 37.32 aorist 10, 48 apophony 3.9, 33, 37.4, 44.14-15, 50.1 aspect 5.1, 5.7 a-stem nouns (= fem. twofold) 4.1, 4.1011, 39.1, 39.2 BdC 29.2, 29.41, 44.361 bě 10.91, 21.21 běaše 9.6. 21.21 blocked 1.2302 bysta 10.51, 10.91, 21.21, 21.211 c 3.4, 29.41 cases 4.02, 37.1 usage 18.0-18.6 nominative 18.1 accusative 18.2-.21 genitive 18.3 locative 18.4 dative 18.5 instrumental 18.6, 39.2, 39.3

classifier 5.41, 5.602 comparatives, formation 4.7, 39.71 long 4.31 short 4.19 complex words 3.0, 24.4 compound declension 4.03, 4.3-.312 compound tenses 14.0-.02 compound 24.4, 24.42 conditional 14.3-.31 conjunctions, usage 22.1-.3 consonant-stem declension 4.410 consonants 2.12-2.2 clusters 2.521-.523 liquids 2.521 soft 2.413 contractions, vowel 2.22, 4.3012, 6.5, 15.41 cyrillic 1.02-.03 čыо 'what?' 4.24, 40.21 da, usage 7.5, 22.1-.12 dative 18.5 absolute 18.5c possession 18.5h reflexive 18.5c with infinitive 18.5d, 21.4 declension mixtures 4.32, 4.5 definite adjectives 4.3-4.304 dentals 2.12, 3.311, 3.3131a, c, 3.6-.62, 11.212, 15.82, 26.1 diacritics 1.31 distribution, consonant 2.51 dj 26.23 e ~ o 3.5a e ~ u 4.11 ě 1.238, 3.5c *ě* ~ *a* 3.5c1, 7.21, 9.111–2, 15.31 *ĕ* ~ *i* 3.5c2, 4.101, 7.101 e ~ en 4.414  $e \sim y$ : see  $y \sim e$  $e \sim \ln/\ln 10.22, 10.6021, 11.212, 11.32$ 

eda 22.2 epenthetic n 3.10, 29.851 epenthetic *l* 2.521, 3.71 feminine nouns: nom. sg. -*i* 4.18, p. 223 n. 4 nom. sg, -6 4.4031 foreign sounds 1.215-.2161 Freising 1.04, 43.173 fricatives 2.12, 35-35.14 future, expression of 21.11 future perfect 14.4 g 1.213 genitive 18.3 form for acc. 4.13, 18.21 partitive 18.3a4, 18.3f w. comparatives 18.3f w. negation 18.3b w. supine 18.3c, 21.5 gerund 21.33 glagolitic 1.0-.01 hard stems, nominal 4.05 pronominal 4.2 heteroclitic 37.31 imperative 7.0-7.5, 46 imperfect 9.-9.6, 47 usage 21.2 infinitive 13, 49 instrumental: see case instr. sg., irregular 4.15, 4.17 iod (j) 1.24, 2.03, 2.2, 2.22, 2.5221, 2.61, 3.6, 27.1, 30.31-.34 initial lack of 2.23, 36.1, 36.6 intervocalic loss of 2.22, 4.3012, 6.5 prothetic iotation 3.6, 6.13-.23, 29.51 iotized letters 1.24, 1.32 i-stem (= simple nominal) declension 4.4, 38.51, 38.8 j: see iod ra 1.238 iako 22.3 je je 1.24 jers 1.237, 2.6, 38.41, 39.5 neutral jers 2.63-.631 tense jers 2.61 jer-shift 2.62 jb 1.235, 2.03

\**jь* 'he' 4.25, 40.22 KAI 3.4, 3.5c2, 31 KI 3.4. 3.5c1. 29.41 kyi 'which' 4.323 l, epenthetic 2.521, 3.71 labials 2.212, 2.521, 26.1, 27.1 laryngeals 27.1 lax vowels 2.11, 28.1 length, vowel 28.1 liquid diphthongs 26.51, 27.41 liquids (with jers) 2.63-.631 locative: see case long adjectives: see adjectives loss of jers: see jer-shift loss of iod 2.22, 4.3012, 6.5 *l*-participles 11.2–11.23 masculine, acc. = gen. form 4.13, 4.16, 18.21 masculine nouns of "fem." declension 4.16 metathesis 15.643 n., 15.86 n., 26.51, 36.25 nasal vowels 1.234, 2.11, 26.52, 29.81 nasal before nasal 29.813 negation 23. nominative: see case nominal suffixes: see suffixes normalized forms 1.101 noun declensions 4.021 *n*-stem neuters 4.414 numerals 1.5, 4.4, 4.402, 41 usage 20. obstruents 2.12, 10.11, 33, 34.1, 35.1 o-stem (hard masc.-neut. twofold) nouns 4.1 ov ~ u 15.501, 15.841, 29.91 *q* 1.2341–2, 29.811, 29.814 o or u 2.71 paradigms: see noun declension pronouns verbs palatal consonants 1.31, 26.1 palatalized consonants 26.11 palatalization 26.12 partitive genitive: see genitive

### SUBJECT INDEX

passive 21.31, 21.322, 21.62 passive participle: see verb, participle past tenses: see aorist, imperfect person, verbal suffixes 5.9, 43 personal accusative 4.13 pleophony, polnoglasie 26.52 pluperfect 14.2 possessive adjective 18.3d, 24.84 possessive dative 18.5h prefixes 3.1, 5.3, 5.31 prepositions, use of 19.-19.32 present tense 6 formation 6.11-12, 43.1 use 21.1 preterite (past tenses) 43 proclitics 2.31, 52 pronominal declension 4.2, 38.1, 40, 40.3 prosodic features 52 punctuation 1.4 reconstructed forms 1.102 "reduced vowels": see jers reflexive dative 18.5c pronoun 4.6 verbs 21.6 roots 3.21, 24.4-.44, 33 rounding 28.1 r-stem feminines 4.413, 38.44 ruki rule 27.3, 29.6 supine 13, 49 usage 21.5 syllabic structure 2.52 šč, št 2.121, 3.311, 26.23 ś 2.123, 40.12 tense jers: see jers tenses, compound 14.-.4 conditional 14.3-.4 future perfect 14.4 perfect 14.1 pluperfect 14.2

thematic vowel 37.5 tj 26.23 truncation 5.5, 5.6 twofold nominal declension 4.1, 38.52 unblocked 1.2302 u-stems 4.145, 38.2, 38.51, 38.7 v: see wvelar 26.1 verb usage 21.0-21.6 future 21.11 infinitive 21.4 past tenses (aor., impf.) 21.2 participles 21.3-.322 present tense 21.1 se verbs 21.1 supine 21.5 vocalic alternations 3.5, 3.9 vocative 4.11, 17.1, 38.9 voicing assimilation 2.522, 3.311 vowels 2.11, 2.51-2, 3.5 V656 (W656) 2.123, 4.21, 40.12 w (= v) 2.12, 3.72, 27.1 prothetic 3.25, 30.2 x-aorist: 10.602, 48.2 y 1.236, 2.03, 2.51, 3.5a-b y ~ zv 3.84.415, 11.341 *v* ~ *e* 3.5b, 39.42 y ~ i 3.5a, 29.72 zero 4.415, 5.601, 37.22 zero verb classifier 5.41, 15.8 žd 2.121 žť 2.121 ź 2.122 3 (z s) 1.211, 1.5, 2.122

This description of the structure of Old Church Slavonic is intended to present fully the important data about the language, without citing all the minutiae of attested variant spellings. The facts have been treated from the point of view of structural linguistics, but pedagogical clarity has taken precedence over the conciseness required for elegant formal discription.

Chapter Six is an entirely new addition to the original text. It contains a sketch of the development from Late Indo-European to Late Common Slavic and will be of interest not only to Slavists but to linguists in general.

> ISBN 3 11 016284 9 www.deGruyter.com